Bone-resident mesenchymal stromal cells in healthy ageing and osteoarthritis: enumeration and gene expression analysis in uncultured cells

Payal Ganguly

Submitted in accordance with the requirements for the degree of Doctor of Philosophy

The University of Leeds

Faculty of Medicine and Health

December 2019

The candidate confirms that the work submitted is her own, except where work which has formed part of jointly authored publications has been included. The contribution of the candidate and the other authors to this work has been explicitly indicated below. The candidate confirms that appropriate credit has been given within the thesis where reference has been made to the work of others.

The work in Chapter 3 and 4 of the thesis have appeared in publication as follows:

The analysis of *in vivo* aging in human bone marrow mesenchymal stromal cells using colony-forming unit-fibroblast assay and the CD45^{low}CD271⁺ phenotype Stem Cells International, August 2019

Authors: Payal Ganguly, Jehan J El-Jawhari, Agata N Burska, Frederique Ponchel, Peter V Giannoudis and Elena A Jones

I was responsible for: Experimental design, sample collection, performing experiments, analysis of data and manuscript writing.

Work attributed to others: study conception and design (JEJ, ANB, FP, and EAJ); sample collection (PVG), analysis and interpretation of data (JEJ, ANB and EAJ), manuscript preparation and final draft (EAJ), response to reviewers and revisions following peer review (EAJ).

This copy has been supplied on the understanding that it is copyright material and that no quotation from the thesis may be published without proper acknowledgement.

© 2019 The University of Leeds and Payal Ganguly

Acknowledgements

I would like to begin with a big thank you to my main supervisor Dr Elena A Jones not only for giving me the opportunity to work on such a globally relevant project but also for guiding, encouraging, criticising, appreciating and understanding my approach at different stages of the project. Apart from being a thoroughly professional and an extremely efficient scientist, I wish to thank Elena for her commendable leadership qualities that has maintained a successful research group of PhDs, postdocs, consultants and experts from across the globe. She is one of those very rare bosses who have the right mix of compassion, strength, vision and motivation and the world desperately needs more of people like her, more leaders like her.

I am very grateful to Leeds Institute of rheumatic and musculoskeletal medicine (LIRMM) for offering me the scholarship and the opportunity to work with people who are best in the field and also with people who will undoubtedly, be the best in the field, in the coming years (contemporary PhD students). A big thank you to University of Leeds (UOL) for being among the most accepting and inviting Universities I have been a part of. UOL pride themselves on their diversity of students and my personal experience will vouch for that. I have met some of the most brilliant minds of the future pursuing a PhD in subjects varying from literature to molecular medicine, from childhood studies to chemical engineering, from water law to plant genetics and many, many more. May this diversity of knowledge keep growing every year!

I would like to extend a big thank you to my postdoctoral supervisors Dr Jehan J El-Jawhari and Dr Agata N. Burska for their unending patience during my laboratory work, analysis and writing phase and for supporting me through and through during the difficult phases of the PhD. Jehan also played a significant role in my understanding of flow cytometry and Agata's inputs were vital for the gene expression and the novel type 1 interferon based chapter. None of this work would have been possible without Professor Peter V. Giannoudis, his patients and the team of surgeons and nurses in Leeds General Infirmary (LGI).

Next, I want to thank the entire MSC research group for being my 4th, but easily, the best research group I have worked with till date. Dr Dennis McGonagle, Dr Thomas Baboolal, Dr Richard Cuthbert, Dr Ala Altaie, Dr Weichao Rachel Zhai, Dr Clara Sanjurjo-Rodriguez and Dr Charlie Bridgewood, Dr Yasser-El-Sherbiny for constructive criticism during lab meetings. My colleagues (and friends) Dr Heather Owston, Dr Dragos Ilas for Mr Tobias Russell, Miss Hannah Rowe, Dr Daniel Perez Witzke, Dr

Thomas Scambler, Dr. Devon Smith, Mr Samuel Lara Reyna, Mr Jonathan Holbrook, Mr Georgi Georgiev and Dr.Katrina Moisely. Dr Heledd Haf Jarosz-Griffith, Dr Emily Caseley, Miss Zoe Wigston, Miss Chen Xia-Dong, Miss Rujiraporn Pitaksalee, Mrs Rekha Parmar and Dr Chi Wong for being very friendly and helpful and co-workers. Miss Priyanka Dutta for her encouragement and warmth and Miss Shelly Pathak, Miss Tejal Bhatt and Mrs Rekha Parmar for all the times they have pampered me, introduced me to the lovely members of their families and have been there to listen and counsel me through difficult situations.

I wish to thank my colleagues from the haematology department for encouragement, guidance, support and help at every stage needed. Dr Sophie Stephenson, Dr Amel Saadi, Dr Marjorie Davison, Dr Katie Holmes, Dr Rebecca Ross, Dr Adewonoula Alase, Dr Mario Cocco, Dr Claire Mckinley, Dr Ruth Kennedy, Dr Jorge Jimenez Lucias, Miss Naomi Gibson, Miss April Baral, Miss Bethany Kuszlewicz, Miss Emma Adams, Mr Nate Lee, Mr Fraser Thomas, Miss Nicole McDermott and Mr Kieran Walker, it has been an absolute pleasure knowing them. I want to convey a huge thank you to the flow cytometry facility in-charge Miss Liz Strszynski and Dr Adam Craig Davison, Dr Claire Taylor for very patiently going through the intense process of using a new equipment and guiding Dr Raechel Barlow for carrying out the enormous task of a laboratory manager for a busy floor as ours. Mrs Angela Greenbank for being very patient and always helping us out with reimbursement forms, meeting room bookings and for always greeting everyone with a bright smile.

Outside of work, I owe a generous thank you to Prof Claire Chambers not only for her insights on PhD life in general but also for introducing the benefits of Zumba in my life. It helped me keep sane through the entire PhD process and today, I cannot imagine my life without it. She has also played a vital role in talking me through the difficult times of the PhD. I would like to thank Dr Kathleen Wright, Dr Amrisha Pandey, Dr Lana Lai and Dr Rohit Thakur for their constant inspiration and motivation. Miss Kamla for her warmth, monthly supply of hand-made desserts and for introducing me to so many people, whom I now call friends. Miss Ambika Kapoor, Miss Richa Yeshevekar, and Miss Arththi Sathnanthar for all the regular catch-ups. Dr Kanhu Charan Pattnayak, his wife Lipsa and their daughter Samvidha for keeping me in their prayers and for looking out for me during the PhD. Dr Wei Goh and Mr Joey Mark Diaz, Dr Gnosi Michiladou and Miss Shuang Wang for their friendship.

Apart from the friends at University, I owe a generous thank you to all my house mates over the last 4 years. Miss Fay, Miss Marta Baltruszewicz, Miss Yixin Ma, Miss Holly

Ingram and Mr. Dominic Flower, Miss Lorienne-Doyle Maw, Miss Aysha Panter and Miss Sophie Roche for being very considerate housemates. It is going to be very hard to find housemates like all the people mentioned above.

Miss Shelly Pathak for keeping up with nearly every phase of my PhD, for introducing me to her lovely family, offering help of every possible kind and for ensuring I had a home away from home in Leeds, especially during the festive seasons. I am very grateful for her friendship. Dr Akhil Krishnan, Miss Manasi Inamdar, Miss Nanda Rohra, Mr Saurabh Patil and Miss Tejal Pant for the continuous flow of care and positivity, for the warmth and love, for ensuring that we caught up every month, for being the source of laughter and for being a guardian angel since 2014, respectively, even from 4,528 miles away! Dr Susmita Basak for her unconditional support and Miss Anusha Sundaram for being in touch nearly every week in good and bad times.

Finally, my family back in Mumbai without whom this would have in no way, been possible. My father, Mr Ramkrishna Ganguly for believing in me even when I did not, for being a constant source of motivation, inspiration, critique and for giving my sister and me the best life lessons on hard work and not giving up. My mother, Mrs Sumana Ganguly for her never ending strength, resilience, and unconditional love. Her dedication for dance inspires anyone and everyone who knows her and I cannot thank both my parents enough for putting up with me for nearly 3 decades now. Last, but not the least, a very big thank you to my younger sister, Miss Bitan Ganguly for being the light of my life, my strength during my weakness and my dose of good thoughts when situations were not the best. Her ability to comprehend situations at her tender age and her passion towards things she believes in, have always pleasantly surprised me. I miss and love her the most.

Abstract

Ageing considerably reduces the quality of life with osteoarthritis (OA) being the most common age-related degenerative disorder. Whilst majority investigations have focussed on cartilage changes in OA, damage in the bone preceding cartilage damage, is increasingly being acknowledged. Bone and cartilage arise from progenitor cells called mesenchymal stromal cells (MSCs) resident in the bone marrow (BM). It was hypothesised that changes in BM MSCs contribute to OA development. It was aimed to investigate changes in BM MSCs in healthy ageing and then explore if these changes were aggravated in OA using flow cytometry for the CD45^{low}CD271⁺ phenotype and gene expression.

Minimally-expanded/uncultured MSCs were used to avoid effects of *in vitro* ageing due to culture expansion. BM aspirates from 51 healthy donors (19-89 years old) were processed for MSC quantification using colony forming unit fibroblast (CFU-F) assay. For OA investigations, MSCs from donors with hip OA (56-83 years old) were recruited. Gene expression of MSC multipotentiality genes, genes associated with cell senescence and type 1 interferon (IFN1) pathway genes was compared between CD45^{low}CD271⁺ MSCs and donor-matched control CD45⁺CD271⁻ haematopoietic lineage cells (HLCs).

MSC numbers declined with advancing age as measured by both assays but a more prominent decline was noted using CFU-F assay. Colony size and integrated density significantly reduced in old donor MSCs indicating age-related decline in MSC proliferation. When cultured in media with old donor serum, young and old donor MSCs displayed lower proliferation. *IL6* expression from old donors displayed 4-fold increase in both MSCs and HLCs. IFN1 genes displayed strikingly high expression but no age-related changes in MSCs. In OA, number of genes displayed significant differences including *LepR*, *CXCL12* and *IL6* as compared to healthy old donor MSCs. Surface marker expression of CD106 and CD295 were found to decline significantly in MSCs and a similar trend for CD295 decline was observed in HLCs. Age-related increase in *IL6* expression was aggravated more notably in MSCs.

In summary, age-related decline was observed in MSC number and proliferative capacity while gene and surface marker expression displayed non-significant differences as compared to donor-matched HLCs. These data indicate that BM MSCs are potentially more resistant to ageing stimuli *in vivo* compared to other BM resident hematopoietic lineage cells.

List of abbreviations

Abbreviation	Full form
2D	Two dimensional
3R	Replacement, reduction and refinement
7-AAD	7-aminoactinomycin D
A-Fib	Anti-Fibroblast microbeads
AGM	Aorta gonad mesonephores
ALP	Alkaline phosphatase
AP2	Adipocyte protein 2
ATM	Ataxia telangiectasia mutated kinase
BID	BH3 interacting domain
BM	Bone marrow
BMA	Bone marrow aspirate
BMD	Bone mineral density
BT	Bone thickness
BV	Bone volume
C.elegans	Caenorhabditis elegans
CAR	CXCL12 abundant reticular
CASP1	Caspase 1
CCND2	Cyclin D2
CD	Cluster of differentiation
cDNA	Complementary DNA
CFU-F	Colony forming unit-fibroblasts
COX2	Cyclooxygenase 2
Ct	Cycle threshold
Cu/Zn SOD or SOD1	Copper/Zinc Superoxide dismuatse
Cx43/GJ1	Connexin43/ gap junction protein alpha 1
CXCL12	CXC motif chemokine ligand 12
DDR	DNA damage response
DMEM	Dulbecco's modified Eagle's medium
DMP1	Dentin matric acidic phosphoprotein 1
DMSO	Dimethyl sulfoxide
DNA	Deoxyribonucleic acid
DNAse	Deoxyribonuclease
dNTPs	Deoxyribonucleotide triphosphates
DXA	Dual energy X-ray absorptiometry
ECM	Extracellular matrix
EC-SOD or SOD3	Extracellular Superoxide dismutase
EDTA	Ethylenediamintetraacetic acid
ELISA	Enzyme linked immunosorbent assay
ES	Embryonic stem
FABP	Fatty acid binding protein
FABP4	Fatty acid binding protein 4
FACS	Fluorescent activated cell sorting
FCS	Foetal calf serum
FH	Femoral head

FITC	Fluorescein isothiocyanate
FM	Freezing medium
Fzd	Frizzled proteins
GCT	Global clinical trial
GDB	Global burden of diseases
GJIC	Gap junction intercellular communication
Gx	Glutathioneperoxidasecatalase
HLA	human leukocyte antigen
HLCs	Hematopoietic lineage cells
HPRT1	Hypoxanthine Phosphoribosyltransferase 1
HSCs	Hematopoietic stem cells
IC	lliac crest
ICD	International classification of disease
ID	Integrated Density
IFC	Integrated fluid circuits
IFN1	Type 1 interferon
IFNA1	Interferon alpha 1
IFNAR1	Interferon alpha receptor 1
IFNAR2	Interferon alpha receptor 2
IFNB1	Interferon beta 1
IGF	Insulin growth factor
IL6	Interleukin6
IL7	Interleukin7
IL8	Interleukin8
iPSCs	Induced pluripotent stem cells
IRF3	IFN regulatory factor 3
IRF9	IFN regulatory factor 9
IRG	Interferon regulated genes
ISCT	International society for cellular therapy
ISG20	Interferon stimulating gene 20
ISGF3	Interferon stimulated gene factor 3
ISRE	Interferon stimulated response elements
IVIVC	In vitro in vivo correlation
JAK1	Janus kinases 1
K/L	Kellgren-Lawrence
KI	Klotho
LAIR1	Leukocyte-associated immunoglobulin-like receptor 1
LepR/CD295	Leptin receptor
LNGFR	Low affinity nerve growth factor
MACS	Magnetic activated cell sorting
MAPK	Mitogen activated protein kinase
MCAM/CD146	Melanoma Cell Adhesion Molecule
MEPE	Matrix extracellular phosphogylcoprotein
MFI	Median fluorescence intensity
MMPS	Matrix metallopeptidases
Mn-SOD or SOD2	Manganese-Superoxide dismutase
MRI	Magnetic resonance imaging
MSCA1	Mesenchymal stem cell antigen 1
	8

MSCs	Mesenchymal stromal cells
MTCH2	Mitochondrial homolog carrier 2
NC	Nucleated cells
NH₄CI	Ammonium Chloride
NK	Natural killer
NSAIDs	Non-steroidal anti-inflammatory drugs
OA	Osteoarthritis
ONS	Office of national statistics
OPG	Osteoprotegerin
OS	Old serum
PBS	Phosphate buffer saline
PDGFRα	Platelet derived growth factor receptor alpha
PDGFRβ	Platelet derived growth factor receptor beta
PHEX	Phosphate regulating neutral endopeptidase X-linked
PPAR-γ	Peroxisome proliferator activated receptor-gamma
QCT	Quantitative computed tomography
QOL	Quality of life
qPCR	Quantitative polymerase chain reaction
RANK	Receptor activator of nuclear factor kB
RANKL	Receptor activator of nuclear factor kB ligand
RKOA	Radiographic knee osteoarthritis
RNA	Ribonucleic Acid
RNAse	Ribonuclease
RNF213	Ring finger protein 213
ROS	Reactive oxygen species
RT	Room temperature
Rt	Reverse transcription
RUNX2	Runt related transcription factor 2
SAMP6	Senescence accelerated mice –P6
SASP	Senescence associated secretory phenotype
SCE	Stem cell exhaustion
SFRP1	Secreted frizzled related protein 1
Sirt6	Sirtuin6
SM	StemMACS
SOD	Superoxide dismutase
SOST	Sclerostin
SOX2	Sex determining region Y- box 2
SPARC	Secreted protein acidic and rich in cysteine
STAT1	Signal transducer and activator of transcription 1
STAT2	Signal transducer and activator of transcription 2
STING	Stimulator of interferon genes
TBHP	Tert-butylhydrogenperoxide
TF	Transcription factors
THR	Total hip replacement
ТМ	Thawing medium
Тр53	Tumour protein 53
TRAP	Tartrate resistant acid phosphatase
TYK2	Tyrosine kinase 2

UK	United Kingdom
UN	United nations
VCAM-1/CD106	Vascular cell adhesion molecule-1
WHO	World health organisation
YS	Young serum

Table of Contents

Acknowled	igements	3
Abstract		6
List of abb	reviations	7
Table of Co	ontents	11
List of Figu	res	14
List of Tab	es	17
Chapter 1	ntroduction	
1.1 A	geing	18
1.1.1	Theories of ageing	19
1.1.2	Age-related conditions – clinical need to understand healthy ageing	21
1.1.3	Bone and bone marrow	23
1.1.4	Bone remodelling and age-related bone loss	25
1.2 N	ASCs	
1.2.1	Definition, characterisation and CD45 CD271 ⁺ phenotype	
1.2.2	Age-related changes in MSCs	
1.3 C	Osteoarthritis	
1.3.1	Epidemiology	
1.3.2	Current understanding of OA	
1.4 F	lypotheses and objectives	
Chapter 2	Materials and methods	40
2.1 A	ge groups distribution	40
2.2 E	thical approval	
	••	
2.3 C	onor selection and sample processing	41
2.3 C 2.4 C	onor selection and sample processing ell freezing and defrosting	41 42
2.3 C 2.4 C 2.5 N	oonor selection and sample processing ell freezing and defrosting /ISC Enrichment using Anti-Fibroblast microbeads	41 42 42
2.3 C 2.4 C 2.5 N 2.6 F	onor selection and sample processing ell freezing and defrosting /ISC Enrichment using Anti-Fibroblast microbeads low cytometry and Fluorescence activated cell sorting (FACS)	
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1	onor selection and sample processing ell freezing and defrosting ASC Enrichment using Anti-Fibroblast microbeads low cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry	41 42 42 43 43
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2	onor selection and sample processing ell freezing and defrosting ASC Enrichment using Anti-Fibroblast microbeads low cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry FACS.	41 42 42 43 43 43 44
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3	Ponor selection and sample processing The freezing and defrosting	41 42 42 43 43 44 44 45
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4	Conor selection and sample processing The format is a selection and sample processing ASC Enrichment using Anti-Fibroblast microbeads ASC Enrichment using Anti-Fibroblast microbeads Iow cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry Principle of flow cytometry FACS Sample preparation and cell staining for cell sorting FACS and sample collection	41 42 42 43 43 44 44 45 48
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5	Conor selection and sample processing cell freezing and defrosting ASC Enrichment using Anti-Fibroblast microbeads Iow cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry FACS Sample preparation and cell staining for cell sorting FACS and sample collection FACS data analysis	41 42 42 43 43 43 44 45 48 48
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C	Ponor selection and sample processing The format is a selection and sample processing	41 42 42 43 43 43 43 44 45 48 48 48 48
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1	Conor selection and sample processing cell freezing and defrosting ASC Enrichment using Anti-Fibroblast microbeads Iow cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry FACS Sample preparation and cell staining for cell sorting FACS and sample collection FACS data analysis Sene expression Principle	41 42 42 43 43 43 44 45 45 48 48 48 48 48 48
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2	Ponor selection and sample processing	41 42 42 43 43 43 43 44 45 48 48 48 48 48 48 48 49 49
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3	Conor selection and sample processing	41 42 42 43 43 43 43 44 45 48 48 48 48 48 48 49 50
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4	Ponor selection and sample processing	41 42 42 43 43 43 44 45 45 48 48 48 48 48 48 49 49 50 50
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5	Ponor selection and sample processing rell freezing and defrosting ASC Enrichment using Anti-Fibroblast microbeads Iow cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry FACS. Sample preparation and cell staining for cell sorting FACS and sample collection FACS data analysis iene expression Principle Fluidigm and Integrated fluid circuits (IFCs) Sample preparation RNA extraction Reverse transcription (Rt) for cDNA preparation	41 42 42 43 43 43 44 45 48 48 48 48 48 48 48 49 50 50 51 51
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6	Description and sample processing	41 42 42 43 43 43 44 45 48 48 48 48 48 48 49 49 50 50 51 52 52
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7	Ponor selection and sample processing	41 42 42 43 43 43 44 45 48 48 48 48 48 48 49 49 49 50 51 51 52 52 52
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7 2.7.8	Denor selection and sample processing	41 42 42 43 43 43 44 45 48 48 48 48 48 48 49 50 51 50 51 52 52 52 52 52
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7 2.7.8 2.7.8	Ponor selection and sample processing	41 42 42 43 43 43 44 45 48 48 48 48 48 48 48 49 49 50 50 51 52 52 52 52 52 52 53 54 54
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7 2.7.8 2.8 C Chapter 3	Ponor selection and sample processing	41 42 42 43 43 43 44 45 48 48 48 48 48 48 48 49 49 50 51 52 51 52 52 52 52 52 53 54 54 0rming
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7 2.7.8 2.8 2.8 2.8 2.8 2.8 2.8 2.8 2.8 2.8 2	As the second series of the second se	41 42 42 43 43 43 44 45 45 48 48 48 48 48 48 49 49 49 50 51 51 52 52 52 52 52 53 54 54 54 54 54 54 54 54
2.3 C 2.4 C 2.5 N 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7 2.7.8 2.8 2.8 2.8 2.8 2.8 2.8 2.8 2.8 2.8 2	Ponor selection and sample processing	41 42 42 43 43 43 44 45 48 48 48 48 48 48 48 48 49 50 51 50 51 52 52 52 52 52 53 54 54 54 54 54 56 56
2.3 C 2.4 C 2.5 M 2.6 F 2.6.1 2.6.2 2.6.3 2.6.4 2.6.5 2.7 C 2.7.1 2.7.2 2.7.3 2.7.4 2.7.5 2.7.6 2.7.7 2.7.8 2.8 C Chapter 3 C unit-fibrob 3.1 In 3.1.1	Ponor selection and sample processing Sell freezing and defrosting ASC Enrichment using Anti-Fibroblast microbeads Iow cytometry and Fluorescence activated cell sorting (FACS) Principle of flow cytometry FACS Sample preparation and cell staining for cell sorting FACS and sample collection FACS data analysis Sene expression Principle Fluidigm and Integrated fluid circuits (IFCs) Sample preparation Reverse transcription (Rt) for cDNA preparation Pre-amplification Sample run Data collection, analysis and presentation Pata presentation and Statistical analysis Proliferative capacity of human BM MSC from using colony the for last (CFU-F) assay Previous studies investigating age-related changes in number of human	41 42 42 43 43 43 44 45 48 48 48 48 48 48 49 49 50 51 50 51 52 52 52 52 52 53 54 54 6 m BM MSCs

3.1.2	Previous studies investigating age-related changes in colony area and density	in
humar	n BM MSCs based on CFU-F assay	58
3.1.3	Significance of microenvironment in MSC proliferation	59
3.1.4	Hypotheses and objectives	60
3.2 N	Nethods	61
3.2.1	Donor selection	61
3.2.2	Enumeration of MSCs using CFU-F assay	61
3.2.3	Colony area analysis	62
3.2.4	Colony integrated density (ID) analysis	64
3.2.5	Serum collection and preparation of medium with human serum	64
3.2.6	Experimental design for investigating MSC colony formation in human serum	64
3.2.7	Statistical tests	68
3.3 R	Results	68
3.3.1	Age-related changes in number of CEU-E	68
3.3.2	Age-related changes in colony area	70
3.3.3	Age-related changes in colony integrated density	72
3.3.4	Proliferation of MSCs in human serum conditions	74
34 Г	Discussion	
5.4		
Chapter 4	Investigating age-related changes in human BM MSC using the	
CD45 ^{low} CD	271 ⁺ phenotype	92
4.1 l	ntroduction	92
4.1.1	Candidate MSC multipotentiality genes	95
4.1.2	Candidate age-specific surface makers	105
4.1.2	Hypotheses and Objectives	109
4.2 N	Nethods	110
4.2.1	Donor selection	110
4.2.2	Native MSC enumeration by flow cytometry	110
4.2.3	Sample preparation and quantification of ROS in uncultured MSCs	112
4.2.4	Cell sorting	114
4.2.5	Gene expression between young and old donors	114
Hs049	173910_s1	116
4.2.6	Statistical analysis	117
4.3 R	esults	117
4.3.1	Age-related changes in the number of CD45 ^{low} CD271 ⁺ cells	117
4.3.2	MSC specificity of the selected genes	118
4.3.3	Age-related differences in osteogenic and adipogenic genes in MSCs	120
4.3.4	Resistance to oxidative stress	123
4.3.5	Age-related differences in bone remodelling and cellular interaction genes in	
MSCs	124	
4.3.6	Age-related differences in selected gene transcripts in HLCs	125
4.3.7	Expression of the selected surface markers on MSCs	128
4.3.8	Age-related differences in selected surface markers in MSCs	131
4.3.9	Age-related differences in selected surface markers in HI Cs	132
4.4 F	Discussion	133
Chapter 5	Expression of type 1 interferon (IFN1) and senescence genes in human E	3M
MSCs		141
5.1 Intro	duction	142
5.1.1	Senescence, SASP and age-related bone loss	142
5.1.2	Type 1 Interferons (IFN1)	145
5.1.3	Hypothesis and objectives	149
5.2 N	Aethods	149
5.2.1	Donor selection	149

5.2.2	Gene expression for exploratory transcripts analysis	149
5.2.3	Statistical analysis	
5.3	Results	151
5.3.1	The expression of all exploratory transcripts in BM MSCs and HLCs	151
5.3.2	Age-related differences in IFN1 signalling pathway related genes	158
5.4	Discussion	166
Chapter 6	Gene and surface marker expression of human BM MSCs in hin	
osteoarth	ritis	172
6.1	Introduction	
611	Hin ΩΔ	173
612	Senescence SASP ROS and OA	
613	Changes in MSC multinotentiality genes in MSCs from OA donors	
614	Changes in age-related surface markers in MSCs from OA donors	170
615	Hypotheses and objectives	170
6 7	Mothods	101 101
6.21	Deper selection	101
0.2.1	Coll conting	101
0.2.2	Cens oversession	102
6.2.3	Gene expression	
6.2.4	Statistical analysis	
6.3	Results	
6.3.1	Differences in the expression of multipotentiality genes between OA MS	SCs and
healt	hy old donor MSCs	
6.3.2	Differences in the expression of multipotentiality genes between OA HL	Cs and
healt	hy old donor HLCs	186
6.3.3	OA related differences in the expression of senescence, SASP and IFN pa	ithway
relate	ed genes in MSCs and HLCs	189
6.3.4	Biomarkers of age-related OA	198
6.3.5	Differences in surface marker expression in OA MSCs and HLCs compare	ed to
healt	hy old donors	199
6.4	Discussion	202
Chanter 7	Discussion	211
7 1	Vov findings	····· 211
7.1	Clinical relevance	211
7.2	The 'higger picture'	211
7.5	Comprehensive/ recorde	
7.4	Comprehensive research	
7.5	Study novelty and contribution to the field	
7.6	Drawbacks and study limitations	
1.1	Future Directions	
Reference	25	221
Annondiu	1. List of domay calls used severe the different symp of investigation	. in this
Appendix	1- List of donor cens used across the different arms of investigation	
thesis		
Appendix	2 – List of reagents, consumables, equipments, softwares, solution	s and
Tagman r	probes used throughout the thesis	
Appendix	3 – Supplementary Figure 3.1	
Appendix	4 List of tables indicating expression of exploratory genes from Cha	apter 5
and 6		
Appendix	5 Ethics	

List of Figures

Figure 1.1 Projections of global ageing
Figure 1.2 Unclear transition from ageing to diseases
Figure 1.3 Bone remodelling27
Figure 1.4 Age-related factors that can potentially contribute to OA
Figure 2.1 Flow-diagram indicating the steps followed for obtaining gene expression data throughout the thesis
Figure 3.1 Duplicate dishes containing colonies from each age group
Figure 3.2 Initial studies to evaluate difference in colony area in 10 young and old donors
Figure 3.3 Flowchart demonstrating the steps and crucial parameters involved in the selection of donors for testing MSC colony formation in human serum
Figure 3.4 Optimisation of controls for testing the prolferation of MSCs in human serum
Figure 3.5 Optimised conditions and medium controls for testing MSC colony-forming capacity in human serum
Figure 3.6 Age-related changes in number of CFU-F
Figure 3.6 Age-related changes in number of CFU-F
Figure 3.6 Age-related changes in number of CFU-F
Figure 3.6 Age-related changes in number of CFU-F
Figure 3.6 Age-related changes in number of CFU-F.69Figure 3.7 Age-related changes in colony area71Figure 3.8 Age-related changes in colony ID73Figure 3.9 Selection of control media75Figure 3.10 Representative donors colonies in all 3 conditions77
Figure 3.6 Age-related changes in number of CFU-F. 69 Figure 3.7 Age-related changes in colony area. 71 Figure 3.8 Age-related changes in colony ID 73 Figure 3.9 Selection of control media. 75 Figure 3.10 Representative donors colonies in all 3 conditions. 77 Figure 3.11 Colony and cell counts of young and old donor MSCs seeded in dishes with SM, media supplemented with YS and media supplemented with OS
Figure 3.6 Age-related changes in number of CFU-F. 69 Figure 3.7 Age-related changes in colony area 71 Figure 3.8 Age-related changes in colony ID 73 Figure 3.9 Selection of control media. 75 Figure 3.10 Representative donors colonies in all 3 conditions 77 Figure 3.11 Colony and cell counts of young and old donor MSCs seeded in dishes with SM, media supplemented with YS and media supplemented with OS 79 Figure 3.12 The analysis of colony area in SM, media supplemented with YS and media supplemented with OS 81
Figure 3.6 Age-related changes in number of CFU-F. 69 Figure 3.7 Age-related changes in colony area 71 Figure 3.8 Age-related changes in colony ID 73 Figure 3.9 Selection of control media 75 Figure 3.10 Representative donors colonies in all 3 conditions 77 Figure 3.11 Colony and cell counts of young and old donor MSCs seeded in dishes with SM, media supplemented with YS and media supplemented with OS 79 Figure 3.12 The analysis of colony area in SM, media supplemented with YS and media supplemented with OS 81 Figure 3.13 ID of colonies from young donors MSCs grown in SM, media supplemented with YS and media supplemented with OS 81
Figure 3.6 Age-related changes in number of CFU-F. 69 Figure 3.7 Age-related changes in colony area 71 Figure 3.8 Age-related changes in colony ID 73 Figure 3.9 Selection of control media 75 Figure 3.10 Representative donors colonies in all 3 conditions 77 Figure 3.11 Colony and cell counts of young and old donor MSCs seeded in dishes with SM, media supplemented with YS and media supplemented with OS 79 Figure 3.12 The analysis of colony area in SM, media supplemented with YS and media supplemented with OS 81 Figure 3.13 ID of colonies from young donors MSCs grown in SM, media supplemented with YS and media supplemented with OS 83 Figure 3.14 ID of colonies from old donors MSCs grown in SM, media supplemented with YS and media supplemented with OS 83 Figure 3.14 ID of colonies from old donors MSCs grown in SM, media supplemented with YS and media supplemented with OS 84

Figure 4.2 Steps involved in the quantification of ROS in CD45 ^{low} CD271 ⁺ BM MSCs113
Figure 4.3 Flex Six [™] chip showing its different segments as described in the text and shown in manufacturer's protocol
Figure 4.4 Enumeration of MSCs by flow cytometry117
Figure 4.5 Cluster analysis of multipotentiality transcripts in MSCs and HLCs 119
Figure 4.6 Expression of osteogenic and adipogenic transcripts in MSCs from young and old donors
Figure 4.7 Measurement of ROS and anti-oxidant enzyme SOD3 gene in MSCs from young and old donors
Figure 4.8 Expression of transcripts associated with bone remodelling and cellular interactions in MSCs from young and old donors
Figure 4.9 Expression of genes associated with osteogenic and adipogenic differentiation in HLCs from young and old donors
Figure 4.10 Expression of transcripts associated with bone remodelling and cellular interactions in HLCs from young and old donors
Figure 4.11 Quantification of surface marker expression in BM MSCs and in HLCs . 129
Figure 4.12 Expression of candidate age-specific surface markers on MSCs and HLCs from all donors combined
Figure 4.13 Expression of candidate age-related surface markers in MSCs from young and old donors
Figure 4.14 Expression of candidate age-related surface markers in MSCs from young and old donors
Figure 5.1 IFN1 signal activation and signalling pathway associated with ROS and intrinsic factors (adapted from (395) and (34))
Figure 5.2 48.48 gene chip outlay150
Figure 5.3 Expression of exploratory genes in MSCs and HLCs
Figure 5.4 Expression of senescence and anti-ageing genes in MSCs and HLCs 153
Figure 5.5 Expression of SASP-associated cytokines in MSCs and HLCs153
Figure 5.6 Expression of IFNA, IFNB and IFNA receptor genes in MSCs and in HLCs
Figure 5.7 Genes displaying significantly higher expression in MSCs than in HLCs. 155

Figure 5.8 Genes displaying significantly higher expression in HLCs than in MSCs.. 156

Figure 5.9 Expression of genes associated with IFN1>BID>ROS pathway in MSCs and HLCs
Figure 5.10 Expression of senescence and anti-ageing genes
Figure 5.11 Expression of cytokines
Figure 5.12 Expression of IFNA, IFNB and IFNA receptor genes
Figure 5.13 Expression of genes displaying statistical age-related differences 162
Figure 5.14 Expression of genes associated with IFN1>BID>ROS pathway
Figure 6.1 Classification of animal models of OA Error! Bookmark not defined.
Figure 6.2 Quantification of surface marker expression in OA FH MSCs and HLCs 183
Figure 6.3 Expression of genes associated with MSC multipotential functions in MSCs from healthy old donors and OA patients
Figure 6.4 Expression of genes associated with MSC multipotential functions in HLCs from healthy old donors and OA patients
Figure 6.5 Cluster analysis of exploratory genes in OA samples in MSCs and HLCs 190
Figure 6.6 Expression of senescence and anti-ageing genes in MSCs and HLCs from healthy old donors and OA patients
Figure 6.7 Expression of SASP-associated cytokines in MSCs and HLCs from healthy old donors and OA patients
Figure 6.8 Expression of IFNA, IFNB and IFNA receptor genes in MSCs and HLCs from healthy old donors and OA patients
Figure 6.9 Genes expression of IRGs in OA MSCs compared to MSCs from healthy old donors
Figure 6.10 Genes expression of IRGs in OA HLCs compared to HLCs from healthy old donors
Figure 6.11 Expression of genes associated with IFN1>BID>ROS pathway in MSCs and HLCs from healthy old donors and OA patients
Figure 6.12 Expression of genes that displayed age-related differences that were further exacerbated in OA
Figure 6.13 Surface marker expression in MSCs from healthy old donors and OA donors
Figure 6.14 Surface marker expression in HLCs from healthy old donors and OA donors

List of Tables

Table 2.1 List of antibodies, isotype controls and their volumes used for various flow
cytometry investigations47
Table 3.1 List of previous studies investigating age-related change in the number of BMMSCs counted by CFU-F assay
Table 3.2 List of previous studies investigating age-related changes in colony area and density of human BM MSCs
Table 4.1 Previous studies examining age-related changes in the number of BM MSCs measured using flow cytometry
Table 4.2 Previous studies investigating age-related differences in osteogenic,adipogenic and bone remodelling molecules in culture expanded MSCs
Table 4.3 Previous studies investigating candidate age-specific surface markers 108
Table 4.4 List of TaqMan probes used for gene expression study in this chapter 116
Table 4.5 Comparison of number of MSC by CFU-F and flow cytometry 118
Table 4.6 Median fold increase in the expression of multipotentiality in MSCs compared to HLCs 120
Table 4.7 Median expression levels and fold differences in selected genes in CD45 ^{low} CD271 ⁺ MSCs and CD45 ⁺ CD271 ⁻ HLCs between young and old donor groups
Table 5.1 Medians of expression of genes shown in figures in MSCs, HLCs, folddifferences and their <i>p</i> values
Table 5.2 Exploratory genes, age-related fold differences and their <i>p</i> values in MSCs
Table 5.3 Exploratory genes, age-related fold differences and their <i>p</i> values in HLCs
Table 6.1 Differences in the expression of MSC multipotentiality genes in OA MSCsand HLCs as compared to healthy old donors
Table 6.2 Differences in the expression of genes associated with IFN1 pathway, ROS, anti-ageing and senescence, in OA MSCs and HLCs as compared to healthy old donors

Chapter 1 Introduction

1.1 Ageing

Ageing in simple terms, is the process of growing old. This process however has been found to be a result of complex changes within an individual. Growing old has increasingly been associated with decline in strength, loss of physical fitness and increased vulnerability to diseases, all of which eventually lead to a gradual deterioration in the quality of life (QOL) (1). Diet, lifestyle, physical activities, heredity, epigenetics and environment, all of these factors are believed to contribute to this decline in QOL (2, 3). World Health Organisation (WHO) reported in 2018 that the "proportion of the world's population over 60 years old will nearly double from 12% to 22%" within a span of three decades (4). The report also outlines that by 2020, the number of people aged over 60 will outnumber children below the age of 5, suggesting a huge shift in the global demographics. The United Nations (UN) estimates over 3 billion people will be over the age of 60 by 2100, globally (Figure 1.1) (5).



Figure 1.1 Projections of global ageing

Increase in number of people aged over 60 over the next few decades until 2100, adapted from the UN 2017 study (5)

In mid-2014, the United Kingdom (UK) was identified as having an 'ageing population' when the average population age exceeded 40 years old (6). Age UK also points out that 37% of the carers for old people are themselves old and over 72% of them have reported pain or discomfort while providing care (7). The office of national statistics (ONS) predicted in 2017 that by 2039, nearly 74 million of UK population will be over the age of 60 (8). The office for budget responsibility predicts that the public spending between 2019/20 and 2064/65 will increase by £79 billion to meet the needs of the ageing population.

The fact that increasing age has been associated with increased frailty (an elevated risk of decline in health and function (6)) has led to an increased interest in understanding the mechanisms of age-related changes in the past few decades. In 2013, a pioneering paper 'the hallmarks of ageing' was published summarising nine major age-related changes at the cellular and sub-cellular levels. These were grouped into main three themes: cellular senescence, genetic instability and stem cell exhaustion (9). In combination, all these factors caused an increase in individual's vulnerability towards diseases, loss of control over daily activities and overall increased time period in recovering from physical injury.

Considering that increasing age has been linked with several types of cancers, cardiovascular and musculoskeletal diseases, loss in cognitive functions and neurodegenerative diseases, healthcare experts have introduced a concept of 'healthy ageing' (10), which WHO experts have defined as 'the process of developing and maintaining the functional ability that enables well-being in older age' (1). In other words, even though ageing has been associated with a number of diseases, it is not a disease by itself. It is the natural progression of one's life. Hence, to understand any age-related disease, it is vital to first understand the age-related changes in 'healthy ageing'.

1.1.1 Theories of ageing

In biological terms, ageing is a complex process of gradual build-up of numerous molecular and cellular damages over time. To better understand the process, a number of theories have been postulated over the years to explain different aspects of age-related differences.

The theory of stem cell exhaustion (SCE) outlines the role of stem cells that drive tissue regeneration over an individual's lifetime. It suggests that there is an age-related decline in the number and functionality of the stem cell pool within our bodies that

eventually leads to a reduction in body's ability to regenerate tissues or heal when damaged by injury or a disease (9, 11). Stem cells are undifferentiated cells that have the potential to differentiate either into any lineage (have total potency or are 'totipotent') or into multiple lineages ('multipotent'). Most adult stem cells are multipotent stem cells (12). Stem cells can be of different origin depending upon the tissue source and species they belong to. They can also be categorised as embryonic, if sourced from an embryo (ES cells); adult, if sourced from an adult (adult stem cells) or can also be induced by reprogramming from mature adult cells to an ES-like state (induced pluripotent stem cells, iPSCs) (13). SCE theory of ageing pertains to age-related changes in adult stem cells.

An age-related decline in the proliferation and regeneration capacities of stem cells has been established in the last decade (14) and a number of factors have been proposed as contributors. These include deoxyribonucleic acid (DNA) damage and, accumulation of toxic metabolites (harmful free radicals) in stem cells themselves and a notable loss in the balance between damage and regeneration signals in the surrounding microenvironment. Interestingly, one of the main current debates revolves around the preferential roles of intrinsic (intracellular) or extrinsic (extracellular) factors on the number and basic functions of stem cells. Research focusing on the impact of the extrinsic factors within in the stem cell niches (specific anatomies and microenvironments within which the stem cells reside) has gained an increased interest owing to parabiosys experiments showing that stem cells in old mice can potentially be reversed to be as functional as stem cells in young mice, using systemic factors from young mice (15, 16). However, this concept dates back to 1913 when Carrel performed experiments to understand ageing of organs with the aim of life extension (17). The concept has now gained importance potentially due to an increasing ageing population, worldwide.

Damage to the DNA of stem cells is one of intrinsic factors that have been proposed to lead to SCE. The theory of DNA damage response (DDR) is among the oldest theories aimed at understanding age-related changes at the sub-cellular levels (9, 18, 19). It refers to the accumulation of damage in the DNA over time. The structure and integrity of DNA continuously gets challenged from both intrinsic factors (for example, replication errors and mutations) and extrinsic factors (for example, background radiation, cellular microenvironment and reactive oxygen species or ROS), which play the role of causative agents of DDR (9). In young people, all the cellular damage/debris is automatically eliminated by our body by means of autophagy (engulfment of damaged cells and organelles by authophagosomes followed by their degradation by

20

lysosomes (20)), which in older age is shown to becomes less efficient (21). Telomere attrition, irreversible arrest in cell cycle (senescence), inadequate response to cellular stress are also key factors contributing to DNA damage (22). Interestingly, ROS acts a major source of formation of DDR which brings us to the next theory of ageing.

The ROS or the free radical theory was first proposed in 1955 by Harman where he suggested that accumulation of cellular damage along with declining integrity of cellular components lead to oxidative stress owing to impaired metabolic cycles in old age (23). The free radical theory of ageing suggests that damages at the cellular and sub-cellular levels are caused by ROS. ROS are generated continuously during regular metabolism but are scavenged due to the presence of naturally occurring anti-oxidant molecules within our bodies (24). The theory outlines that the accumulation of ROS increases with age and overwhelms the ability of protective anti-oxidant molecules to scavenge ROS, thus increasing cellular damage and eventually leading to inability of cells to regenerate. However, the exact mechanisms that govern age-related changes in humans due to ROS are still unknown. Until now, levels of free radicals and anti-oxidants have been used to estimate the balance between ROS levels and the potential anti-oxidant scavengers of ROS but only in culture expanded stem cells (25, 26).

Recently, ROS, DNA damage and SCE theories have been combined leading to various studies investigating age-related changes in hematopoietic stem cells (HSCs) within the bone marrow (BM) niche (27, 28). In independent studies, type 1 interferon (IFN1) was found to play a role in the clearing of senescent cells (29, 30) and IFN1 pathway was found to potentially be the link between DNA damage, ROS and ageing of HSCs (31). Bone and joint cells are produced and maintained by another type of BM-resident adult stem cells (mesenchymal stem/stromal cells (MSCs)) where they co-exist with HSCs in the same microenvironment. However, the roles of various factors like ROS and IFN1, on these stem cells in healthy ageing and age-related musculoskeletal conditions, remain unknown and require further investigation.

1.1.2 Age-related conditions – clinical need to understand healthy ageing

Age-related conditions are a global health burden and combating them is now the driving force for many scientists and economists alike. Section 1.1.1 above outlined the theories of ageing and potential causes of the age-related damaging effects. In this section, some age-related conditions will be discussed in more depth, primarily to highlight the debilitating impact they have on the suffering individual's QOL. In fact, I

believe, the co-existence of the numerous theories about the process of ageing, in reality, indicates a gap in knowledge on the complexity of ageing and its link to agerelated diseases. However, there is a possibility that in fact, all the theories are complementary and/or collectively responsible for the different levels of damages (subcellular, cellular and organismal).

There is growing body of evidence that connects different diseases to the process of growing old. For example, accumulation of damage over time, via ROS acting at the sub cellular and cellular levels have been found to be common between healthy ageing and many disease conditions including cancer (18, 32, 33), cardiovascular diseases (34-36), neurodegenerative disease (37-39), as well as diseases of the musculoskeletal system (40-42). Figure 1.2 shows how the known factors for age-related changes can transition to distinct disease conditions, but the main drivers for this transition remain unclear.



Figure 1.2 Unclear transition from ageing to diseases

Shared features of ageing and age-related diseases with an unclear path towards disease transition (adapted from (43))

Old age has been linked with many diseases along with the reduction in the QOL of older people. Thus, there are scientists who not only consider ageing to be a degenerative disease (44), but also propose that ageing must officially be classified as a disease by the International classification of disease (ICD) (43, 45). The ICD is the classification used by WHO for epidemiology studies and health care management globally, the 11th revision being the most updated version (ICD-11). It is a tool that aids the monitoring of occurrence of diseases to better understand the healthcare condition in countries and among populations. Ageing was not classified as separate disease in

ICD-11 and rightly so, as the concept of 'ageing is a disease' is controversial owing to the fact that it is a natural progression of every living being.

In fact, some scientists believe that we begin to 'age' right from our conception, even before birth. The process is simply broken down into phases of birth, growth, maturation, adulthood and old age. This idea has been further promoted by Allison *et. al* who found that pregnant rodents that were treated with anti-oxidants, had off-springs that grew old slower than those whose mothers were not given any anti-oxidants (46), suggesting that ageing begins from the moment we are conceived. While advanced age leads to considerable frailty, not every old person is frail. These variations are due to the different environmental factors, genetic make-up and life styles that individuals lead.

Among the diseases associated with old age, diseases of the musculo-skeletal system impacting the bones and joints may be considered the most incapacitating diseases worldwide. The WHO reports that osteoarthritis (OA) is the most common degenerative disease in older population (47). While the disease has primarily been described as the degeneration of cartilage with negative impact on the other joint tissues, recent evidence suggests that the damage to bone plays a key role in the disease progression (48). The disease has largely been identified by degenerating cartilage, stiffness and pain. Increasingly along with these, subchondral bone sclerosis, bone marrow lesions, increase in bone volume but low mineralisation due abnormal bone remodelling (explained in section 1.1.4) have gained recognition as hallmarks of the disease (49, 50), even though the debate as to if these hallmarks are the causes or the effects, are still active.

1.1.3 Bone and bone marrow

The skeletal system consists of bones, cartilage, tendons, ligaments and connective tissues. It forms the framework of our bodies and provides us with basic structure and support. Apart from protecting our internal organs, it is also responsible for our locomotion in daily lives. Bones in particular perform mechanical functions along with serving as the reservoir for the hematopoietic marrow and a reserve for calcium. At this given point of time there are 16,334 active clinical trials globally that are aimed to restore bone and bone related conditions (51).

Structurally, the bone contains inorganic mineral depot and organic extracellular matrix, lipids and water. The organic components include abundant quantities of collagen, proteoglycans, glycolipids and different types of cells. The inorganic component of the

bone is majorly hydroxyapatite ($Ca_{10}(PO_4)6(OH_2)$) (52). The composition involving both, the organic and inorganic components enable the bone to have both, mechanical strength as well as flexibility for movement. The inside of the bone is a very dynamic environment with various types of cells residing and interacting with one another. The bone marrow (BM) within the bone is also where stem cells reside, making it one of the most important organs in the body.

The BM exists within the trabecular spaces in the cancellous bones (such as the iliac crest or IC) and in the medullary cavities of the long (cortical) bones. It is contained by the exterior (cortex) of the bone and includes a complex mesh of blood vessels, cells and fatty tissue. This micro-environment within the bone forms the 'niche', where BM stem cells reside (53). As mentioned, the BM houses two types of adult stem cells and their progenies: HSCs and MSCs. HSCs originate in the aorta-gonad-mesonephors (AGM) region in the embryo and migrate to the liver and then to BM (54-56), where they reside and self-renew for the rest of the adult life. BM HSCs give rise to progenitor cells of both lymphoid as well as myeloid lineages. Lymphoid progenitor cells further give rise to natural killer (NK) cells, T lymphocytes and B lymphocytes. Myeloid progenitor cells on the other hand, give rise to monocytes/macrophages, eosinophils, basophils, neutrophils, erythrocytes and platelets.

HSCs residing in their BM niche replenish billions of cells every day while maintaining themselves over decades, a process known as haematopoiesis (57). The BM or the stem cell niche is the microenvironment within which these stem cells reside. The niche is known to provide the appropriate conditions, growth factors, temperature, oxygen levels to enhance and maintain the cells and their functions within the BM, making it an important factor for stem cell formation, function and survival (56). Within the BM or the stem cell niche, MSCs also exist which originate from the mesoderm and the neural crest, which is a transient embryonic tissue (58).

It has been reported that neural crest cells would migrate to different locations through the blood stream including the BM where their arrival would coincide with the arrival of HSCs. They continued to be there until adulthood and supported HSCs. There have also been reports of these neural crest cells travelling through developing nerves rather than via bloodstream. What is definitely known is that regardless of their embryonic origins, BM resident MSCs can give rise to bone, fat and cartilage lineages. For MSCs specifically, the BM niche can be divided into endosteal and perivascular based on the anatomical location within the microenvironment. Endosteal niche is located around the bone lining in proximity with bone formation cells (osteoblasts), whereas the

24

perivascular niche is located near the sinusoidal vessel within the BM cavity (59, 60). In mice, MSCs have been also topographically linked to CXC motif chemokine 12 (CXCL12) abundant reticular (CAR) cells that form a network of stromal cells connecting endosteal and perivascular niches. MSCs residing at the endosteal niche are most likely to be directly involved in the process of bone remodelling (58).

1.1.4 Bone remodelling and age-related bone loss

To maintain the integrity of the bone, about 10% of the bone is believed to be remodelled each year (61). The process of bone remodelling involves a balance between old bone removal and new bone formation. However, with increasing age, the rate of new bone formation has been found to fall behind the rate of old bone removal leading to the eventual decline in bone formation and reduced functionalities of the bone (62). The mechanisms underlying bone remodelling have been extensively studied and have revealed the involvement of the *RANKL* (Receptor activator of nuclear factor kB ligand)/*OPG* (Osteoprotegerin) pathway (63).

The cells directly involved in bone remodelling process are osteoblasts (the bone forming cells), osteoclasts (the bone resorption cells) and osteocytes, which are fully mature bone cells embedded within the mineralised matrix (64). Osteoblasts are derived from MSCs and are the effector cells largely known for their ability to produce bone matrix by the secretion of alkaline phosphatase, type 1 collagen, proteoglycans, osteocalcin and osteopontin. They are short-lived and exist and function in clusters along the bone surface. Some of the osteoblasts mature over time and get embedded within the matrix of the bone to eventually become osteocytes (64). Over the years, osteoblasts have been extensively studied in relation to their contribution to age-related bone loss and age-related bone diseases (65, 66).

Osteocytes are the most abundant cells embedded deep inside the bone matrix and their half-life is estimated to be around 25 years (67). Owing to their deep seated location (Figure 1.3), isolation and investigation of these cells pose technical challenges. Thus, most of the osteocyte investigations in humans have largely been based on histology of stained and fixed specimens or using gene expression for osteocyte-specific transcripts. Osteocytes are considered to be the final differentiated cells of the osteoblast lineage that eventually perish as a result of apoptosis/necrosis or autophagy (68). By gene expression, osteocytes have classically been identified by the expression of sclerostin (*SOST*), matrix extracellular phosphogylcoprotein (*MEPE*), dentin matric acidic phosphoprotein 1 (*DMP1*) and phosphate regulating neutral endopeptidase X-linked (*PHEX*) (69). Osteocytes have also been acknowledged as

25

'mechanostat' of the bone that sense and respond to mechanical stress by sending signals to osteoblasts and osteoclasts, as well as MSCs, on the bone surface and modulate bone remodelling (70).

Osteoclasts are the cells that are responsible for the process of resorption of bone. These are multinucleated cells that are derived from the mononuclear progenitors of the myeloid lineage from HSCs, that also give rise to macrophages. Discovered back in 1837, they possess a unique 'ruffled border' that is formed by complex finger-like projections from the cytoplasmic membrane, which isolate area under the cells to release high concentration of acid to dissolve bone mineral (71). The acidic environment is achieved by the transfer of H⁺ and Cl⁻ ions across the borders of the ruffles to ensure the dissolution of the non-organic component of the bone to bring about bone resorption (72). They are typically identified due to the presence of tartrate resistant acid phosphatase (TRAP) by immunohistochemistry (71).

Dysregulation of osteoclast activity can lead to both increased bone mass (if there is reduced development or function of osteoclasts) or decreased bone mass (in case of their increased development or function). For the purpose of bone remodelling, the progenitors of osteoclasts are recruited to the surface of the bone, where they mature to form a multinucleated osteoclast, then perform resorption functions and eventually die via apoptosis (73).

The process of bone remodelling is brought about by a number of soluble factors that recruit the above mentioned cells for executing the process. The key players of this process are RANKL, RANK and OPG. RANKL is a protein produced by the osteoblasts as well as MSCs, which binds to its receptor RANK on the surface of osteoclast progenitors. The binding of RANKL to RANK is essential for the formation and function of osteoclasts (71). The formed osteoclasts then begin to resorb the old bone leaving behind a portion of the bone pit known as 'howship's lacunae' exposed. Osteoclasts eventually undergo apoptosis to end resorption (74). Osteoblasts, then form new bone in the bone pit left behind by the osteoclasts for bone formation. OPG, the natural decoy receptor for RANKL is also expressed by osteoblasts and MSCs (69). When OPG binds to RANKL, the latter is no longer available for binding to RANK on osteoclasts progenitors and therefore it stops the formation of any further osteoclasts (Figure 1.3).



Figure 1.3 Bone remodelling

Figure indicating the different cells involved in the process of bone remodelling. Green spindle shaped cells are MSCs that develop into osteoblasts (shown in orange cells with blue nucleus) that are associated with bone formation. Osteoclast precursors give rise to osteoclasts (shown in orange multinucleated cells with ruffled borders) that are involved with bone resorption. RANK-RANKL binding brings about the process of osteoclast formation which will be stopped if OPG (Decoy receptor of RANKL) binds to RANK instead (adapted from (75))

Bone-remodelling process is therefore tightly regulated and balanced. However, with increasing age, this balance appears to be lost and the rate of bone resorption seems to exceed the rate of new bone formation leading to age-related bone loss (65). Why bone formation rate is reduced in old age remains unclear but it can be speculated that it can be due to a decrease in the number or function of osteoblast progenitors, the MSCs.

Age-related bone loss comprises of a gradual decline in bone mass and strength in both, men and women. Clinically, it is seen as an increase in non-traumatic, low energy

fractures in people of older age, especially over the age of 60. Bone loss can be measured as lower bone mineral density (BMD), bone volume (BV) and bone thickness (BT) on quantitative computed tomography (QCT) and in dual energy X-ray absorptiometry (DXA) scans. Of note, age-related bone loss is higher in women after menopause as compared to age-matched men (76, 77). It is also believed that women begin to lose bone earlier than men and at a greater rate as compared to men, potentially due to the role of sex hormones (78, 79).

The age-related bone loss has also been observed parallel to an increase in the adipose content within the BM. In fact, age-related shift from bone formation to fat formation is potentially the best known age-related change in the BM associated with bone loss (80). With the knowledge that bone is a very complex and dynamic organ, it can be said that these age-related changes in bone are potentially a combination of changes in many cells inside the bone, as well as soluble factors they produce. The reduction of growth factors like insulin growth factor (IGF) which is an important regulator of osteoblast proliferation and differentiation in the bone matrix in older age suggests that the BM microenvironment too plays a part in age-related bone loss (81). The bone cells directly involved in the bone remodelling process are mature and fully-differentiated cells. However, the progenitors of osteoblasts (MSCs) are interestingly, also the progenitors of the fat cells within the BM. Thus, a better understanding of age-related changes in BM MSCs is essential for uncovering the cellular mechanisms of age-related bone loss and their contribution to age-related skeletal diseases such as OA.

1.2 MSCs

1.2.1 Definition, characterisation and CD45⁻CD271⁺ phenotype

Mesenchymal stromal cells primarily originate from the mesodermal lineage in the embryo and in the neural crest as outlined in section 1.1.3. MSCs are plastic adherent, give rise to bone, fat and cartilage tissues (tri-lineage differentiation potential) and are positive for cell surface cluster of differentiation (CD) markers of CD73, CD90 and CD105 (82). They were first discovered by Friedenstein et. al in 1970s when he found a population of cells within the BM that was highly proliferative, adherent to plastic and formed colonies of fibroblasts when seeded at low density (83). Hence, due to the growth of these cells as colonies of fibroblasts, they came to be known initially as colony forming unit-fibroblasts (CFU-F). The CFU-F assay remains to date, the most accepted method for MSC quantification. Since their discovery, MSCs gained a lot of interest due to their tri-lineage differentiation potential as well as immune-modulating

abilities (84). MSCs have been subsequently isolated from other tissue sources including bone marrow, adipose tissue, synovial tissue and fluid and umbilical cord (85, 86), but these MSCs remain outside the scope of this thesis.

As of now there are globally over 1150 clinical trials that utilise MSCs for applications varying from the regeneration of the liver (GCT no: 0101062750) to the treatment of broncho-pulmonary dysplasia (GCT no:0103683935) (51). Among all the potential sources of MSCs, BM remains the most preferred choice for *in vitro* expansion and further application in clinical settings. This is possibly because the BM is the first and most reliable described MSC source, even though the frequency of MSCs in the BM is very low and ranges between 0.01-0.001% of total nucleated cells (87). Thus they usually need to be enriched and cultured *in vitro* for separating them from other cells and in order to obtain sufficient number of MSCs for clinical applications.

In spite of the growing interest in MSCs, there remain discrepancies in the definition of these cells. The source of the MSCs, the technique used for their isolation, laboratory conditions and many other factors add to this discrepancy. As lack of uniformity can give different results for the same experiments in different laboratories or clinical trials, there is a need to identify key factors to help define MSCs better.

In 2006, the International society for cellular therapy (ISCT) outlined three minimal criteria for defining MSCs in culture. First, MSCs should adhere to plastic, second they should be positive for the expression of CD73, CD90 and CD105 and not expressing CD45 (pan–leukocyte marker), CD34 (common HSC marker), CD14, CD11b, CD79a, CD19 and human leukocyte antigen (HLA) class II. The final criteria for MSCs is their tri-lineage differentiation capacity *in vitro*. For these differentiation assays, plastic adherent cells are grown in culture in medium containing factors for osteogenic, adipogenic and chondrogenic differentiation for 2-3 weeks. For osteogenic differentiation, the level of the enzyme alkaline phosphatase (ALP) found in high quantities in bone and Von Kossa or alizarin red stains indicative of calcium deposition from developing osteoblasts is quantified (82). For adipogenic assay, the level of oil red stain (that stains fat) is measured and for chondrogenic assay, a pellet is formed which is then weighed and the amount of glycosaminoglycans (GAGs, indicative of chondrocytes) is quantified (88).

While the criteria specified by ISCT have significantly contributed to ensuring the uniformity for defining MSCs, there are many factors that were not clearly addressed by the ISCT. To begin with, the first criteria of defining MSCs is the ability of the cell to

adhere to plastic. MSCs are not the only plastic adherent cells - fibroblasts, epithelial cells, endothelial cells and many others - are also adherent to plastic. The next criteria includes the expression of CD73, CD90 and CD105. Even though the ISCT panel has long been used for the characterisation of MSCs (82, 89) and they (CD73, CD90 and CD105) remain largely stable through in vitro cultures, it has often been criticised due to the lack of their cell specificity and due to their presence on other types of cells. For example, CD73 and CD105 were found to be expressed by dermal fibroblasts (90, 91) and in umbilical vein endothelial cells (92, 93). CD90 too has been found to be expressed on many other cell types in the BM, hence paving way for the need of a more suitable BM MSC marker. CD90 (Thy 1) is known to mediate cell to cell adhesion and is involved in the adhesion of monocytes and leucocytes to endothelial cells and fibroblasts (94). Also, it is a recognised marker for embryonic stem cells (95). Thus, simply combining the first two criteria, the test for ensuring that cells are MSC is unreliable. While some of these markers have been known to decline with in vitro culture expansion (96, 97), the investigations of age-related differences of these surface markers have not provided any conclusive results.

The final criterion is MSC tri-lineage differentiation ability, *in vitro*. While this criterion focusses on the differentiating ability *in vitro*, it critically misses out on the quantitative evaluation of native or uncultured MSC. Also, culturing MSCs have been shown to alter their properties, even at passage zero (P0, cultured but not yet passaged). If the criteria have been assigned to cells that adhere to plastic, it automatically means that they do not define native MSC that are freshly obtained and are uncultured. Changes in cell size, granularity and morphology (98, 99), surface marker expression (97) and gene expression (99) have been shown to alter as a result of *in vitro* culturing. These changes have been referred to MSC '*in vitro* ageing' or ageing induced by culturing cells in non-physiological conditions (100), and are further discusses below in section 1.2.2.

Schneider and Mitsui were first to investigate '*in vitro* ageing' using cultured fibroblasts from young and old donors (101). They found significant difference between fibroblasts from young and old donors with respect to cell proliferation, *in vitro* life span and senescence. They concluded that using cultured cells from young and old donors is a suitable model for investigating age-related changes in cells prior to culture. While the cells used were cultured therefore not devoid of manipulation, their study indicated that *in vitro* culture expansion sensitises cells to two-dimensional (2D) conditions therefore early stages of culture expansion better represent the state of cells *in vivo* compared to late stages.

30

Owing to lack of clarity in defining uncultured BM MSCs using ISCT criteria, several attempts have been made to investigate surface markers that reliably define them and can be used for their isolation prior to culturing. Mesenchymal stem cell antigen 1 (MSCA-1) and CD271 have been proposed to be positive specific markers of MSCs (94, 102). Historically, Stro-1 was the first surface marker found to be specific for BM MSCs (103). However, Stro-1 was found to be cross reactive with erythroblasts (103) and thus it was used to enrich/identify MSCs in combination with other molecules such as CD106 and CD146 (104) or with platelet derived growth factor receptors alpha and beta (PDGFR α and β) (105, 106).

CD271 or Low-affinity nerve growth factor receptor (LNGFR) belongs to the tumour necrosis factor receptor super family and has been used as a marker for BM MSCs by several independent research groups in the last two decades. CD271 is also known by the name of p75 neurotrophin receptor and was among the first receptors of its family to be characterised. It has also been suggested to be involved with the functions of cell survival and is suggested to be present in abundance during the growth phase (107). The phenotype CD45^{low}CD271⁺ has been shown to provide best specificity for BM MSCs compared to other candidate markers or their combinations (108, 109). In contrast to CD73, CD105 or CD90, CD271 is absent on fibroblasts indicating its least cross-reactivity with non-MSC adherent cells.

The CD45^{low}CD271⁺ phenotype has also shown to fulfil all the three ISCT criteria needed to define an MSC, that is, the *in vivo* expression of CD73, CD90 and CD105, plastic adhesion and formation of colonies, as well as tri-lineage differentiation following minimal cultivation (110, 111). CD271 has thus become a marker of choice for phenotyping BM MSCs. However, using CD45 as a negative marker or using other known markers like CD106, CD146 along with CD271 provides better characterisation of BM MSCs (112).

1.2.2 Age-related changes in MSCs

Age-related changes in cultured cells are measured using various techniques as large numbers of cells are available for investigation. Quantification of cell numbers (96, 113), telomere length (114), cell proliferative capacity (115), senescence (116), resistance to oxidative stress (117), metabolism (118), alterations in differentiation potential (80, 119) are some of the methods that have commonly been employed for studying age-related changes in cultured MSCs. As seen from the literature, these

31

investigations have been most commonly performed using cultured MSCs from young and old human donors. However, the effect of passaging on cells' behaviour have not been commonly considered.

Amongst animal models, *Caenorhabditis Elegans* or *C.Elegans* (phylum nematode, round worms with simple/non existing organ systems) has been widely studied owing to its primitive organ system, ease of use and short life-span which enables investigation of age-related changes in the organism during the whole course of its life span (120). However, to understand ageing in humans better, physiologically advanced animals like mice, naked mole rats, rhesus macques (primates) as well as dogs and cats have been used as models (121). Studies like these have immensely contributed to our understanding of age-related differences. Nevertheless, due to ethical considerations, similar experiments cannot be performed in healthy, functional humans. Thus, investigations involving age-related differences in young and old donors remain limited to use of (majorly) culture expanded and rarely, uncultured cells. Next sections will outline the main considerations for the study of age-related changes in human BM MSCs using cultured and uncultured cells.

1.2.2.1 MSC ageing in vitro (passage dependent)

As mentioned, age-related changes in human BM MSCs have been historically studied using culture expanded MSCs. However, it hasn't been commonly considered that all primary cultured and plastic adherent cells 'age' in vitro as a result of rapid cell division (98, 101). As MSCs are plastic adherent cells, their in vitro ageing on tissue culture flasks and similar consumables is relatively well understood and has been studied using proliferation, migration and differentiation assays (98, 122), as well as telomere length (114), senescence assays (99) and gene transcripts measurements (99). A decline in the proliferative capacities of serially-passaged BM MSCs has been observed in various studies (114, 123) along with a change in the expression of certain surface markers with in vitro ageing (97). CD106 and CD146 have been associated with young or aged MSCs based on culture expansion in vitro. CD106 has been shown to either decline in expression (124), increase in expression (125) or to have random oscillations (123) with in vitro passaging. CD146 has been shown to decline in serial passages with culture expansion (124). CD295 has been shown to be higher in cells expanded in culture along with higher levels of Annexin-V indicating increased apoptosis in MSCs of later passages (126). Morphologically, MSCs have been reported to increase in size, appear more flattened and granulated when expanded in vitro. Mauney et al. found about 10-fold increase in the size when they compared cells in late

stages with cells in the initial stages of culture expansion (127) along with an increase in cell granularity (99) indicative of *in vitro* ageing and replicative senescence.

Cells in the later passages are known to show greater positivity for senescence, indicated by assays like SA- β -gal (128, 129). Senescence is generally defined as an irreversible arrest of cell proliferation (130) and over the years various methods have been used to identify senescent MSCs in culture (131). Apart from all of the differences mentioned above, the shift towards adipogenic differentiation from osteogenic differentiation has also been shown when MSCs were cultured *in vitro*. Bonab et al. in 2006 have shown that when donor MSCs are cultured over time *in vitro*, then both, their osteogenic and adipogenic potential decline after 5-8 passages (98).

This clearly suggests that *in vitro* differentiation studies are more indicative of replicative senescence that has accumulated over time rather than resembling the actual differentiation capacity of MSCs *in vivo* towards any lineage. The compromised ability of MSCs in later passages to undergo osteogenic differentiation was also recently shown by Yang et al. by gene expression (123) and by other researchers using *in vitro* differentiation assays (132, 133). Even though there are reports that have shown a decline in *in vitro* adipogenic potential with increasing MSC culture (99), the presence of a 'fatty marrow' with advancing age (134) suggests toward an increasing adipogenic differentiation of BM MSCs *in vivo*.

1.2.2.2 MSC ageing in vitro: donor age dependent

Summarised evidence from literature outlined in the previous section were performed using BM MSCs from early and late passages. The loss of surface some markers or an increase in cell size that have been observed during this exaggerated ageing process, lead to a hypothesis that similar changes may occur in older donor MSCs *in vivo*.

This section focuses on studies that were performed on BM MSCs from young and old human donors at the same passage, in order to mimic their age-related changes *in vivo*, a little better. A number of studies observed a decline in osteogenic potential (135, 136) and telomere lengths in MSCs from older donors compared to younger donors from the same 'culture age' (114). With respect to surface markers, an up-regulation CD264 (137), CD106 (138) and CD295 (126) has been reported from cultured cells of older donors when compared to cultured cells of young donors. Increased production of ROS along with an increase in the proportion of senescent cells was also observed in cultured cells of older donors as opposed to the cultured cells from younger donors (96).

Stenderup et al. examined the various aspects of MSC ageing using cultured cells from young (18-29 years old) and (68-81 years old) donors and found an age-related decline in the life span of MSCs from older donors (129). Another age-related difference observed when MSCs were cultured was the decline in number of MSCs with advancing age, potentially indicating that higher proportion of MSCs in older donors were senescent (139). These findings suggested that there might be an age-related decline in the number of MSCs *in vivo* potentially due to increased ROS levels, which should also be quantified in uncultured cells.

1.2.2.3 MSC ageing in vivo: animal models

Our understanding of age-related changes in uncultured MSCs majorly comes from animal studies. A number of species have been used to understand broader ageing processes *in vivo* using *Caenorhabditis elegans* (or C.elegans due to short life-span) (140), rodents, naked mole rats and primates (121). Investigations specifically regarding the ageing of MSCs have most commonly been restricted to rodents owing to feasibility and ease of handling (141). Among these, the shift from osteogenic to adipogenic differentiation was shown in the 1990s in senescence-accelerated mice–P6 (SAMP6) by Takahashi et al. (142). These mice developed osteoporosis within a few months after birth suggesting that MSC senescence could lead to the imbalance between bone and fat differentiation.

In earlier studies using cultured MSCs, BM MSCs of old C57BL/6 mice (20-26 months old) were found to produce higher amount of fat compared to younger (6-8 months old) mice (143). BM MSCs from the old female Wistar rats (56 weeks) had increased senescence and apoptosis, reduction in proliferative capacity and a decline in bone markers as compared to young (3 weeks) rats (144). Interestingly, the authors did not find any increase in MSC adipogenesis in the old rats when compared to young rats. They found lower number of CFU-Fs in old rats, which was also observed by Katsara and co-workers in adult mice (>3 months) in comparison to young (<1 month) mice (145) and by Josephson and colleagues in old (52 weeks) mice compared to young (12 weeks) and old (146). Studies investigating BM MSCs from young (4 months) and old (15 months) Sprague Dawley rats, found lower tri-differentiation potential in old rat BM MSCs compared to BM MSCs from young rats. The study also observed an age-related decline in the presence of Connexin43 (Cx43) in BM MSCs, (147) which is the most abundant gap junction for intercellular interaction within the bone (148).

In another extensive study, rat BM MSCs were culture expanded to passage 100 and various experiments were carried out to understand the link between *in vitro* ageing and chronological ageing of these animals. BM MSCs from old (12 months) Sprague Dawley rats were compared to young (3 weeks) rats (149) and the results showed that the MSC migration potential and anti-oxidant capacity were the only parameters that decline in both, *in vitro* and chronological ageing. The osteogenic differentiation potential, cell morphology, senescence markers and the expression of genes associated with MSC differentiation showed significant changes only in *in vitro* ageing. Authors concluded that chronological ageing and *in vitro* MSC ageing are distinct processes (149). However, based on accumulated evidence from all presented animal studies, it is possible to hypothesise that *in vitro* MSC ageing (passage dependent) represents an exaggerated ageing process, some features of which (for example, a loss of some surface markers) can be detected in MSCs *in vivo*. Animal studies have also indicated towards a decline in the number and proliferative capacity of BM MSCs, mainly in rodents.

1.3 Osteoarthritis

1.3.1 Epidemiology

Osteoarthritis (OA) is among the most incapacitating degenerative diseases in the world. In the United States, it is the most common joint disorder and 10% of men and 13% of women over the age of 60 suffer from OA (150). It is also cited as the most common rheumatic condition with a huge impact on population and demographics, worldwide (151). Of the 291 conditions examined in the global burden of diseases (GBD) study, hip and knee OA are ranked the 11th major contributor, just after diabetes and falls (151). Incidentally, it is also the most common reason for total hip and knee replacement (152).

The global cost of OA has been estimated to be very high. Chen et al., described total costs as direct costs (medication, surgery), indirect costs (loss of productivity, absence from work) and intangible losses (suffering, decreased QOL and depression). In 2012 in the UK alone, just the direct OA costs were recorded as 168 million for drug therapies and 872 million for joint replacement surgeries (153). The direct and indirect costs of other studies in America and Europe have also revealed considerable amount of cost incurred due to OA. As for intangible loss, the level of anxiety, depression and loss in personal morale along with decline in QOL is unimaginable (154). And, this impacts not only the patient suffering from OA, but also close family members and the loved ones of the patient.

Considering the global burden OA imposes, it has been classified under the diseases of the musculoskeletal system in four further subparts in the ICD by the WHO. The four major categories of OA in the ICD-11 include OA of the hip, OA of the knee, OA of the wrist and hand and OA of other joints. Each of these categories are then further subdivided on the basis of the cause/origin. These are primary, post traumatic, secondary and unspecified causes.

Unfortunately, by the time a patient is diagnosed with the disease, the condition has usually progressed considerably. This means, that by the time, the patient has come to the doctor with complaints of pain and stiffness, chances are, the patient will certainly be given pain relief drugs for pain management and suggested lifestyle changes with immediate effect. The other outcome is that the patient will need surgery. Tissue engineering applications have seen an increased interest owing to the possibility of minimal invasion for joint repair and many biomaterial scaffolds are now available in the market (155); however, the attempts are far from becoming the first choice of therapy on a global scale. The diamond concept (156) of tissue regeneration requiring a scaffold, cells, growth factors and mechanical strength of a given tissue engineering platform often faces challenges to be translated from 'bench to bed' successfully (157).

1.3.2 Current understanding of OA

OA is a degenerative disease that may have several underlying causes. Pain in joints is the key symptom for medical attention. WHO's scientific group on rheumatic diseases estimates that 10% of the world's population aged 60 years or older have significant clinical problems that could be attributed to OA (158).

Commonly recognised by stiffness and pain in joints, on pain persistence, radiological examination involves measuring joint space narrowing and the presence of osteophytes., thinner cartilage lining and inflammation in the area of pain. While there are different techniques for imaging in OA, the technique of magnetic resonance imaging (MRI) has become a very useful method owing to its ability to evaluate the pathology in structures that are not easily visible in traditional radiographies (159). Due to these observations, initially, OA was approached as the disease of the cartilage. However, it has been recognised as a degenerative disease of the joint owing to the involvement of the cartilage, bone, synovium as well as meniscus and ligaments (160).

In a study in 2008 by Quintana and Azkarte, the authors found the prevalence of hip and knee OA in older population (161). Obesity (162, 163), genetics (164), mechanical
stress (165), physical trauma and age (166), all have been associated with the progression of the disease. While it is understood that over 45% of men and 55% of women over the age of 60 are likely to face OA, it is also established that factors like diet, lifestyle, physical exercise, genetics and environment also play an important role. A number of common factors have been suggested to link old age and OA. These include loss of normal bone structure and increased stiffness in joints (167), oxidative stress due to ROS (168), cellular senescence (169) and reduced QOL (170) outlining that OA is complex and multifactorial. However, majority of these links have been found using cultured chondrocytes and investigations using bone cells remain minimal. Considering OA is multifactorial and the number of links found between ageing and OA, one may hypothesise that these age-related factors are potentially exacerbated in OA patients (Figure 1.4).

Healthy ageing

Increased ROS Increased degeneration Increased inflammation Reduced immunity

Osteoarthritis

Figure 1.4 Age-related factors that can potentially contribute to OA

The most common way to define OA radiographically is by the use of Kellgren-Lawrence (K/L) radiographic scoring method that defines OA ranging from grades 0-4 depending on the level of severity, presence of osteophytes and joint deformity in general. While a school of thought believes that it is the secretion of pro-inflammatory factors that leads to the degradation of the ECM in cartilage and eventually leads to bone remodelling in OA; another school of thought supports the hypothesis that subchondral bone remodelling and joint inflammation precedes the articular degeneration in OA (171). Another fact supporting our lack of understanding of OA pathophysiology is the presence of over a million scientific publications on 'Osteoarthritis' in google scholar and the disease still being a major health care burden across the globe with pain management and surgery as the last medical interventions available to patients.

Due to the fact that OA is a multi-factorial disease involving articular cartilage, the subchondral bone as well as the synovium and synovial fluid, the exact mechanisms of the degradative events up to now have been difficult to dissect. Largely, the development of OA may be segregated as ageing dependent processes and ageing independent processes. Ageing dependent processes include the wear and tear mechanisms, such as age-related changes in cartilage matrix, loss of viable chondrocytes due to apoptosis and increase in age-related chondrocyte senescence. As mentioned, age-related process within OA subchondral bone, including MSCs however, remain poorly understood (172). Ageing independent processes include the processes of mechanical load and physical injury, genetic and environmental factors along with stress on lifestyle and obesity which impact individuals, irrespective of their age.

Current approaches for treatment of OA include pharmacological approaches using medication for pain relief like non-steroidal anti-inflammatory drugs (NSAIDs) and others (173) and non-pharmacological approaches. These include surgeries, physical alternatives like orthosis, insoles and foot wear (174), changes in lifestyle (175), use of physiotherapy (176) and an increase in the use of tissue engineering with MSCs (155) for the treatment of OA. Lifestyle changes have also been shown to prevent and potentially reduce the occurrence of OA (177). OA of the hip is the leading cause of the hip replacements in the elderly (178) and with projections of an ageing population worldwide, the incidence of hip replacement in only expected to increase. Thus to investigate the factors in ageing that may be aggravated in OA, samples of hip OA patients were investigated in this thesis.

This project first aimed at investigating changes in number, gene and surface marker expression of BM MSCs in healthy ageing. The same parameters would then be examined in MSCs from donors with hip OA to appreciate any trends in healthy ageing that would be further aggravated in OA.

1.4 Hypotheses and objectives

Hypotheses:

1) There is an age-related decline in the number of BM MSCs in humans.

- There are age-related differences in the levels of expression of multipotentiality genes, genes associated with senescence and IFN1 pathway in uncultured old donor MSCs compared to young donor MSCs.
- 3) These age-related differences are further aggravated in uncultured MSCs of patients with primary OA, a major age-related musculoskeletal condition.

Objectives:

- To quantify the number of BM MSCs across the broad age range (19-89 years old) and between young (19-40 years old), intermediate group (41-59 years old) and old (60-89 years old) donors, using the CFU-F assay and the CD45^{low}CD271⁺ phenotype.
- To investigate the level of expression multipotentiality genes, genes associated with senescence and IFN1 pathway in uncultured BM MSCs (CD45^{low}CD271⁺) from young (19-40 years old) and old (60-89 years old) donors.
- To investigate the level of expression of multipotentiality genes, genes associated with senescence and IFN1 pathway in uncultured BM MSCs (CD45^{low}CD271⁺) in patients with primary hip OA.

Chapter 2 Materials and methods

The methods described in this chapter were performed in experiments presented in more than one results chapter. The methods specific to a result chapter have been described in the respective methods section of that chapter. List of reagents, consumables, equipments, softwares used, prepared solutions are listed in Appendix 2 Table 1, 2, 3, 4, 5 respectively.

2.1 Age groups distribution

The term ageing or defining at what age someone can be called old is more than often, subjective. While many references focussed on ageing refer to 'old' as the age of 60 and above (Figure 1.1) globally (5), skeletal ageing become noticeable around the fifth decade of life (62, 96). In both cases, there are individuals, whose biological age (defined as how young their bodies are, how physically active their organs are) do not match their chronological age (actual age in years) (179). Thus to examine age-related changes in BM MSCs in human adults, in this project, three age groups were investigated: young (19-40 years old); intermediate age group (41-59 years old) and old (60-89 years old).

In order to investigate and understand differences in old donors relative to young, this PhD included donors beginning from adulthood (>18 years old). Paediatric donors (<18 years old) were excluded from this study as they are in early stages of growth which includes elongation of bones, sexual development and reaching skeletal maturity (180). Considering this, an obvious difference would be expected in factors like the number of MSCs and gene expression of osteogenic differentiations in old (>60 years old) donors when compared to young, eventually leading to a bias. While an adult over the age of 18 is considered to have attained skeletal maturity, the age of 40 has been suggested to start displaying a decline in BMD (62). Incidence of fractures and bone associated conditions become prevalent around the age of 60 and thus 60 and above was chosen as the cut-off for old donors (1, 181).

Based on the evidence presented above, it can be proposed that in order to understand the process of ageing in BM MSCs that is relevant to their *in vivo* biological ageing, MSCs from human donors must be used prior to any culture expansion. This will avoid the changes in the cells induced artificially through passaging and cultivation-induced replicative senescence. When cultivation cannot be avoided (such as in CFU-F assay), it should be reduced to a minimum. Based on these considerations, uncultured

40

or minimally-cultured MSCs will be used in this thesis for the purpose of investigating age-related changes in human BM MSCs *in vivo*.

2.2 Ethical approval

Ethical approval for the collection of human tissue samples was obtained from Yorkshire and Humberside National Research Ethics Committee (ethics reference 06/Q1206/127 for bone marrow aspirate (BMA) collection, 14/YH/0087 for femoral head (FH) collection and 04/Q1206/107 for the collection of blood from healthy volunteers (Appendix 5). Participants' details and sample distribution across different study arms are presented in separate tables outlined in Appendix 1.

2.3 Donor selection and sample processing

The age range for the three groups were as follows: 19-40 years old were referred to as the young age group, 41-59 years old were indicated as donors of the intermediate age, and 60-89 years old were referred to as the old age group. The ages of recruited OA donors were between 56-83 years old.

Human BM aspirates (BMA) were collected in 4ml Ethylenediaminetetraacetic acid (EDTA) tubes from the iliac crest (IC) of patients undergoing surgeries for fracture fixation or for removal of metal work in Leeds General Infirmary, Major Trauma Unit. The patients were reported as otherwise healthy. BMA was collected from n=58 donors; 30 males and 28 females with the age range of 19 to 89 years old, median age of 47.5 years old. BMA median volume was 8 ml (range 4-24ml).

In the laboratory, BMA was first passed through a sterile nylon mesh 70μ m cell strainer to exclude small clots, if any. The total aspirate volume was recorded and collected in a 50ml sterile falcon tube. Enumeration of MSCs was performed using CFU-F assay explained in Chapter 3 (Appendix 1, Table 1) and flow cytometry (Appendix 1, Table 2) on fresh samples (n=51 and n=32, respectively) followed by experiment for testing resistance to oxidative stress (Appendix 1 Table 2) the next day. The remaining sample was then processed for freezing (section 2.3) for testing MSC proliferation in the presence of human serum (Appendix 1, Table 1) explained in Chapter 3 or for use in cell sorting followed by gene expression (Appendix 1, Tables 2 and 3).

FHs from OA donors (Appendix 1, Table 4) was collected, processed and frozen by Dragos Ilas according to the published methods (182, 183). In brief, frozen cells from

the femoral head were defrosted, stained and then processed for cell sorting as explained in section 2.3 and 2.5 below.

2.4 Cell freezing and defrosting

For freezing of the nucleated cells (NCs), BMA was treated with ammonium chloride (NH₄Cl) solution (BMA: NH₄Cl=1:8, v/v) for 10 minutes at room temperature (RT) for the lysis of erythrocytes and then centrifuged at 650g for 5 minutes at RT to pellet NCs. The supernatant was discarded and the cell pellet was washed twice with PBS to remove any remaining traces of NH₄Cl. The pellet was then re-suspended in 5-10ml of Dulbecco's minimum essential medium (DMEM) supplemented with 10% Fetal calf serum (FCS).

The number of NCs were counted using a haemocytometer by mixing 10μ I of cell suspension with 10μ I of trypan blue dye. At least 2 large squares with 16 grids each were counted and the average count was multiplied by 10^4 , the dilution factor for trypan blue and the total volume of cell suspension (for example, $56\times10^4 \times 2$ (dilution factor for trypan blue) x 5mI (total volume of suspension)). The cells were centrifuged again at 650g for 5 minutes at RT. The supernatant was discarded and the cells were resuspended in the required volume of freezing medium (FM) (Appendix 2, Table 5). 10- 20×10^6 cells were frozen in each cryovial in 1mI of FM, labelled and first stored in -80°C freezer for at least 24 hours followed by transfer into liquid nitrogen for future use.

To revive frozen cells for future experiments, vials were defrosted in water bath (37°C) and the cell suspension was added to 15ml falcon tubes. Thawing medium (TM) was prepared by adding 20μ l DNAse to 20ml of complete medium to prevent accumulation of cells into clumps due to released DNA from damaged cells. 5ml TM was added for every ml of defrosted cell suspension drop by drop. Cells were pelleted at 600g for 5 minutes at RT to remove DMSO and were slowly re-suspended in TM up to desired volume. For any study using these cells, if a clump was observed at any given point of time, the cell suspension was filtered using 70 μ m cell strainer to obtain a homogenous cell suspension.

2.5 MSC Enrichment using Anti-Fibroblast microbeads

MSCs are a rare population in the BMA (87, 108) and the MSC enrichment step is needed prior to fluorescent activated cell sorting (FACS) to pre-enrich MSC thus saving time and cost of cell sorting. Enrichment of MSCs was performed by three different methods in the first year of this project including Low-affinity nerve growth factor or (LNGFR) beads and RosetteSep cocktail, and owing to no differences in the proportions and absolute numbers of enriched MSCs obtained, the most feasible method of using Anti-Fibroblast (A-Fib) microbeads was chosen for future experiments. A-Fib beads are magnetically conjugated to anti-D7-Fib antibody allowing MSCs as D7-Fib positive cells (108) to be retained in the column during separation and hence isolating them from D7-Fib negative cells.

For this, defrosted NCs were pelleted at 650g for 5 minutes at RT and were resuspended in 500 μ l to 1ml of MACS buffer (Appendix 2, Table 5) to achieve average concentration of 1x10⁷ cells/ml as per the manufacturer's instructions. 40 μ l of A-Fib beads per 10⁷ cells was added to the cell suspension, mixed well and incubated at RT for 30 minutes. The cell-bead suspension was next washed with 1-2ml of MACS buffer and centrifuged at 650g for 5 minutes at RT. The supernatant was carefully removed and the cells were re-suspended in1.5-2ml of MACS buffer. The cell suspension was passed through a cell strainer to ensure that there were no clumps.

MACS column was positioned in multi-stand with the MACS Separators attached, these are powerful magnets and MACS columns amplify the magnetic field by nearly 10,000 folds. MACS buffer alone was first run through the column to prime the column before adding the cell suspension carefully. The fraction of cells retained in the column included the magnetically labelled cells while the cells leaving the column (negative fraction) did not contain the magnetic beads. The column was washed with 2-4ml of MACS buffer to wash out residual negative cells and the negative fraction was collected in a 15ml sterile tube.

Once the negative fraction had been collected, the column was removed from the magnetic field and placed on top of a separate 15ml sterile tube. 2-3 ml of MACS buffer was added into the column and the bead-labelled cells (positive fraction) were collected by firmly pushing the plunger into the column.

2.6 Flow cytometry and Fluorescence activated cell sorting (FACS)

2.6.1 **Principle of flow cytometry**

Flow cytometry is the technique that uses fluid dynamics (flow) to study and quantify or measure (metry) the different properties of cells (cyto) within a heterogeneous cell population (184). It has five key components that include the sample, fluidics, optics, detectors and output. The core of the instrument is the fluidic system consisting of the

flow chamber with sheath fluid and pressurised lines, both of which are responsible for the transportation of the particles, or in this thesis, cells (185). Excitation optics include lenses which help to direct and focus the other segment of the excitation optics, the lasers. The lasers produce light using high voltage electricity. The point at which the stream of cells interacts with the laser is known as the interrogation point. It is also the point where the optics detect the light scattering (186). At the interrogation point, the light emitted by cells is collected by the collection lens and passed through dichoric mirrors (mirrors with separate reflection/transmission property at two different wavelengths) and filters that ensure the wavelengths of specific lights to be detected by their respective detectors.

The light that travels along the path of the laser is in the forward direction and is assembled by collecting lens as 'forwards scatter' or FSC. FSC is indicative of the cell size or surface area and is the result of diffraction of light. The light then travels to the photodiodes which is responsible for the conversion of light into electronic signals to be recorded by the computer system (185). When light comes in contact with cells and refracts or reflects and is collected at 90 degrees to the laser, the scattering pattern is called 'side scatter' or SSC. SSC is indicative of granularity and the internal complexity of cells. (185). The light signals captured by both photo diodes and photomultiplier are converted to electronic currents which travel to the amplifier to be converted into voltage pulse. This analogue measure is further amplified by linear and logarithmic amplifiers, converted into digital signals by digital converters which may then be used to produce histograms or plots for data representation (185).

2.6.2 FACS

This is where the use of fluorescent labelling comes in, which combines the functionality of the flow cytometers with fluorescent labelled antibodies used to stain cells. These florescence labelled antibodies can bind to specific cell types and help in distinguishing different types of cells from a heterogeneous mixture of cells. The distance between the absorption and emission maxima (peak) in the spectra of a fluorochrome is known as 'Stoke's shift'. The higher the Stoke's shift, the greater is the separation between the absorbance and the emission of a given fluorochrome (187).

The total photons being absorbed by a fluorochrome are associated with the excitation wavelength. Thus, while FITC is known to absorb light at 400-550 nm, its highest absorbance is usually at 490nm at which it absorbs more photons ensuring that the emission will be more intense (187). Using the fluorochromes and the principle of dynamic light scattering, some flow cytometers can segregate labelled cells from a

heterogeneous cell population. This is referred to as cell sorting. When fluorochrome labelled cells pass the laser beam, they are excited by the correct wavelength of the laser light. After excitation, the emitted light is directed towards emission filters allowing for the detection of multiple fluorochromes emitted by a cell.

The fluorochromes used are usually for cell surface staining or for intracellular staining. Selection of the appropriate fluorochromes for a given study is essential and the panel of fluorochromes to be used is dependent upon the aims of the study. In this thesis, different fluorochromes were used to identify and enumerate MSCs using the CD45^{low}CD271⁺ phenotype (Table 2.1). The same phenotype formed the base of cell sorting into two separate populations of MSCs (CD45^{low}CD271⁺) and HLCs (CD45⁺CD271⁻). Studies investigating levels of potential age-related surface markers and for detecting oxidative stress in young and old donor uncultured BM MSCs were performed using fluorescent antibodies and dyes listed in table 2.1.

2.6.3 Sample preparation and cell staining for cell sorting

Vial containing FH digest was defrosted in TM, cells were counted and distributed in 10 tubes to be used for cell sorter instrument settings (8 tubes) and cell sorting (2 tubes). Owing to significantly smaller numbers of MSCs in BMA compared to FH digests (182), FH cells were used for setting of the machine for unstained cells, isotype controls and spectral compensation. These procedures were performed by the technical staff of the Flow Cytometry and Imaging Facility in St.James's University Hospital. Cell sorting was usually performed using two BMA samples (from one young and one old donor) on a single day, in addition to a FH sample.

For sorting, the enriched cells from BM samples were counted and distributed equally into 2 tubes, one to be stained with CD271 PEVio770, CD45 V450, CD106 PE and CD295 APC combination and another one for CD271 PEVio770, CD45 V450, CD146 PE and Cx43 APC (Table 2.1). Violet laser (405 nm excitation) was used for the V450 fluorochrome, blue laser (488 nm excitation) was used for PE, PECy7 and PEVio770 and PerCP fluorochromes and the red laser (633 or 640 nm excitation) was used for APC as indicated below in table 2.1.

Average cell numbers for isotype controls were $3x10^5$ while for tubes with cells for sorting, contained up to $3x10^6$ cells. Both, FH and BM cells were centrifuged to form a pellet. The supernatant was discarded and the cells were re-suspended in 100μ l of FACS buffer. The cells were then treated with Fc block for 10minutes to prevent non-specific binding.

The cells were subsequently incubated with the antibodies in dark for 15 minutes. After staining, the cells were washed in FACS buffer, re-suspended in 500 μ l of the same buffer and passed through the filtered-cap flow cytometry tubes to remove the presence of any cellular clumps. Finally, 10 μ l 7-aminoactinomycin D (7-AAD) was added to all tubes containing cells (except unstained) to distinguish between live and dead cells and run through the FACS machine. The list of antibodies and their volumes, isotype control antibodies and their volumes used is summarized in Table 2.1.

Experiment	Cells	Antibodies /dye	Instrument	Excitation (nm)	Laser used	Emission (nm)	Isotype control	Vol (µl)
Enumeration*	MSCs	CD271 APC		650	633	660	lgG1	20
	MSCs	CD45 PECy7	LSR II	496	488	785	lgG1	20
	MSCs	7-AAD PerCP		482	488	647	NA	10
ROS production*	MSCs	CD271 PEVio770		488	488	775	lgG1	20
	MSCs	CD45 V450	LSR II	404	405	448	lgG1	20
	MSCs	CellROX FITC		508	488	525	NA	2
	MSCs	Sytox APC		640	633	658	NA	2
FACS*, [#]	MSCs and HLCs	CD271 PEVio770		488	488	775	lgG1	20
	MSCs and HLCs	CD45 V450		404	405	448	lgG1	20
	MSCs and HLCs	CD106 PE	BD Cell sorter	565	488	575	lgG1	20
	MSCs and HLCs	CD295 APC		650	640	660	lgG2a	10
	MSCs and HLCs	CD146 PE		565	488	575	lgG1	10
	MSCs and HLCs	Cx43 APC		650	640	660	lgG2b	10
	MSCs and HLCs	7-AAD PerCP		482	488	647	NA	10

Table 2.1 List of antibodies, isotype controls and their volumes used for various flow cytometry investigations

MSC: CD45^{low}CD271⁺ cells, HLC: CD45⁺CD271⁻ cells, *data presented in Chapter 4, [#]data presented in Chapter 6

2.6.4 FACS and sample collection

Once instrument settings were adjusted, stained samples were sorted to collect CD45^{low}CD271⁺ cells (MSCs) and CD45⁺CD271^{low} cells (HLCs) which would be the control population for the population of interest (MSCs) (188). While sorting, data was simultaneously acquired for the selected surface markers of interest – CD106/Vascular Cell Adhesion Molecule (VCAM-1), CD295/Leptin receptor (LepR), CD146/ Melanoma Cell Adhesion Molecule (MCAM) and Connexin43/ gap junction- (Cx43/GJA-1). Sorted cells were collected into 1.5ml eppendorf tubes containing 350µl Buffer RL (Appendix 2, Table 1) that has guanidinium salts for lysing cells and exposing the genomic contents (189). The salts also ensure the inhibition of RNAse to prevent the degradation of RNA into smaller components. The eppendorfs were frozen at -80°C until further use.

2.6.5 FACS data analysis

Data from FACS was analysed using FlowJo software (version 10). The gating strategy and the histograms for each of the surface markers are shown in the respective chapters. First, the raw files were loaded on to the workspace and each tube within a given experiment was analysed in detail. Dot plot display was chosen to better visualise the cells of interest. 7-AAD negative cells were gated as live cells. Using this live cell gate, the cells with the phenotypes CD45^{low}CD271⁺ (MSC) and CD45⁺CD271^{low} (HLC) were identified. On each of these cell phenotypes, the expression of surface markers (CD106, CD295, Cx43 and CD146) was analysed using respective histogram displays. MFI was recorded for each of the surface markers for MSCs and HLCs for all samples. The MFI values were then compared and analysed for any differences in relation to donor age or the presence of OA, as described in Chapter 4 and Chapter 6, respectively.

2.7 Gene expression

Genes are the fundamental unit of heredity and information about a person at the genetic level. Measurement of the expression of genes have long been used to identify differentially expressed genes in diseases or other anomalies. To understand if there are any differentially expressed genes in old donor MSCs as compared to young donor MSCs, or in OA, gene expression experiments were carried out using sorted CD45^{low}CD271⁺ MSCs and CD45^{high}CD271⁻ HLCs as the control.

2.7.1 Principle

Polymerase chain reaction (PCR) may be defined as a chain reaction that occurs to increase the final product exponentially with the help of an enzyme polymerase. Thus the reaction enables the production of billions of copies of DNA from a single copy of DNA. The essential components needed for the production of billions of copies from a single reaction include the enzyme, primers, DNA template and nucleotides. Initially, the technique was developed for DNA but was soon applied to RNA as well. However, the technique is qualitative, consumes time for the preparation of gels using hazardous chemicals, is limited in terms of number of samples and allows the detection of the amplified product only after the end of all the cycles.

Quantitative PCR (qPCR) technique is built on the same basic principles of PCR but has several advantages. Not only is the method quantitative but the reaction can be quantified at every step of amplification. This is achieved by the use of fluorescence labelled probes (in this thesis, TaqMan probes). The fluorescent labelled probes have a fluorophore that is not active in the beginning of the reaction as it is attached with a quencher. As the reaction proceeds, the fluorochrome on the 5' end and the quencher on the 3' end of the TaqMan assay (in this thesis FAM-MGB respectively) are separated by taq DNA polymerase. This separation activates the fluorochrome which is now fluorescent ensuring the amplification dependent increase in the fluorescence is tracked and measured by the amplification curve. The sample preparation does not involve the assembly of gels using hazardous materials and allows a large number of sample reactions at the same time (generally up to 386 samples). These factors make qPCR a preferred technique to quantify gene expression over normal PCR.

2.7.2 Fluidigm and Integrated fluid circuits (IFCs)

Due to the population of interest MSCs being a rare population, it was vital to use a method for gene expression that would allow high quality and consistent outputs even with input of sample with low cDNA concentration. Gene expression analysis in sorted MSCs and HLCs were performed using integrated fluid circuits (IFC) chip technology which is based on the principles of microfluidics. These devices are based on the principle of qPCR but allow further miniaturization and integration of liquid handling components on a single chip. Such microfluidic devices utilise technology that combines networks of channels (fluid lines of extremely small dimension (10⁻⁶m)), chambers (holding extremely small volumes of liquids (10⁻⁹l)) and Nanoflex valves (that can withstand very high pressure (175mmHg)) in a single chip (190). IFC chips allow reactions with small quantities of cDNA and TaqMan primers carrying out numerous PCR reactions (for example up to >9000 reactions for IFC96*96) depending on the size

49

of the chip, all at once. These chips are aimed at high throughput data collection and analysis of biological samples with low concentration of starting materials.

Two types of IFC chips were used in this thesis based on the number of genes investigated. The Flex Six^{TM} chip (Section 4.2) was used for performing gene expression in Chapter 4 which investigated 12 genes, owing to the 12 segments in each of the 6 compartments of the chip. For investigation of genes in Chapter 5, 2 of 48.48 chips (Section 5.2) were used to examine 96 genes. At one time in one 48.48 chip, 2,304 reactions were carried out. The difference between the two chips used lies in the number of assays and samples that may be run and the final volume of reaction mixtures. The concept, principle and application remains the same. A simple flowchart (Figure 2.1) shows the steps that were followed from the sample preparation to the gene expression data generation, irrespective of the chip used.



Figure 2.1 Flow-diagram indicating the steps followed for obtaining gene expression data throughout the thesis

2.7.3 Sample preparation

As indicated in section 2.3, frozen NCs from young and old donor cohorts (Appendix 1, Tables 2 and 3) were defrosted and then subjected to cell sorting based on CD45^{low}CD271⁺ and CD45⁺CD271⁻ phenotypes for MSCs and HLCs, respectively. HLC fraction included BMA cells of the hematopoietic lineage and served as a control to study specificity of the selected panel of genes to MSCs. Cell counts were recorded for

each population fraction. The cells were collected in the RL buffer (lysis buffer) in an eppendorf and the cell lysates were then kept at -80 until further use.

2.7.4 RNA extraction

To minimise the degradation of RNA, the working bench was wiped with RNAse away. RNA extraction was performed using Norgen single cells RNA isolation kit (Appendix 2, Table 1) which uses the column technique for better yield owing to the low number of cells in the fraction of interest (range: 647–22,229 with a median of 4,694 CD45^{low}CD271⁺ MSC). The technique essentially utilises spin column chromatography using resins as the separation matrix which ensures the purification of RNA without the use of phenol or chloroform (189).

Frozen cell lysates were defrosted on ice and 200μ l of 100% ethanol was added to each lysate for every 350μ l of the sample volume. The mix of cell lysate and ethanol was then gently mixed by vortexing for 10 seconds. The mix was added to a column contained inside a collection tube and centrifuged for 2 minutes at a speed 3,500g at 15° C. This ensures the binding of the RNA to the resin column while proteins are eliminated in the flow-through liquid (189). The flow-through was discarded and the column was washed with 400μ l of wash solution by centrifugation for 1 minute at 14,000g to ensure the removal of any remaining impurities.

Genomic DNA digestion ensures the removal of any DNA contamination that was not eliminated in the previous steps and could lead to false positives and lower detection sensitivity in samples. It was performed by adding 100μ l of DNAse I digestion mix (DNAse I and DNAse I reaction buffer) to the column containing the RNA and centrifuged for 1 minute at 14,000g ensuring the entire volume had passed through the column. The flow-through was added back onto the column and the column was incubated for 15 minutes at RT. Following incubation, the column was washed twice with 400μ l of wash solution by centrifuging for 1 minute at maximum speed. The flow-through was discarded and the column was centrifuged twice to thoroughly dry the resin.

The dry column was placed in a new elution tube and 12µl of elution solution was carefully added (drop-wise) to the centre of the column and incubated for 2-3 minutes to allow the hydration of RNA attached to the column. The column was centrifuged for 1 minute at 200g followed by 1 minute at 14,000g to ensure that the elution solution had passed through pulling the purified RNA along with it. The column was then discarded and the concentration of the eluted RNA was measured using Nanodrop.

Nanodrop measures the amount of a variety of molecules including nucleic acids based on the principle that different molecules can absorb lights at different wavelengths (spectrophotometry). Proteins were measured at 280nm, nucleic acids (sample of interest) at 260nm and organic compounds at 230mn. The eluted RNA was quantified by recording the generated concentration of the RNA in ng/µl by the spectrophotometer. The purity of the RNA samples was checked using the 260/230 ratio indicative of presence of impurities belonging to organic compounds for values below 2 and the 260/280 ratio indicative of protein impurities in the RNA sample below 1.8. The extracted and quantified RNA was then placed in -80°C until further use.

2.7.5 Reverse transcription (Rt) for cDNA preparation

Rt is needed to convert single stranded RNA to cDNA. The Rt component mix was prepared by mixing 14 μ l of Rt master mix and 28 μ l of nuclease free water. Rt master mix contains all the components required to prepare cDNA which include deoxyribonucleotide triphosphates (dNTPs), mix of oligo-dTs and random primers, RNAse inhibitor and reverse transcriptase along with buffers. The dNTPs act as the building blocks for the new strand being built, the primers direct the synthesis of the strand, RNAse inhibitor prevents the degradation of RNA in the process and reverse transcriptase is the enzyme that is needed for Rt for cDNA preparation. 3 μ l of the Rt mix was added to each well in a 96-well plate followed by 2 μ l RNA sample to each well making the total volume 5 μ l. The plate was sealed, mixed by vortexing, centrifuged and placed in the thermocycler for the reverse transcription using following the settings: 5 minutes at 25°C, 30 minutes at 42°C, 5 minutes at 85°C and then hold at 4°C, for approximately 50 minutes. The produced cDNA (5 μ l) was then stored at -20°C for future use.

2.7.6 Pre-amplification

As BM MSCs are a rare population, the number of cells (CD45^{low}CD271⁺) obtained after cell sorting were relatively low (section 2.6.4) and so was the concentration of the RNA from these cells (5.6-32.5 ng/µl with a median of 11.55ng/µl). Limited quantities of RNA and cDNA often restrict gene expression analysis, ultimately generating no results (191). Thus to ensure detectable levels of starting material (cDNA), pre-amplification (PA) step was needed. This is basically a qPCR with limited cycles before the actual qPCR begins (192). The limited number of cycles (usually 10-18) ensure the amplification of cDNA without significant bias. PA allows for amplification for up to 96 target genes and a pool of primers is prepared form the same gene expression assays that are to be used for the qPCR (191). PA is an established method and is increasingly being used for similar work (193).

TaqMan assays used in different study arms are indicated in respective chapters. For PA, pooled TaqMan assay mix was prepared by mixing 2μ l of each TaqMan assay (indicated in respective chapters) and the final volume was made up to 200µl using nuclease-free water. Next, 1µl of PA master mix was added to 1.25µl of the prepared pooled TaqMan assay, 1.5µl of nuclease-free water and 1.25µl of cDNA prepared (section 2.6.5). Total volume of 5µl was transferred to individual wells in a 96-well plate in a pre-determined sequence. The plate was carefully sealed and the samples were pre-amplified on the thermocycler using 18-cycle protocol (2 minutes at 95°C, 15 seconds at 95°C, 4 minutes at 60°C and then hold at 4°C). The sample were then prepared for qPCR in Fluidigm. 18-cycle PA is the recommended number of cycles for samples with a concentration range of 5-20ng of RNA (193). The PA samples were diluted with 15µl Tris-EDTA (TE) buffer and then loaded on the IFC chip for qPCR on the same day.

2.7.7 Sample run

qPCR was performed using Fluidigm Flexsix[™] IFC and 48.48 IFC. Flex Six[™] chip has 6 independent compartments. Each compartment has 12 assay and 12 sample inlets. Each compartment can be run independently as a separate experimental run or simultaneously. Before the first run on the chip, the complete volume (150µl) control line fluid was carefully added to the accumulator region only, pushing past the spring for the purpose of priming the chip. The 48.48 chip is a single-use chip and needs to primed using the control fluid just as in the Flex Six[™] chip. For both the chips during priming, the barrier plugs were kept on and the chip was placed in their respective priming instrument and the prime script was run for 15 minutes.

After priming, the chip was removed from the machine and the barrier plugs of the compartments intended for use were removed. For the assays, 2μ I of each TaqMan assay and 2μ I of assay loading buffer were mixed and 3μ I from this mix was added to the 12 wells in the assay compartment in a pre-determined sequence. Similarly, the sample mix was prepared by adding 28μ I of TaqMan universal master mix, 2.8μ I of sample loading buffer and 1.8μ I of pre-amplified cDNA, and 3μ I of the sample mix was added to the sample side of the compartment. The volumes used for 48.48 chip are indicated in Chapter 5, section 5.2.

2.7.8 Data collection, analysis and presentation

The qPCR run was performed on the BioMark[™] HD System with the 'Data collection' software and GE Flex Six[™] Standard cycling protocol (35-40 cycles) for approximately 60 minutes cycles. Owing to the large number of reactions that can occur simultaneously within extremely small volume (10⁻⁹ I) in a single run, it surpasses the other quantitative assays for qPCR (193). After completion of the run, IFC was removed from the Biomark and the post run script was run for 5 minutes on IFC Controller HX to relax the valves. This allows the unused compartments to be used in further experiments within a span of 3 months. 'Biomark Real-Time PCR Analysis' software was used for the analysis of the data. First, the cycle threshold (Ct values) were generated for each gene using the software for analysis and the data was exported to Microsoft excel for further calculations. The data were then normalised with hypoxanthine-guanine phosphoribosyl transferase 1 (HPRT1) as the housekeeping gene to generate the ΔCt values for each gene using the formula [Ct target gene - Ct housekeeping gene]. Finally, the ΔCt values were converted to 'relative expression' for each gene using the formula [2^{-ΔCt}] and the values were compared between young and old donor cohorts, or in OA cells and age-matched controls, in both MSC and HLC populations.

Cluster and Treeview softwares were used to generate clusters of gene expression for the samples and observe any trends between the expression of genes in CD45^{low}CD271⁺ MSCs and CD45^{high}CD271⁻ HLCs. The softwares were also used to get an overview of gene expression data distribution across the entire donor cohort. Grey squares in clusters represent data that was below detection. Comparative investigations between young and old donors or old donors and OA patients were analysed using GraphPad Prism software (version7.0a).

2.8 Data presentation and Statistical analysis

Various data representation techniques were used for displaying the results obtained. Cluster analysis was used to observe any differences in gene expression across the entire donor cohorts. Dot plots were used to indicate individual values, frequency distribution curves were used to display the data frequency across a range. Bar graphs were used to indicate the results of a subgroup, histograms were used to indicate MFI generated using flow cytometry.

Statistical analysis and graphs preparation were performed using GraphPad Prism software. The distribution of the data was assessed using the Shapiro-Wilk and Kolmogorov-Smirnov tests for normality. As the data were found to be non-parametric,

54

and unpaired, Mann-Whitney U test was used to compare data between young and old donor groups, or between MSC and HLC fractions. The tests used for each experiment are specified in the respective chapters. For all data, *p* value of <0.05 was considered significant (**p*<0.05, ***p*<0.01, ****p*<0.001 and *****p*<0.0001).

Chapter 3 Proliferative capacity of human BM MSC from using colony the forming unit-fibroblast (CFU-F) assay

3.1 Introduction

As stated in the general introduction (section 1.2.1), BM aspirate from the iliac crest (IC) is a reliable source for MSCs but they exist as a rare population with a frequency between 0.01-0.001% of nucleated cells (87, 108). CFU-F assay has been the oldest method for counting the number of MSCs ever since its discovery by Friedenstein and colleagues (194). MSCs are enumerated by counting the number of single cell derived colonies that are formed on petri dishes based on first, their ability of adhering to plastic and second, their ability to proliferate into a complete colony. Each colony is representative of a single MSC and it takes up to two weeks before the colonies can be visualised. In spite of it being a time consuming method, it is still performed today and remains the most feasible way to enumerate MSCs based on plastic adhesion. Flow cytometry based methods and phenotypes used for enumeration of MSCs will be discussed in Chapter 4. The plastic adherent cells that are not proliferative, either do not form a colony or will form a colony with very few cells. To ensure that a single MSC derived colony that was counted as a colony was a true representation of its proliferative capacity, a group of at least 50 cells was considered a colony in this project (195). Thus, each colony from the CFU-F assay represented a single MSC.

3.1.1 Previous studies investigating age-related changes in number of human BM MSCs based on CFU-F assay

Several groups have performed CFU-F experiments to examine the age-related changes in number of MSCs. These experiments have yielded inconclusive results with some of them showing a decline (96, 114, 132) while some not showing any age-related difference (113, 196, 197) in the number of colonies. This can be due to the lack of uniformity in laboratory protocols used for the CFU-F assay. Table 3.1 lists investigations in the number of MSCs as observed in cultured cells and in uncultured cells. From Table 3.1, it can be seen that many of the studies for investigating the age-related changes in number of MSCs using CFU-F assay have been performed on cultured cells. When grown in cultures, *in vitro* ageing sets in within the MSCs (98, 114), as stated in section 1.2.2, which is not representative of uncultured cells *in vivo* (100). Thus, the actual biological age-related difference in number of native MSCs requires further elucidation.

MSC source	Age (years)	Isolation	Media	Colony definition	CFU-F	Ref
Trabecular BM	14-87; no groups	PA	α-MEM+ 10%FCS	> 1mm in diameter	No change	(113)
BM from iliac crest	Y: 22-44; O: 66-74	DC	α-MEM+ 10%FCS	>16 cells	No change	(197)
BM from iliac crest	Y: 0-18; O: 59-75	DC	DMEM+ 10%FCS	ND	Decline	(114)
BM from iliac crest	Y: 19-40; O: >40	DC	DMEM+ 10%FCS	>50 cells	Decline	(96)
BM	Y: 6m-16y; O: 29-76	PA	α-MEM+ 20%FCS	ND	NS decline	(198)
BMA	1-52; no groups	DC	DMEM+ 20%FCS	>50 cells	No change	(199)

Table 3.1 List of previous studies investigating age-related change in the number of BM MSCs counted by CFU-F assay

BM: Bone marrow, BMA: Bone marrow aspirate, O: Old, Y: Young, PA: Plastic adhesion, DC: Density centrifugation α-MEM: Alpha – Minimum essential medium, DMEM: Dulbecco's minimum essential medium, FCS: Foetal calf serum, ND: Not defined, NS: Non significant

When uncultured cells have been used, different methods in extraction of BMA, further processing of the aspirate for MSC isolation, and using different media, culture conditions as well as colony definition and scoring criteria can all explain the variation in the results (Table 3.1).

3.1.2 Previous studies investigating age-related changes in colony area and density in human BM MSCs based on CFU-F assay

While CFU-F assay has commonly been used to count MSCs, few studies have also analysed the colony area and fewer studies have evaluated the colony density as a measurement of proliferative capacity of MSCs. Ultimately, the number of studies correlating colony area and density to donor age are sparse (listed below in Table 3.2). Colony area measures how big a colony is and larger colony areas have been associated with younger donor samples (132). However, the variations in the methods adapted in independent laboratories and donor variation, make it difficult to come to a consensus with respect to age-related changes in colony area.

Even though colony area measures how big the colony is, it does not take into account the density of cells within a colony. The denser a colony, the more the number of cells that will be closely packed within a colony, indicating higher proliferative rates of colony initiating MSCs. As per my knowledge, there has been only one study that has mentioned colony density in relation to donor age in uncultured BM MSCs (Table 3.2). The study used uncultured cells that observed no change in colony density (200) but they did not present the data for the same.

Quantification of colony size has either been performed by comparing diameters of the colonies or by comparing the area of colonies in millimetre square (mm²) or related units (132, 201). Quantification of density of a colony is relatively more challenging considering the variation in sizes of the colonies, the patterns of cells in which they may overlap making it difficult to calculate the density. This is potentially why, only 2 studies have mentioned density of colonies in their studies and only one of them reported qualitative observations (Table 3.2). This indicates that a study of BM MSC colony area and density, using a large cohort of donors, is needed.

Table 3.2 List of previous studies investigating age-related changes in colony area and density of human BM MSCs

MSC source	Age (years)	Cultured	Colony size criteria	Change in colony area	Change in colony density	Ref
Trabecular BM	14-87; no groups	UC	NR	Decline	NC (no data)	(113)
Vertebral spine BM	Y: 3-36; O: 41-70	UC (cadavers)	Y-10.23mm ² ; O- 8.23 mm ²	Decline	NR	(132)
BM from iliac crest	Y: 22-44; O: 66-74	UC	Y- 7.2mm ² ; O- 6.1 mm ²	NC	NR	(197)

BM: Bone marrow, Y: Young, O: Old, UC: uncultured, NR: Not reported, NC: No change

3.1.3 Significance of microenvironment in MSC proliferation

As mentioned previously (in section 3.1.2) CFU-F assay can not only directly measure the number of MSCs from a donor, but it also gives information about the proliferative capacity of these MSCs. MSCs from BMA *in vivo* exist within a complex matrix composed of other types of cells and various other soluble and extracellular matrix factors forming its own micro-environment (Figure 1.3). Stenderup et al. investigated the *in vivo* bone forming capacity of human MSCs obtained from 5 young (24-30 years old) and 5 old (71-81 years old) donors in immuno-deficient mice. Interestingly, they found that the bone forming capacity was maintained even in the MSCs obtained from old donors (202). This could be due to the fact that mouse microenvironment was able to boost the function of human MSCs from old donors.

Microenvironment includes various factors that guide the cells to their fates. Factors like extracellular matrix (ECM), other non-cellular components like soluble proteins, growth factors present in the serum have all known to influence the direction of adult stem cells. Presence of growth factors, conditioned media from older donors (139) as well as serum from older donors (203) have been shown to have an inhibitory effect on culture of MSCs as compared to the effect of growth factors, conditioned media and serum from young donors. These investigations suggested that the microenvironment of old donors also contribute to age-related effects in the cells of older donors. These components play an important role in MSC fate decisions with respect to proliferation, differentiation, migration as well as molecular interactions with other cells. However, the investigation of how these factors influence the area or density of an MSC colony

and thus the proliferative capacity of MSCs within the CFU-F assay, remains unexplored.

Autologous serum flows from blood vessels to the BM where MSCs reside, therefore *in vivo* MSCs from older donors are continuously exposed to 'old donor' serum and vice versa for young donors. Keeping this in mind, several studies have investigated the effect of autologous serum on the proliferative capacity of MSCs, but this has been performed only on culture expanded MSCs (203, 204). Furthermore, such studies have been performed for various aims primarily, finding a human alternative to foetal calf serum for supporting MSC expansion, differentiation and proliferation (205, 206). Only limited studies have used human serum aimed at investigating age-related changes (203), and none in CFU-F assays from uncultured BM MSCs.

This chapter aimed at investigating the age-related changes in the number of BM MSCs using the CFU-F assay. The same assay would be used to investigate the proliferative capacity of BM MSCs from young and old donors by further dissecting the CFU-F assay. The proliferative capacities of BM MSCs from young and old donors would also be investigated when exposed to young and old serum to understand the role of microenvironment (serum) in MSc proliferation.

3.1.4 Hypotheses and objectives

Hypotheses:

- 1. The number of MSCs measured by CFU-F assay declines with donor age.
- 2. The proliferative capacity of MSCs measured by colony area and density declines with donor age.
- 3. The proliferative capacity of MSCs measured by colony area and density is lower when exposed to serum from older donors.
- 4. Subjecting old donor MSCs to media supplemented with young donor serum will increase their proliferative capacity.

Objectives:

- 1. To enumerate BM MSCs by CFU-F assay.
- 2. To investigate the proliferative capacity of MSCs using CFU-F assay.
- 3. To test the proliferative capacity of MSCs grown in media supplemented with young and old human serum.
- 4. To compare the proliferative capacity of old donor MSCs grown in young and old human serum.

3.2 Methods

3.2.1 Donor selection

BMA was collected from donors undergoing surgeries for fracture fixation or for removal of metal work in Leeds General Infirmary, Major Trauma Unit in 4ml EDTA tubes as outlined in section 2.2.2. Donors were reported as otherwise healthy and were aged between 19-89 years old (median age=48) with n=27 males and n=24 females with a total of n=51 donors.

3.2.2 Enumeration of MSCs using CFU-F assay

CFU-F assay was performed as previously optimised in our laboratory (111, 207). To perform CFU-F assay, 80µl of freshly collected and filtered BMA was added into duplicate 60mm petri dishes containing 4ml of StemMACS (SM) MSC expansion medium. It was later adapted to 200µl of BMA in duplicate 100mm petri dishes containing 10ml of SM medium to obtain more colonies per dish. Following BMA and SM medium mixing, dishes were placed in the incubator in 37°C and 5%CO₂ and maintained in culture for two weeks. After 48 hours, the dishes were gently washed twice with 10ml of Phosphate buffer saline (PBS) using sterile stripette and 10ml of fresh SM medium was added. Half medium change was performed for these dishes twice a week and the dishes were checked for the growth of colonies under the inverted microscope. On the 14th day, the medium was aspirated completely and the cells were fixed for an hour using 10ml of 3.7% formaldehyde prepared previously (Appendix 2). The dishes were then stained with methylene blue stain for one hour and washed carefully to visualize the colonies. The stained dishes were allowed to air dry, then scanned using the Infinity scanner at 1200dpi resolution. Images of the cells within the colonies were captured using microscope equipped with the Infinity Analyze software. A colony was defined as consisting of at least 50 cells (207). The colonies in each duplicate dishes were manually counted (Figure 3.1) and CFU-F/ml was calculated according to the formula ((1000µl x average number of colonies from 2 dishes)/200µl) and then averaged.

Figure 3.1 illustrates duplicate CFU-F dishes with colonies grown from BM aspirates from representative donors from each age group as outlined in section 1.2.3. ImageJ software was used to investigate the colony area and density which are together indicative of the proliferative capacity of MSCs (201, 208).



Figure 3.1 Duplicate dishes containing colonies from each age group.

The dishes in the figure illustrate colonies after staining with methylene blue on day 14. The numbers on the top right of every panel indicate the number of colonies counted in each dish.

3.2.3 Colony area analysis

To investigate age-related differences in MSC colony area, dishes were chosen from the young and old donor groups. Initially, CFU-F dishes from 10 young and 10 old donors were selected for analysis using ImageJ. Visually, colonies of various sizes were observed in CFU-F dishes from both young and old donors (Figure 3.1), and there was no clear evidence for separate groups of large or small colonies, as categorised by Gothard et al. (201).

The images of scanned dishes were converted to 'grey-scale' to be recognized by ImageJ software and then measurements for colony area was performed. By default, ImageJ measures area in (pixel)². The measurement was then calibrated for millimeter² (mm²) and the software generated an outline for each of the colonies. To avoid the influence of the background disturbances and to minimize software error (for example, by counting two close colonies as one), the colonies on the edges of the dish and two or more colonies that almost merged as one, were given an outline manually. This enabled the measurement of colony areas for each individual colony. Details about the samples, on which colony area analysis was performed can be found in Table 1, Appendix 1.

Upon initial analysis of colony area, it was observed that the data for colony area was not normally distributed (Figure 3.2). In some donors, dishes contained numerous colonies including very large colonies (for example; donors: 31/F and 62/M). It was decided that selection of CFU-F dishes with low number of colonies would potentially miss out on these rare large colonies leading to a biased analysis of colony areas. Thus, dishes with a minimum of 25 colonies were chosen for the final analysis of colony area and density (Figure 3.2).



Figure 3.2 Initial studies to evaluate difference in colony area in 10 young and old donors.

Dots indicate individual colonies and horizontal lines indicate median values from young donors (top panel) and old donors (bottom panel). Age/Sex of the donors is indicated within brackets.

The final colony area studies were performed on dishes from n=7 young and n=7 old donors that had generated 25-60 colonies per dish. The results were analysed for young and old donor groups using different approaches as discussed in section 3.3.2.

3.2.4 Colony integrated density (ID) analysis

The data for colony ID was generated along with the data for colony area for all colonies from the same 7 young and 7 old donors, for which the colony area had been calculated. The ID was measured by the software (area * mean grey value) for each of the colonies. The value of ID was always generated by the software, irrespective of the outline for areas done automatically or manually. The values for colony area and ID were generated simultaneously. Once the data for colony area and ID were recorded, the data were compared between young and old donor groups.

3.2.5 Serum collection and preparation of medium with human serum

To generate human serum, 30 ml of blood was collected in 6 ml serum clot activator tubes from 4 young and 4 old healthy donors after obtaining their consent. The tubes contain silica micro particles coated along the inside of the tubes that activate clotting of blood. Once the blood was collected, the tubes were agitated for 5 seconds to allow the homogenous mixing of the blood with the coated silica micro particles. The mix was allowed to stand for 30 minutes and then centrifuged at 2000g for 15 minutes. This allowed the separation of the serum from the cells which was seen as a clear bilayer. The serum from the upper segment of the separated mix was collected carefully to avoid extraction of cells. However, if a small quantity of cells was extracted due to the proximity of the two liquids at the separating junction, the mix was further centrifuged in 1.5ml eppendorfs to ensure complete separation of serum from the cells. This was done for blood from each donor. Subsequently sera from four young donors (equal volumes) were pooled together to generate stocks of 'young' serum (YS). Similarly, 'old' serum (OS) was also prepared and both YS and OS were then aliquoted in 1ml vials and frozen at -20°. Media containing either 10%YS or 10%OS were prepared fresh using DMEM as the basal media and supplemented with 1% Penicillin-Streptomycin for each experiment. Optimisation experiments were performed using serum from a single young donor for the ease of availability of young healthy donors.

3.2.6 Experimental design for investigating MSC colony formation in human serum

The aim of these experiments was to investigate how young and old donor MSCs responded to young and old donor microenvironment (represented by YS and OS), in comparison to SM, as well as how their colony numbers, areas and densities were affected when grown in 'old' human serum (OS) compared to 'young' human serum

(YS). These experiments were performed using frozen cells (for freezing and thawing, refer to section 2.2) from 6 and 4 donors from 'young' and 'old' groups, respectively (Appendix 1, Table 1).

3.2.6.1 Medium optimisation

To test the effect of 'young' and 'old' human serum on MSCs, it was important to ensure the right controls for the experiments. The first medium chosen as a positive control was SM medium that is industrially optimised for isolation, adhesion and growth of MSCs from BMA. However, the serum in SM is foetal calf serum (FCS) but it does not specify the type of basal media used.

As the aim of this experiment was to test the effect of human serum on the colonyforming capacity of BM MSCs, it was important to have a positive control with human serum supplement added to a known basal media. Mesencult supplement or Mcult is a human serum supplement that is optimized for isolation of MSCs *in vitro*. Thus, it was chosen as another positive control added at a concentration of 10% in basal Dulbecco's minimum essential medium (DMEM). DMEM with standard FCS (not optimised for MSCs growth) was used as negative control. All the media were supplemented with respective serum and 1% penicillin/streptomycin antibiotic.

3.2.6.2 Experimental design for control media evaluation

Frozen NCs (predicted to generate approximately 25-30 colonies based on the initial CFU-F assays, (Figure 3.2) were defrosted at 37°C in thawing medium (TM, Appendix 2). The cell suspension was centrifuged to remove the DMSO from the suspension and was re-suspended in TM. The cells were then equally divided into 12 60-mm dishes (3 for each of the medium of choice, Figure 3.3) containing 4 ml SM for an initial 48-hour attachment period.

Subsequently, the media was removed, the non-adherent cells were washed with PBS and complete media changes were performed as follows: 3 dishes were supplemented with 4 ml SM (control), 3 other dishes with 4 ml DMEM/10%YS, 3 other dishes with DMEM+10%Mcult and 3 final dishes with 4ml of DMEM/10%FCS. The dishes were then placed at 37°C and 5%CO₂ for 2 weeks, and half medium changes were performed using the respective media twice a week. On day 14, all media were aspirated and for each condition, two dishes were stained with methylene blue to visualise the colonies, while the adherent cells from the third dish were lifted off using trypsin (Appendix 2) and counted. The stained dishes were allowed to air dry and then analysed for colony area and density, as described in the above section.The steps

65

involved in selection of appropriate donors and the criteria in testing the effect of MSC colony formation in human serum are shown in a flowchart format in Figure 3.3 followed by the representation of medium change for all conditions in Figure 3.4.



Figure 3.3 Flowchart demonstrating the steps and crucial parameters involved in the selection of donors for testing MSC colony formation in human serum.

Steps have been written as a flow chart to simplify the complexity of factors that were involved in executing the plan of the experiment with human serum. NCs: nucleated cells, TM: thawing medium, SM: StemMACS media, DMEM: Dulbecco's minimum essential medium, FCS: Foetal calf serum, MCult: Mesencult supplement, YS: young serum

NCs seeded to get 25-30 colonies for 48 hours in StemMACS for adhesion



Medium was changed as indicated after 48 hours



Figure 3.4 Optimisation of controls for testing the proliferation of MSCs in human serum

Equal number of defrosted MSCs were plated in 60mm petri dishes for 48 hours in SM medium to ensure uniform adhesion of MSCs in all dishes. After 48 hours, SM medium was aspirated and the corresponding medium was added (shown in the coloured panel). Half medium change was peformed twice in a week and on day 14, 2 dishes from each medium condition were stained and one was trypsinised for obtaining cell counts.

3.2.6.3 Final experimental design

The final experimental design included pooled YS and OS from n=4 donors from each group. As the optimisation experiments were performed from a single young donor, serum for final experiments were pooled to avoid bias. Observation from results from optimisation experiments (discussed further and shown in Figure 3.10) lead to finalisation of 3 serum conditions for the experiments shown in Figure 3.5. This included the use of SM, DMEM+10%YS (pooled) and DMEM+10%OS (pooled).

NCs seeded to get up to 25-30 colonies for 48 hours in SM for adhesion



Medium was changed as indicated after 48 hours



Figure 3.5 Optimised conditions and medium controls for testing MSC colonyforming capacity in human serum.

3.2.7 Statistical tests

Statistical analysis and graphics were performed using GraphPad Prism software (version 7.0a). The normal distribution of the data was assessed using the Shapiro-Wilk and Kolmogorov-Smirnov tests for normality. As no data were found to be normally distributed, Mann-Whitney test and Kruskal-Wallis test with Dunn's correction for multiple comparisons were used to compare two and three groups, respectively. Spearman test was used to analyse correlations. The results were considered significant at *p* value of <0.05.

3.3 Results

3.3.1 Age-related changes in number of CFU-F

As mentioned in section 3.2.2, the colonies were visualised after staining the cells with methylene blue solution on day 14. Initial observation included scanning the petri dishes for images and manual counting of the colonies. There was no age-related pattern observed in terms of colony size in general when colonies were counted

indicating different types of colonies existed even within the duplicate dishes of a single donor (Figure 3.1).

Once all the colonies were visualised and counted, the data for n=51 donors (n=27 males, n=24 females), was compiled and then correlated with donor age (Figure 3.6A). There was a significant age-related decline (p=0.0153) in the number of colonies. While there were donors with a relatively higher number of colonies among the older donors, an overall decline in the number of colonies (indicative of MSCs) was evident from the figure.



Figure 3.6 Age-related changes in number of CFU-F

(A) Age-related change in the number of CFU-F per ml of BMA across entire donor age range (n=51) and (B) in between age groups. Each dot indicates individual donor, black dots represent male donors and empty circles represent female donors. The black line on the left indicates the slope and the on the right, the median values. Spearman non-parametric test was performed for A and Kruskal-wallis test with Dunn's correction was performed for B, *p<0.05.

When the number of colonies was segregated based on the donor age groups (Figure 3.6B), a decline of over 3-folds (p=0.0245) was found between young donors (median=163 CFU-F/ml) and donors of intermediate age group (median=53 CFU-F/ml). When young and old donors (median=38 CFU-F/ml) were compared, a decline of over 4-folds (p=0.0512) was observed. It must be noted that while the overall median was the lowest in the older donors, the decline was the steepest between young donors and donors of intermediate age group. This indicted towards the considerable decline in the number of MSCs in the donors who are over the age of 40. The decline in the number of colonies continued after 60s as well, however, some of the donors over the age of 60 were also observed to have relatively large number of CFU-Fs suggesting young biological age in terms of MSCs despite belonging to the old chronological age group.

To understand if there were age-related changes with respect to gender, male and female donors were split and analysed separately for age-related correlations (Appendix 3, supplementary Figure 3.1A and C). There was a significant decline found in males (p=0.0319; r=-0.41) and a non-significant decline in females (p=0.128; r=-0.32). To understand the variation in each gender, they were further split into three age groups (Appendix 3, supplementary Figure 3.1B and D). The trend for decline in males and females was found to be similar to 3.6A above. However, owing to the limited number of donors in old males and intermediate group females, it was decided to keep the focus on age-related changes without segregating data for males and females further in the thesis.

3.3.2 Age-related changes in colony area

Until now, data from this chapter has established that there is significant decline in the number of colonies (indicative of number of MSCs) with increased donor age. While counting colonies and analysing them, colonies of different sizes were observed (Figure 3.7A). From literature, it was known that each colony is indicative of a single MSC and that colony area and colony morphology has the potential to indicate the proliferative capacity of MSCs (201, 208). However, none of these studies had measured the colony area on uncultured MSCs in relation to age ranging from 20-89 years old or quantified colony density. While the general idea related to the decline in sizes of MSC colonies has been linked with lower MSC proliferation in aged donors (132, 144), investigations on age-related differences in colony density have not been performed.

CFU-F dishes from 14 donors (n=7 young and n=7 old) containing a minimum of 25 colonies each were scanned. The scanned images of the dishes were analysed for colony area as described in section 3.2.3 and all colonies from each age group were combined to increase the statistical power of the analysis. The comparison of colony area between all colonies from young donors (n=321) and all colonies from old donors (n=266) suggested that the area of colonies in old donors (median=10.3 mm²) was significantly 10% lower (p=0.0289) than the area of colonies in young donors (median=11.2 mm²) (Figure 3.7 B). However, it was not clear if the decline in the colony area was due to a predominant loss of the larger colonies or due to a shift from large to small colonies in old donors.



Figure 3.7 Age-related changes in colony area

(A) Presence of different sizes of colonies from a single donor CFU-F dish (B) Comparison of colony areas between all colonies from 7 young (321 colonies) and 7 old (266 colonies) donors. Each dot indicates a single colony with median values indicated in horizontal lines (C) Frequency distribution of colony area of young donors (black curve) and old donors (grey curve). *p<0.05, Mann-Whitney U test.

To investigate if there is a loss of bigger colonies in old donors, a frequency distribution curve was generated for the colony areas of both young and old donor MSCs (Figure 3.7C). The range of colonies was set in gaps (referred to as 'bin centres' in GraphPad

Prism) of 5mm² in order to analyse the differences in sizes as small as 5mm². The frequency distribution of colony areas was next performed using GraphPad Prism software by generating a non-linear fit line based on modal values of pre-defined bin widths of 5 mm². This analysis revealed unimodal colony area distributions in both young and old cohorts, however, a prominent shift towards smaller colonies was observed in the old group. The frequency distribution in the figure shows a tendency of young donors to have larger colony areas whereas, the data from old donors indicate a shift towards smaller colonies with colony areas less than 10 mm². To segregate the colonies as 'large' or 'small', the median of all 587 colonies from both groups of donors (median=10.64 mm²) was used as the cut off limit.

3.3.3 Age-related changes in colony integrated density

Colony density data was collected along with the colony area by ImageJ software and was measured in integrated density (ID) units. It was calculated as a product of the area and the mean grey value of a colony. The high mean grey value reflects the closeness of the cells that are stained within the colony, while zero grey value indicates un-stained empty space, that is, space with no cells. The greater the contact between the cells within a colony, the greater would be the mean grey value of the colony.

It was hypothesised that the colonies with the highest IDs would be the biggest and the most dense (indicating the most highly proliferative MSCs) categorised by Oreffo and colleagues in 2013 as 'large dense colonies' (201). Conversely, the colonies with the lowest IDs would be equivalent to the 'small sparse' colonies (208), indicating the least proliferative MSCs. The first comparison was performed between the ID of all colonies from young donors (n=321) and all colonies from old donors (n=266) (Figure 3.8A). The comparison indicated a significantly 32% lower (p<0.0001) ID of colonies in old donors (median=1116 units) as compared to the ID of colonies in young donors (median=1643 units). The next comparison was performed between all the small colonies (area less than median area of 10.64 mm² described in the section above) from both young and old donors with all the large colonies from both young and old donors (Figure 3.8B). The data indicated a significant drop (p < 0.0001) of over 3-folds in the ID of small colonies (median=777 units) when compared to the ID of large colonies (median=2661 units). As the formula for ID was (area*mean grey value), both, the contact between the cells in a colony and the area of the colony are directly proportional to the actual ID calculated. The fact that the larger colonies generated higher values for ID than small colonies, it was therefore expected and indicated that such segregation may provide further insights on the proliferation potentials of MSCs from Y and O donors.

72


Figure 3.8 Age-related changes in colony ID

(A) Comparison of colony ID of all colonies from young and old donors (B) Comparison of all small and large colonies from both age groups based on median area (C) Comparison of ID of small colonies in young and old donors (D) Comparison of ID of large colonies in young and old donors. Black dots represent young donors (n=7) and grey dots indicate old donors (n=7). Median values in all the figures is represented by the black horizontal line.**p<0.01 and ****p<0.0001, Mann-Whitney U test

To further understand if the ID of each colony type (large or small) had similar agerelated trends, the colony ID data for small colonies was first compared between young and old donors (Figure 3.8C). The ID of small colonies from old donors (median=563 units) was significantly lower (p<0.0001) than the ID of colonies from young donors (median=981 units). Similar comparison was applied to large colonies for the two age groups (Figure 3.8D) and a similar trend of lower ID (p=0.0029) was observed in old donors (median=2325 units) as compared to the ID of large colonies in young donors (median=2850 units). The difference in ID was however greater in the small colonies (1.7-folds) as compared to large colonies (1.2-folds). This suggested that there were proportionally more small sparse colonies in old donors, rather than less large dense colonies. Irrespective of the colony size, if they were from young donors, they had higher colony area and ID indicating their higher potential to proliferate as compared to colonies from old donors.

3.3.4 Proliferation of MSCs in human serum conditions

3.3.4.1 Optimisation experiments

The first step of testing the hypothesis that the proliferative capacity of MSCs is further reduced in 'aged' microenvironment was to ensure that the right controls were being used for comparison. Section 3.2.6 outlines the choices of medium used and the rationale behind these choices. To identify the best control(s), young donors with more than 100 CFU-Fs/ml from fresh BMA and with over 10⁷ frozen cells (Figure 3.3) were selected to generate approximately 25-30 colonies in 12 dishes each (3 dishes for each of the four different media). Young serum (YS - from 1 young male) was used to optimise the controls as it was more practicable and also because the volume of old serum (OS) was limited owing to rarity of the donors in this age group and so it was saved for the main experiment. Once the young donor cells were added to the dishes and the medium was changed as shown in Figure 3.4, after 14 days, dishes were stained and the number of colonies were counted.

Figure 3.9 shows the results of the optimisation experiment where the dishes containing cells in DMEM+10%FCS on top panel and in DMEM+10%Mcult (bottom panel) did not have visible colonies. While there were groups or clusters of cells seen in both – DMEM+10%FCS and DMEM+10%Mcult, they had less than 50 cells to be considered as a colony for colony analysis. This was repeated with two other donors and the observation for both DMEM with FCS or Mcult was consistent throughout the three experiments. The cell count in dishes containing either DMEM with FCS or with Mcult also consistently revealed fewer or no cells at all (range 0-2 colonies per dish as compared to 30-60 colonies per dish in SM).



Figure 3.9 Selection of control media

Top and bottom panel (outlined in red) medium conditions did not generate colonies resulting in their elimiation and finalising SM as the positive control. Numbers on the top right corner of every dish indicate the number of colonies in each dish.

Colonies were expected in SM as it is commercially optimised for the adhesion, growth and proliferation of cultured and uncultured MSCs *in vitro* and was shown to generate colonies in previous experiments (section 3.2.1). Number of colonies counted in SM were greater than those in DMEM+10%YS. However, colonies in the latter appeared larger in size as compared to the colonies grown in SM. Mcult is also optimised for adhesion, growth and proliferation of MSCs but only for cultured cells. This could explain the absence of colonies in this experiment with uncultured cells. DMEM+10%FCS has previously been used to grow MSCs and compare with human serum (206). However, the lot used for this experiment was not specifically selected for uncultured MSC growth. This can explain the lack of colonies from uncultured BMA.

Based on these observations, only SM was chosen as the positive control for subsequent experiments. It can be argued that DMEM+10%FCS could have been still kept as negative control. However, keeping in mind that the number of MSCs declined in older donors (section 3.3.1) and the number of frozen cells required to set up these experiments (9 dishes x 25-60 colonies per dish), it was decided to finalise the experimental set up on 3 sets of conditions: human young serum, human old serum and SM and not perform negative controls.

3.3.4.2 Proliferation of MSCs in media supplemented with human serum

The aims of following experiments were to test if there was a decline in the most proliferative MSCs when grown in media containing old serum and also to evaluate the rejuvenation potential of growing old donor MSCs in young serum as compared to growing in old serum.

As the number of old donors was limited and so were their number of MSCs in most cases, the old donors with 30 CFU-Fs/ml and above were considered for this experiment. Additionally, all of the frozen cells from the old donors were plated as described in section 3.2.5. Altogether, the colony formation from NCs from n=6 young donors and n=9 old donors was evaluated. As expected, seeding old samples that had less than 100 CFU-F/ml and less than 10⁷ frozen NCs did not yield any colonies and thus could not be considered for further analysis. The results of the donors that generated colonies have been presented and discussed in detail in the sections below.



Figure 3.10 Representative donors colonies in all 3 conditions

Top panel shows the scanned image of the stained dishes following culture in SM medium followed by media supplemented with 10%YS in the centre panel and in media supplemented with 10%OS in the bottom panel. Left side indicate representative dishes from a young donor and the right side indicate dishes from an old donor. The numbers on the top right corner of the dishes indicate the number of colonies in each dish.

Figure 3.10 (left side) shows colonies obtained from young donor following culture in SM, media supplemented with 10%YS and media supplemented with 10%OS. In SM (top panel), 36 and 34 colonies were observed. Comparing these colonies with those obtained in media containing human serum (10%YS or 10%OS), they were fewer in number as well as appeared smaller in size. Colonies from young donor MSCs grown in media supplemented with YS appeared merged due to larger colonies in close proximity. Colonies of varying sizes were observed in all 3 conditions.

Figure 3.10 (right side) shows colonies obtained from old donor following culture in SM, media supplemented with 10%YS and media supplemented with 10%OS. In SM (top panel), 36 and 61 colonies were observed. Comparing these colonies with those obtained in media containing human serum (10%YS or 10%OS), they appeared smaller in size. However, colonies in SM were more than the number of colonies obtained from human serum, especially in media supplemented with OS (bottom panel). Colonies grown in media supplemented with YS appeared larger than colonies obtained from other 2 conditions. Colonies of varying sizes were observed in all 3 conditions.

Visually, Figure 3.10 suggests that the colonies are generally bigger in size in media containing human serum as compared to SM. While the difference between the number of colonies or their sizes was not strikingly evident between YS and OS, the general observation of larger colonies seen by the naked eye remained constant for all young donors (n=6) in YS as compared to SM (Figure 3.10).

Using SM as the positive control, all the data for colony area from young and old donor MSCs grown in media supplemented with 10%YS or 10%OS were compared to SM. To confirm if the colonies grown in media supplemented with YS were actually denser than those grown in SM or in media supplemented with OS, the ID of all the colonies were also quantified. Due to the observation that SM generated good number of colonies but smaller in size and media in YS demonstrated otherwise, one of the 3 dishes from each condition was trypsinised and counted to complement the ID data.



Figure 3.11 Colony and cell counts of young and old donor MSCs seeded in dishes with SM, media supplemented with YS and media supplemented with OS

(A) Number of CFU-Fs as counted from stained dishes in young donors (left) and old donors (right) grown in SM, media supplemented with YS and media supplemented with OS (B) Cell counts from the third dish in young donors (left) and old donors (right) grown in SM, media supplemented with YS and media supplemented with OS. Bars indicate median and error bars indicate interquartile range. n=6 young donors and n=4 old donors.

Figure 3.11 represents the colony-forming capacity of young and old donor MSCs in SM, media supplemented with YS and media supplemented with OS. The data in Figure 3.11A shows no significant difference in the number of colonies between young and old donors grown in any of the 3 conditions. However, while the total number of colonies remained similar in SM, a non-significant decline in old donor MSCs was observed in both, media supplemented with YS and media supplemented with OS human serum conditions.

The cells counted from the third dish (Figure 3.11 B) showed interesting additional trends. First, a trend for decline in the number of cells in media supplemented with OS was observed in both young and old donor MSCs compared to their growth in SM and YS conditions, in spite of plating equal number of cells in all conditions for each donor. This suggested that there were colonies with fewer cells in OS conditions when compared to SM and YS conditions. This was further tested by evaluating the colony areas and IDs using methods described in section 3.2.3 and 3.2.4.

Figure 3.12 shows the comparison between colony areas of young and old donor MSCs in the 3 conditions, along with the frequency distribution of the colony areas. Figure 3.12 A, left panel represents the colony areas of all colonies from young donors grown in SM, media supplemented with YS and in media supplemented with OS. For young donors, YS conditions had produced the largest colonies with median area of 11.89mm². This was significantly higher than observed for the colonies grown in OS conditions (median area=11.08mm², *p*=0.0062) or SM conditions (median area=8.42mm², *p*<0.0001). Interestingly, the colonies grown in OS were also significantly larger than those grown in SM (*p*=0.0016).

Frequency distribution of colonies from young donor MSCs (Figure 3.12 A, right panel) grown in SM (dashed line) peaked in between 5-10 and 10-15 mm² area boundaries indicating that most frequent colonies in SM were more within this boundary. Frequency distributions of colonies from young donors MSCs grown in media supplemented with YS or OS did not show much difference, consistent with their median values being very close, as mentioned above. Both the curves appeared to have similar range of large and small colonies indicated by the nearly overlapping pattern of the black and grey line. As mentioned, the frequency distribution curve of the colonies from the same donors grown in SM appeared to have a strong shift towards the colonies of smaller size.



Figure 3.12 The analysis of colony area in SM, media supplemented with YS and media supplemented with OS

(A) Comparison of colony areas in young donors (n=6, left) in SM (n=345 colonies), in media supplemented with YS (n=286 colonies) and in media supplemented with OS (n=301 colonies). Frequency distribution of colony area (right) based on bin centres of 5 mm² of colonies from young donors grown in SM (dashed line), media supplemented with YS (solid line) and media supplemented with OS (dotted line) (B) Comparison of colony areas in old donors (n=4, left) in SM (n=241 colonies), in media supplemented with YS (n=175 colonies) and in media supplemented with OS (n=165 colonies). Frequency distribution of colony area (right) based on bin centres of 5 mm² of colonies from young donors grown in SM (dashed line), media supplemented with YS (n=165 colonies). Frequency distribution of colony area (right) based on bin centres of 5 mm² of colonies from young donors grown in SM (dashed line), media supplemented with YS (solid line) and media supplemented with OS (dotted line). Median is indicated by black horizontal line. Data was considered significant when **p*<0.05, ***p*<0.01, *****p*<0.0001, Kruskal-Wallis test with Dunn's correction.

Figure 3.12 B represents the colony areas of all colonies from old donors grown in SM, media supplemented with YS and in media supplemented with OS. For old donors too, YS conditions had produced the largest colonies with median area of 11.19 mm². This was significantly higher than for the colonies grown in OS conditions (median area=8.38 mm², p=0.0141). While there was no difference found between the colony area of colonies grown in SM (median area=10.14 mm²) and colonies grown in media supplemented with YS, median area of colonies in SM was significantly larger than the

colonies grown in media supplemented with OS (p=0.0215). Thus, in old donor MSCs, colonies grown in OS showed the lowest colony areas.

Frequency distribution of colonies from old donors in SM (dashed line) once again peaked in near 10-15 mm² area indicating that most of the colonies in SM were relatively small in size. However, very different from results in young donors, colonies from old donors in media supplemented with OS shifted further towards the left indicating the presence of colonies smaller in size than those grown in SM (Figure 3.12B, right panel). As expected, colonies grown in media supplemented with YS demonstrated a prominent shift to the right (larger colonies), almost as seen in young donors (Figure 3.12A, right panel).

Overall, this section showed that in young donor MSCs, colonies grown in SM are significantly smaller in size, while those grown in media supplemented with YS or OS do not show any differences in size indicating that OS does not have a detrimental effect on the growth of young donor MSCs. However, old donor MSCs grown in media supplemented with OS generated the colonies of the smallest size. Very interestingly, the same old donor MSCs, when grown in media supplemented with YS generally showed a shift towards larger colonies in MSCs from both young and old donors. These data indicated that YS supplementation had a positive effect on the colony size from old donor MSCs.

3.3.4.3 Age-related differences in colony ID in MSCs grown in SM, media supplemented with YS and media supplemented with OS

Next, to understand the age-related differences in density of the colonies, the IDs of the colonies were then analysed for MSCs grown in all the 3 conditions. Colonies from young donors and old donors were segregated as small or large based on the median area of all colonies (colonies grown in media supplemented with YS and with OS) from young as well as old donors (9.4 mm²). Colony area evaluation in human serum suggested an age-related decline when MSCs were exposed to OS and this was observed for young donor MSCs as well as for old donor MSCs. Next, the data acquired for colony ID from ImageJ was analysed for small and large colonies, first in young donors and then in old donors. It was hypothesised that a similar trend for decline in colony ID would be seen in OS conditions, and particularly for old donor MSCs All

colonies were segregated into small and large based on the median area of all colonies (grown in all conditions and from young as well as old donors, 9.4 mm²).

Figure 3.13A represents colony IDs for small colonies from young donors grown in SM, media supplemented with YS and media supplemented with OS. Small colonies grown in media supplemented with YS had the highest IDs (median=1110 units) which was significantly higher than the IDs of colonies grown in SM (median=937 units, p=0.0208) or media supplemented with OS (median=887 units, p=0.0003). Even though the IDs of colonies grown in media supplemented with OS were the lowest, it was not statistically significantly lower than IDs of colonies in SM.



Figure 3.13 ID of colonies from young donors MSCs grown in SM, media supplemented with YS and media supplemented with OS

(A) ID of small colonies in SM (n= 156 colonies), in media supplemented with YS (n=108 colonies) and in media supplemented with OS (n=129 colonies) (B) ID of large colonies in SM (n=153 colonies) in media supplemented with YS (n=191 colonies) and in media supplemented with OS (n=172 colonies). Each dot represents a single colony and the median is indicated by black horizontal line across the data set. *p<0.05, **p<0.01, ***p<0.001 and ****p<0.0001, , Kruskal-Wallis test with Dunn's correction.

Similar trend was observed for the large colonies from young donors (Figure 3.13 B) showing a significantly higher ID in colonies grown in YS (median=2598 units) as compared to the colonies grown in SM (median=2047 units, p=0.0080) or in media supplemented with OS (median=2200 units, p<0.0001). Altogether, the data presented in Figure 3.13 indicated that in young donors, OS conditions induced a significant decline in the ID of colonies as compared to those grown in SM and YS. This was true for both, small as well as large colonies. This implied that proliferation of MSCs was the lowest when grown in media supplemented with VS. Using colony ID as an indication of MSC proliferation, it can be concluded that growing MSCs in OS conditions reduced MSC proliferation capacity, and this was observed for both, large and small areas.



Figure 3.14 ID of colonies from old donors MSCs grown in SM, media supplemented with YS and media supplemented with OS

(A) ID of small colonies in SM (n=103 colonies), in media supplemented with YS (n=76 colonies) and in media supplemented with OS (n=98 colonies). (B) ID of large colonies in SM (n=138 colonies) in media supplemented with YS (n=99 colonies) and in media supplemented with OS (n=67 colonies). Each dot represents a single colony and the median is indicated by black horizontal line across the data set. p<0.05^{*}, , Kruskal-Wallis test with Dunn's correction.

Small colonies from old donors did not show any significant difference between the ID of colonies grown in SM, YS or OS (Figure 3.14A). However, the IDs remained lower in OS conditions (median=1062 units) compared to YS conditions (median=1105 units) while being the lowest in SM (median=937 units).

For the large colonies grown from old donor MSCs (Figure 3.14B), the IDs of colonies grown in media supplemented with YS (median=2729 units) were significantly higher than the IDs of colonies grown in SM (median=2441 units p=0.0170) and non-significantly higher that the IDs of colonies grown in OS (median=2501 units). This indicated that, while media supplemented with YS encouraged the proliferation of MSCs from old donors, the effects were small and statistical significance was not achieved. ID analysis of colonies from both young and old donors confirmed that growing MSCs in aged serum induced a decline in the overall proliferation capacity of MSCs (Figure 3.13 and 3.14). The data also indicated that proliferation capacity of MSCs in old donors could be slightly improved when grown in media supplemented with YS as compared OS (Figure 3.14B). The trends for a change were the same for large and small colonies suggesting that YS or OS factors were equally affecting all colony types and were therefore not specific to small or large colonies.

3.4 Discussion

Literature regarding age-related differences in BM MSCs with respect to their numbers has been controversial (Table 3.1). Furthermore, while the number of CFU-Fs have been evaluated for counting MSCs, the size of the colonies has been discussed less often and their density, rarely. In this thesis, it was hypothesised that the number of MSCs in the BM shows an age-related decline. Additionally, the colony size and colony density measured were also hypothesized to show an age-related decline with the loss of the most proliferative MSCs that in this assay, formed the largest and densest colonies. Furthermore, it was hypothesised that MSC 'niche', modelled here with the use of human serum, would additionally impact on MSC colony formation, leading to smaller and less dense colonies when grown in old donor serum.

The data presented in this chapter showed a significant age-related decline in the number of MSCs quantified by CFU-F assay, confirming our hypothesis and indicating an age-related decline in BM MSCs *in vivo*. Furthermore, the MSCs from older donors had significantly lower proliferative capacity than younger donor MSCs as indicated by colony area and colony ID analysis. The potential of human serum to support proliferation of MSCs was clearly demonstrated in the presented experiments with

human serum and confirmed previous studies with culture expanded MSCs (206, 209). The colony area and colony ID results from human serum experiments indicated that 'aged' serum (OS) provided conditions for lower proliferative capacity of MSCs, from both young and old donors. Finally, colony area and ID results indicated a possibility for using young serum to potentially rejuvenate MSCs from old donors, by making CFU-F colonies significantly bigger and slightly more dense.

CFU-F assay has been performed by many groups to investigate the change in the numbers of MSCs. However, the techniques of cell isolation, media used and other factors have varied in different laboratories (Table 3.1). Fewer groups (including our group) have performed the CFU-F assay on fresh BMA to detect the age-related changes in number of MSCs. The results of the current study indicated a significant age-related decline in the number of MSCs (Figure 3.6A), a 4-fold decline in MSC numbers between the young donors (163 colonies/ml BMA) and old donors (38 colonies/ml BMA) using CFU-F assay (Figure 3.6B). These data in principle support previous data by Stolzing et al. (96) where they have shown a significant 4-fold decline in MSC numbers between paediatric (0-20 years old, median of 100 colonies/ml) and adult donors (21-40 years old, median of 25 colonies/ml) by CFU-F assay. However, the groups used in these two studies were different, spanning an age range of 5-55 years old in the Stolzing et al., study and 19-89 years old in the present thesis. The numerical difference in the number of colonies in the common 40-60 years old age group in both the studies could be due to the fact that Stolzing et al., plated 5 x 10^4 cells/well for CFU-F assay while fresh BMA was used for the current study. Even though the trend of age-related decline in the number of MSCs is common in both the studies, there is notable difference in the age range of donor age used, CFU-F plating method and in the number of donors recruited in the studies. Stolzing et. al showed a prominent drop in MSCs after the completion of skeletal growth which is also observed in the present study.

D'Ippolito et al., performed similar studies and found a significant decline in the number of colonies (66.2 ± 9.6 per 10^6 cells) in young (3-36 years old) donors as compared to old (41-70 years old) donors (14.7 ± 2.6 per 10^6 cells) (132). However, they used vertebral BM and counted CFU-ALP (alkaline-phosphatase positive), which could explain the variation in the number of colonies. Muschler et al., performed age correlations studies on iliac crest BMA and also counted CFU-ALP, and found a significant age-related decline in the number of CFU-ALP for both the genders. They also found a decline in the number of CFU-ALP per 10^6 NCs, sharper in females from 13 to 27 in female age groups as compared to 17 to 38 in the male age groups (210).

Using a smaller cohort of patients, similar trend was recently observed by our group with a significant decline in the number of colonies in females as compared to a non-significant decline in males, across the whole age range between 22 and 80 years old (111). Both these studies were in contrast to the gender related data obtained in this project. However, owing to the lower number of female donors in the young and intermediate groups and lower number of males in the old donor groups (Appendix 3, supplementary Figure 3.1), no firm conclusions can be made in terms of age-related declines in the number of MSCs in separate genders. An equal number of donors from both the genders in each age group (minimum n=10) will be required for a firmer conclusion.

The present results indicating a decline in the number of MSCs is in contrast to work done previously by Justessen et al., where they found no significant difference in the number of MSCs in normal conditions with respect to both, age and gender (196). They found a significant difference in the MSC colony number with age only in those treated with osteogenic medium (4+/-4 colonies of seeded 100,000cells/cm² NCs for young donors aged 18-42 and 2+/-4 of seeded NCs for old donors aged 66-78). This study also used density centrifugation for erythrocyte removal, scored colonies with 16 or more cells in healthy donors and in osteoporosis patients, in contrast to this study where NH₄Cl was used for erythrocyte lysis, colonies with only more than 50 cells were scored and only healthy donors were recruited.

Jing Li and team used uncultured cells from BM of donors aged 1-52 years old and found no age-related decline in the number of colonies. In contrast, this study analysed uncultured BM MSCs from adults ranging from 19-89 years old and categorised donors in 3 age groups. It used direct BM plating technique which prevented the loss of cells that occurs in centrifugation. It also used a media that was standardised for growing uncultured BM MSCs along with ensuring that a colony always had >50 cells. Overall, in this study, a significant age-related decline was observed in the number of colonies obtained from uncultured BM MSCs with advancing age and a 4-fold decline in old donors (>60 years old) when compared to young donors (<40 years old). Importantly, the decline in the number of MSCs was notable from the age of 40 years old and not 60 years old as expected. While the numerical value was the lowest in old donors, the decline was the steepest in the intermediate age group suggesting the skeletal ageing in vivo begins in the fifth decade of life. The other important observation was the presence of old donors with a relatively higher number of MSCs as compared to the others in the same age group, which was comparable to MSC numbers obtained from young donors. This suggested that in vivo, these old donors (with high number of

MSCs) potentially had a younger 'biological age' in comparison to others with the same chronological age (same age group).

Colony area and colony ID were analysed next to evaluate the age-related differences in the proliferative capacity of MSCs. Age-related changes in proliferative capacities of MSCs have been studied by different methods using cell proliferation assays, cumulative population doublings calculations and population doubling time in cultured MSCs. Some have shown an age-related decline in MSC proliferation (211) and others, not shown a difference (197). As mentioned in this thesis Introduction (section 1.2.2), such studies are limited by the fact that cultured MSCs undergo passage-dependent *in vitro* ageing process that can over exaggerate or negate any age-related changes in MSCs *in vivo*. On the other hand, the studies that have investigated MSC proliferation capacity prior to passaging are very limited. While they have measured colony area and the cellular contact between the cells in the colony (201), these factors have not been studied in relation to age-related changes. Therefore, the data from this study on any differences in colony sizes and IDs in relation to donor age are entirely novel.

Gothard et al., proposed a segregation of BM MSC colonies into large (>2.5mm diameter) and small (< 2.5mm diameter) based on median diameter chosen form the entire diameter range observed. They also segregated colonies as dense or sparse based on cellular contact confluency within a colony. High cell contact (>80%) was defined as dense and low cell contact (<80%) was defined as sparse colony. In the present study, different sizes of colonies were observed, which did not indicate a clear visual segregation by size or density, and generated a unimodal distribution following frequency distribution analysis by the GraphPad software (Figure 3.7A). To make a fair comparison with the Gothard et al. study, in this study, the colonies were segregated based on the median area from all the colonies (10.64mm²) obtained from ImageJ software analysis which was equivalent to average diameter of 3.76mm. The diameter was larger than the average colony diameter described in Gothard et al. study and the area was similar to colony areas presented in D'Ippolito study (10.23mm² for young donors, 8.64 mm² for old donors (132)).

Even though colony area measures the size of the colony indicating the spread of the cells, it does not measure the density of the cells within a colony. A colony can be large and dense or large and sparse as defined by Gothard et al., 2013. A large dense colony has more cells that are in close contact with each other indicating high proliferation and a small sparse colony will have fewer cells that are comparatively less compact in their cellular contact. A large dense colony will thus be derived from an

MSC with high proliferative capacity and a small sparse colony will originate from the least proliferative or pre-senescent MSCs (208). Hence the area of a colony alone is not a true indicator of the proliferative capacity of the MSCs and the colony ID, which is a combined measure of area and density, is thus an improved measure of the proliferative capacity of the MSCs. As listed in Table 3.2, only 2 studies have mentioned colony density in their studies. While one of them found no age-related difference in their research using uncultured cells (200) (data was not shown), the more recent study using cultured cells by Travis et al., in 2017 observed an age-related decline in the colony density visually (no quantification) (139).

ImageJ analysis of colony area and ID in this study helped to generate numerical ID values for all the colonies. The median values obtained from the analysis of colony area measured for all the colonies were used to categorize the colonies as large or small and then compare their ID in young and old donors. The results indicated a significantly higher ID of all colonies in young donors as compared to old donors (*p*<0.0001) (Figure 3.8A). Small and large colonies were observed in both, young and old donors but young donor colonies always had higher ID irrespective of the size of the colony (Figure 3.8C and D). The decline in IDs was more prominent in small colonies than in large colonies suggesting that old donor MSCs generate proportionally more small sparse colonies than proportionally less large dense colonies. This was opposite to the original hypothesis of fewer large dense colonies originating from most proliferative cells in old donor BM. These are important new data that highly-proliferative MSC remain to be present in old donor BM.

The presented data confirms the concept of decrease in the rate of MSC proliferation with increase in donor age in uncultured cells (212). The analysis of colony area and density has been previously done only by Gothard et al., (201), but they did not investigate these parameters on a large number of colonies with respect to donor age. Furthermore, no CFU-F study, as per my knowledge, has dissected the CFU-F assay at the single colony level to understand the age-related changes in colony area and ID making data from this chapter novel, in comparison to others to study age-related changes in proliferation of minimally cultured individual MSCs. Once the age-related decline in colony area and ID were established, next it was hypothesised that colony area and ID of MSCs when exposed to old donor serum (OS) will show an age-related decline as compared to media supplemented with YS. The other hypothesis with exciting potential applications was to test if the presence of young microenvironment (media supplemented with YS) could reduce the loss of proliferative capacity seen in old donor MSCs. *In vitro*, MSCs are grown in rather artificial conditions that differ from

their native conditions *in vivo* (100). To mirror the biological conditions of old donors, media supplemented with old and young donor serum was used instead of standard media supplemented with FCS. Preliminary experiments established optimal conditions and controls for these investigations, and established that human serum supported colony formation from MSCs better than negative control FCS (Figure 3.9).

In this study, a decline in proliferative capacity of MSCs from both young and old donors was observed when they were grown in media supplemented with OS. Using both colony area and colony ID as parameters to measure MSC proliferation at single-colony level, the data suggested that the use of OS negatively impacted the proliferative capacity of MSCs as seen by the presence of lower frequency of large colonies and higher frequency of small colonies in Figure 3.13 and 3.14. This was also shown by the low colony ID in young donors when grown in OS in Figure 3.13. This can be explained by the increase of ROS in cells as shown in a rat model (117) and by the presence pro-inflammatory factors (213) in OS that inhibit proliferation and eventually lead to MSC senescence. Increased levels of pro-inflammatory cytokines and SASP have been associated with increased age in human donors (214-216). Similar investigation in serum from human donors used for culturing young and old donor MSCs in this thesis, will help to explain the loss of proliferative capacity of MSCs from old donors. It will also provide evidence for the impact of aged microenvironment (SASP released from aged cells in medium containing human serum) *in vivo*.

Abdullah and colleagues studied the effect of aged serum on hMSC-tert cell line and found that the gene expression of MSC multipotentiality markers in old serum was significantly lower (p < 0.01) than the cells grown in young serum emphasising the impact of aged microenvironment on MSC-like cells (203). Josephson and colleagues performed experiments in mice and found a significant decline in the number of colonies from young (12 weeks) animal grown in serum from middle-aged (52 weeks) animals. They found an increase of SASP in MSCs grown in serum from middle aged animals (146). Quantification of pro-inflammatory cytokines and SASP molecules (like IL6, IL8, TNF, Tp53) from the media of old donor MSCs would further explain the loss of proliferative capacity of MSCs in OS. Additionally, proliferation of old donor MSCs grown in media supplemented with YS could be only slightly improved (limited rejuvenation). Human serum was used to reflect physiological conditions in vivo and indicated towards limited rejuvenation of old donor MSCs in YS. Serum is the fluid that remains after removal of all the clotting factors. The use of other human supplements like PL (platelet lysate) could potentially display stronger rejuvenation, owing to the presence of clotting factors (217). PL is a cell free lysate prepared by disruption of

platelets by freeze/thawing. Investigations using PL have shown that the osteogenic differentiation of MSCs grown in young donor PL was more pronounced as compared to their differentiation when grown in PL from old donors (217, 218).

The concept of MSC rejuvenation with culture in YS was not as pronounced in the present experiments, but positive trends were observed. The colony area measurements of old donor MSCs showed a trend for significantly bigger colonies when grown in media supplemented with YS as compared to OS (Figure 3.12B). Also a trend for higher ID of colonies from old donors grown in YS as compared to OS was found (Figure 3.14B). The fact that colony area was increased more than colony ID can be explained by the fact that YS increased cell migration out of the colony more than cell proliferation, this can be also useful for MSC function *in vivo*. This hypothesis can be further tested by using holographic microscopy by monitoring of cellular movement in real time (219). Growing old donor cells on extracellular matrix derived from younger donors as shown previously (139, 220), could also demonstrate noteworthy conclusions with possible clinical applications of rejuvenation. Overall, the present data indicate that exposing old donor MSCs to supplements (serum, PL, extracellular matrix- collagen fibres and proteins) from young donors can help aged MSC not lose their proliferative capacity any further, and possibly improve their migration capacity. Treatment of uncultured MSCs with anti-oxidants prior to clinical use (46), use of senolytic drugs that target senescent cells using MSC ageing surface markers (221) along with use of supplements to provide a 'young' microenvironment could also aid rejuvenation and delay the process of ageing.

In conclusion, this chapter has shown a significant age-related decline in the number of BM MSCs in older donors compared to young resolving the existing controversy. Colonies from the CFU-F assays showed a significant age-related decline in their area and ID. For the first time, colony sizes and densities were quantified (measured in ID) along with colony area alone and correlated with donor age, and showed an age-related decline in the proliferative capacity of individual MSCs from old donors. The proliferative capacity of young and old donor MSCs were also investigated in more physiological conditions with human serum to mirror *in vivo* conditions in which MSCs reside. OS negatively impacted the proliferative capacity of young and old donor MSCs and exposing old donor cells to YS stopped further loss in their proliferative capacity. These data, for the first time, suggest the importance of the microenvironment (MSC niche) in their functions and showing that both intrinsic and extrinsic factors should be considered in MSC ageing studies.

Chapter 4 Investigating age-related changes in human BM MSC using the CD45^{low}CD271⁺ phenotype

In the previous chapter, age-related changes in number and proliferative capacity of BM MSCs were investigated using the classical colony-forming assay. The results showed a significant decrease in MSC numbers across the whole age range (19-89 years old) (section 3.3.1), as well as their reduced proliferative capacity measured by colonies' integrated density (section 3.3.3). Reduced proliferative capacity was also observed in MSCs when they were grown in media supplemented with old donor serum as compared to media supplemented with young donor serum. This chapter will explore any potential changes in BM MSC gene expression with a focus on the molecules involved in MSC multipotentiality. For this, uncultured BM MSCs were sorted based on the CD45^{low}CD271⁺ phenotype, as previously described (109, 188, 222) and gene expression in young (19-40 years old) and old (60-89 years old) donor MSCs were investigated and compared.

4.1 Introduction

The identification and characterisation of uncultured BM MSCs has been previously performed using flow cytometry and a number of surface markers including Stro-1 (133, 197), SSEA-4 (139, 223), MSCA-1 (224) and others, outlined in section 1.2.1. Over the years, new markers such as CD140a (225), CD140b (226), CD146 (227), and CD295 (228) have also been proposed, sometimes as single markers but usually in combination with ISCT panel markers (105, 229) for ensuring appropriate identification of BM MSCs.

CD271 or Low affinity nerve growth factor (LNGFR) was proposed as a very specific BM MSC marker in 2002 (108, 230). The combination of CD45 (pan-hematopoietic lineage cell marker) and CD271 was later proven to be most useful for the identification of BM MSCs, as it provided the highest resolution and the least cross-reactivity with other BM cells. Since then, CD45^{low}CD271⁺ phenotype has been used by many independent laboratories where CD271 was consistently found to be present in BM MSCs making it a preferred choice for studying uncultured BM MSCs (109, 231, 232). In one study, Torales and colleagues carried out flow cytometry and compared the percentage of cells that expressed CD73, CD90, and CD105, and were negative for CD45 and CD34, with the percentage of cells that were CD271⁺ (233). They found that while the traditional ISCT panel showed an average 0.54% of mononuclear cells as putative MSCs, CD271⁺ alone was highly specific and 0.53% of BM cells expressed CD271. Nevertheless, the work from our laboratory (234), as well as others (109)

showed that the addition of CD45 as a 'negative' gating marker is necessary for a clearer MSC identification, and the removal of CD271^{low} 'passenger' cells that appear to be of a hematopoietic origin (233).

Considering that CD45^{low}CD271⁺ cell phenotype has been extensively used for identification of BM MSCs (section 1.2.1), this phenotype was chosen to be used in this chapter to further investigate age-related changes in the numbers and the gene expression in native BM MSCs without any culture manipulations. As shown in the previous chapter, the age-related decline in the number of BM MSCs quantified using the CFU-F assay was significant (Section 3.3.1), however, the CFU-F assay remains culture-dependent. It was thus logical to explore whether the same decline would be observed using flow cytometry, with no culturing steps involved, and to evaluate the results obtained by both methods. This is of particular importance as the studies that have investigated age-related changes in number of BM MSCs using flow cytometry methods remain limited (Table 4.1).

Table 4.1 includes studies that have investigated the age-related changes in number or percentage of BM MSCs using flow cytometry. While most of the studies (4 out of 6) have indicated a decline in the number of BM MSCs with increasing age, a number of variations can be observed. This includes disparities in grouping methods of 'young' and 'old', additionally, differences in cell isolation techniques and the markers used to identify BM MSCs were most notable. Interestingly, large donor variation is another factor that must be considered, especially when dealing with uncultured cells (235, 236).

Source of MSCs	Age groups	Markers used Age related change		Ref
BM from iliac crest	Y: 22-44; O: 66-74	STRO-1	No change	(197)
Femoral head BM	Y:<50; I:50-65; O:>65	STRO-1	No change	(237)
ВМ	13-80; Y:<45	Multiple	Multiple Decline	
Femoral neck bone	Y:28-31; O:80-97	CD271⁺ SSEA-4⁺	Decline	(223)
BMA	22-80; no groups	CD45 ^{low} CD271 ^{high}	Decline (F only)	(111)
lliac crest bone graft	24-89; Y:<50; O:>50	CD45 ^{low} CD271 ^{high}	Decline	(146)

Table 4.1 Previous studies examining age-related changes in the number of BM MSCs measured using flow cytometry

BM: bone marrow, BMA: bone marrow aspirate, Y: young, I: intermediate, O: old, F: females

4.1.1 Candidate MSC multipotentiality genes

As mentioned in section 1.2.2, the adipogenic bias of aged BM MSCs is among the most established age-related changes described to date. Table 4.2 lists previous studies that have investigated the shift from osteogenic to adipogenic differentiation in MSCs in aged donors. Of importance, all these investigations have been performed using cultured MSCs. The disruption in bone formation has been suggested to be due to the impaired ability of MSCs towards osteogenic differentiation in older donors. This loss in bone formation is compensated by excessive adipocyte formation. Different pathways and molecules have been suggested to be responsible for this shift, however, the exact mechanism is not yet completely understood. Of note, bone homeostasis is not only a result of bone formation, but also of bone resorption (section 1.1.4). In one study, an age-related increase in the expression of a bone resorption molecule *RANKL* in pre-adipocytes along with an increase in osteoclastogenesis, was observed within the BM (238). In part age-related bone loss is brought about by disruption of bone resorption, described in section 1.1.4.

Apart from the shift towards adipogenic differentiation and bone remodelling, the other factor that is potentially affected by older age in MSCs, is their communication with other BM cells with the use of trophic factors including chemokines. CXCL12 (C-X-C motif chemokine 12) is one such chemokine known to be critical to maintain the HSC pool within the BM (239, 240) as well as being involved in MSC migration and apoptosis (241). CXCL12 is regulated by Cx43 and Cx45 (242), where Cx43 is the most common gap junction within the bone (148). However, age-related changes in CXCL12 production by native BM MSCs remain unexplored. In terms of direct cellular interactions, connexins are intercellular communication molecules and as an entire family of molecules, have been suggested to have an age-related decline in many tissues with a potential link to human ageing and cancer (243). Both of these molecules will be discussed further in 4.1.1.5.

As BM MSCs are very rare cells (87), the numbers of purified cells are not sufficient to perform standard differentiation assays, which normally use a range of 2x10⁵-2x10⁶ cells/ assay (108). Therefore, gene expression in CD45^{low}CD271⁺ cells was studied as the first step to explore their differentiation and cell communication potentials.

4.1.1.1 Osteogenic-lineage transcripts

RUNX2 (Runt related transcription factor 2) is the master regulator of bone formation and belongs to RUNX family of transcription factors (TFs). All molecules of this family

share a common runt domain. *RUNX2* is the earliest indicator of osteogenic differentiation and is also known to tightly regulate chondrocyte hypertrophy and vascular invasion of developing skeletons. In MSCs, levels of *RUNX2* expression have been mainly investigated to detect their osteogenic differentiation capacity and MSC commitment towards osteo-progenitor lineage (244). With respect to age-related changes, overexpression of *RUNX2* has been suggested to contribute to bone resorption in osteoblastic lineage cells by increasing *RANKL* expression (245). However, when narrowing to MSCs from human donors, investigations are sparse. In older donors, the expression of *RUNX2* has either been observed to decrease or have no change in cultured MSCs providing no substantial evidence for age-related differences in uncultured MSCs (Table 4.2). Nevertheless, it could be hypothesised that it may be reduced based on the idea of 'MSC adipogenic bias' with advancing age.

Table 4.2 Previous studies investigating age-related differences in osteogenic, adipogenic and bone remodelling molecules in cultureexpanded MSCs

Function	Source of cells/ MSCs	Age groups	Genes tested	Age related changes	Reference
	BM from humans	17-90; Y:<50; O:>50	RUNX2	Decreased	(133)
Osteogenic differentiation	BMA from IC in humans	Y:<50; A:50- 65 ; O:>65	RUNX2, OC, ALP, OPN	No changes in <i>RUNX</i> 2	(246)
	Bone graft from IC	Y:18-49; O:>50	RUNX2, OC, ALP, OPN	No changes in <i>RUNX</i> 2	(247)
Adipogenic differentiation	BMA from IC in humans	13-80; no group	LPL, PPAR- γ	No changes in <i>PPAR-γ</i>	(97)
	Human ACL	Y:17-27; O:69-79	LPL, PPAR- γ	No changes in <i>PPAR-</i> γ	(248)
	BM from iliac spine	Y: 18-42; O: 66-78	$PPAR-\gamma$	No changes	(196)
	Bone from IC	7-78; no group	Leptin receptor CD295	Increased	(126)*
Bone remodelling	BM from humans	Y:<50; O:>55	RANKL, OPG	RANKL increased, OPG decreased	(249)
	BM from humans	Y:<55; O:>55	RANKL	RANKL increased in females	(250)

BM: Bone marrow, IC: Iliac crest, Y: young O: old, A:adult, *: investigated at protein level

SPARC (Secreted protein acidic and rich in cysteine) or more commonly known as 'Osteonectin' is a non-structural matricellular glycoprotein in the bone that binds calcium. It is secreted by osteoblasts and pre-osteoblasts during bone formation and is needed for bone calcification (251). Both, over expression (252) and selective deletion of SPARC (253) have been associated with impaired motility, suggesting that its normal functioning is vital for bone growth and development. With respect to MSCs, the precursors of osteoblasts, it is not surprising that MSCs too express certain levels of SPARC, shown previously by our group (188). Its role in the functions outlined above suggests that it plays an important part in the MSC transition towards the osteogenic lineage. It is thus expected with increasing age, as the fat content increases with a compromise in bone formation, the levels of SPARC potentially decline. However, the number of studies investigating this also remain limited, with no study to date investigating any changes in its expression in uncultured MSCs. Interestingly, our previous study using cultured MSCs have shown that SPARC expression significantly increased in late-passage, pre-senescent MSCs (222) therefore the present evidence on the association of with MSC ageing or senescence remains controversial.

SFRP1 (Secreted frizzled related protein 1) is a member of the SFRP family of molecules containing cysteine rich glycoproteins, homologous to the putative Wnt (wingless integrated)-binding site of the frizzled proteins (fzd receptors). It acts as a soluble modulator of the Wnt signalling pathway. Wnt signalling pathway is a major pathway involved in skeletal development, bone mass regulation and is often regarded integral for osteogenesis (254). SFRP1 is well known as a Wnt antagonist that acts by directly binding to Wnt proteins, making it a target for therapeutic approaches for bone remodelling and repair. SFRP1 has been extensively explored in bone and boneforming cells (254). The deletion of SFRP1 reduced osteoblast and osteocyte apoptosis in vivo and proliferation and differentiation capacity of osteoblasts in vitro in mice (255). It has also been reported to block osteoblast induced osteoclastogenesis by inhibiting RANKL dependant osteoclast formation (256). SFRP1 was also found to suppress Wnt signalling in an immortalised osteoblast cell line (257). Overall, previous studies indicate that SFRP1 inhibits osteoblast formation and their coupling with osteoclasts. Based on these findings, SFRP1 inhibitors were developed as potential therapies with the aim of increasing bone formation (258). While their potency has been tested in vitro, in vivo efficacy in appropriate animal models or in human cells are yet to be examined.

The effect of age on *SFRP1* expression has not been extensively studied to date. Rauner and co-workers investigated the level of expression of *SFRP1* in the osteoblasts of young (6w), adult (6mo) and old (18mo) male C57BL/6J mice and found a significant age-related decline (259). Considering the above stated findings and the fact that the gene has been found on osteoblastic lineage, it can be hypothesised that MSCs, expression of *SFRP1* could show an age-related decline. Indeed, this was observed in our previous study where *SFRP1* displayed an age-related decline in BM CD45^{low}CD271⁺, cells but this was observed in paediatric donors compared to adult donors (188).

Owing to their involvement in MSC osteogenic differentiation, these three transcripts were selected for investigating age-related differences in uncultured MSCs in relation to their osteogenic differentiation potential.

4.1.1.2 Adipogenic-lineage transcripts

An increase in the transcripts associated with adipogenesis have been observed in MSCs aged *in vitro* (section 1.2.2) or in aged animals *in vivo* (80, 143, 260). Among many transcripts, an increase in *PPAR-* γ (Peroxisome proliferator activated receptor-gamma) *and FABP4* (fatty acid binding protein 4) expression has been associated with the adipogenic differentiation of MSCs (261). *LepR* (Leptin receptor) plays an important role in bone-fat balance (262, 263) and its expression was shown to decline in uncultured cells by our group previously (188). However their donor range included paediatric samples and the oldest donor was 72 years old. Age-related changes in adults and in donors aged above 75, with respect to expression of these transcripts in uncultured MSCs is yet to be explored.

PPAR- γ gene encodes a molecule termed the PPAR- γ receptor. There are two different isoforms of the protein, named PPAR- γ 1 and PPAR- γ 2 that exist due to alternative splicing of the gene. While PPAR- γ 1 is expressed in various cell types including osteoblasts, PPAR- γ 2 is restricted to adipocytes including those within the BM (264). The receptor is found in abundance not only in adipose cells but also in pre-adipogenic cells, endothelial cells and vascular smooth cells (265). Owing to its requirement in adipose tissue formation, PPAR- γ is considered a key regulator of adipogenic differentiation. PPAR- γ in MSCs can be both, anti-osteogenic and pro-adipogenic, therefore it can play the role of osteoblastic/adipogenic 'switch' (266). It has thus been described to show 'tremendous potential in novel strategies for bone tissue engineering and clinical applications' (267). As mentioned above, there is published evidence of increased expression of *PPAR-\gamma* in cultured MSCs from aged

donors as well as in cells aged *in vitro* through culture expansion (143). Based on previous literature , it was hypothesised that there would be an age-related increase in the expression of PPAR- γ in native BM MSCs from older donors.

Leptin receptor (LepR) is another transcript for which there is an increasing body of evidence on its role in the bone-fat balance (262, 263). *LepR* gene encodes for the surface receptor for leptin, a hormone secreted by adipocytes and known to regulate body weight and metabolism (268, 269). To understand the gene encoding for leptin receptor, it is worth taking a step back to understand the role of leptin.

Leptin is an adipokine predominantly produced by adipose cells and is known to contribute to energy balance by acting on hunger/satiety centres located in the hypothalamus (270, 271). Its level increases with weight gain and decreases with reduced weight (272). The adipokine has also been shown to be secreted by skeletal muscles and bone cells and is an established growth factor for muscle and bone in early life (268). The receptor for the leptin adipokine (CD295) has been shown to be present in musculo-skeletal tissues (263). Initially, it was thought that leptin hormone contributed to age-related decline in bone mass via the hypothalamus and the betaadrenergic receptors (273). However, more recent studies suggest that the hormone may have a more direct role on bone lineage cells including MSCs (274). However, the role of CD295, the leptin receptor in MSC differentiation to bone is not very well understood. With respect to ageing, Laschober and colleagues showed an age-related increase in the level of expression of LepR at the protein level (CD295) in BM MSCs (126). Their study also indicated that the CD295 positive BM MSCs in old donor MSCs were more apoptotic than young donor MSCs, quantified using Annexin-V staining. On the other hand, our group has previously shown an age-related decline of LepR gene expression in CD45^{low}CD271⁺ cells in a small number of donors (n=8) (188). We therefore hypothesised an age-related difference in the transcript of this gene in native BM MSCs from old donors.

FABP4 belongs to the FABP (fatty acid binding protein) family of transport proteins that are responsible for the transfer of fat or lipophilic molecules across the cellular membranes and is a well-established marker for differentiated adipocytes (275). It was first detected in adipose tissues and mature adipocytes and has also been termed as adipocyte P2 (ap2) due to similarity with myelin P2 protein (276), which is also a part of the FABP family. It is highly expressed during adipogenic differentiation of MSCs and is transcriptionally controlled by *PPAR-* γ agonists, fatty acids and insulin. With respect to MSCs, *FABP4* has been detected in high levels in MSCs expanded in culture over a

period of 21 days undergoing adipogenic differentiation in parallel to the expression of PPAR- γ (277). Based on these findings, an age-related increase in its expression in uncultured BM MSCs was hypothesised.

4.1.1.3 Oxidative stress in MSCs and anti-oxidant enzyme SOD3

Presence of reactive oxygen species (ROS) has been suggested as one of the potential drivers of the adipogenic bias of MSCs (278, 279). ROS are by-products of metabolism inside a cell, mitochondria being the major site of ROS production. Oxidative stress is the consequence of an excessive ROS production and limited activity of antioxidant enzymes to balance the negative impacts of ROS. A decline in the anti-oxidant enzymes with increasing age leads to reduced capacity of scavenging of ROS (280). Excessive ROS causes accumulation of DNA damage leading to cellular senescence in ageing and associated diseases (281). This ultimately drives the cell onto the apoptosis path and is believed to be one of the major causes of age-related differences in MSCs.

During metabolism, ROS intermediate molecules like superoxide and peroxides are formed as a part of cell metabolism. These intermediate ROS products are scavenged by anti-oxidants enzymes like superoxide dismutase (SOD) and glutathione peroxidase catalase (Gx) to prevent the interaction of ROS with lipid membranes and proteins within the body (282). SOD converts superoxide ion to hydrogen peroxide, which is further converted into water and oxygen molecules under catalase and glutathione peroxidase activities (282).

Anti-oxidants thus play an important role in critically maintaining the appropriate ROS levels. SOD or superoxidedismuatse form the first line of defence in scavenging the superoxide molecule (283). 'Dismutation' by definition is a redox reaction that involves the formation of two products of different oxidation states from a single molecule of intermediate oxidation state. Superoxide molecule (O_2^-) has a negative charge which causes it to react with lipids and proteins nearby causing damage. SOD thus gets its name from the function of carrying out 'dismutation' of superoxide molecule to hydrogen peroxide (H₂O₂). H₂O₂ further is scavenged by Gx into water and oxygen molecule. SOD enzymes include SOD1, SOD2 or SOD3. SOD1, also known as Copper/Zinc SOD (Cu/Zn SOD) is mainly located in the cytosol of the cells. SOD2 is also known Manganese-SOD (Mn-SOD) is majorly located in the mitochondria of cells. As mitochondria is the major site of ATP production, SOD2 has been investigated in a large number of studies to understand the impact of this anti-oxidant enzyme in oxidative stress (283).

SOD3 is also known as extracellular SOD (EC-SOD) and gets its name from its location. SOD3 has been shown to be secreted by MSCs (284), while SOD2 and SOD1 do not have any such cellular specificity. Due to its location, it is the only SOD that scavenges ROS in the extracellular compartment of cells and catalyses the dismutation of superoxide molecule. Within the BM, MSCs exist within a hypoxic BM niche with oxygen concentrations as low as 2-8% O_2 (285). Culture expansion in artificial conditions automatically expose these cells to a 'hyperoxic' condition but it is usually referred to as 'normoxic' or normal oxygen concentration due to the widespread practice of *in vitro* expansion. An increase in the oxidative stress in MSCs has often been suggested as a potential cause for the senescence and their adipogenic bias *in vitro* (278, 286).

Mohd Ali and colleagues compared the proliferative capacities of BM MSCs from young (<30) and old (>60) donors in hypoxic (5% O₂) and normoxic (20% O₂) conditions. They found that MSCs from older donors demonstrated reduced proliferative and differentiation abilities along with increase in cellular senescence. Interestingly, they found that MSCs cultured under hypoxic conditions in both age groups were able to demonstrate enhanced self-renewal and proliferative capacity (287). Another study investigated the changes in *SOD3* at the mRNA and protein level during differentiation of cultured BM MSCs. The study found significantly higher levels of *SOD3* during adipogenesis, lower levels during chondrogenesis but no difference during osteogenesis (288).

Considering that increased adipogenic differentiation bias is also age-related (section 1.1.4) and that increase in oxidative stress is among the oldest theories of ageing (section 1.1.1), it can be said that SOD3 may be affected in old donors. However, more clarity in how it alters in uncultured BM MSCs in old donors is required. It may be hypothesised that the level of ROS in old donor MSCs will be greater and the levels of anti-oxidant enzyme *SOD3* in old donor MSCs will be lower than in young donor MSCs.

4.1.1.4 Transcripts associated with bone remodelling

Bone remodelling is a tightly regulated mechanism, which under normal conditions perfectly balances the rate of bone resorption by osteoclasts and the bone formation by osteoblasts. The bone remodelling cascade and the different cells involved was described in detail in section 1.1.4.

RANKL (Receptor activator of nuclear factor kappa beta ligand) is expressed by osteoblasts and binds to its target RANK that is expressed by osteoclast progenitors (64, 74). RANKL is a type II membrane protein from the tumour necrosis factor (TNF) super family. It is a homotrimeric protein and exists in both, soluble and membrane form. However, the soluble form is known to have lesser ability to form osteoclasts (289). MSCs have been known to contribute towards both, bone resorption by expression of *RANKL* and bone formation as precursors of bone forming cells (290). While culture expanded MSCs are able to express *RANKL* and have been shown to contribute to osteoclastogenesis (291), its expression in MSCs in relation to *in vitro* ageing remains controversial (Table 4.2) and in uncultured MSC, unexplored.

RANK (Receptor activator of nuclear factor kappa beta) is the receptor for its ligand, the *RANKL* and is also designated *TNFRSF11A*. *RANK* is expressed on the surface of osteoclast precursors and *RANKL* binding to *RANK* initiates the process of osteoclast formation. It is expressed on osteoclast precursors (292) which upon maturation form bone resorbing osteoclasts. However, osteoclast formation takes place only after the activation of TNF associated receptor factor 6 (TRAF6) signalling (293). This then mediates the MAPK/NF-Kb pathway which eventually brings about osteoclastogenesis.

OPG (Osteoprotegerin or Tumour necrosis factor super family member 11B/ *TNFSFR11B*) is expressed widely by a number of cells and is the natural decoy/inhibitor of *RANKL*. If *OPG* (instead of *RANKL*) binds to *RANK*, then osteoclast formation stops and bone resorption does not take place. It is highly expressed in tissues including osteoblasts, pre-osteoblasts and other cells of the bone lineage in the BM (71). It also belongs to the TNF receptor superfamily but is different from the other members as a secreted protein that does not contain a trans-membrane domain (294).

An age-related increase in *RANKL* and decrease in *OPG* in MSCs was hypothesised, potentially contributing to increased bone resorption rates *in vivo*, accounting for the age-related bone loss in old age.

4.1.1.5 Transcripts associated with cellular interactions

CXCL12 (Chemokine motif 12 also known as SDF-1 or stromal derived factor-1) is a major chemokine facilitating the chemotaxis of HSCs, MSCs and other BM cells (295). It is a member of a large family of chemokines that are structurally related. It binds to its receptor CXCR4 on target cells and is known for its role in the maintenance of the hematopoietic compartment of the BM niche (239, 240) and in chemotaxis of MSCs (296). In the BM, *CXCL12* is expressed by osteoblasts and endothelial cells as well as

stromal cells lining the endosteal surface (297). Most HSCs within the BM are usually in contact with cells expressing high amounts of *CXCL12* which have been named *CXCL12*-abundant reticular cells (or CAR cells) in mice (239). *CXCL12* is key to maintain the healthy pool of HSCs. The counterpart of CAR cells in human BM are likely to be adventitial reticular cells, topographically overlapping with CD45^{low}CD271⁺ MSCs (298).

In addition to its support for hematopoietic cell activities, such as their retention and egression from the BM, *CXCL12-CXCR4* signalling axis has also been implicated in the processes related to osteoclast recruitment, chemotaxis and bone resorption (299). There are also reports describing the impact of *CXCL12-CXCR4* signalling in metastases of breast cancer, lung cancer and some hematopoietic malignancies to the bone (300).

Investigation in culture expanded MSCs to understand any differences with the advancement of *in vitro* ageing, to the best of my knowledge, does not yet exist. The information regarding the levels of its expression before culturing MSCs, also remain limited. Our group found very high levels of *CXCL12* expression in uncultured CD45⁻ CD271⁺ BM MSCs as compared to CD45⁺CD271⁻ HLCs but there was no age-related differences observed with 8 donors. As red (hematopoietic) marrow is replaced by yellow (fatty) marrow as a function of age, it was hypothesised that there could potentially be an age-related decrease in the levels of *CXCL12* in uncultured BM MSCs from aged individuals.

Among the connexins, *Connexin43 (Cx43)* is the most abundant gap junction protein in bone cells, mainly associated with the intercellular interaction between osteocytes and osteoblasts (148). It is essential for intercellular interaction, normal bone formation and for the viability of osteocytes (301). As the main gap junction in mature bone-lineage cells, it is therefore an important molecule to investigate with respect to *in vivo* MSCs. In the BM, MSCs interact with each other forming an intricate net or scaffold upon which other BM cells including HSCs reside, these networks are facilitated by *Cx43* gap junctions. The gap junctions also regulate the transcription of *CXCL12* in a cell contact dependant manner (242).

Previous studies investigating age-related differences in the expression of Cx43 have found a significant decline in native CD45^{low}CD271⁺ BM MSCs (188). Studies in human derived cultured expanded MSCs to investigate changes with *in vitro* ageing do not exist. With respect to animal studies, Donahue's group have investigated the significance of *Cx43* in bone formation and skeletal homeostasis in rats. They have found no age-related differences in *Cx43* gene expression in cultured BM MSCs (302). Although the data remains controversial, these studies present *Cx43* a potential molecule affected by age in native BM MSCs (303).

At the protein level, Cx43 or gap junction protein alpha 1 (GJA1) is an intercellular membrane spanning channel that mediates direct exchange of small molecules, ions, nucleotides between neighbouring cells. The role of Cx43 in skeletal metabolism and homeostasis is now well established as gap junction intercellular communication (GJIC) (304). In relation to the bone, a growing body of evidence exists on the importance of Cx43 in modulating bone cells to respond to mechanical signals and growth factors to enable bone healing post fracture (148, 305). Absence or reduced Cx43 function have been associated with reduced levels of osteogenic genes (306) and reduced level of osteoclast formation leading to decreased bone resorption (307). These findings have established that Cx43 is vital for growth, development and maintenance of healthy bone health. Owing to the various important roles played by this molecule in the BM, Cx43 was chosen to be tested at both, the gene and protein levels and was hypothesised to display an age-related decline in BM MSCs in this project.

4.1.2 Candidate age-specific surface makers

The knowledge of age-specific markers for uncultured BM MSCs remains limited, especially when investigating BM MSCs from young and aged human donors. It has been shown in various studies that BM MSCs from younger donors have good therapeutic potential. However, the use of BM MSCs from older donors for clinical applications usually requires further in depth investigation. Identifying a candidate surface marker that could differentiate young from old donor MSCs could be next considered as a marker to indicate the ageing status of native MSCs before further investigations.

The previous chapter showed that there is a decline in the proliferative capacity of BM MSCs from old donors, irrespective of whether they formed large or small colonies. Identification of a marker that could indicate low-proliferative (aged) MSCs, would be very useful for cell sorting to remove these poorly proliferative cells before clinical and therapeutic applications of MSCs.

4.1.2.1 CD106

CD106 also known as vascular cell adhesion molecule (VCAM-1) is a cytokineinducible cell surface protein capable of mediating adhesion and immunomodulation (308, 309). It is a member of the immunoglobulin family and is expressed on BM MSCs (Table 4.3). However, the level of expression in human BM MSCs has shown considerable experimental variation, whereby decreasing expression (182), increasing expression (125) and random alternations (123) have been shown with *in vitro* culture expansion.

Thus the information on what happens to the expression of CD106 with *in vitro* ageing remains controversial. Cellular adhesion, T cell activation and recruitment of lymphocytes at the site inflammation is known to be mediated by CD106. CD106 also plays an important role in MSC mediated immune-suppression (310). Considering that MSC display their immune-regulatory properties using cell-to-cell adhesion (311) and a general decline in the immune-suppression potential of BM MSCs with age (303), it was hypothesised that the expression of CD106 would display an age-related decline.

4.1.1.6 CD146

CD146 or Melanoma cell adhesion molecule (MCAM) is a 113kDa cell adhesion molecule which was originally described as an endothelial cell marker (312). It is also expressed in a variety of cells including lymphocytes (313). With respect to BM, CD146 identifies the population of the perivascular MSCs (314). A 2011 study by Tormin et al., showed that native BM MSCs with both phenotypes CD45^{low}CD271⁺CD146⁺ and CD45⁻CD271⁺CD146⁻ formed comparable number of colonies in the CFU-F assay and generated similar cultures *in vitro* with similar cell morphology, FACS and gene expression profiles (227). In terms of their location, CD45⁻lowCD271⁺146⁺ cells were found in the perivascular region whereas CD45⁻lowCD271⁺146⁻ cells were present as bone lining cells.

With respect to age-related changes, a few studies listed in Table 4.3, have shown down regulation of the marker in culture. Previous work from our laboratory has shown a decline during *in vitro* MSC ageing in the expression of CD146, as well as CD106 (182). Siegel and co-workers found an age-related decline in the geometric mean of CD146 expressing BM MSCs. However, no data on the expression of CD146 in uncultured BM MSCs in relation to donor age is available. Based on previous evidence of decline in expression of CD146 in culture expanded BM MSCs (Table 4.3), it was hypothesised that CD146 displays an age-related decrease in the expression in old donors.

The aim of this chapter was to investigate any age-related changes in the number of uncultured BM MSCs using the CD45^{low}CD271⁺ phenotype in flow cytometry. The same phenotype would be used to segregate MSCs (CD45^{low}CD271⁺) and HLCs (CD45⁺CD271⁻) cells and investigate any age-related changes in multipotential gene and surface marker expression.

Surface marker	Source of MSCs	Age groups	Passage/ donor age dependant	Isolation	Medium	Passage/ Age- related change	Ref
CD106	BMA	2-61	Passage dependant	Density centrifugation	NH expansion medium	Decreased	(182)
	IC	T:0-20; YA:20- 40; AA:40-60; O:>60	Donor age dependant	Plastic adhesion	MEM +20%FCS	Increased	(138)
	BMA	13-80; Y:<45; I:45-65; O:>65	Donor age dependant	Density centrifugation	Mesencult	Decreased	(97)
CD146	BMA	2-61	Passage dependant	Density centrifugation	NH expansion medium	Decreased	(182)
	BMA	13-80; Y:<45; I:45-65; O:>65	Donor age dependant	Density centrifugation	Mesencult	Decreased	(97)
	BM	NA	Passage dependant	Plastic adhesion	DMEM+ 10%FCS	Decreased	(123)

Table 4.3 Previous studies investigating candidate age-specific surface markers

BMA: Bone marrow aspirate, IC: Iliac crest, T: teenage, YA: young adults, AA: ageing adults, I: intermediate, O:old, NA: not applicable
4.1.2 Hypotheses and Objectives

Hypotheses:

1. The number of CD45^{low}CD271⁺ uncultured BM MSCs demonstrates an age-related decline.

2. The transcripts for osteogenic and adipogenic differentiation measured in CD45^{low}CD271⁺ BM MSCs demonstrate an age-related decline and increase, respectively.

3. The level of ROS in CD45^{low}CD271⁺ BM MSC increases and the expression of antioxidant enzyme SOD3 decreases with increasing age.

4. The transcripts for bone remodelling measured in CD45^{low}CD271⁺ BM MSCs demonstrate an age-related bias for bone resorption over bone formation.

5. The transcripts related to cellular interactions measured in CD45^{low}CD271⁺BM MSCs demonstrate an age-related decline.

6. The candidate surface markers measured by flow cytometry show age-related decrease.

Objectives:

1. To enumerate uncultured BM MSCs by flow cytometry using the CD45^{low}CD271⁺ phenotype across a broad donor age range.

2. To measure transcripts associated with MSC osteogenic (*RUNX2, SPARC, SFRP1*) and adipogenic differentiation (*PPAR-\gamma, FABP4, LepR*) in CD45^{low}CD271⁺ BM MSCs in young and old donor groups.

3. To quantify ROS and to examine the expression of anti-oxidant enzyme SOD3 in CD45^{low}CD271⁺ BM MSCs in young and old donors.

4. To test the transcripts associated with bone remodelling (*RANK, RANKL, OPG*) in CD45^{low}CD271⁺ BM MSCs in young and old donors.

5. To test transcripts associated with cellular interactions (*CXCL12, Cx43*) in CD45^{low}CD271⁺ BM MSCs in young and old donors.

6. To investigate the expression of surface markers of CD106, CD295, Cx43 and CD146 in CD45^{low}CD271⁺ BM MSCs in young and old donors.

4.2 Methods

4.2.1 Donor selection

Fresh BMA from n=30 donors was used to perform enumeration of BM MSCs using the CD45^{low}CD271⁺ phenotype. Frozen BMA from a total of n=20 donors (10 young and 10 old donors) were used for gene expression and surface marker expression experiments. Donors used in this chapter were n=10 young donors (19-40 years old, median age=27 years old) and n=10 old donors (59-89 years old, median age=68 years old). Samples were defrosted as outlined in section 2.3 of the thesis and the exclusion criteria remained the same as described previously in section 3.2.1

4.2.2 Native MSC enumeration by flow cytometry

Flow cytometry was used as the method is automated, can be performed with minimal manipulation and without any culture expansion. For native BM MSC enumeration, 100µl of fresh BMA was placed in a FACS tube and incubated for 15 minutes at RT with the following antibodies: 20µl CD271-allophycocyanine (APC) and 10µl CD45-phycoerythrin (PE)-cyanine dye 7 (Cy7); and 10µl of 7-Aminoactinomycin D (7-AAD) was added to distinguish between live and dead cells. After staining, erythrocytes were lysed by using 500µl of ammonium chloride (NH₄Cl) solution (Appendix 2, Table 2 Buffers) per the tube for 3-5 minutes at 37°C. Finally, 50µl of Count bright absolute counting beads were added at RT after vortexing. The cell and bead suspension (total volume 700µl) was run using the Beckton Dickinson (BD) LSRII 4 laser flow cytometer.

Unstained and single antibody stained samples were used to optimize the cytometer voltage settings and spectral compensation. Isotype controls for the antibodies IgG1 APC and PE-Cy7 (Appendix 2, Table 2) were used to set the gating strategy on the population of interest (MSCs). A minimum of 100 MSCs (CD45^{low}CD271⁺ cells) were collected for each sample. The acquisition time was dependent on the cellularity of the sample. Data analysis and gating strategies were optimized based on the method described previously by Cuthbert *et. al* (207) and also shown in figure 4.1.



Figure 4.1 Uncultured BM MSC enumeration by flow cytometry using CD45^{low}CD271⁺ phenotype

Gating strategy for enumeration of CD45^{low}CD271⁺ MSCs. Forward and side scatter plot was used to eliminate debris followed by separation of counting beads and cells based on granularity and fluorescence. Then live cells were identified based on 7-AAD negativity from where CD45^{low}CD271⁺ were identified as MSCs. Figure indicates representative donor.

The analysis of the flow cytometry data was performed using the FACS Diva software. Cell debris were eliminated based on forward and side scatter and the counting beads were distinguished based on their granularity and fluorescence. The dead cells were excluded by the uptake of 7-AAD dye. The cell population of interest (MSCs) were identified as CD45^{low}CD271⁺ and the number of MSCs per μ l of BMA was calculated using the bead manufacturer's formula [number of live MSC events/number of beads events x number of beads per 50 μ l/ total sample volume]. The number of MSCs per μ l was multiplied by 700 to get the total number of MSCs in 100 μ l of BMA which was then multiplied by 10 to get the number of MSCs per ml of BMA.

4.2.3 Sample preparation and quantification of ROS in uncultured MSCs

BMA samples from young and old donors were collected as mentioned in section 2.2. RBCs were lysed using NH₄Cl and $2x10^6$ NCs were added to each of the flow tubes. While one tube was not exposed to any chemical, the other tube was exposed to 8mM tert-butylhydrogenperoxide (TBHP) for the generation of ROS under extreme oxidative stress as per optimised protocol (315). Both tubes were then incubated for an hour at 37°C and 5% CO₂. Cells that were not exposed to TBHP served to measure 'basal' levels of ROS in the MSCs.

After incubation, the cells were washed with 2ml 1X PBS and centrifuged at 400g for 5 minutes. The supernatant was discarded and the cells were re-suspended with 100µl of medium containing DMEM+10%FBS. The cells were stained with 20µl CD45 V450 and 20µl of CD271 PE-Vio770 and 2µl of CellROX (FITC) for staining MSCs at the same time for 15 minutes. To identify ROS, cells were stained with CellROX which is a non-fluorescent dye under reduced state but exhibits strong fluorescence upon oxidation localized within the cell. Exposure to TBHP causes additional formation of ROS resulting in further oxidation which was detected by the dye and measured by flow cytometry. The entire process has been summarized in Figure 4.2 below. After 15 minutes of incubation, the cells were washed with PBS for 5 minutes at 400g. The supernatant was discarded and the cells were re-suspended with 100µL of medium containing DMEM+10%FBS. Finally, 2µl of Sytox (APC), dead cell dye was added to distinguish live cells from the dead cells and after 15 minutes of incubation, the sample was quantified for measurement of ROS using flow cytometry.

MFI for CellROX that was indicative of the amount of ROS in the sample was recorded and compared between pre and post stimulation with TBHP. The values were then compared between young and old donors for age-related differences.



Figure 4.2 Steps involved in the quantification of ROS in CD45^{low}CD271⁺ BM MSCs

(A) BMA was treated with ammonium chloride to eliminate RBCs. This was followed by treatment of cell suspension with tert-butylhydrogenperoxide (TBHP) or no treatment (basal) for an hour in incubator. The cells were then washed, stained with antibodies and quantified using flow cytometry. (B) Gating strategy for measuring ROS in CD45^{low}CD271⁺ MSCs, recently published by Jawhari et al., (315). Forward and side scatter plot was used to eliminate debris (left panel) followed by identification of live cells using Sytox negativity (middle panel). Live cells were identified as CD45^{low}CD271⁺ MSCs. Figure is indicative of representative donor.

4.2.4 Cell sorting

The technique for cell sorting has been described in section 2.4. In brief, frozen nucleated BM cells from both donor groups with at least 40x10⁶ frozen cells were defrosted using TM (section 2.3). The cell suspension was then centrifuged at 400g for 5 minutes for the removal of DMSO. The supernatant was discarded and the cells were re-suspended in 5 ml of the TM. The MSCs were then enriched using Anti-fibroblast microbeads that positively select for cells with D7-fib receptor to enrich the percentage of CD45^{low}CD271⁺ MSC by positive selection, as previously developed in our laboratory (108) and described in section 2.4.

After enrichment, the cells were distributed in 10 tubes for staining. For isotype controls tubes, a minimum of 2x10⁵ cells were used for each antibody and for cell sorting, up to 3x10⁶ cells were used to collect sufficient number of MSCs. The duration of cell sorting from setting up the spectral compensation and up to the completion of cell sorting for 3 samples varied from a minimum of 180 minutes up to a maximum of 270 minutes depending upon the cellularity of the samples. As a rare population in the BM, the number of CD45^{low}CD271⁺ cells collected after sorting ranged from a minimum of 999 cells to a maximum of 15,000 cells depending on donor age, technical challenges during defrosting and cell loss due to enrichment. Simultaneously, a maximum of 70,000 cells of the control population of CD45⁺CD271⁻ HLCs were also collected. Their collection was usually accomplished within 45-60 minutes owing to the relatively larger proportion of these cells within the BMA. The number of HLCs collected was thus considerably higher than the number of MSCs collected.

Analysis of flow cytometry data was performed using FlowJo software. Unstained and single antibody stained tubes were used as controls. Gating strategy as described by Cuthbert et al., (207) was followed to record the number of cells and MFI in a selected gate as shown in Figure 4.11. The recorded values of MFI for each surface markers were plotted as a dot plot for young donors (black) and old donors (grey) using GraphPad Prism with median values to illustrate and compare any difference in the MFI.

4.2.5 Gene expression between young and old donors

The general method of qPCR has been described in section 2.5. The experiments were performed in integrated fluid circuit Flex Six^{TM} chip (Fluidigm) which has 2 partitions with 6 compartments each (Figure 4.2). One side of the chip is for placing the TaqMan assays (assays inlet) and the other side is for samples' cDNA (samples inlet). Each compartment can hold up to 12 samples/assays. The compartments are covered using

barrier plugs to prevent any external contamination before being used (Figure 4.3, top panel). The accumulator is the region where the control fluid is added that eventually enables the mobilisation of the assay and the corresponding sample to their reaction depot (Figure 4.3, bottom panel) based on the principles of microfluidic technology.



Figure 4.3 Flex Six[™] chip showing its different segments as described in the text and shown in manufacturer's protocol

The volumes of reagents, assays and samples used have been described in section 2.5. The TaqMan probes for genes presented in this chapter are shown in Table 4.4

below. This technique was chosen over traditional RT-PCR owing to the lower volume of sample and low concentrations of RNA needed for the experiment. As mentioned earlier, CD45^{low}CD271⁺ is a rare population in the BM and numbers of cells obtained after cell sorting were as low as 999 cells. The collection and elution of RNA was performed to ensure minimal dilution of the RNA content and at the same time to get sufficient RNA quantity for performing experiments. The guidelines provided by the manufacturers were followed in order to carry out the experiment.

For analysis, Fluidigm software was used to set the Ct values and the expression of genes were exported as a Microsoft excel file. In the excel file, first the data was normalised with *HPRT1* as the housekeeping gene to generate the Δ Ct values for each gene using the formula [Ct target gene - Ct housekeeping gene]. Finally, the Δ Ct values were converted to 'relative expression' for each gene using the formula [2^{- Δ Ct}] and the values were compared between young and old donor cohorts. To observe trends across the whole donor cohort, cluster analysis was performed using Cluster and Tree view softwares. The software generated a cluster in red (+3 units) indicating genes and samples with highest expression and green (-3 units) indicating genes and samples with comparatively lower expression. Relative expression values closer to the centre of the defined range (-3 to +3) were depicted in black and the values that were below detection level were depicted in grey (shown further below in Figure 4.6).

Gene symbol	Genes (full name)	Assay
RUNX2	Runt related transcription factor 2	Hs00231692_m1
SPARC	Secreted protein acidic and rich in cysteine/ Osteonectin	Hs00277762_m1
SFRP1	Secreted frizzled related protein 1	Hs00610060_m1
PPAR-y	Peroxisome proliferator activated receptor - gamma	Hs01115513_m1
FABP4	Fatty acid binding protein 4	Hs00609791_m1
LepR	Leptin Receptor, encoding CD295 protein	Hs00174492_m1
SOD3	Superoxide dismutase 3	Hs04973910_s1
TFNSF11A/RANK	Receptor activator of nuclear factor kappa-B	Hs00921372_m1
TNFSF11/RANKL	Receptor activator of nuclear factor kappa-B Ligand	Hs01092186_m1
TNFRSF11B/OPG	Osteoprotegerin	Hs00900360_m1
Cx43	Connexin43	Hs00748445_s1
CXCL12	C-X-C motif chemokine 12	Hs00171022_m1
HPRT1	Hypoxanthine phosphoribosyl transferase (housekeeping)	Hs99999909_m1

Table 4.4	List of	TaqMan p	orobes	used for	gene expressi	on stud	y in this	chapter
-----------	---------	----------	--------	----------	---------------	---------	-----------	---------

4.2.6 Statistical analysis

The normal distribution of the data was assessed using the Shapiro-Wilk and Kolmogorov-Smirnov tests for normality. Cluster analysis was used to observe any differences in gene expression across the entire donor cohort. Statistical analysis and graphics were performed using GraphPad Prism software (version 7.0a). As no data were found to be normally distributed, Kruskal-Wallis test with Dunn's correction was used for multiple comparisons, Wilcoxon rank test was performed for paired data and Mann-Whitney test was performed for unpaired data. Spearman test was used to analyse correlations. The results were considered significant at p value of <0.05.

4.3 Results

4.3.1 Age-related changes in the number of CD45^{low}CD271⁺ cells

Using the CD45^{low}CD271⁺ phenotype for native BM MSCs, their numbers were quantified for n=30 (10 young, 11 intermediate age, 9 old) donors. Once data collection was completed, the numbers of MSCs were correlated with the age of the donors. The correlation showed an age-related decline in the number of MSCs. However, the decline was non-significant (p=0.21).



Figure 4.4 Enumeration of MSCs by flow cytometry

(A) Age-related correlation of number of MSCs (CD45^{low}CD271⁺ cells) measured by flow cytometry. (B) Age-related change in number of MSCs between the age groups. Data were compared using Kruskal-Wallis with Dunn's correction. Each dot represents an individual donor. Black line in A indicates slope and in B indicate median values for number of cells per milliliter of BM

When the same data was divided into age groups (Figure 4.4B, right panel), a steeper decline was observed in the number of cells per ml of aspirate in intermediate age group (median=1326) as compared to the young donor group (median=5339) than in

the donors of the old age group (median=3006) compared to the young donor group. This was in agreement with the CFU-F findings where the number of MSCs showed a considerable decline commencing form the 5th decade of life. The trend was similar as seen in the number of MSCs detected by the CFU-F assay (as shown previously on Figure 3.4) but the changes in the number of MSCs measured by flow cytometry were found to be non-significant. This could be due to the fact that not all CD45^{low}CD271⁺ cells identified by flow cytometry may attach to plastic and proliferate into a colony of more than 50 cells.

To find how many cells identified by CD45^{low}CD271⁺ phenotype could potentially form a colony, the number of MSCs counted by CFU-F assay and by CD45^{low}CD271⁺ phenotype in donor matched samples were compared (Table 4.5). Across the whole age range, the ratio CD45^{low}CD271⁺/CFU-F suggested that 1 in every 38 CD45^{low}CD271⁺ cell (median value) detected by flow cytometry would form a colony in our laboratory conditions. However, the ratio of CD45^{low}CD271⁺/CFU-F was found to be 1 in 26 in young donors, 1 in 27 in the donors of intermediate age and 1 in 73 in old donors (Table 4.5).

	19-40 years	41-60 years	61-89	
Parameters	old	old	years old	Median
CFU-F (colonies/ml,				
medians)	206 (n=19)	48 (n=18)	41 (n=15)	81
CD45 ^{low} CD271⁺ (cells/ml,		. ,		
medians)	5339 (n=10)	1326 (n=11)	3006 (n=9)	3094
Ratio (CD45 ^{low} CD271 ⁺ / CFU-			. ,	
F, medians)	25.92	27.63	73.32	38.20

Table 4.5 Comparison of number of MSC by CFU-F and flow cytometry

This suggested that the number of colony forming MSCs within CD45^{low}CD271⁺ cell phenotype declines with age. This supported data on colony IDs presented in section 3.3.3 indicating that MSC proliferative capacity declines with age. The next step was to investigate any age-related changes in the transcripts chosen to reflect their multipotential functions.

4.3.2 MSC specificity of the selected genes

Prior to the investigation of any age-related changes in uncultured BM MSCs, the MSC specificity of the chosen transcripts was tested by comparing their expression with

CD45⁺CD271⁻ HLCs. The list of genes included transcripts for osteogenic differentiation (*RUNX2, SPARC, SFRP-1*), adipogenic differentiation and bone-fat balance (*FABP4, PPAR-\gamma* and *LepR*), bone remodelling (*RANK, RANKL and OPG*), and cell communication (*CXCL12* and *Cx43*). All of the transcripts were tested for their level of expression in both MSCs and HLCs and a high level of specificity for MSCs was observed for all the molecules in both young and old donors combined, as shown in the cluster analysis in Figure 4.5.

The cluster analysis clearly grouped all the MSCs on the left hand side and the donor matched HLCs on the right hand side (Figure 4.5). Interestingly, within each of the populations of MSCs and HLCs, there was no segregation associated with the donor age. Both, young and old donors were placed randomly by the algorithm of the cluster analysis within the cell subset specific clusters. The grey boxes indicate samples that were below detection. In the figure below, it can be seen that *RANK* was expressed in very few MSCs in MSCs while *OPG*, *SFRP1* and *SOD3* and were almost absent in MSCs. The fold changes and p values indicating the specificity of the individual transcripts for MSCs are shown in Table 4.6.



Figure 4.5 Cluster analysis of multipotentiality transcripts in MSCs and HLCs

MSCs form the left side of the cluster and HLCs form the right side of the cluster. Grey squares indicate values that were below detection and black areas indicate values that were closest to the middle of the range (-3 to +3). Dendogram on the top indicate donors and the genes investigated are shown to the right of the cluster.

Transcript	Young don	ors	Old donors		
	Fold difference	p value	Fold difference	<i>p</i> value	
RUNX2	2.5	0.0541	3.2	0.0002	
SPARC	139.5	0.0002	557	<0.0001	
SFRP-1	ND in HLCs	NA	ND in HLCs	NA	
PPAR-γ	55.2	0.0019	154.6	<0.0001	
FABP4	269.2	0.0012	122.1	<0.0001	
LepR	245.3	0.0002	2050	0.0002	
SOD3	ND in HLCs	NA	ND in HLCs	NA	
RANKL	103.8	0.0003	24.2	0.0047	
RANK	1.6	NS	10.8	NS	
OPG	ND in HLCs	NA	ND in HLCs	NA	
Cx43	21.3	0.0047	11.3	0.0005	
CXCL12	63428	0.0002	56182.2	<0.0001	

Table 4.6 Median fold increase in the expression of multipotentiality genes inMSCs compared to HLCs

Gene expression levels were measured relative to *HPRT1*. MSCs: Mesenchymal stromal cells, HLCs: Hematopoietic lineage cells, ND: not detected, NA: Not Applicable, NS: Non significant, Mann-Whitney U test

4.3.3 Age-related differences in osteogenic and adipogenic genes in MSCs

After establishing MSC specificity of the selected transcripts, the next investigation involved comparing the transcripts levels reflecting multipotentiality of MSCs between young (19-40 years old) and old (59-89 years old) age groups. The median fold differences for each of the transcripts between young and old groups are shown in Table 4.7 below. A minimum of n=8 donors in young and old group were tested for each transcript.

Table 4.7 Median expression levels and fold differences in selected genes in CD45^{low}CD271⁺MSCs and CD45⁺CD271⁻HLCs between young and old donor groups

				Fold	
	-	Medians	Medians	difference	
Population	Transcript	in YD	in OD	(O/Y)	<i>p</i> value
	RUNX2	1.40	1.10	0.79	NS
	SPARC	17.41	20.51	1.18	NS
	SFRP1	1.41	2.76	1.96	NS
	$PPAR-\gamma$	2.60	6.20	2.38	NS
	FABP4	1.71	1.76	1.03	NS
MSCs	LepR	1.93	4.12	2.13	NS
10003	SOD3	0.58	0.29	0.50	NS
	RANKL	0.16	0.35	2.19	NS
	OPG	0.99	2.43	2.45	NS
	RANK	1.36	0.76	0.56	NS
	CXCL12	1051.00	1554.00	1.48	NS
	Cx43	36.64	31.54	0.86	NS
	RUNX2	0.56	0.34	0.61	NS
	SPARC	0.12	0.03	0.25	0.0281
	SFRP1	ND	ND	NA	NA
	FABP4	0.006	0.014	2.33	NS
	$PPAR-\gamma$	0.05	0.04	0.85	NS
HLCs	LepR	0.0080	0.002	0.25	NS
	SOD3	ND	ND	NA	NA
	RANKL	0.0016	0.0146	9.13	0.0047
	OPG	ND	ND	NA	NA
	RANK	0.10	0.07	0.71	NS
	CXCL12	0.02	0.03	1.69	NS
	Cx43	1.72	2.78	1.62	NS

Gene expression levels were measured relative to *HPRT1*. MSCs: Mesenchymal stromal cells, HLCs: Hematopoietic lineage cells, YD: Young donors, OD: Old donors, ND: not detected, NA: Not Applicable, NS: Non significant; Mann-Whitney U test

Relative expression of *RUNX2, SPARC* and *SFRP-1* in MSCs showed no statistical difference between young and old donor groups (Figure 4.6). However, while the expression of *RUNX2* and *SPARC* remained largely the same based on the median values, *SFRP-1* was expressed nearly 2-fold higher in old donors MSCs compared to young donors MSCs (Table 4.7). Taken together, transcripts for osteogenic

differentiation in BM MSCs from old donors did not show significant differences in their expression compared to young donor MSCs.



Figure 4.6 Expression of osteogenic and adipogenic transcripts in MSCs from young and old donors

(A) Expression of *RUNX2*, *SPARC* and *SFRP-1* in MSCs. (B) Expression of *PPAR-\gamma*, *FABP4* and *LepR* in MSCs. Young and old donors are indicated in black and grey dots, respectively. Horizontal lines indicate median values.

For the selected adipogenic transcripts, $PPAR-\gamma$ showed 2-fold higher expression in old donors but *FABP4* did not show any age-related differences (Figure 4.6 and Table 4.7). The level of *LepR* also was 2-fold higher in old donors however the differences failed to reach statistical significance (Figure 4.6 and Table 4.7). Overall, data for selected adipogenic molecules in uncultured BM MSCs indicated a potential trend for increase in some transcripts which merits further investigations in a larger cohort of donors. Interestingly, small increases in *PPAR* γ and *LepR* were not mirrored by similar increases in *FABP4*.

4.3.4 Resistance to oxidative stress

Resistance to MSC to oxidative stress was measured by quantification of basal intracellular ROS in CD45^{low}CD271⁺ cells by flow cytometry as well as ROS generated by exposing MSCs to TBHP using methods previously established in our group (315). It was hypothesised that the levels of basal ROS and ROS generated using TBHP would display an age-related increase in MSCs. To potentially explain any differences in ROS, the expression of the anti-oxidant enzyme SOD3 was also quantified by qPCR. *SOD3* expression levels were expected to display an age-related decline that could explain the increase in ROS with advancing age.



Figure 4.7 Measurement of ROS and anti-oxidant enzyme SOD3 gene in MSCs from young and old donors

(A) Increase in ROS before and after stimulation with TBHP measured by CellROX median fluorescent intensities (MFI) in young donors (left) and old donors (right), Wilcoxon rank sum test. (B) Comparison of ROS levels between young and old donors before stimulation with TBHP (left) and after stimulation (right). (C) Comparison of expression of anti-oxidant enzyme *SOD3* in young and old donors by gene expression. Young and old donors are indicated in black and grey dots, respectively. Wilcoxon rank sum test, *p<0.05.

Comparing stimulated and unstimulated levels of ROS in young and old donor MSCs (Figure 4.7A) showed a significant increase in stimulated state indicating that the exposure to TBHP led to the production of more free oxygen radicals that were detected as higher levels of CellROX. After showing that ROS can be measured by this method in MSCs before and after stimulation, basal levels of ROS were first compared between young and old donors. The basal levels of ROS is indicative of the inherent mitochondrial activity in cells and showed no difference in old donor MSCs compared to young donor MSCs (Figure 4.7B, left panel). When the level of ROS was quantified in the stimulated state, indicating a cells ability to combat extreme oxidative stress, there was similarly no statistically significant difference (Figure 4.7B, right). However, there was a 2-fold decline in the level of stimulated ROS when old donor MSCs were compared to young donor MSCs, based on median values (Figure 4.7B, right panel) and the expected increase in ROs was not observed.

SOD3 the anti-oxidant enzyme is the first line of defence in presence of harmful ROS and is known to be specifically secreted by MSCs. To investigate if this gene declined with advancing age, its expression was compared between young and old donor MSCs using qPCR. Here again, the difference was not statistically significant but the level of SOD3 transcript was 2-fold lower in old donor MSCs (Figure 4.7C).

4.3.5 Age-related differences in bone remodelling and cellular interaction genes in MSCs

The transcripts associated with bone remodelling also showed no statistically significant differences between young and old donor MSCs. As seen from the cluster analysis (Figure 4.5) *RANK* expression was specific for HLCs and barely detected in MSCs. This was expected, as *RANK* is expressed primarily on T cells (section 1.1.4) (63, 316). In contrast to *RANK*, MSCs from old donors showed an over 2-fold increase in the expression of both, *RANKL* and *OPG*, as compared to young donors (Table 4.6). Of note, *RANKL* was detected in only 6 out of the 8 donors MSCs and *OPG* was detected in 7 out of 8 donors MSCs.

Overall, even though statistical significance was not achieved, there was a trend indicating an age-related increase in *RANKL* and *OPG*, in uncultured BM MSCs from old donors (Figure 4.8).



Figure 4.8 Expression of transcripts associated with bone remodelling and cellular interactions in MSCs from young and old donors

(A) Expression of *RANK*, *RANKL* and *OPG* (B) Expression of *CXCL12* and *Cx43*. Young and old donors are shown in black are grey dots, respectively. Horizontal lines across the data indicate median values.

The transcripts associated with cellular interactions were next investigated. *CXCL12* is involved in chemotaxis of cells within the BM (section 4.1.1.5), and its expression level was among the highest in CD45^{low}CD271⁺ MSCs in comparison to other transcripts as seen in the cluster (Figure 4.5) suggesting its presence in high quantities in uncultured BM MSCs. No difference in the level of its expression was found between young and old donors (Figure 4.8B, left panel). *Cx43* is involved in intercellular interactions in cells within BM (section 4.1.1.5) and also showed no age-related differences in its gene expression levels in CD45^{low}CD271⁺ MSCs.

4.3.6 Age-related differences in selected gene transcripts in HLCs

The HLCs were sorted as CD45⁺CD271⁻cells and included the lymphoid and myeloid cell lineages (188, 222). In this project, donor matched CD45⁺CD271⁻ cells served as a control group for uncultured BM MSCs. As shown in Figure 4.5, the specificity of the

chosen transcripts was very high for MSCs, with the exception of *RANK*. However, with prior knowledge, that most of these transcripts are also expressed by other cells (even if at the low levels, section 4.1.1) their gene expression was next investigated in HLCs and compared between young and old donor groups.

For osteogenic transcripts, while *RUNX2* and *SPARC* were detected in low levels in HLCs, the expression of *SFRP1* was below detection in HLCs further illustrating selectivity of these genes for MSCs. There was 1.6-fold age-related decline in the expression of *RUNX2*, however not statistically significant, and *SPARC* (4-fold, p=0.0281) in older donors HLCs (Figure 4.9A)



Figure 4.9 Expression of genes associated with osteogenic and adipogenic differentiation in HLCs from young and old donors

(A) Expression of *RUNX2*, *SPARC* and *SFRP1* (B) expression of *PPAR-\gamma*, *FABP4* and *LepR*. Horizontal lines across the data indicate medians. Young and old donors are shown in black are grey dots, respectively. *p*< 0.05^{*}, Mann-Whitney U test.

The transcripts related to adipogenic differentiation and bone-fat balance were next analysed. All three transcripts were detected in higher levels in MSCs compared to donor matched HLCs, as seen in cluster (Figure 4.5). *PPAR-\gamma* did not show any age-related trends in the HLCs. *FABP4* exhibited a slight trend towards an age-related

increase, whereas LepR displayed considerable 4-fold age-related decline in HLCs, which narrowly missed statistical significance (Table 4.6, p=0.0541).

Bone remodelling transcripts were next scrutinised (Figure 4.10A). *RANKL* showed a significant 9-fold increase (p=0.0047) in old donor HLCs. As mentioned, *OPG* was not detected in HLCs (confirming it's high-level specificity for MSCs). The last group of genes to be investigated in HLCs were those involved in cellular interactions. Both the genes, *CXCL12* and *Cx43* were detected in young and old donor HLCs but showed no difference in expression (Figure 4.10B).



Figure 4.10 Expression of transcripts associated with bone remodelling and cellular interactions in HLCs from young and old donors

(A) Expression of *RANK*, *RANKL* and *OPG*. (B) Expression of *CXCL12* and *Cx43*. Horizontal lines across the data indicate medians. Young and old donors are shown in black are grey dots, respectively. $p < 0.01^{**}$, Mann-Whitney U test.

Similar to the observations in MSCs, the expression of all transcripts in HLCs exhibited large donor variation within the same age group. Altogether, there were no statistically significant differences observed in transcript expression levels in young and old donor HLCs for most transcripts, with the exception of *SPARC* which decreased and *RANKL* which increased in old donors. This was in contrast to MSCs, in which only a trend for increase in *PPAR-*^{γ}, *LepR*, *RANKL* and *OPG* was found in older donors.

Two of these transcripts were also investigated at the protein level (LepR/CD295 and Cx43) along with other potential MSC age-related surface molecules CD106 and CD146 (section 4.1.2). The next section includes the results obtained from the investigation of these surface markers in uncultured BM MSCs and control HLCs.

4.3.7 Expression of the selected surface markers on MSCs

Frozen-thawed nucleated cells from young and old donors (Appendix 1, Table 2) were processed for native MSC isolation and characterisation using the CD45^{low}CD271⁺ phenotype. As a control population, CD45⁺CD271⁻ cell fraction (HLCs) that contained the lymphoid and myeloid lineage cells was also analysed (188). First, the specificity of the selected surface markers for MSCs (CD45^{low}CD271⁺ cells) was tested. For the measurement of the surface marker expression the median fluorescent intensity (MFI) was used. Representative histograms from each donor group are shown in Figure 4.11.



Figure 4.11 Quantification of surface marker expression in BM MSCs and in HLCs

(A) Gating strategy for identification of live cells based on 7-AAD negativity followed by CD271 and CD45 gating for the selection of populations of interest (MSC and HLCs)
(B) Histograms for individual surface markers in MSCs and HLCs from representative donor. Numbers in the top-right corner indicate MFI.

The MFI of each surface marker was first compared between MSCs and HLCs. Figure 4.11 below shows the expression of each surface molecule in donor matched MSCs and HLCs from all young and old donors, combined.



Figure 4.12 Expression of candidate age-specific surface markers on MSCs and HLCs from all donors combined

(A) Expression of CD106 (B) Expression of CD295 (C) Expression of CD146 (D) Expression of Cx43. Each panel shows the expression of the respective marker in MSCs and HLCs in donor-matched cells. Each dot represents an individual donor. *p<0.05 and ****p<0.0001, Wilcoxon rank sum test

All surface markers were expressed in significantly higher levels in MSCs compared to HLCs from all donors. The differences in CD106 (p<0.0001), CD295 (p<0.0001) and CD146 (p<0.0001) were highly significant indicating very high specificity for MSCs. CD106 and CD295 showed the highest specificity in terms of fold differences (10-fold each, Figure 4.12A and B), followed by CD146 (nearly 4-fold, Figure 4.12C). While Cx43 showed the lowest fold differences between expression in MSCs and HLCs (less than 2-fold) and the p value (p=0.0266), but remained significantly specific for MSCs, nevertheless (Figure 4.12D).

The expression of CD295 in MSCs was the highest (ranging between 10^1 and 10^3 MFI) while the expression of CD146 and Cx43 were the lowest, both of which were below 10^2 MFI. All antibodies were used at the manufacturer's recommended concentrations. CD295 and Cx43 specificity at the protein level was also reflected at

the gene level (Figure 4.5 and Figure 4.11, Table 4.6) suggesting consistency in techniques used and results obtained.

4.3.8 Age-related differences in selected surface markers in MSCs

Expression of all four surface markers was then compared between young and old donors, in MSCs and then HLCs separately. No difference in the expression levels in any of the markers in MSCs was found between young and old donors (Figure 4.13). Large donor variation was observed in the expression of these markers within each donor group.



Figure 4.13 Expression of candidate age-related surface markers in MSCs from young and old donors

(A) Expression of CD106 (B) Expression of CD295 (C) Expression of CD146 (D) Expression of Cx43. Young and old donors are represented by black and grey dots, respectively. Each dot represents an individual donor. Horizontal lines across the data represent median values.

Neither of the markers demonstrated any trends when compared between young and old donor MSCs suggesting no age-related difference in the expression of these markers in uncultured BM MSC. The expression of these markers were then investigated in young and old donor HLCs.

4.3.9 Age-related differences in selected surface markers in HLCs

Expression of the four surface markers in HLCs also displayed no age-related difference in their expression as shown in figure 4.14 below.



Figure 4.14 Expression of candidate age-related surface markers in MSCs from young and old donors

(A) Expression of CD106 (B) Expression of CD295 (C) Expression of CD146 (D) Expression of Cx43. Young and old donors are represented by black and grey dots, respectively. Each dot represents an individual donor. Horizontal lines across the data represent median values.

Even in HLCs, the surface markers showed no differences in the level of expression in young and old donors. Overall, section 4.3.7 to 4.3.9 displayed significantly higher expression of surface markers in MSCs than in HLCs. With respect to age-related differences, the expression of these age-specific markers displayed no age-related differences or trends in either of the cell types (MSCs and in HLCs).

4.4 Discussion

In this chapter, an age-related decline in the number of CD45^{low}CD271⁺ native BM MSCs using flow cytometry was established confirming the results from previous chapter based on the CFU-F assay. Gene expression of all the selected molecules (apart from *RANK*) showed very high specificity for MSCs as compared to HLCs. The osteogenic transcripts' expression showed no age-related trends in MSCs. Adipogenic transcripts also did not reach significance between young and old groups but displayed a non-significant 2-fold increase in the expression *LepR* and *PPAR-* γ in the old donor group. Interestingly, unlike in the MSCs, *LepR* displayed a significant 2-fold decrease in the HLCs. Quantification of ROS and anti-oxidant enzyme *SOD3* gene expression also did not show any statistically significant difference in old donors, however, the level of *SOD3* detected in old donors as well as the stimulated level of ROS detected in old donors, declined by 2-folds. Transcripts for cellular interactions also did not show any age-related changes in the MSCs or the HLCs. Similarly, surface marker expression exhibited no difference between young and old donor groups and each surface marker was significantly higher-expressed on MSCs compared to HLCs.

Flow cytometry data revealed a trend for native BM MSC to decline with age. However, the decline was not as significant as indicated by the CFU-F assay. Flow cytometry analysis depends upon the gating strategy which can be subjective. However, the analysis and gating strategy used in this study were the same as previously published by our group (111, 234). As shown in previous studies (111, 207), not every CD45^{low}CD271⁺ cell forms a colony in standard CFU-F conditions, and consistent with these publications, flow cytometry data revealed higher number of MSCs per ml of BMA compared to CFU-F assay. The age-related decline was lower as measured by flow cytometry possibly because it identifies all the CD45^{low}CD271⁺ cells as MSCs, irrespective of their ability to proliferate into a colony from a single cell; while a colony having less than 50 cells was not scored as a colony in the CFU-F assay. Overall, the data on BM MSC enumeration by flow cytometry in this study confirmed their age-related decline as measured by CFU-F assay. This has not been shown before when other surface markers for uncultured MSCs were used (237).

Use of CD45^{low}CD271⁺ cells for enumerating MSCs has been established on fresh BM cells before and used by our group (111, 207) and the data was compared to the total colony area showing a positive correlation. The studies however did not look into age-related differences in MSC number by both CFU-F and flow cytometry. Recently, our group also measured CD45^{low}CD271⁺ cells by flow cytometry and CFU-F assay and observed an age-related decline in females but not in males by both assays (111).

However, the study was aimed at comparison of two flow cytometry techniques for evaluation of CD45^{low}CD271⁺ MSCs and was not specifically focussed on age- or gender-related differences. Furthermore, it used a smaller cohort of donors ranging from 22-80 years old. In agreement with data presented in this chapter, recently, Josephson and co-workers also showed a decline in CD45^{low}CD271⁺ cells by flow cytometry when compared across the entire age range (24-89 years old) as well as between young (<50 years) and old (>50) donors (146). However, their group definitions were different and they were not able to show the bigger drop in MSCs occurring from early 40s rather than later in life (from 60s).

A group of transcripts was next chosen to explore if various functions in MSCs may be affected by donor age. Many of the molecules have been previously shown to be specific for MSC by our group (188). In fact, an age-related decline in *LepR and SFRP1* was observed in our previous study using a cohort of patients with ages ranging from 2 to 74 years old. This was not seen in this study with uncultured BM MSCs, most likely due to different age-range of this study that did not include paediatric donors, as in the former.

In this project, there were no or non-significant age-related trends in the transcripts related to osteogenic and adipogenic differentiation. Based on previous studies, the general consensus has been that there is an age-related decline in the osteogenic potential of MSCs paralleled to an increase in their adipogenic potential (Table 4.2). Studies that supported this concept have been based on *in vivo* assays by examining trabecular bone volume, bone mineral density (BMD), total bone volume and marrow adipocyte content. These parameters have been discussed in the introduction in section 1.1.4. These studies provide robust *in vivo* data supporting the process of 'age-related bone loss' and increased bias towards marrow adipogenicity, and formed the foundation of our initial hypothesis for reduced osteogenic and increased adipogenic commitment of uncultured BM MSCs. As these cells were not expanded in culture, they better represent the *in vivo* MSCs in BM, making the data novel and reliable.

However, *in vitro* data relating to both donor-related and passage-related MSC ageing in terms of their osteogenesis remained controversial. Yang and co-workers in 2018 compared the differentiation potential of BM MSCs between passage 4 and passage 8 (123). They tested the differentiation potential using both, *in vitro* differentiation assay and by using gene expression of classic osteogenic and adipogenic markers (*RUNX2* and *PPAR-* γ , respectively). They found that cells from passage 8 were able to maintain the adipogenic potential, but the osteogenic potential was higher in passage 4 MSCs. It

was concluded that the process of adipogenic differentiation is 'better preserved' in BM MSCs than osteogenic differentiation and the growth medium had a strong impact on their differentiation potential, further highlighting the limitation of using cultured MSCs in MSC ageing studies. The present data using a limited number of transcripts have not provided strong evidence for any bias in MSCs, however interesting trends in *PPAR-* γ in particular were observed that supported our original hypothesis. Similar investigation of osteogenic and adipogenic molecules at the protein level using with more donors would help to understand the age-related shift towards adipogenic differentiation in bone *in vivo*.

Another molecule that plays a role in osteogenic-adipogenic balance in the BM is *LepR* or CD295. In this project, LepR was tested at both the transcript and protein levels. By both techniques, LepR showed a trend for a two-fold increase in MSCs from old donors. Before discussing the biological relevance of this data, it is important to note that CD295 was proven to be the most specific of the all tested molecules for human CD45^{low}CD271⁺ MSCs (10-fold higher in MFI compared to control HLCs). Yue and colleagues published a paper on LepR+ BM MSCs in mice showing that *LepR* is not needed for the maintenance of MSCs but acts on MSCs to inhibit osteogenesis and promote adipogenesis (262). They also found that the mice fed with a high fat diet not only had increased levels of plasma Leptin, but their bone-lineage progenitors formed significantly lower CFU-Fs in comparison to the mice that were not fed the high fat diet. This indicated that high serum Leptin may directly inhibit MSC colony formation ability. This was the first study that had connected intake of high fat diet with the impact of LepR+ cells on the regeneration ability of these cells. However, as only adult mice (6mo) were used, no correlations with donor age could be made in this study.

In 2009, Laschober and co-workers studied the expression of CD295 by flow cytometry in cultured BM MSC obtained from donors aged between 7-78 years old without any age grouping. Their findings suggested that CD295 was up-regulated in the 'dying cells' marked by Annexin V (marker for cell death by apoptosis) and that it was found to be higher in old donor BM MSC cultures. They suggested that CD295 is an indicator/marker for apoptotic cells (which are higher in old donor cultures) in BM MSCs (126).

The data from the present study showed a tendency of both, the receptor expression and the gene encoding for the receptor to increase in BM MSCs in older donors, but the differences failed to reach statistical significance. A similar trend was observed for the adipogenic molecule $PPAR-\gamma$. Should this be statistically confirmed in a larger cohort study, CD295 expression should be further tested for predicting the adipogenic fate of BM MSCs *in vivo*. Even though the change was not significant, it must be noted that these experiments were performed with nearly no laboratory manipulation of donor MSCs precluding any artefacts due to *in vitro* culture conditions and thus better reflecting MSC age-related changes *in vivo*. The present findings are interesting and complement previous studies showing that its ligand, the hormone Leptin itself, shows an age-related increase in its blood levels (317). This may lead to an overall enhancement of Leptin signalling in MSCs and negatively affect their colony-formation potential.

Knowledge in the increased adipogenic bias of BM MSCs *in vivo* and its potential link to age-related changes of BM MSCs is vital for developing new treatment for diabetes, obesity, osteoarthritis and osteoporosis, the risks of which are much higher in the aged population. The risk of obesity is higher for new-borns that are born to obese mothers and the MSCs from these babies show a greater potential to become adipogenic (318). While the exact mechanism for MSC fate decisions remains to be discovered, literature from past and present projects shed new light on the topic. Future investigations should include obese and non-obese donor cohorts, which quantify both circulating Leptin levels along with *LepR* from BM MSCs at the gene and protein level. These future studies will help to dissect the age-related and diet/obesity-related changes in bone homeostasis due to Leptin and its impact of age-related skeletal disease like osteoarthritis.

As previously stated (section 4.1.1.3), several studies have reported increased ROS during adipogenic differentiation of MSCs as compared to osteogenic differentiation. Accumulation of ROS is also one of the main features of aged cells (281). Based on these considerations, this project investigated the levels of ROS and the expression of anti-oxidant enzyme SOD3 in BM MSCs in young and old donors. *SOD3* transcript was highly specific to MSCs and was not detected in HLCs (Table 4.6). Therefore we expected an age-related increase in the level of ROS in old donors, parallel to a decrease in *SOD3*.

Even though the results were not statistically significant, the decline in the induced levels of ROS in old donor MSCs as compared to young donor MSCs, was unexpected. A 2-fold decline was observed in the expression of *SOD3* in old donor MSCs but it did not reach statistical significance. Recently, Agrahari et al., found that MSCs engineered to overexpress SOD3 were better adapted for survival under serum starved conditions, possibly due to enhanced autophagy (319). Culture expanded

MSCs from old donors (96) as well as long-term cultured BM MSCs have shown to possess lower levels of anti-oxidants such as SODs, as well as increased vulnerability to oxidative stress (320, 321). However, measurement of active ROS and anti-oxidant enzyme expression in uncultured BM MSCs has not been carried out before, making the data from this chapter very novel.

Overall, with respect to the idea that MSCs display an age-related bias towards adipogenic differentiation and there is increased ROS in older donors, this chapter suggests no or only subtle differences in uncultured BM MSCs. While larger donor group numbers could potentially display statistically significant results, the data suggests that at the transcript level, BM MSCs are not necessarily or strongly biased towards adipogenic differentiation. An additional sample set previously treated in hypoxic conditions or with anti-oxidant before inducing oxidative stress in BM MSCs would help explain the decline in both ROS levels and *SOD3* as seen this project. Additionally, exploring other anti-oxidant molecules like *Gx*, *SOD1* and *SOD2* at gene and protein levels would also help in understanding age-related differences in dealing with oxidative stress.

In regards to bone remodelling transcripts, subtle non-significant increases in both *RANKL* and *OPG* in old donors were found. Principally, increased levels of *RANKL* suggest increased bone resorption, therefore the obtained data are contradictory but remain interesting in relation to BM MSC adipogenesis. The idea that increased BM adipogenesis and bone remodelling are inter-related and impact each other is not completely new. The evidence for this has been furnished in Takeshita *et. al* who suggested that the increased BM adipogenesis is linked to the age-related increased expression of *RANKL* (238). They used uncultured whole BM cells from mice and isolated stromal cells and macrophages. They found increased bone resorption and adipogenic bias *in vivo*. They also found an increase in the expression of *PPAR-* γ and other adipogenesis related markers with a 3-fold age-related increase in the level of *RANKL* in whole BM cells. Investigating macrophages from human donors will provide with more comprehensive knowledge about age-related differences in bone remodelling within the BM.

Apart from testing this on uncultured cells, they also used murine BM stromal cell line ST2. The baseline expression of *RANKL* in this cell line was low and the level of *OPG* expression was very high. When the cell line was subjected to adipogenic differentiation, the level of *RANKL* increased and *OPG* decreased, completely

reversing their expression levels. While this study confirmed the age-related adipogenic bias and increased bone resorption in C57BL/6 mice aged 2 months (young) and 2 years (old), animal studies are often criticised for the lack of IVIVC (in vitro in vivo correlation) with humans. This is because organ systems and genetic makeup of animals are not identical to humans. In the presented study, the *RANKL* increase in old donor MSCs was expected, but *OPG* increase was unexpected highlighting such differences between animals and humans.

The transcripts associated with cellular interaction have also shown no age-related changes in the present study. Recent examination by Gomariz and co-workers, suggested that CAR and sinusoidal endothelial cells within the BM compartment of mice remain mostly unaltered and show no age-related difference when inspected for quantitative and structural analysis (322). While they did note an insignificant agerelated decline in the number of CARs in old mice (20-24 months old) compared to young (8-12 weeks old) mice, the difference was minor suggesting that CXCL12 abundant (CAR) cells remain largely preserved in old age. Interestingly, they defined CAR cells by the CD295+ phenotype, which confirms similar cell identity to MSCs as identified in the present study. While there were a good number of studies outlining the CXCL12-CXCR4 signalling and its potential role in BM homeostasis in mice (239, 240), it is difficult to find many studies aimed at investigating age-related changes in this chemokine in cultured or uncultured human BM MSCs. Against our original hypothesis, the data presented in this chapter displayed no age-related decline in the expression of CXCL12. Considering the high levels at which CXCL12 is expressed, detection and quantification of the transcript in large number of donors should not be a hurdle and will be interesting as the present data indicate a very subtle age-related increase of CXCL12 expression in BM MSCs.

With respect to Cx43, the most abundant connexin within bone, no age-related differences were found in this study both at transcript and protein levels. Kar et al., showed age-related decline in mouse osteocytes (323) and more recently, Davis and colleagues have suggested that microRNA21 (miR21) is potentially responsible for this osteocyte Cx43-mediated age-related bone loss (305). Cx43 in osteoblasts has been shown to play an important role in bone remodelling by controlling both bone resorption and bone formation (324) and another study suggested that Cx43 in bone development acts in coordination with RUNX2 (325). All of the aforementioned studies were performed on differentiated cells, suggesting that techniques for detecting age-related changes in uncultured progenitors are often not attempted and need validation in humans.

Cx43 has been also implicated in controlling the damaging effects of ROS within the stem cell niche. Grayson et al., observed a significant increase in the *Cx43* immunocytochemistry staining in human BM MSCs cultured in hypoxic conditions. They suggested that *Cx43* potentially contributes to maintaining MSC proliferation and 'stemness' (326). On a similar line of thought, Ishikawa et al have shown the protective role of *Cx43* for HSCs within the BM of mice. They suggested that *Cx43* mediates the transfer of ROS from HSCs to the stromal cells within the BM thus protecting HSCs from damage. Similarly, Kar and colleagues had shown that *Cx43* contributes to the protection of osteocytes against oxidative stress (323).

Based on these studies, our proposed hypothesis was for an age-related decline in Cx43 in uncultured BM MSCs. However, the investigation at both, transcript and protein level did not reveal any age-related differences. Examination of the molecule using a different technique may be needed to further strengthen our conclusion of no age-related differences. Owing to the presence of *Cx43* at cellular junctions, immunostaining techniques or the use of confocal microscopy that has labelled Cx43 at cellular junctions would be a useful tool to investigate and quantify the intensities of stains between young and old donor MSCs.

Investigations of multipotentiality genes in HLCs displayed a significant decline in *SPARC* and a significant increase in *RANKL* in old donors. SPARC is also known to be an immunomodulatory molecule and thus it is not unexpected to observe an agerelated decline in *SPARC* in HLCs (316). *RANKL*, as mentioned earlier is expressed on T cells (316). An increase in *RANKL* suggests increased bone resorption. Higher bone resorption with advancing age is expected owing to overall age-related bone loss that has been established in literature (62, 81). The data presented in this chapter with respect to gene expression in HLCs was thus expected and supported the decline in immunomodulatory properties and increasing bone resorption observed with advancing age.

With regards to surface markers, CD106 and CD146 were chosen based on previous evidence of their altered expression in *in vitro* aged MSCs (Table 4.3). For CD106, the results were far from conclusive, with all the studies reported very high donor variation, even within the same age groups (123). Liu *et. al* used human BM MSC cell lines and found significant downregulation of CD106, at both transcript and protein levels, when cells were expanded *in vitro* (327). In contrast, Laschober and co-workers found an upregulation of CD106 at both transcript and protein levels in donors aged over 60

years old (138). They also found that CD106 was highly expressed in MSCs when exposed to inflammatory cytokines TNF- α or IFN- γ (138). Other studies have linked CD106 in MSCs with various immune and inflammation related changes (311, 328), suggesting a potential link between CD106 and inflamm-ageing within BM niches. Finally, Siegel and co-workers found a significant age-related decline in CD106 across a broad age range (20-80 years old) (97).

Since the literature pertaining to cultured MSCs remained contradictory, it was interesting to investigate CD106 expression in uncultured MSCs. The current study showed no age-related difference in the level of CD106 expression in uncultured CD45^{low}CD271⁺ BM MSCs. In agreement with previous studies, this study too found the highest donor variation in the expression of CD106 compared to other markers. In future, investigations in larger number of donors and at the protein level will further increase our understanding.

Similar to CD106, CD146 expression on CD45^{low}CD271⁺ MSCs in the present study also showed no age-related differences. This is in contrast to cultured MSC data, which commonly showed passage- or donor-age related decline. For example, Siegel *et. al* found a donor age-related decline in CD146 expression in BM MSCs at passage 1 (97). More recently, Yang and co-workers examined CD146 expression in culture expanded BM MSCs between passages 4 and 8 and found a passage-related decline in its level expression (123). In contrast, Hagmann and co-workers did not observe any age-related change in the expression of CD146 in culture expanded MSCs (329). Only one study looked at the CD146 MSC expression *in vivo* and found that the CD271⁺CD146⁺ cells were reduced in frequency in old donors (>55 years old) compared to young donors (19-55 years old) (229).

In conclusion, the present data on CD106 and CD146 suggests that using cultureexpanded MSC data for the search of MSC 'ageing' marker *in vivo* is unlikely to be very fruitful. Other methods, for example using cell surface proteomic approach (330) and combining histological staining with flow cytometry for quantifying cell surface target proteins present on senescent MSCs (331) are more likely to provide interesting data (131).

140

Chapter 5 Expression of type 1 interferon (IFN1) and senescence genes in human BM MSCs

The previous chapter described a non-significant age-related decline in number of BM MSCs measured using the CD45^{low}CD271⁺ phenotype and also examined differences between young and old donor MSCs in level of ROS, anti-oxidant enzyme SOD3 transcript, and selected multipotentiality genes and surface markers. Investigation of SOD3 in MSCs from old donors interestingly revealed a two-fold decline but differences were not statistically significant. Similarly, some trends were observed in the expression of *PPAR-* γ , *LepR*, *RANKL and OPG* (over 2-fold increase in older donors), however they too did not reach statistical significance. Unlike MSCs, HLCs displayed a 4-fold decline in osteogenic gene *SPARC* and 10-fold increase in bone remodelling gene *RANKL* in old donors, both of which were statistically significant. This indicated towards the fact that age-related gene expression differences for multipotentiality and possibly other genes could be more pronounced in HLCs as compared to MSCs.

Age-related changes in HSCs and their lineages are better understood owing to the larger number of investigations performed as compared to MSCs. Among these age-related changes, the decrease in lymphoid cells and increase in myeloid cells (myeloid skewing), loss in proliferation and increased DNA-damage in old donors have been well documented (332-334). The theories explaining these differences in aged HSCs are the same that describe MSC ageing (DNA damage, ROS accumulation, increased senescence and others, as explained in section 1.1.1). More recently, the IFN1 pathway has attracted an increasing interest as a potential link between DNA-damage response (DDR), oxidative stress, senescence and ageing in HSCs (335, 336).

As mentioned, HSCs are the progenitors for HLCs which eventually give rise to lymphoid and myeloid lineages. Considering that HSCs give rise to HLCs and that these cells share the same BM niche with MSCs, this chapter aims at investigating new exploratory genes emerging from the HSC 'ageing' field and including those associated with senescence and the IFN1 pathway, in MSCs as well as in HLCs. As in the previous chapter, this chapter also explores any age-related changes in the expression of these genes in uncultured cells and compares young donor and old donor cohorts. Finally, the chapter also aimed at inspecting if age-related changes were more pronounced in HLCs compared to MSCs, as observed in the previous chapter.

5.1 Introduction

Due to the relatively higher number of studies in HSCs, age-related differences in this cell type have been explored to a greater extent than for MSCs. Previous investigations in HSCs, have revealed an age-related increase in the myeloid lineage cells along with a decline in HSC proliferative capacity (337). The tendency to produce less lymphoid lineage cells and more myeloid lineage cells is known as 'myeloid skewing' and potentially contributes to age-related increase in vulnerability to infections (332). Interestingly, the hypothesised mechanisms of these changes that include DNA damage, stem cell exhaustion and the effects of ROS (28, 338) are no different from the hypothesised mechanisms of age-related differences in MSCs that were described in Chapter 1 Section 1.1.1 in theories of ageing. Considering that both MSCs and HSCs reside within the same BM niche (section 1.1.3), it is possible that these damaging factors act on both cell populations (339).

5.1.1 Senescence, SASP and age-related bone loss

Residing within the same niche, MSCs are known to support activities of HSCs including their myeloid and erythroid differentiation and early lymphopoiesis (340). MSCs reside together with other cells in a regulated BM niche that helps maintain the functioning of the microenvironment (Chapter 1, section 1.1.4). MSCs are also known to directly and indirectly modulate and control T cell responses (341) and to inhibit the proliferation and differentiation of B cells (342, 343).

With advancing age and accumulation of ROS and DNA damage in ageing cells, an increasing number of BM cells become senescent (344). Senescence is the state of irreversible growth arrest and is a regulated response to cellular insults like ROS and DNA damage (345). Senescent cells have been identified by various methods including an increase in SA- β -gal, Prelamin A, cell surface lipids like Lipofuscin, Tp53, critically short telomeres as well as differences in gene methylation patterns (131). Tp53 is a transcription factor well known for its function in cell-cycle, DNA-repair, and inducing cell growth as well as apoptosis (346, 347). While the presence of Tp53 is vital for maintenance of cells including BM MSCs (348), excessive levels of the transcription factor are known to be detrimental to healthy ageing (346). Aligned with this, our group found an age-dependant increase in the levels of Tp53 quantified by flow cytometry in culture expanded BM MSCs (96). Considering the aforementioned knowledge, Tp53 was hypothesised to be expressed in higher quantities in old donor cells as compared to young donor cells.

Along with increase in ROS and DNA damage, senescence is also characterised by the senescence associated secretory phenotype (SASP) (130), which includes the release of pro-inflammatory cytokines (like IL6, IL8) and metalloproteinases (MMPs) from senescent cells (345), and can then induce senescence and inflammatory responses in neighbouring cells (131). Senescent MSCs release higher amount of SASP, which reduces the multifunctionality of MSCs (80, 349) as well as accelerates the rate of ageing in younger cells (350). Interestingly, the level of serum IL6 has more than often been found to be higher in older donors as well as in donors with age-related diseases (214, 351, 352) and is an accepted marker of SASP and inflamm-ageing. Inflamm-ageing is defined as inflammation associated with increasing age and age-associated diseases and is explained further below (215, 353). With respect to MSCs, over-expression of IL6 has been shown to disrupt their role in maintaining HSC functionality and homeostasis (354). Thus in this project, the expression of *IL6* was hypothesised to display an age-related increase in both cell populations.

IL7 is essential for B cell development, survival and proliferation (355) and has long been known to be secreted in high quantities by BM MSCs (356). While it is not classified as a pro-inflammatory marker, its role in supporting B cell development in the BM and thus contributing to BM niche homeostasis makes it an important cytokine to investigate. In culture expanded BM cells, IL7 activity was found to drop dramatically in old mice (357) while in synovial fluid of old human donors, the level of IL7 was found to increase (358). Thus, in spite of knowing that IL7 is majorly produced by BM MSCs and its role in B cell formation and development, what happens to its levels in relation to advancing age remains unclear. Knowing that B cells functionalities decline with increasing age even though the change in number remains debateable (359), it was hypothesised that *IL7* expression will show an age-related decline in both cell types.

Along with IL6, IL8 too has been identified as an indicator of SASP (360, 361). While the number of investigations in relation to ageing are relatively fewer for IL8 compared to IL6, it is a known pro-inflammatory marker associated with high stress levels, is increased proportionately with age in urine samples (362) and in cancer patients (363). In BM MSCs co-cultured with colorectal cancer cells, IL8 production was found to enhance the proliferation tumour angiogenesis (364). Thus in this project, the expression of IL8 was hypothesised to be higher in old donor cells as compared to young donor cells.

Within the bone, increase in SASP has been shown to correlate with advancing age in myeloid as well as in osteoblastic cells of aged mice (24 months) when compared to

young mice (6 months) (365). Increased senescence, adipogenic shift of MSC differentiation, increased oxidative stress are all associated with age-related factors giving rise to increased levels of inflammation, which is now referred to as 'inflamm-ageing' (353). Inflamm-ageing conceptually postulates that increasing age is associated with an intrinsic decline of the proliferation potential of immune cells, their increased senescence and DNA damage. This is accompanied with the reduction of protective extrinsic mechanisms within the cellular microenvironment that aim at removal of senescent cells and accumulated damage, that is, autophagy (366). As mentioned in section 1.1.1, autophagy is the body's internal mechanism for the removal of senescent cells and damaged cell organelles by lysosome degradation. This is known to prevent the accumulation of DNA damage in the young and healthy (21). However, with the passage of time and increasing age, efficiency of autophagy declines. This means that the body's inherent capacity to identify and eliminate senescent cells/damaged organelles, also declines (21).

As autophagy declines, the number of senescent cells increase and so does the accumulation of damaged cells and organelles. These senescent cells release proinflammatory cytokines and growth factors (SASP), that ultimately contribute to inflamm-ageing (366). Expectedly, the understanding of senescence, autophagy, inflamm-ageing and their contribution in age-related diseases, makes senescence a promising target for anti-ageing strategies (366). These strategies may use seno-therapeutics aimed at removal of senescent cells by killing senescent cells (senolytics), or by modulating SASP (senomorphics) or by the clearance of senescence cells mediated by the body's immune system (seno-inflammation) (367). While senescence and SASP can indicate 'ageing' when they increase in body, anti-ageing genes also play an equally important role in maintaining 'youthful' characteristics of cells (368).

Sirt6 is an anti-ageing gene (369) recognised for regulating DNA breaks and damage repair (370) and its over-expression was found to expand the lifespan of male mice (371). In human dermal fibroblasts, an age-related decline in the expression of *Sirt6* was observed along with higher resistance towards reprogramming. However, the addition of *Sirt6* in old donors dermal fibroblasts (>50 years old) was found to enhance the reprogramming efficiency (372). In relation to MSCs, *Sirt6* has been shown to regulate osteogenic differentiation of BM MSCs in rats (373) and to protect embryonic MSCs from oxidative stress characterised by dysregulated redox metabolism in *Sirt6*^{-/-} human MSC cell lines (374). Another study in 2016 found that knocking out *Sirt6* gene accelerates the process of senescence in human BM MSCs (375). From all of the

144
aforementioned observations, the expression of *Sirt6* was hypothesised to show an age-related decline in both cell populations.

Investigations with another anti-ageing gene *Klotho* (*Kl*) have shown that its overexpression leads to extended life-span (376), while its mutations lead to decreased lifespan in mice (377) and promoted the process of ageing by damaging telomerase activity (378). The Klotho protein (379) was found to be inversely correlated with increasing age in the serum of human donors (380). Considering these findings, it is not surprising that genetically modified BM MSCs expressing *Kl* has been patented aimed at cell based therapies for many age-related diseases in humans (381). Interestingly, the potential mechanism of *Kl* to reduce senescence and extend life-span has been thought to be via the Tp53/p21 pathway (382). With this collective knowledge, it was hypothesised that there would be a decline in the expression of *Kl* in cells form older donors as compared to young donors.

5.1.2 Type 1 Interferons (IFN1)

IFNs are a group of cytokines that are known to elicit pleotropic biological effects such as immuno-modulation and cell differentiation (383). They were initially recognised for their interference with viral replication, however other broad range of properties beyond their antiviral actions have been recognised recently in various cell types including fibroblasts, dendritic cells and HSCs (384). In stem cells in particular, they have been known to play a role in differentiation, immune-modulation, and their anti-proliferative potentials (385). IFNs may be activated due the presence of extrinsic factors like virus/infection or due the presence of intrinsic stress factors leading to DNA damage.

In health, the levels of type 1 IFNs and IFN responses are fine-tuned to maintain the immune homeostasis between protection against viruses (386) and its toxic effects, which are widely associated with pathological conditions such as autoimmune diseases (i.e. Systemic lupus erythematosus or SLE) (387), chronic infection (388) and cancers (389). IFN1 binds to its receptor that is heterodimeric cell surface receptor, consisting of two transmembrane spanning subunits IFNAR1 and IFNAR2. Most cells types are known to express the receptor (390) including fibroblasts, dendritic cells, and peripheral blood cells including monocytes (391).

The canonical signalling pathway involves IFN1 production and binding to the IFNAR. The receptor is phosphorylated and triggers downstream signal transduction cascade via activation of Janus kinases 1 (JAK) and non-receptor tyrosine kinase 2 (TYK2), which then induces phosphorylation of transcription factors STAT1 and

STAT2. Phosphorylated STAT1 and STAT2 dimerise, dissociate from IFNAR2 and form a trimolecular complex with IFN regulatory factor 9 (IRF9) known as Interferon stimulated gene factor 3 (ISGF3) (392, 393). This complex then translocates to the nucleus where it binds to DNA sequence motif known as IFN stimulated response elements (ISRE) to eventually activate the transcription of a wide range of IFN regulated genes, also known as the IRGs, in an autocrine and paracrine way (Figure 5.1, top panel).

Recently, a study from our group examined the immune-modulating property of MSCs from cancellous bone fragments on CD3/CD28- stimulated CD4+ T cells. It found a 'dose dependant suppression of CD4 T-cell proliferation' along with an increase in the levels of TGF β in co-cultures (316). It also identified a list of candidate genes that potentially were responsible for eliciting MSC immunomodulation. Interestingly, many transcripts belonged to the IFN1 responsive family of genes, raising interest in understanding age-related differences in the expression of these genes in uncultured BM MSCs and HLCs, and their potential link to inflamm-ageing.

In an early study in 1984, the levels of IFN1 secreted from mononuclear cells were shown to decline in donors aged over 50 (394). In contrast, study in 2011 described DDR-induced expression of IFN genes after double stranded DNA breaks were introduced in HeLa cell lines (395). An intracellular molecule *STING* was increasingly becoming the core molecule mediating IFN expression induced by DDR (396). In health, DNA remains localised to the nucleus or the mitochondria. Thus, the presence of the DNA in the cytoplasm of the cell is expected to trigger DDR. In 2013, cGAS was found to detect the presence of the DNA in the cytoplasm and undergo conformational change. This conformational change in cGAS could now convert ATPs to the second messenger cyclic GMP-AMP (cGAMP) (397) which is a high affinity ligand to the adaptor protein stimulator of IFN genes (*STING*) (398). STING translocates from the endoplasmic reticulum to the golgi bodies which activates IFN regulatory factor 3 (IRF3) which leads to the production of IFN1 (399).

Based on this evidence, the cGAS-cGAMP-STING pathway has been reported to be the link between inflammation-based DDR, senescence as well as cancer (399). Considering that the presence of extrinsic/intrinsic stress factors illicit IFN1 response via STING, cGAS-STING-IFN1 pathway has been proposed as one of the pathways for autophagy (400-402). As mentioned, there has been an increased interest in unravelling the links between DDR, ROS, cell senescence and IFN1 signalling in HSC (30, 403), that share the same niche with the MSCs, giving a strong rationale to study these molecules in MSCs (Figure 5.1, bottom panel). Furthermore, the connection between DNA damage and the ATM-BID-MTCH2 pathway has been recently proposed to mediate the negative impact of ROS in HSCs (31). In HSCs from a mouse model deficient for Mixed Lineage Leukemia 5 or MLL5), BID (BH3 interacting domain) regulated mitochondrial ROS in DNA damage pathway downstream IFN1 signalling. Another protein - MTCH2 (mitochondrial carrier homolog 2) - protected cells from oxidative stress-induced death, and BID which is the precursor of MTCH2 in the ATM (ataxia telangiectasia mutated kinase)-BID pathway, worked as a MTCH2 antagonist (404).



Figure 5.1 IFN1 signal

activation and signalling pathway associated with ROS and intrinsic factors (adapted from (393) and (31))

(A) Left: General IFN1 signalling pathway activation by extrinsic factors. IFN1 binds to its receptor and conformational changes begin the cascade of phosphorylation of STAT1 and STAT2 to form ISGF3 that ultimately leads to IFN response elements in the nucleus. Right: IFN1 may also be activated by intrinsic factors like cytoplasmic DNA written in red. (B) Intrinsic factors leading to IFN1 response in MSCs: Intrinsic ROS causes DNA damage which leaks out of nucleus and becomes cytosolic DNA. This hypothetically activates IFN1 signalling, potentially leading to mitochondrial ROS.

Under low level of DNA damage, ATM through BID phosphorylation, reduces the level of BID and consequently, the level of ROS in HSCs. However, during DNA damage; loss of BID phosphorylation directs it toward the mitochondria and results in massive ROS production by interacting with MTCH2 (403). To date these investigations have not been performed on human cells. Moreover, the presence of ATM-BID-MTCH2 pathway molecules in MSCs has not been explored before. Because HSCs and MSCs coexist in BM niche, this pathway may be potentially affected in MSCs exposed to oxidative stress and DNA damage (both previously associated with age-related changes).

The role of IFNs and related signalling pathways in the complex and dynamic interplay between MSCs and HLCs in the BM niche remains unexplored. Taking the work from our group that recently found the expression of many IRGs in MSCs, forward along with some more exploratory transcripts related to senescence and anti-ageing, this chapter examined age-related changes in these transcripts in MSCs and HLCs. It was expected that DNA damage and senescence associated genes could potentially be higher in old donor cells, and possibly more in HLCs than in MSCs. List of analysed transcripts is shown in Table 5.1. Furthermore, considering that IFN1 signalling has been more commonly linked to innate immune cells, which are derived from HSCs, it was hypothesised that genes associated with IFN1 pathway would be expressed in higher levels in HLCs as compared to MSCs. Moreover, senescence and SASP genes were hypothesised to increase in older donors in MSCs and in HLCs. Age-related changes in anti-ageing genes, IFN receptors as well as IFN-associated genes would also be expected to be more pronounced in HLCs as compared to MSCs.

This chapter aimed at exploring if uncultured BM MSCs expressed any IFN1 responsive genes in comparison to HLCs along with the expression of senescence associated genes. The chapter also aimed at investigating any age-related changes in these genes in both MSCs and in HLCs.

5.1.3 Hypothesis and objectives

The following hypotheses and aims were postulated for this chapter

Hypotheses:

- 1. Senescence and SASP associated genes display an increased expression in old donor cells, with greater increase in BM HLCs than in BM MSCs. Antiageing genes display the opposite trend.
- Both BM MSCs and HLCs express the exploratory genes from the IFN1 responsive family of genes, but their expression is higher in HLCs, including selected molecules linking IFN pathway with DNA damage, ROS and senescence.
- 3. These exploratory transcripts would show age-related decline in cells from older donors, with possibly higher declines in HLCs as compared to BM MSCs.

Objectives:

- To quantify the expression of senescence, SASP and anti-ageing genes in uncultured BM CD45^{low}CD271⁺ MSCs and CD45⁺CD271⁻ HLCs from young and old donors.
- 2. To compare the expression of exploratory genes linking IFN pathway with DNA damage, ROS and senescence in uncultured BM MSCs and HLCs.
- 3. To investigate the age-related changes in these exploratory genes in uncultured BM MSCs and HLCs.

5.2 Methods

5.2.1 Donor selection

A total of n=12 donor BM cells, 6 young (19-40 years old, median age=27 years old) and 6 old (59-89 years old, median age=68 years old) were used for investigation in this chapter. The description of donor samples used in this chapter have been outlined in Appendix 1, Table 3.

5.2.2 Gene expression for exploratory transcripts analysis

The general method has been described in section 2.5. Briefly, frozen nucleated cells from BM of donors were defrosted (Section 2.3), enriched using anti-fibroblast beads (Section 2.4) and then sorted to collect CD45^{low} CD271⁺ MSCs and CD45⁺CD271⁻ HLCs and frozen in lysis buffer at -80°C for further use. Next, these cell lysates were defrosted and the RNA was extracted (Section 2.6.4). The RNA was then reverse transcribed to cDNA (Section 2.6.5), pre-amplified for 18 cycles (Section 2.6.6) and

then qPCR was performed on BioMark[™] (Section 2.6.7) using 48.48 dynamic array (Figure 5.2). The principle of these reactions are the same as those described in Section 2.6, but this chip accommodates up to 48 candidate genes.



Figure 5.2 48.48 gene chip outlay

The quantification was performed during analysis wherein the Ct values for each gene of interest were normalised to Ct value of *HPRT1* as the housekeeping gene to generate the Δ Ct values for each gene using the formula [Ct _{target gene}- Ct _{housekeeping gene}]. Finally, the Δ Ct values were converted to 'relative expression' for each gene using the formula [2^{- Δ Ct}] and the values were compared between young and old donors for both BM cell populations (MSCs and HLCs).

5.2.3 Statistical analysis

Cluster analysis was used to observe any differences in gene expression across the entire donor cohort. Statistical analysis and graphics were performed using GraphPad Prism software (version 7.0a). The normal distribution of the data was assessed using the Shapiro-Wilk and Kolmogorov-Smirnov tests for normality. As data were found to be not normally distributed, Mann-Whitney U test was performed for unmatched data. The results were considered significant at *p* value of <0.05.

5.3 Results

5.3.1 The expression of all exploratory transcripts in BM MSCs and HLCs

The expression of transcripts for type 1 IFNs, IFN Receptors and IRGs, as well as genes associated with IFN1>BID>ROS pathway and senescence, was investigated in sorted uncultured BM MSCs and HLCs, based on the CD45^{low} CD271⁺ and CD45⁺CD271⁻ phenotypes, as described in the previous chapter for the whole donor cohort. Cluster analysis of gene expression data revealed the presence of two distinct clusters for MSCs and HLCs, with the exception of one sample (Figure 5.3). High variation in gene expression between the donors was noted, but surprisingly, an overall pattern of higher expression (more red colour) in s compared to HLCs was found. The higher expressed IRGs in MSCs relative to MSCs included *IFITM1, IFITM3, IFI27, SERPING* and IRGs with higher expression in HLCs included *RGS1, IL8* and BID (Figure 5.3).

The statistical analysis including the fold differences in the expression of gene shown in figures are shown in Table 5.1. The median values and differences between expression in MSCs and HLCs of the complete list of genes is shown in Appendix 4, Table 5.1.



Figure 5.3 Expression of exploratory genes in MSCs and HLCs

Expression of *IFNA1, IFNB1, IFN receptors,* IRGs, genes associated with IFN1>BID>ROS pathway, senescence and other exploratory genes grouped by cluster as MSCs (left) and HLCs (right). Colour key in shown at the bottom right of the figure. Grey squares indicate gene expression values that were below detection.

5.3.1.1 Analysis of expression of senescence and anti-ageing associated genes in MSCs and HLCs

For the whole donor cohort, analysis of transcripts associated with senescence and anti-ageing showed that *Tp53* and *Sirt6* displayed no statistically significant differences (p<0.05 for both genes) between MSCs and HLCs, although *Sirt6* had slightly higher expression in MSCs as compared to HLCs (2.86-fold). However, the expression of *KI* was significantly (25-fold) higher in MSCs (p=0.0083) as compared to HLCs (Figure 5.4). In combination, the pattern of expression for all three genes indicated a possibility for stronger anti-ageing defence mechanisms in MSCs compared to HLCs.





Expression of *Tp53*, *Sirt6* and *KI* in donors in MSCs and HLCs. Horizontal line across data set indicates median values. *p*<0.01**, Mann-Whitney U test.

5.3.1.2 Analysis of expression of SASP-associated genes in MSCs and HLCs

Two out of three SASP cytokine genes were also differentially expressed between MSCs and HLCs. Cytokine *IL6* showed no difference in expression between the two cell populations (Figure 5.5). As expected, *IL7* displayed significantly higher expression in MSCs (p=0.0005) in comparison to HLCs, and *IL8* displayed higher expression in HLCs (p=0.0022) as compared to MSCs.



Expression of *IL6, IL7* and *IL8* in MSCs and HLCs. Horizontal line across data set indicates median values. p<0.01** and p<0.001***, Mann-Whitney U test.

5.3.1.3 Expression of IFN1, IFN Receptors and IRGs in MSCs and HLCs

Expression of *IFNA1* and *IFNB1* did not show differences in their expression levels between MSCs and HLCs (Figure 5.6A). Very interestingly, the levels of expression of *IFNAR1* and *IFNAR2* showed a significantly higher expression in MSCs (p=0.0018 and p=0.0332, respectively) compared to HLCs (Figure 5.6B), which was unexpected, given their expression was historically linked to immune-lineage cells.



Figure 5.6 Expression of IFNA, IFNB and IFNA receptor genes in MSCs and in HLCs

(A) Expression of *IFNA1* and *IFNB1* in MSCs and HLCs. (B) Expression of *IFNAR1* and *IFNAR2* in MSCs and HLCs. Horizontal line across data set indicates median values. p<0.05* and p<0.01**, Mann-Whitney U test.

Among the IRGs, many of them including *USP12*, *IFITM1* and *IFITM3*, displayed higher expression levels in MSCs (Figure 5.7). Similarly, there were genes including *PRDM1*, *LAIR* and *CASP1* that were significantly higher expressed in HLCs (Figure 5.8). Overall, and as indicated by the cluster analysis shown in Figure 5.3, a higher proportion of the genes displayed significantly higher expression in MSCs than in HLCs. This was unexpected, given IFN1 gene expression was historically linked to the immune-lineage cells.



Figure 5.7 Genes displaying significantly higher expression in MSCs than in HLCs

Expression of most significantly differentially expressed genes in MSCs over HLCs. Horizontal line across data set indicates median values. $p<0.01^{**}$, $p<0.001^{****}$ and $p<0.0001^{*****}$, Mann-Whitney U test.



Figure 5.8 Genes displaying significantly higher expression in HLCs than in MSCs

Expression of most significantly differentially expressed genes in HLCs over MSCs. Horizontal line across data set indicates median values. *p*<0.05*, Mann-Whitney U test.

5.3.1.4 Analysis of expression of genes involved in IFN1>BID>ROS pathway

As discussed in the introduction to this chapter, damage to the DNA due to external or internal factors is known to activate IFN1 signalling via *STING*, *IRF3* and *STAT1* through *IFNA* receptors. This has eventually been shown to mobilize *BID*, which exerts its expression via *MTCH2*, which is the mitochondrial receptor for *BID*. Mobilisation of *BID* has been shown to lead to accumulation of mitochondrial ROS leading to functional defects and senescence in the HSCs. Because HSCs give rise to HLCs and MSCs coexist with HSCs in the BM niche, it was thus in the interest of this project to investigate whether this pathway is also engaged in MSCs.



Figure 5.9 Expression of genes associated with IFN1>BID>ROS pathway in MSCs and HLCs

Expression of *STING, IRF3* and *STAT1* (top panel), *BID* and *MTCH2* (bottom panel) in each group. Horizontal line indicates median values. *p*<0.05*, Mann-Whitney U test.

Both *IFNAR1* and *IFNAR2*, displayed higher expression levels in MSCs as shown in figure 5.6B. *STING* showed no significant difference in expression between MSCs and HLCs although a trend for higher levels in MSCs was noted (Table 5.1). *IRF3* and *STAT1* displayed higher expression levels in MSCs as compared to HLCs (p=0.0205 and p=0.0242 respectively, Figure 5.9, top panel). *BID* displayed significantly higher (p=0.0188) expression in HLCs as compared to MSCs while *MTCH2* showed no differences (Figure 5.9, bottom panel).

		Madian	Madian	Fold	
Function	Gene	(MSCs)	(HLCs)	(MSCs/HLCs)	<i>p</i> value
IFN1 and receptors	IFNA1	6.97	3.01	2.32	NS
	IFNB1	1.50	1.59	0.95	NS
	IFNAR1	4.28	0.88	4.86	0.0018
	IFNAR2	5.82	1.84	3.16	0.0332
IFN1-BID- MTCH2 pathway	STING	2.64	1.54	1.72	0.0684
	IRF3	2.051	1.90	1.08	0.0242
	STAT1	4.33	2.26	1.92	0.0205
	BID	0.21	1	0.21	0.0188
	MTCH2	1.43	1.06	1.35	NS
Anti-ageing and senescence	KI	0.51	0.02	25.50	0.0083
	Sirt6	0.50	0.17	2.86	NS
	Тр53	2.5	1.59	1.57	NS
SASP	IL6	0.15	0.27	0.56	NS
	IL7	1.48	0.016	93.73	0.0005
	IL8	2.92	67.21	0.04	0.0022

Table 5.1 Medians of expression of genes shown in Figures 5.4 to 5.6 and 5.9 in MSCs, HLCs, fold differences and their p values

NS: Non-significant, Gene expression is normalised to HPRT1, Mann-Whitney U test.

Overall, this gene expression pattern potentially indicated a higher level of constitutive IFN1 signalling in MSCs, as well as their possession of the necessary molecular machinery to combat oxidative damage via IFN1>BID>ROS pathway.

5.3.2 Age-related differences in IFN1 signalling pathway related genes

The median values for each gene shown in the figures below in young and old donors, the fold difference with advancing age and their p values are indicated in Table 5.2 and Table 5.3. Similar tables for the all the exploratory genes are shown in Appendix 4, Table 5.2 and 5.3 for age-related differences in MSCs and in HLCs, respectively

5.3.2.1 Age-related differences in the expression of senescence and anti-ageing associated genes in MSCs and HLCs

Senescence-associated gene *Tp53* and anti-ageing genes of *KI* and *Sirt6* did not show any differences between young and old donor MSCs (Figure 5.10A). In HLCs, the expression of all of these genes declined in old donors (Figure 5.10B) with *Tp53* showing a statistically significant decline (p=0.0043). At least in relation to anti-ageing genes, these data indicated a possibility for stronger 'ageing-resistance' molecular mechanisms in MSCs compared to HLCs.



MSCs

А

Figure 5.10 Expression of senescence and anti-ageing genes

(A) Expression of *Tp53, Sirt6* and *KI* in young and old donors MSCs (B) Expression of *Tp53, Sirt6* and *KI* in young and old donor HLCs. Young and old donors are indicated in black and grey dots, respectively. Horizontal line across data set indicates median values. p<0.01^{**}, Mann-Whitney U test.

5.3.2.2 Age-related differences in SASP-associated cytokine transcripts in MSCs and HLCs

IL6 transcript levels were found to be 4-fold higher in old donor MSCs as compared to MSCs from young donors however the differences failed to reach statistical significance. *IL7* and *IL8* did not show any difference between young and old donor MSCs either (Figure 5.11A). In HLCs, *IL6* transcript was similarly nearly 4-fold higher in from old donors as compared to young donors however the differences did not reach statistical significance. *IL7* showed a slight decline and *IL8* did not show any difference between young and old donor HLCs (Figure 5.11B).





(A) Expression of *IL6, IL7* and *IL8* in young and old donor MSCs. (B) Expression of *IL6, IL7* and *IL8* in young and old donor HLCs. Young and old donors are indicated in black and grey dots, respectively. Horizontal line across data set indicates median values.

5.3.2.3 Age-related differences in IFN1, IFN Receptors and IRGs in MSCs and HLCs

Cluster analysis between young and old donor MSCs revealed no patterns for differences in type 1 IFN, IFN receptors, IRGs and cytokines. Similarly, investigation of differences between young and old donor HLCs using cluster analysis also revealed no particular trends.

IFNA, IFNB and *IFNAR* transcript levels showed no significant differences or particular trends between young and old donors MSCs (Figure 5.12A). However, in HLCs, some differences were significant. *IFNA1* transcript levels displayed a trend for over 2-fold increase in old donors HLCs compared to young donors. *IFNB1* showed no differences between young and old donor HLCs (Figure 5.12B). *IFNAR1* and *IFNAR2* transcript levels showed declines in their expression in old donor HLCs, which was statistically significant for *IFNAR2* (*p*=0.041, Figure 5.9B).



Figure 5.12 Expression of IFNA, IFNB and IFNA receptor genes

(A) Expression of *IFNA1* and *IFNB1* (top panel) and *IFNAR1* and IFNR2 (bottom panel) in young and old donor MSCs (B) Expression of *IFNA1* and *IFNB1* (top panel) and *IFNAR1* and *IFNAR2* (bottom panel) in young and old donor HLCs. Young and old donors are indicated in black and grey dots, respectively. Horizontal line across data set indicates median values. *p*<0.05^{*}, Mann-Whitney U test.

Out of all the exploratory IRGs, *RNF213* was the only gene that showed statistically significant lower expression in MSCs from old donors as compared to young donors (p=0.041, Figure 5.13A). In the HLCs, more IRGs displayed significant decline in old donors (Figure 5.13B) and Appendix 4, Table 2. This potentially indicted reduced IFN1 signalling in old donor HLCs, which wasn't the case for old donor MSCs.

Α

MSCs





HLCs





(A) Expression of differentially expressed genes in MSCs in young and old donors and (B) Expression of differentially expressed genes in in HLCs in young and old donors. Young and old donors are indicated in black and grey dots, respectively. Horizontal line across data set indicates median values. p<0.05* and p<0.01**, Mann-Whitney U test.

5.3.2.4 Age-related differences in genes associated with IFN1>BID>ROS pathway

The analysis of genes associated with the IFN1>BID>ROS pathway also revealed interesting findings. Again, investigation in MSCs did not show any significant agerelated differences, although *STAT1, IRF3* and *STING* showed a trend for an increase in MSCs from old donors. BID expression demonstrated a non-significant, nearly 3-fold decline in old donor MSCs (Figure 5.14A).

Interestingly, investigation of expression of the same genes in young and old donor HLCs revealed different results. While *STAT1* expression showed no difference, *IRF3* displayed a significant decline in old donors HLCs (p=0.033) (Figure 5.14B). Significant declines in old donor HLCs were also seen for *MTCH2* and *STING* expression (p=0.0152 and p=0.041, respectively) while *BID* did not show any age-related differences.

Altogether, investigation of IFN1 family of genes, SASP cytokine genes and other genes linking IFN pathway and cell senescence did not reveal any significant differences between young and old donor MSCs, except for *RNF213* gene, which is an IRG with yet unknown function. While expression of these genes in MSCs did not show any age-related differences, the same genes in HLCs revealed interesting expression patterns. In particular, *IFNAR2*, many IRGs and genes linking IFN1 pathway with ROS and cell senescence displayed a decrease in their expression in old donor HLCs. Altogether, these data indicated a potential reduction in constitutive IFN1 signalling in old donor MSCs.



MSCs



(A) Expression of *STING*, *IRF3*, *STAT1*, *BID* and *MTCH2* in young and old donor MSCs (B) Expression of *STING*, *IRF3*, *STAT1*, *BID* and *MTCH2* in young and old donor MSCs. Young and old donors are indicated in black and grey dots, respectively. Horizontal line across data set indicates median values. *p*<0.05*, Mann-Whitney U test

Gene	Medians (young)	Medians (old)	Age-related difference*	p value
IFNA1	5.136	9.15	1.78	NS
IFNB1	1.247	1.725	1.38	NS
IFNAR1	4.278	4.757	1.11	NS
IFNAR2	5.82	5.76	0.99	NS
STING	2.63	4.88	1.86	NS
IRF3	1.946	3.13	1.61	NS
STAT1	3.42	5.16	1.51	NS
BID	0.3	0.11	0.37	NS
MTCH2	1.43	1.27	0.89	NS
KI	0.51	0.48	0.94	NS
Sirt6	0.506	0.406	0.80	NS
Tp53	2.25	2.88	1.28	NS
IL6	0.05	0.2	4.00	NS
IL7	1.481	1.169	0.79	NS
IL8	2.24	5.14	2.29	NS

Table 5.2 Exploratory genes, age-related fold differences and their *p* values in MSCs

*Age-related difference calculated by median expression in old donor/median in young donor. NS: Non-significant, Gene expression is normalised to *HPRT1*. Mann-Whitney U test

	Median	Median	Age-related	
Genes	(young)	(old)	difference*	<i>p</i> value
IFNA1	0.295	7.443	25.23	NS
IFNB1	1.53	1.591	1.04	NS
IFNAR1	1.1	0.48	0.44	NS
IFNAR2	2.75	1.38	0.50	0.041
STING	2.293	1.033	0.45	0.026
IRF3	1.561	0.66	0.42	0.0411
STAT1	2.17	2.4	1.11	NS
BID	1	1.03	1.03	NS
MTCH2	1.61	0.79	0.49	0.0152
KI	0.04	0.011	0.28	NS
Sirt6	0.28	0.15	0.54	NS
Тр53	2.93	1.21	0.41	0.0043
IL6	0.16	0.6	3.75	NS
IL7	0.038	0.013	0.34	NS
IL8	46.47	67.21	1.45	NS

Table 5.3 Exploratory genes, age-related fold differences and their p values in HLCs

*Age-related difference calculated by median expression in old donor/median in young donor. NS: Non-significant, Gene expression is normalised to *HPRT1*. Mann-Whitney U test.

5.4 Discussion

This chapter presents novel data with respect to the presence and expression of genes associated with senescence, SASP and anti-ageing genes in uncultured BM MSCs. There were no differences observed in their expression between young and old donor MSCs but all their expression declined in HLCs with *Tp53* displaying a significant decline in expression. In SASP, while *IL6* showed a non-significant increase in old donors in both MSCs as well as in HLCs, *IL7* and *IL8* showed no particular differences in old donors in both, MSCs as well as in HLCs.

With respect to a the exploratory IFN1 family of genes, a number of novel findings were observed. Firstly, a large number of genes were found to be differentially expressed. Second, there were more highly expressed genes in MSCs over HLCs rather than the other way around, which was unexpected. This indicated towards the potential immune-related functions of BM MSC via IFN1 pathway as indicated by cancellous bone MSCs previously (316), that will require further explorations. Interestingly, *IFNA1, IFNB1* and the *IFNA* receptors, were expressed significantly higher in MSCs as compared to HLCs. Genes involved in the IFN1>BID>ROS pathway (403) also revealed interesting trends with *STAT1* and *IRF3* indicating greater expression in MSCs, while *BID* displayed higher expression in HLCs as compared to MSCs and this supported the hypothesis of this chapter (31). Among the senescence associated and anti-ageing genes, *KI* displayed higher expression in MSCs as compared to HLCs. Finally, in cytokines, *IL7* displayed higher expression in MSCs as compared to HLCs.

When compared for age-related differences; *IFNA1, IFNB1* and *IFNA* receptors showed no differences in MSCs, but *IFNAR2* displayed a significant decline in HLCs from old donors. The only IRG that showed an age-related difference in MSCs was *RNF213*, while a number of IRGs were found to display a decline in their expression in HLCs. Analysis of genes involved in the IFN1>BID>ROS pathway also showed no changes in expression between young and old donors in MSCs, while in HLCs, *STAT1, IRF3* and *MTCH2* displayed a significant decline in expression between young and old donors.

A recent study investigating the effect of retrotransposon (RT) in human fibroblasts found that the RT in human (long-interspersed element-1 or L1) becomes transcriptionally depressed in senescent cells which activates IFN1 response (336). The senescent cells were prepared by culturing them *in vitro* until their proliferation ceased. They found a high percentage (68%, 57 out of all the 84 tested genes) to be

highly upregulated in senescent cells which were IRG or associated with the IFN1 signalling.

Over 30 of these genes were also tested in this thesis but in relation to age in uncultured BM MSCs and did no observe any age-related grouping in cluster analysis. Also, while other studies found a number of IFN1 associated genes to be upregulated in replicative senescent cells (336), this thesis found a significant decline in only one of the IFN1 associated genes in MSCs. This could be due to the fact that firstly, the cells described in this thesis were MSCs and not skin fibroblasts and secondly, the MSCs and HLCs used in this thesis were uncultured and did not undergo replicative senescence by *in vitro* culture expansion.

The same study also examined the presence of L1 in aged mice and found a significant increase in IFN1 associated genes. They treated old mice with nucleoside reverse transcriptase inhibitors known to antagonise the effect of L1 reverse transcriptase. In doing so, they observed that not only did the IFN1 associated genes showed a downregulation but the SASP pro-inflammatory state was also alleviated. Their *in vivo* mice study thus linked senescence to ageing in mice. The only SASP gene that was found to increase in this project was *IL6* while *Tp53* and IFN1 associated genes were mainly found to decline in older donors, unlike in the above mentioned study. Similar treatments using nucleoside reverse transcriptase inhibitors in uncultured BM cells in a larger number of donors could shed light on the senescence profile of BM cells in association with IFN1.

In this project MSCs did not display any difference in the level of expression of any of the above genes between young and old donors except RNF213 that showed a significant decline. RNF213 or ring finger protein has also been known as 'mysterin' owing to the lack of knowledge curated about this gene. It is known as the causative gene for Moyamoya disease which is a disease of blocked arteries within the brain. The disease is more prevalent in East Asian countries of Japan, Korea and China and is more common in women (405). Not much is known about the gene and more exploratory studies are needed to reveal its function in MSCs, as well as in HLCs.

There were a good number of transcripts with over 2-fold age-related differences in MSCs. For example, pro-inflammatory cytokines *IL6 and IL8* were found to be 4-fold and 2-fold higher in the old donors MSCs as compared to young donors, respectively. Of note, while literature suggests that *IL6* is higher by 2-4 fold in older donors (MSCs), much higher the fold difference is suggested to be in circulating levels of this cytokine

in old donors with chronic inflammation (406). Age-related differences were more prevalent in the HLCs, wherein genes like IFN regulatory factor 2 and 9 (*IRF2* and *IRF9*) and cyclin D2 (*CCND2*) displayed a significant decline in old donor HLCs. Similar decline in IRGs has been observed in the peripheral blood mononuclear cells in older donors (394).

BID and MTCH2 regulate the oxidative metabolic state of HSC and thus manage their oxidative stress, which was discovered not too long ago. The hypothesised mechanism behind it was described by Gross et al., in 2017 where DNA damage in the nucleus lead to accumulation of ROS in mitochondria via the IFN>BID>MTCH2 pathway. HSCs as a population have extensively been studied for understanding their biology and ultimately for use in several cell therapies. Not only did this project confirm the expression of these transcripts but also compared it between young and old donor MSCs. While no age-related differences were observed in MSCs, there was a significant age-related decline in *MTCH2* observed in the HLCs.

An age-related decline in IFN1 signature genes was also observed in activated CD4⁺ T cells from donors of older age (65-85 years old) as compared to younger donors (20-35 years old) when investigated in cells isolated from peripheral blood mononuclear cells (407). Considering that IFN1 are an integral part of the innate immune system that is responsible for triggering responses in bacterial and viral infections, it may suggest that the decline in IFN1 in old age compromises their ability to respond to these infections. With the knowledge that DDR can activate IFN1 signalling and that DNA damage increases in the older age, it may also explain the reduced ability of recognising this damage in older donors owing to the decline in IFN signature genes. Even though the number of investigation of IFN1 on MSCs remain limited, the existing studies have shown the presence of IRGs in culture expanded BM MSCs. The study also showed an increase in the number and size of colonies when BM MSCs were treated with monoclonal antibody against the IFNAR1 chain of human type 1 (385).

The transcripts associated with anti-ageing and senescence namely *KI*, *Sirt6* and *Tp53* showed no age-related differences in uncultured BM MSCs. *KI* has been a well-known anti-ageing gene, the absence or deletion of which observed ageing-like syndrome in mice. In general, overexpression of *KI* has been observed to increase lifespan, whereas repressing the transcript has been associated with premature ageing. *Tp53* is crucial for DNA repair and induction of apoptosis and senescence and has previously been shown to be increased in MSCs from old donors (96). Age-related decline in *Tp53* in HLCs potentially indicates decline in autophagy ability of cells.

A study in 2016 aimed at identifying senescent cells within mice bone microenvironment and performed varying investigations in T cells, B cells, myeloid cells, MSCs, osteoblasts as well as in osteocytes of young (6mo) and old (24mo) mice (365). They observed an increase in senescence and SASP profile in old mice. While *P16* increased in all the cell types, *Tp53* was highly expressed in osteocytes and myeloid cells and not in MSCs. However, in this thesis, *Tp53* was found to expressed higher in MSCs as compared to HLCs and declined in older donors in both MSCs and in HLCs which was unexpected as *Tp53* increases in old senescent cells (408).

With respect to studies in humans, De Oliveira found that *KI* down-regulation induced premature senescence in human fibroblasts and downregulation of *p53* in *KI* attenuated cells restored normal growth in the cells (382). Overall their study demonstrated that *KI* regulated cellular senescence via the p53/p21 pathway highlighting on the anti-ageing role of *KI*. More recently, Zhang et al., found that addition of KI protein to MSC cell lines, attenuated their osteogenic differentiation ability (409). Another human study examined the serum of healthy volunteers and found that KI declined with increasing age. Their study had a broad range of donor age (0.1-88 years old) including toddlers as well as old donors. While they did observe a decline in KI across the entire age range, they segregated their young and old groups age <17 as young donors and over 20 years old as old donors (380). The specificity of the *KI* in MSCs has never been showed before making the data from this project very novel. As an anti-ageing gene, this transcript could potentially be used to understand ageing *in vivo* in MSCs.

Along with *Kl, Sirt6* is another transcript that is recognised for its anti-ageing potential, the deficiency of which has been associated with age-associated degenerative processes (410). It is known to be expressed in BM MSCs and Sun and colleagues found that while its knockout has resulted in decreased osteogenic differentiation and proliferation, its overexpression increased osteogenesis in mice (373). They also found that Sirt6 was doing so via partial suppression of the NF-kb pathway in old mice. Zhang et al., also found the Sirt6 promoted osteogenic differentiation in mice (411), however, their results contradicted the results of Sun and colleagues, and demonstrated the involvement of the BMP pathway instead. Around the same time, another study that generated MSCs from human embryonic stem cells found that Sirt6 played a protective role against ROS in MSCs (374). They found that cells that were depleted of Sirt6 not only were more susceptible to accelerated degeneration but were also more vulnerable to the negative impacts of ROS. Considering these findings, we expected an age-

related decline in the expression of *Sirt6* in uncultured MSCs but did not observe any differences in this project.

Taken together, the anti-ageing transcripts and transcripts associated with senescence showed significant differences in HLCs, but no difference in MSCs. This was as expected in the hypothesis suggesting the MSCs *in vivo* are relatively more resilient to ROS and DNA damage in comparison to HLCs. Previous investigations of these transcripts in uncultured MSCs is sparse, making data from this thesis novel.

Among the cytokines, *IL6* showed no significant difference in its expression level between MSCs and HLCs. This was unexpected as IL6 has closely been associated with inflammation (351). In contrast, *IL7* exhibited very high expression levels in MSCs and *IL8* displayed higher expression in HLCs. *IL6* has been a well-known indicator of inflamm-ageing and has been reported to be higher in old donors and significantly higher in old donor with inflammation (406). Thus it was no surprise that the expression of *IL6* was found to be higher in old donors in both cell types, MSCs and HLCs. IL6 and IL8 are also associated with SASP and have been shown to increase in MSCs that were DNA damaged (354). However, that study was based on MSCs and HSCs in co-culture conditions. In this thesis, the expression of *IL7* showed no age-related differences in either cell populations while *IL8* displayed a non-significant increase in HLCs.

Considering that both HSCs and MSCs originate within the BM (section 1.3.1), they are bound to impact each other. The protective role of MSCs in preventing HSCs from ROS with the help of *Cx43* has already been discussed in section 4.4. It is a wellknown fact that age-related changes of increased adipogenesis in MSCs and shift towards myeloid lineage (myeloid skewing, section 1.3.1) in HSCs occur within the same BM niche. However, as the two belong to different stem cells types, they have often been examined separately. Among the few scientists to have linked the two and acknowledge the relationship, were Kovtonyuk and colleagues in their 2016 study (412). Referring to previous studies, they outlined that adipocytes negatively impacted HSC functions supporting B-cell lymphopoiesis. As adipogenesis increased in old age, there is a high possibility that it enhanced the process of age-related myeloid skewing in old individuals.

In summary, this chapter was the first to quantify the expression level of IFN1 associated and other genes in uncultured BM cells. Interestingly, a large number of genes were found to be highly expressed in MSCs as compared to HLCs and this

finding is novel. Furthermore, a number of genes also displayed age-related differences, more in HLCs than in MSCs, and displayed the potential link to DNA damage, IFN1, aging, senescence and oxidative stress. HLCs appear more vulnerable to age-related damage at cellular level as compared to MSCs *in vivo*. Future work would involve exploring the protein levels and functions of the genes that displayed significantly higher expression in MSCs than in HLCs. Ideally, this would require recruitment of a larger donor cohort and controls including investigation of cells that were exposed to extreme oxidative stress prior to analysis of gene and protein expression. Correlation of this data with DNA damage performed by comet assay and senescence assay from the same donors would provide extremely useful links between IFN1, ageing, ROS, DNA damage and senescence.

Chapter 6 Gene and surface marker expression of human BM MSCs in hip osteoarthritis

6.1 Introduction

Osteoarthritis (OA) is a classic age-related degenerative disease which, some researchers think, is inevitable in old age. It is by far the most common joint disease in the world. It leads to a poor QOL by causing debilitating pain in the joint which in turn impacts gait, all of which reduces the daily functions (100). In spite of considerable variation in the definition of the condition, age has been an underlying factor for most of the cases observed (413). While the frequency of OA incidences have been found to increase with age, it is known that not all old people suffer from the disease (414). Increasing age does however, escalates the chances of someone suffering from OA due to decrease in body immunity and accumulation of damage at the cellular level including MSCs (100).

Section 1.3 of this thesis outlined that the impact of OA is not only confined to the patients but also affects the family members and the economy in general, considering the high rate of occurrence of the disease, globally. While OA may have different aetiology and be caused by various factors like lifestyle, mechanical load on the joints, genetics, environmental factors, physical damage and injury; the management of the disease usually involves the same approach. This indicates our lack of understanding in the disease, especially in terms of its pathophysiology at the *in vivo* cellular level. The other example supporting our lack of understanding of the disease are many different theories of OA development and disease progression, which vary between research scholars and OA clinical experts alike.

Early detection of the disease can significantly prevent and/or alleviate the pain and the poor QOL in patients, before it gets to the point where surgery is the only option available. This outlines the need for an increase in the intensity of basic research and applied science research for early OA diagnostics. The last decade has seen a rise in the idea that OA is not a single disease, but a culmination of a number of progressive conditions. One of these conditions may indeed be the premature ageing of bone resident MSCs, leading to altered bone homeostasis and the inability of the subchondral bone to support cartilage nutrition and shock absorbance properties (415, 416). To better understand these individual contributing factors and their pathologies, a number of OA animal models that mimic different aspects of the disease in humans have been developed (417). Similar to the animal models used for understanding the

ageing of MSCs (or even ageing in general), animal models of OA too are varied in several parameters. A number of models exist, each of which have their own advantages, limitations and their applicability for OA ageing studies is discussed below.

6.1.1 Hip OA

In the last 15 years, the aetiology of hip OA has been proposed as different to knee OA. While a number of factors like sex, obesity, genetics, occupation and local injuries are known to contribute to the progression of OA (418), age has strongly been associated with hip OA displaying a sharp increase of incidence from 0.7% in people aged 40-45 to 14% in people aged over 85 (419). Many of the strategies provided for hip OA management have been derived from those developed for knee OA. This has been due to the higher prevalence of knee OA worldwide and due the fact that knee OA is relatively easier to image, scan and analyse as compared to OA of the hip (420).

Pain and stiffness in the hips are the first symptoms of OA. Diagnosis of hip OA is either by radiographic evidence or by clinical examination or a combination of both. Radiographically, K/L scale (section 1.1.5) is used to determine the extent of the damage in the hip joint. Higher K/L score indicates greater extent of bone and cartilage damage including osteophyte formation, narrowing of joint space and subchondral bone sclerosis (421). In spite of radiographic evidence providing significant information about OA of the hip, it has often been criticised when used by itself for the diagnosis of OA. There is strong evidence in the literature that suggests that not every patient with hip pain has radiographic signs of OA. Also, most elderly patients with higher risk of hip OA did not show its evidence by radiography (422). Another study in 2013 'compared the sensitivity of physical examination with radiographs' in the diagnosis of clinically significant hip OA. Interestingly, they found that physical examination was significantly more accurate in diagnosis of hip OA as compared to radiography (423).

Treatment of OA of the hip does not follow an approach any different from that of OA of another part of the body. Despite of the different theories/hypotheses existing around the progression of the disease, the treatment is fundamentally symptomatic. Use of NSAIDS, pain management and changes in lifestyle is often suggested to the patients (424). These include pharmacological intervention similar to those described earlier in section 1.1.5 along with other medications such as anti-resorptives (425). NSAIDs work by inhibiting cyclooxygenase 2 (COX2) which blocks the production of prostaglandins (PGs) at sites of inflammation and tissue damage (426) while anti-resorptives, as the name suggests, function largely by supressing bone resorption.

Unfortunately, most of the cases of a detected hip OA often result in a much invasive hip replacement surgery. The extent of hip damage is what determines whether or not an adult needs to undergo the hip replacement surgery. The surgically added hip implant usually stays intact for up to 15 years, after which there might be a revision surgery. The revision surgery often has poorer impact on supporting the patient after 5 years of revision as compared to the primary total hip replacement (THR) surgery. THR with metal-on-polyethylene bearing surfaces remain the gold standard for the treatment of end-stage hip OA. It provides good relief from pain along with steady longevity as observed across patients (427).

From the literature presented in the above two sections, it is evident that a better understanding of hip OA pathophysiology, including the 'age component' of the disease, could have a significant socioeconomic impact, and that there is no suitable animal models faithfully replicating its progression in humans. The study of MSCs in hip OA joints, and comparing their characteristics to MSCs from healthy aged individuals may shed new light on the progression of this condition potentially leading to new therapies.

6.1.2 Senescence, SASP, ROS and OA

Senescence is a process of irreversible growth arrest that has been observed in ageing cells, and has been outlined in this thesis in sections 1.2.2 and 5.1.1. Considering OA has been associated with older age, a number of studies have investigated cell senescence in OA (428) largely focussing on OA cartilage (429). The presence of senescent cells and the associated secretory phenotype has been found to be significantly higher in OA cartilage compared to healthy cartilage. The SASP detected in OA has largely been attributed to pro-inflammatory cytokines like IL1, IL6 and IL8, and matrix metalloproteinases (MMPs) (430). The initiating factors are believed to be increased DNA damage (431) due to increased ROS (432, 433) leading to increased inflammatory mediators. The presence of senescence in OA is further confirmed by the development of OA-like condition in mice that were injected with senescent fibroblasts in their knees (434).

Basic research has steadily contributed to our knowledge of the multi-factorial aspects of OA including the role of age-related senescence (428) and inflammation in the disease pathogenesis (435). Even though the pieces of the puzzle of OA and senescence are far from complete, there has been significant increase in our understanding of the two. The fact that both, senescence and OA are multifactorial and need more research at the cellular level, challenge the process of collection of relevant evidence in spite of the hypothesised potential link between the two. Nevertheless, there is a growing body of evidence connecting the impact of cell senescence to primarily degenerative diseases, like OA.

IL6 has been shown to be expressed in high quantities in the serum of patients with OA (436, 437). The Chingford study (Livshits et al., 2009) on middle aged British women with radiographic knee OA (RKOA, K/L grade>2) confirmed that higher circulating levels of IL6 in obese patients were indicators for potential OA in future. They found that first, the circulating levels of IL6 in patients with RKOA were consistently higher than in non-OA donors, which increased proportionally with increase in OA severity, and second, that the levels of IL6 increased with increasing BMI (437). Thus, they suggested that higher BMI and increasing serum levels of IL6 are both predictive biomarkers of RKOA and outlined the need for future research on IL6 in OA. This study importantly highlighted the impact of lifestyle (obesity) as a contributor to the disease.

Obesity results in an increased load on joints along with the secretion of adipogenic factors (like adipokines, adipogenic hormones) that mediate inflammatory responses. Also, the fact that there is a well-known shift from muscle mass and bone formation towards increased adipogenic tissue formation also suggests that obesity contributes to both, inflamm-ageing and OA. Another factor that crops up when discussing adipogenic factors and OA, is the presence of harmful reactive oxygen species (ROS) as data from previous research indicates that ROS may be key contributor to the adipogenic bias of MSCs at the expense of their osteogenic potential (438).

The role of ROS in most diseases has been found to be fundamental, owing to the fact that ROS is a by-product of cellular metabolism. It has been closely linked to a number of diseases occurring in almost every organ in the human body (281). This is expected as every organ is made of functionalised tissues which are basically, groups of highly specialised cells. As it may be expected, ROS has been found to be a key contributor of age-related articular cartilage degradation, which has been a major evidence correlating ageing to OA (439-441). In particular, ROS was suggested to contribute to the loss of proliferative capacities of chondrocytes, senescence of chondrocytes, and increase in the production of inflammatory markers along with an increase in oxidative stress in OA cartilage (441, 442).

Alterations in cartilage proteoglycans and other proteins within the cartilage matrix have been discussed at length in several publications over the last two decades (443, 444). The increase in alterations that eventually lead to the drastic increase in ROS

within cells has been thought to be due to the imbalance of natural anti-oxidants in the body (section 4.1.1.3.) Considering this imbalance, several research groups have assessed ROS damage by quantifying ROS in OA samples or by measuring the anti-oxidant enzymes within the cells (445-447). Theoretically, an increase in the ROS levels in OA (433) is expected to be accompanied by a decrease in the levels of anti-oxidant enzymes (446, 447) that ultimately lead to genomic instability and cell senescence of chondrocytes (448).

Regan and colleagues examined the levels of extracellular SOD (SOD3) in the cartilage of hip OA patients by ELISA and found a 4-fold decline in the levels of the SOD3 in OA donors (445). Similarly, other studies have confirmed the decline in the levels of the different naturally occurring anti-oxidant enzymes in OA, but for all of these studies, the focus has been on cartilage/ chondrocytes (449, 450). Altogether, evidences discussed above suggest that age and lifestyle choices taken together may play an important role in OA pathogenesis. These changes manifest themselves as inflamm-ageing and oxidative stress that lead to cellular senescence, which in itself may be a strong contributor to OA. While current research focuses mostly on chondrocytes, similar processes may also occur in MSCs residing in the underlying subchondral bone.

Results from previous chapter indicated the potential involvement of certain IFN1 transcripts in age-related changes within the BM niche. It was thus worth examining if these IFN1 pathway transcripts have any role in OA pathogenesis, even though there has been no such investigation so far. Interestingly, in Lupus, an auto-inflammatory disease extensively studied for the role of IFN1 in its progression, BM MSCs have been shown to be senescent and express altered IFN1 signature. Based on the outlined literature evidence, it could be hypothesised that not only chondrocytes, but also MSCs from OA patients could possess an increased expression of senescence and SASP associated genes, compared to non-OA old age donors. On the other hand, the expression of anti-ageing genes *KI* and *Sirt6* could be lower. With respect to IFN1 and associated genes, there are no studies that have investigated them with respect to OA. A number of studies, however, have indicated an association between increased IFN signalling in Lupus with increased MSC senescence in the disease, which may also take place in OA MSCs (387, 391, 451).

6.1.3 Changes in MSC multipotentiality genes in MSCs from OA donors

Previous investigations of age-related changes in OA have focussed on the articular cartilage where the first noticeable physiological alterations of OA have been historically found. Only relatively recently, OA has begun to be acknowledged as a disease of the entire joint including considerable changes in the subchondral bone (452), as well as cartilage (outlined in section 6.1). The limited research performed on primary OA guinea pig model indicated that alterations in microstructures of subchondral trabecular bone pointed towards changes in the bone preceding the visible changes in the cartilage (171). However, investigations performed on human bone/bone cells and consequently, the data available for human bone cells including MSCs remains very limited.

A recent study from our laboratory has established that the numbers of MSCs measured by both CFU-F assay and flow cytometry was significantly higher in OA-affected bone (in bone marrow lesions, BML) as compared to a less affected bone in hip OA patients. Compared to non-BML MSCs, the proliferative capacity of BML-MSCs was lower, and they had altered expression levels of bone remodelling molecules *RANKL* and *OPG* (183). While this was the first study to demonstrate numerical, topographical, gene expression and functional alterations in MSCs from OA hips, these examinations were performed on culture expanded MSCs. Furthermore, no study has been yet performed to compare gene expression profiles of MSCs from OA patients with age-matched healthy individuals.

A subsequent study from our group has shown increased expression of bone lineagerelated transcripts *RUNX2, OPG* and *SPARC* in hip OA MSCs compared to healthy bone (69), however while their OA MSC donors were in the age-range of 55-89 years old, their healthy controls ranged from a much younger 38 years old up to 93 years old and not age-matched (section 1.3.1).

This chapter aimed at exploring MSC multipotentiality genes, senescence and antiageing genes, IFN1 genes and other exploratory genes previously explored in Chapter 4 and Chapter 5, in uncultured MSCs from OA donors compared to MSCs from a cohort of age-matched healthy old donors. The investigation was also performed to find whether genes that displayed age-related trends in the previous two chapters were further aggravated in OA.

The list of transcripts chosen for investigation in this chapter is presented in Table 6.1 including available literature indicating any alterations in these molecules in OA patients.

Gene	Gene (full name)	Evidence in OA
RUNX2	Runt-related transcription factor 2	(69, 453)
SPARC	Secreted Protein Acidic And Rich in Cysteine	(69)
SFRP1	Secreted frizzled-related protein 1	
PPAR-γ	Peroxisome proliferator activated receptor gamma	(69)
FABP4	Fatty acid binding protein 4	(69)
LepR	Leptin receptor/CD295	(454)
SOD3	Superoxidedismuatse 3	(445)
RANKL	Ligand for Receptor Activator of Nuclear factor-Kappa B	(69)
OPG	Osteoprotegerin	(69)
RANK	Receptor Activator of Nuclear factor-Kappa B	
CXCL12	C-X-C motif chemokine ligand 12	(455, 456)
Cx43	Connexin43	(457)
IFNA1	Interferon alpha 1	
IFNB1	Interferon beta 1	
IFNAR1	Interferon alpha receptor 1	
IFNAR2	Interferon alpha receptor 2	
STING	Stimulator of interferon genes	
IRF3	Interferon regulatory factor 3	
STAT1	Signal transducer and activator or transcription 1	
BID	BH3 interacting domain	
MTCH2	Mitochondrial carrier homolog 2	
KI	Klotho	(409)
Sirt6	Sirtuin 6	(458, 459)
Тр53	Tumour protein 53	(460)
IL6	Interleukin 6	(437, 461)
IL7	Interleukin 7	
IL8	Interleukin 8	

Table 6.1 List of genes investigated for study in MSCs from OA donors

List of transcripts investigated and discussed in this chapter. The top panel includes MSC multipotentiality genes and the bottom panel includes exploratory genes that were investigated in chapter 5 and will be explored in this chapter.

6.1.4 Changes in age-related surface markers in MSCs from OA donors

Among the age-specific surface markers discussed previously in this project (Chapter 4), CD106 expression levels on OA synovial fibroblasts has been reported to be highly elevated as compared to non OA synovial fibroblasts (462). Schett and colleagues in 2009, examined the level of CD106 in the serum from control and hip OA donors and found significantly higher levels of CD106 in the OA donors (463). Not too long after that, Pulsatelli and colleagues found that the serum levels of VCAM-1/CD106 were

significantly higher in donors with erosive hand OA compared to non OA hands (464). Another study showed that the levels of serum CD106 was an indication of the number of joints affected by OA (465). More recently, a study investigated CD106 in hip OA and found no significant increase in men or women or in the total population with respect to the CD106 levels (466). Serum CD106 could be shed from many different cell types inside the body, and no study has yet explored CD106 expression on uncultured hip OA MSCs or HLCs. Based on this evidence, it was hypothesised that expression of CD106 levels on hip OA MSCs or HLCs would be higher in OA.

The growing evidence connecting obesity to OA in the last two decades has witnessed an increased interest in the hormone Leptin and its potential contribution to OA (467). However, the hormone Leptin has been studied considerably more than its receptor. An increase in both, the levels of the hormone leptin and its receptor CD295 in the synovial fluid (468), subchondral osteoblasts (469) and cartilage of OA patients has been observed, correlating positively with the severity of the disease (454). More recently, Voultaneeho et al., in 2014 and Mogi Yan and colleagues in 2018 have found leptin with its receptor contribute to obesity in OA development (470). Voultaneeho et al., suggested that in an obese person, there is an unbalanced leptin signalling which results in an increased production of pro-inflammatory factors (270). Their work, in some way connects Leptin to OA along with the pro-inflammatory factors which have often been associated with inflamm-ageing. Co-incidentally on similar lines, Stannus and co-workers investigated the 'association between Leptin, IL6 and radiographic hip OA in older people' (471). While they did not mention the term 'inflamm-ageing' in their article, their data set forms the baseline for future studies to investigate the association between OA, obesity and inflamm-ageing. Based on all the evidence, it was hypothesised that Leptin receptor/CD295 levels in OA MSCs would be increased.

Cx43 was another surface marker that was investigated in this project due to its high prevalence on bone cells and related functionalities (section 4.1.3.3). Apart from being closely associated with intercellular interactions within bone cells, Cx43 has also been identified as a crucial player for cell communication in OA. Andrew and colleagues in 2004, found that the number of gap junctions in synovial biopsies of OA donors were increased (4-fold) as compared to the non OA donors (472). In 2014, Gupta et al., showed that the overexpression of Cx43 in human synovial fibroblast cell lines enhanced the expression of genes associated with inflammation in OA and a decrease in the expression of Cx43 in these cells, reduced the expression of many of the catabolic and inflammatory genes (473). In agreement, Casagrande and colleagues found a significantly higher level of Cx43 in the cartilage of shoulder OA patients

compared to healthy controls. Higher levels of Cx43 at the protein level were also observed in human chondrocytes (457). In another investigation, Shichomura et al., compared the levels of Cx43 in patients with RA, OA and in healthy controls (474). In contrast to the above mentioned studies, they found that Cx43 was 'hardly detected' in patients with OA. Based on this evidence Cx43 was expected to be most likely elevated in OA MSCs.

CD146 or MCAM, as discussed previously (Chapter 4) has been used to define MSCs in a number of human tissues (227, 229). The sparse data available for this marker and how its expression alters with the progression of OA, suggests that CD146 positivity was an indication of late stage OA (475). However, another study did not find any other major differences between OA patients and controls with respect to CD146 expression (476). More recently, another group examined the MSCs from synovial fluid of OA patients and found a significant increase in the number of colonies and colonies of larger diameter from OA donors (477). They also inspected the levels of CD146 in both control and OA MSCs by flow cytometry, but found no difference in its expression. Based on this contradicting evidence, CD146 expression on OA BM MSCs was expected to be similar to age-matched healthy MSCs and display no OA-related differences.

The investigations on cellular senescence in OA remain largely limited to cartilage and chondrocytes and information on bone cells in OA related senescence has a long way to go before catching up with the amount of information available on cartilage. Obesity, DNA damage, senescence, ROS - all of these factors appear to play a key role in OA, therefore any alterations in the signalling molecules connecting these pathways in OA MSCs would be a highly novel finding.

This chapter aimed at investigating the expression of all the genes in Chapter 4 and 5 and the surface markers in MSCs and HLCs from hips of OA donors, in comparison to healthy ageing to explore any trends of healthy ageing that could be further aggravated in OA.
6.1.5 Hypotheses and objectives

The following hypotheses and aims were postulated for this chapter

Hypotheses:

1. The transcripts for BM MSC osteogenic and adipogenic differentiation potential, stromal functions and bone remodelling, in CD45^{low}CD271⁺ MSCs are altered in OA patients compared to age-matched healthy individuals.

2. The transcripts related to senescence and IFN1 pathways in CD45^{low}CD271⁺ MSCs are altered in OA patients, with possible increases in senescence and IFN1 related genes as compared to age-matched healthy individuals.

3. Surface markers CD106, CD295, CD146 and Cx43 are altered in OA patients, with possible increases in CD106 and Cx43 as compared to age-matched healthy individuals.

4. Based on findings from the previous chapter, that HLCs were more susceptible to age-related differences than MSCs, OA-related trends in CD45⁺CD271⁻ HLCs were expected to be more pronounced compared to MSCs.

Objectives:

1. To quantify the level of transcripts indicating BM MSC osteogenic and adipogenic differentiation potentials, stromal functions and bone remodelling, in MSCs from OA patients by qPCR and compare with old donor MSCs.

2. To quantify the level of transcripts related to senescence and IFN1 pathway in BM MSCs and HLCs from OA patients by qPCR and compare with old donor MSCs.

3. To quantify surface markers CD106, CD295, CD146 and Cx43 in MSCs from OA patients by flow cytometry and compare with old donor MSCs.

6.2 Methods

6.2.1 Donor selection

Appendix 2, Table 4 outlines OA samples used. The donor cohort included 5 males and 8 females ranging from 56-83 years old with a median age of 74. Exclusion criteria included any history of cancers, previous surgery of the hip or any other disorder directly affecting the bone. As these donors were mostly above the age of 60, they were more likely to have primary (non-traumatic) hip OA. The FHs once obtained after surgery, were digested in collagenase for 4 hours and frozen by a colleague (183).

6.2.2 Cell sorting

Frozen vials with up to 10⁶ nucleated cells from digested OA FHs were defrosted using TM (section 2.2) and washed with PBS for the elimination of any residual DMSO (section 2.4). Once washed, cells were counted and distributed into FACS tubes and stained (refer section 2.5, Table 2.1). Owing to the large number of MSCs in the FH digest (183), the cells from FHs were used not only for the purpose of OA MSC and HLC sorting, but also for instrument setting for MSCs from BMA.

For this purpose, these cells were used as both unstained cells and cells stained with isotype controls to establish the sorting gates for positively-stained cell populations. In contrast to rare BM MSCs, these cells did not undergo any enrichment (section 2.4) prior to cell sorting. This experimental design helped to preserve rare healthy MSCs for sorting (as none were wasted for instrument settings) and ensured consistency as healthy and OA MSCs were always sorted on the same day. The cells were treated with a blocking buffer for 10 minutes followed by staining with respective antibody or isotype control (section 2.5, Table 2.1).

In the end, 7-AAD was added to the cells and they were then passed through a filtercapped FACS tube to prevent any cellular aggregates to avoid blocking of the flow channel. The instrument settings and the isotype controls were set as explained in section 2.4 and the sorted cells were collected directly into a tube containing lysis buffer. The populations collected included the same two populations of CD45^{low}CD271⁺ (MSCs) and CD45⁺CD271⁻ (HLCs) as previously described for healthy BM MSCs (section 2.4).The gating strategy and the histograms for single markers are shown in Figure 6.2 below



Figure 6.1 Quantification of surface marker expression in OA FH MSCs and HLCs

(A) Gating strategy for identification of live cells based on 7-AAD negativity followed by CD271 and CD45 gating for the selection of populations of interest (MSC and HLCs)
(B) Histograms for individual surface markers from representative donors in MSCs and HLCs from representative donor. Numbers in the top-right corner indicate MFI.

6.2.3 Gene expression

Transcripts were investigated (Table 6.1) out of which the first 12 were MSC multipotentiality transcripts and the next 9 were associated with senescence and antiageing, IFN1 pathway, and ROS. The procedures followed for the quantification of MSC multipotentiality transcripts has previously been described in Section 2.5. In brief, RNA was extracted from cell lysates (obtained from cell sorts) and stored in -80°C until further use. 2µI of RNA was reversely transcribed to cDNA which was stored in -20°C until further use. All samples were then preamplified in the thermocycler using 18 cycles and were stored at −20°C before processing on the BioMark HD. Quantitative PCR was performed using Fluidigm Flex Six[™] (MSC multipotentiality genes) and 48.48 IFC (genes associated with senescence and anti-ageing, IFN1 pathway, and ROS) on BioMark HD. Samples and assays were loaded into the reaction chambers of the Flex Six[™] using the IFC Controller HX, and then transferred to the BioMark HD for qPCR (95°C for 10 min; 40 cycles of 95°C for 15 seconds and 60°C for

60 sec). The Ct values for genes of interest were normalised to the endogenous control *HPRT1* using the formula [Δ Ct = Ct _{target gene}- Ct _{housekeeping gene}] and relative expression was calculated as 2^{- Δ Ct} and used for statistical analysis.

6.2.4 Statistical analysis

Cluster analysis was used to observe any differences in gene expression across the entire donor cohort. Statistical analysis and graphics were performed using GraphPad Prism software (version 7.0a). The normal distribution of the data was assessed using the Shapiro-Wilk and Kolmogorov-Smirnov tests for normality. As no data were found to be normally distributed, Wilcoxon rank sum test were performed for paired data and Mann-Whitney test was performed for unpaired data. The results were considered significant at p value <0.05.

6.3 Results

The results presented below include evaluation of the MSC multipotentiality gene expression, as first described in chapter 4 (section 4.3) followed by a panel of the exploratory genes, that were covered in chapter 5 (section 5.3). OA patients (median age 74 years old, range 56-83), were compared to the healthy old donor group (median age 68, range 61-89), as described in section 3.3.1. Number of MSCs sorted from control healthy donors ranged from 999 to10,000 cells (median 4,700 cells) and an average 70,000 MSCs were sorted from OA FHs. HLCs from healthy old donors and OA patients were collected from a minimum of 50,000 cells up to a maximum of 70,000 cells.

6.3.1 Differences in the expression of multipotentiality genes between OA MSCs and healthy old donor MSCs

First, the expression of genes indicating MSC multipotentiality was compared in MSCs from clinically diagnosed hip OA patients and healthy old donor MSCs (Figure 6.3, Table 6.1, top panel).



Figure 6.2 Expression of genes associated with MSC multipotential functions in MSCs from healthy old donors and OA patients

(A) Expression of *RUNX2*, *SPARC* and *SFRP-1*, (B) Expression of *PPAR-\gamma*, *FABP4* and *LepR*, (C) Expression of *RANK*, *RANKL* and *OPG* (D) Expression of *CXCL12*, *Cx43* and *SOD3*. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data indicate median values. $p<0.001^{***}$ and $p<0.0001^{****}$, Mann-Whitney U test.

With respect to osteogenic transcripts, there was no significant differences in the level of their expression between OA and old donor MSCs. Expression of *RUNX2* was 2.43-fold higher in OA patients and expression of *SFRP1* was 4-fold lower in OA patients but the differences failed to reach statistical significance (p=0.062). Expression of *SPARC* was similar (Figure 6.3A).

Next, the transcripts associated with adipogenic differentiation and bone-fat balance were examined. Both *LepR* and *PPAR-\gamma* showed significant decline (55-fold, *p*<0.0001 and 16-fold, *p*<0.0001, respectively) in OA MSCs compared to healthy old donor MSCs. *FABP4* showed a tendency of lower expression in OA MSCs (2.5-fold), however, the differences failed to reach statistical significance. Overall, all the three transcripts displayed a decline in OA patients indicating a decline in the formation of adipose tissue in OA patients as compared to healthy old donors (Figure 6.3B).

There was a non-significant decline in the expression of *RANKL* and a significant decline (10-fold) in *OPG* expression in OA patients as compared to healthy old donors (Figure 6.3C) potentially indicating changes in the rate of bone remodelling in OA patients. Among the transcripts associated with MSC stromal function, *Cx43* showed no difference in expression between OA patients and healthy old donors but *CXCL12* was found to be significantly 58-fold lower (p<0.0001) in OA patients (Figure 6.3D). This indicated that the stromal functions in MSCs from OA donors were potentially impaired as compared to MSCs from healthy old donors. The expression of *SOD3* was found to increase by 2-fold in OA MSCs as compared to MSCs from healthy old donors. While this was unexpected, future evaluation of this gene along with ROS quantification in FH MSCs would help in gaining better understanding of oxidative stress in uncultured BM MSCs.

Overall, these data potentially indicated a lesser impairment in the osteogenic capacity of OA MSCs, compared to their other functions (such as adipogenesis and stromal support).

6.3.2 Differences in the expression of multipotentiality genes between OA HLCs and healthy old donor HLCs

The investigation of transcripts associated with MSC osteogenic differentiation in HLCs is indicated in Figure 6.4 and the median values for the same is outlined in Table 6.1, bottom panel. No differences were observed in the expression of *RUNX2* and a significant 13-fold increase in the expression of *SPARC* (p=0.0003) in OA HLCs compared to healthy old donors HLCs (Figure 6.4A). Adipogenic transcripts *LepR* and *PPAR-* γ did not display any difference in OA HLCs compared to old donors HLCs, but *FABP4* showed significantly 8-fold higher levels in OA HLCs (Figure 6.4B).

With respect to bone remodelling transcripts, *RANK* and *RANKL* showed no difference in HLC expression levels between healthy old donors and OA patients. OPG was not detected in healthy old donors, therefore it could be reasonably assumed that its expression in OA donors HLCs was elevated (Figure 6.4C).

When the transcripts associated with MSC stromal support were investigated, there was no significant differences in the HLC levels of either *Cx43* or *CXCL12* transcripts between OA patients and healthy old donors. Nevertheless, the expression of *CXCL12* was found to be 2-fold higher in OA HLCs whereas, expression of *Cx43* was 2-fold lower in OA HLCs as compared to HLCs from healthy old donors (Figure 6.4D).

Differences in multifunctionality genes in OA were not aggravated from ageing and thus were only associated with OA. Increase in the expression of *SOD3* in both MSCs and in HLCs in OA was unexpected as OA is often associated with declined SOD3 and increased ROS (446). This suggests need for future work including the detection of ROS from uncultured OA donors alongside the quantification of a larger panel of anti-oxidant enzymes.



Figure 6.3 Expression of genes associated with MSC multipotential functions in HLCs from healthy old donors and OA patients

(A) Expression of *RUNX2*, *SPARC* and *SFRP-1*, (B) Expression of *PPAR-\gamma*, *FABP4* and *LepR*, (C) Expression of *RANK*, *RANKL* and *OPG* (D) Expression of *CXCL12*, *Cx43* and *SOD3*. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data indicate median values. *p*<0.001* and *p*<0.0001***, Mann-Whitney U test.

Overall, HLCs displayed a different pattern of differences to MSCs between OA patients and healthy old donors. Fold differences and p values of all of the above mentioned transcripts in healthy old donors and OA patients are shown in Table 6.1 and 6.2

Cells	Genes	Medians in Old	Medians in OA	Fold difference*	p value
MSCs	RUNX2	1.09	2.65	2.43	NS
	SPARC	20.51	24.54	1.20	NS
	SFRP1	2.83	0.71	0.25	NS
	$PPAR-\gamma$	6.20	0.37	0.06	<0.0001
	FABP4	1.77	0.69	0.39	NS
	LepR	4.13	0.07	0.02	<0.0001
	SOD3	0.29	0.66	2.28	NS
	RANKL	0.35	0.20	0.56	NS
	OPG	2.44	0.25	0.10	0.0008
	RANK	ND	ND	NA	NA
	CXCL12	1554.00	26.19	0.02	<0.0001
	Cx43	31.55	24.51	0.78	NS
HLCs	RUNX2	0.35	0.33	0.95	NS
	SPARC	0.04	0.51	13.86	0.0003
	SFRP1	ND	0.002	NA	NA
	FABP4	0.0145	0.1175	8.12	0.0409
	PPAR-γ	0.04	0.02	0.48	NS
	LepR	0.0023	0.0023	0.99	NS
	SOD3	ND	0.009	NA	NA
	RANKL	0.35	0.19	0.54	NS
	OPG	ND	0.09	NA	NA
	RANK	0.10	0.07	0.71	NS
	CXCL12	0.03	0.06	2.21	NS
	Cx43	2.79	1.05	0.38	NS

Table 6.1 Differences in the expression of MSC multipotentiality genes in OA MSCs and HLCs as compared to healthy old donors

Top panel of genes include expression of MSC multipotentiality genes in MSCs and the bottom panel includes expression of HLCs . All expressions are relative to *HPRT1*, Mann-Whitney U test. * Medians in OA/ medians in old, NS: non-significant

6.3.3 OA related differences in the expression of senescence, SASP and IFN pathway related genes in MSCs and HLCs

An overall cluster analysis of all exploratory genes revealed some segregation of MSCs and HLCs within FH samples, but not as clear as seen in section 5.3.1, Figure 5.3.



Figure 6.4 Cluster analysis of exploratory genes in OA samples in MSCs and HLCs

Expression of *IFNA1, IFNB1, IFN receptors,* IRGs, genes associated with IFN1>BID>ROS pathway, senescence and other exploratory genes grouped by cluster analysis. Dendogram on the top indicate samples and dendogram on the right indicates genes. Colour key in shown at the bottom right of the figure. Grey squares indicate gene expression values that were below detection.

The median values for complete list of genes and differences in old healthy donors and OA patients is presented in Appendix 4, Tables 6.1 and 6.2 for gene expression in MSCs and HLCs, respectively.

With respect to senescence and anti-ageing genes, the following results were obtained. Irrespective of the cell population, the anti-ageing gene *KI* expression decreased by 4.8-fold in MSCs and by 5.5-fold in HLCs in OA patients as compared to healthy old donors. MSCs from OA patients showed over 3-fold decline in *Tp53* but no change in OA HLCs was observed. The expression of *Sirt6* remained unchanged in both OA MSCs and HLCs. High level of donor variation was observed in the expression of *KI* in OA patient samples in both, MSCs and in HLCs, and none of these trends were found to be statistically significant (Figure 6.6).



Figure 6.5 Expression of senescence and anti-ageing genes in MSCs and HLCs from healthy old donors and OA patients

(A) Expression of *Tp53*, *Sirt6* and *KI* in MSCs. Grey dots indicate old donors and empty circles indicate OA patients. (B) Expression of *Tp53*, *Sirt6* and *KI* in HLCs. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data set indicates median values.

When the expression of SASP-associated cytokines was analysed, *IL6* displayed a significant 132-fold increase in OA MSCs (p=0.0101) and also showed a 6-fold

increase in HLCs (not significant), these trends were consistent with its hypothesised changes in OA. *IL7* showed a non-significant 5-fold decline in MSCs and no change in HLCs. *IL8* did not reach statistical significance but showed opposite trends in MSCs and HLCs. While in MSC, it displayed nearly 8-fold increase in OA, in HLCs it had a nearly 4-fold decline (Figure 6.7).



Figure 6.6 Expression of SASP-associated cytokines in MSCs and HLCs from healthy old donors and OA patients

(A) Expression of *IL6*, *IL7* and *IL8* in MSCs, (B) Expression of *IL6*, *IL7* and *IL8* in HLCs. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data set indicates median values. p<0.05*, Mann-Whitney U test.

Overall, anti-ageing genes and senescence associated genes did not display any significant OA-related differences even though *KI* declined in OA patients as displayed in both cell populations. SASP associated cytokine genes displayed significant increase in OA MSCs and a non-significant increase in HLCs.

Cells	Genes	Medians in Old	Medians in OA	Fold difference*	p value
MSCs	IFNA1	9.15	2.23	0.24	NS
	IFNB1	1.72	0.27	0.16	NS
	IFNAR1	4.75	1.26	0.27	NS
	IFNAR2	5.76	1.36	0.24	0.041
	STING	4.88	1.25	0.26	NS
	IRF3	3.13	0.85	0.27	0.051
	STAT1	5.16	1.47	0.28	0.0221
	BID	0.11	0.47	4.27	NS
	MTCH2	1.27	0.47	0.37	NS
	KI	0.48	0.1	0.21	NS
	Sirt6	0.4	0.3	0.75	NS
	Тр53	2.88	0.81	0.28	NS
	IL6	0.2	26.37	131.85	0.0101
	IL7	1.17	0.22	0.19	NS
	IL8	5.14	40.45	7.87	NS
HLCs	IFNA1	7.44	1.17	0.16	NS
	IFNB1	1.59	0.39	0.25	NS
	IFNAR1	0.48	0.65	1.35	NS
	IFNAR2	1.38	0.66	0.48	0.035
	STING	1.033	0.457	0.44	NS
	IRF3	0.66	0.43	0.65	NS
	STAT1	2.4	1.57	0.65	NS
	BID	1.032	0.62	0.60	NS
	MTCH2	0.79	0.34	0.43	NS
	KI	0.011	0.002	0.18	NS
	Sirt6	0.15	0.1	0.67	NS
	Тр53	1.21	0.75	0.62	NS
	IL6	0.59	3.92	6.64	NS
	IL7	0.013	0.015	1.15	NS
	IL8	67.21	27.35	0.41	NS

Table 6.2 Differences in the expression of genes associated with IFN1 pathway,
ROS, anti-ageing and senescence, in OA MSCs and HLCs as compared to
healthy old donors

Top panel of genes include expression of MSC multipotentiality genes in MSCs and the bottom panel includes expression of HLCs. All expressions are relative to *HPRT1*, Mann-Whitney U test. * Medians in OA/ medians in old, NS: non-significant.

Differences in *IFNA1*, *IFNB1* and *IFNA* receptors were next analysed for understanding OA associated changes in relation to old age (Figure 6.5). In OA patients, *IFNA1* showed a non-significant decline (4-fold and 6-fold) in MSCs and HLCs, respectively. *IFNB1* showed similar trends with 6- and 4-fold declines in OA MSCs and HLCs,

respectively. *IFNAR1* showed a 3-fold decline in OA MSCs but its expression was unaltered in OA HLCs. *IFNAR2* showed a statistically significant decline in both MSCs (p=0.041) and HLCs (p=0.035) in patients with OA in comparison with old donors.



Figure 6.7 Expression of IFNA, IFNB and IFNA receptor genes in MSCs and HLCs from healthy old donors and OA patients

(A) Expression of *IFNA1* and *IFNB1* (top panel) and *IFNAR1* and IFNR2 (bottom panel) in MSCs (B) Expression of *IFNA1* and *IFNB1* (top panel) and *IFNAR1* and *IFNAR2* (bottom panel) in HLCs. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data set indicates median. p<0.05*, Mann-Whitney U test.

Among the IRGs, a number of genes were found to be differentially expressed in both MSCs and in HLCs in OA patients compared to old donor populations. Figure 6.9

below shows some of the differentially expressed genes in MSCs. While most of the genes declined in OA patients, *ISG20, LAMP3* and *IFI27* were significantly higher in OA patients MSCs. The medians, fold differences and statistical significance (if p<0.05) of all the investigated genes are shown in Table 6.2 and Appendix 4.



Figure 6.8 Genes expression of IRGs in OA MSCs compared to MSCs from healthy old donors

Examples of differentially expressed IRGs in MSCs from OA donors compared to healthy old donors. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data set indicates median. p<0.05* and p<0.01**, Mann-Whitney U test.

Figure 6.10 displays examples of genes that were differentially expressed in HLCs in OA patients as compared to healthy old donors. Interestingly, *CASP1*, *ISG20* and *LAIR1* were differentially expressed in OA patients irrespective of the population examined (*ISG20* was higher in OA and *CASP1* and *LAIR1* were lower in OA).



Figure 6.9 Genes expression of IRGs in OA HLCs compared to HLCs from healthy old donors

Examples of differentially expressed IRGs in HLCs from OA donors compared to healthy old donors. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data set indicates median. p<0.05* and p<0.01**, Mann-Whitney U test.

The genes associated with the IFN1>BID>ROS pathway were next investigated for any differences in OA patients as compared to healthy old donors. In MSCs, all the genes except *BID* displayed a decline in OA with *STAT1* showing a significant decline (p=0.0221). However, in HLCs, the trends for decline were more subtle, even though a minor decline stayed constant for almost all the genes (Figure 6.11).



Figure 6.10 Expression of genes associated with IFN1>BID>ROS pathway in MSCs and HLCs from healthy old donors and OA patients

(A) Expression of *STING*, *STAT1* and *IRF3* (top panel) and *BID*, *MTCH2* (bottom panel) in MSCs, (B) Expression of *STING*, *STAT1* and *IRF3* (top panel) and *BID*, *MTCH2* (bottom panel) in HLCs. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data set indicates median values. p<0.05^{*}, Mann-Whitney U test.

In summary, section 6.3.3 investigated differences in exploratory gene expression including anti-ageing and IFN1>BID>ROS pathway genes, anti-ageing, SASP and cytokines, and identified novel molecules that displayed significant differences in their expression in OA MSCs. *IFNAR2* declined significantly in OA MSCs along with a number of IRGs and *STAT1*, a downstream molecule through which IFNs induce the expression of IRGs. As expected, *IL6* increased significantly in OA while anti-ageing gene *KI* decreased by nearly 5-fold in OA patients. Altogether, these expression patterns indicated a possibility for a reduced IFN1 signalling in OA MSCs, which could further impact on their potency and senescence status.

6.3.4 Biomarkers of age-related OA

In the course of this study, a number of genes that displayed a trend towards agerelated differences, were also found to have a similar and aggravated trend in OA. While the expression of some of these genes was found to decline with both increasing age and OA, the others showed an increase in their expression. In MSCs, the statistical significance in these 'aggravated' genes was only observed for *IL6* (Figure 6.12A), while more such genes were found in HLCs. In itself, this indicated that MSCs were more resistant to ageing and OA, compared to HLCs (Figure 6.12B). This section discusses the genes in both cell populations that have displayed distinct trends and statistical significance with respect to age-related differences that were further exacerbated in OA.

The on transcript that displayed a distinct age-related increase followed by a further increase in OA MSCs was *IL6* (p=0.0005), as shown in Figure 6.12A. Genes that displayed a age-related decrease followed by a further decrease in expression in OA HLCs included the anti-ageing gene *Sirt6* (p=0.02), senescence associated *Tp53* (p<0.0001). Interestingly, *IFNAR2* (p=0.0026) also displayed and age and OA-related decrease in HLCs, potentially suggesting an overall decline in the IFN1 cascade with advancing age, that is aggravated in OA. Thus, it was not surprising that the stimulator of IFN genes or *STING* was also found to display an age-and OA-related decline (p=0.008) along with *MTCH2* (p=0.0019), all shown in Figure 6.12B.



Figure 6.11 Expression of genes that displayed age-related differences that were further exacerbated in OA

(A) Expression of *IL6* in MSCs (B) Expression of *RANKL*, *Sirt6*, *Tp53*, *IFNAR2*, *STING* and *MTCH2* in HLCs. Young donors, old donors and OA patients are indicated by black dots, grey dots and empty black circles, respectively. The black line indicates median values. *p<0.05, **p<0.01 and ****p<0.0001, Kruskal-wallis test with Dunn's correction.

Overall, this section outlines the potential markers of age-related OA reflecting the *in vivo* BM niche in uncultured BM MSCs and in HLCs. Further examination of these genes at both gene and protein level in a larger donor cohort would be extremely useful in translating these genes as biomarkers.

6.3.5 Differences in surface marker expression in OA MSCs and HLCs compared to healthy old donors

While none of the surface markers displayed any age-related changes in MSCs (Chapter 4, figure 4.14), CD106 and CD295 showed a significant decrease (4-fold, p=0.0003 and 4-fold, p=0.0043, Figure 6.13A and B, respectively) in their expression in

OA MSCs as compared to old donors MSCs. Expression of CD146 and Cx43 remained unchanged (Figure 6.13C and D). Interestingly, the decline in *LepR* both at the transcript level as well as in the protein level was found to be consistent and statistically significant, suggesting an imbalance between osteogenic and adipogenic differentiation abilities in patients with OA.



Figure 6.12 Surface marker expression in MSCs from healthy old donors and OA donors

(A) CD106 (B) CD295 (C) CD146 and (D) Cx43 from old healthy donors and OA patients. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data indicate median values. *p<0.05 and ***p<0.001, Mann-Whitney U test.

Following comparisons in MSCs, surface marker expression was investigated in HLCs. CD106 and CD146 displayed a trend for increased expression in HLCs (2-fold in each) but the differences failed to reach significance. CD295 and Cx43 on the other hand, displayed a decline in expression in OA patients, but again failed to reach significance (2-fold each) (Figure 6.14). CD295 was found to decline in both MSCs as well as in HLCs.



Figure 6.13 Surface marker expression in HLCs from healthy old donors and OA donors

(A) CD106 (B) CD295 (C) CD146 and (D) Cx43 from old healthy donors and OA patients. Healthy old donors and OA patients are indicated in grey dots and black empty circles, respectively. Horizontal line across data indicate median values.

Overall, compared to healthy aged MSCs, MSCs from OA patients displayed a significant decline in *PPAR-*₇, *LepR*, *OPG* and *CXCL12* expression. None of these changes could be described as 'aggravated ageing' since all of these genes showed no age-related differences in healthy donor MSCs (section 4.4.3). *LepR* was also investigated at the protein level and a similar significant decline was confirmed in OA patients. This was also unique to OA since no age-related changes in LepR/CD295 were found in healthy old donors (section 4.3.5). Furthermore, like CD295, CD106 was another protein that displayed a very significant decline in MSCs from OA patients, which was likely to be associated with other local or systemic factors contributing to OA since no age-related differences in its expression were found in the previous chapter (section 4.3.5).

In OA HLCs, anti-ageing genes *Sirt6 and KI*, declined significantly and the expression of IFN-related genes of *IFNAR2*, *STING*, *MTCH2* and *IRF3* also declined significantly.

These genes were found to decline in healthy ageing as well. From these findings, it is tempting to speculate that in HLCs, age-related trends particularly in relation to IFN pathway, are further exacerbated in OA. But this is not the case for MSCs. Interestingly, the pro-inflammatory and SASP-associated cytokine *IL6* transcript was found to increase by over 100-fold in OA MSCs and by 7-fold in OA HLCs. *IL6* expression also increased by 4-fold in healthy ageing in both MSCs and HLCs (section 5.3.1). As *IL6* was found to increase in healthy ageing and further aggravated in OA, in both MSCs and HLCs, it would be a good biomarker of age-related OA.

6.4 Discussion

The results from this chapter reveal very interesting alterations in MSCs from OA FH bone compared to healthy aged BM MSCs. Gene expression results indicated a decline in the adipogenic potential of MSCs from OA FHs, no alteration in the levels of osteogenic differentiation transcripts and a significant decline (nearly 10-fold) in the expression of OPG in MSCs from OA patients. A significant decline (60-fold) in the level of *CXCL12* implicated in cell recruitment was also observed. Interestingly, there was no significant change in the transcript encoding an anti-oxidant enzyme *SOD3* in OA MSCs compared to healthy old donors MSCs. The study of exploratory transcripts encoding cytokines revealed an increase in the level of *IL6* transcript in OA, as expected from literature (461, 478). *STAT1* declined in OA patients while no significant changes were observed in ageing and senescence associated genes (*KI, Sirt6, Tp53*). Interestingly, the anti-ageing gene *KI* decreased in OA patients by nearly 5-fold but statistical significance wasn't achieved due to high donor variation in its expression levels.

Transcripts associated with osteogenic differentiation potential showed no difference in OA MSCs compared to healthy old donor MSCs. A recent study from our group (Ilas et al., 2019) investigated the level of *RUNX2* in BM MSCS from the FH of OA patients and found a significant increase in the expression of *RUNX2* in MSCs from OA patients as compared to MSCs from healthy donors (69). Even though this was the most relevant study for comparison with the PhD project, the OA patients were not age-matched with healthy donors in the Ilas et al., study. In this project, the genes associated with osteogenic differentiation showed no difference or non-significant 2-fold increase (*RUNX2*) in OA MSCs as compared to healthy old MSCs. Future work with genes associated with osteogenic differentiation in OA must firstly consider the different stages OA and analyse the gene expression separately. Investigating these

molecules at the protein level would further enhance the current understanding of OA progression.

The expression of adipogenic transcripts in MSC population in this project suggested a decline in the fat forming capacity of MSCs from OA donors. This is in agreement with a previous study by Murphy et al. that have shown that the adipogenic potential of MSCs from OA donors is compromised. Interestingly, they also found a decline in the osteogenic potential of MSCs from OA donors which was not what was observed in this project. The variation could be potentially due to the fact that Murphy et al., used cultured cells as opposed to uncultured cells as in this project. Also, while they measured the differentiation potential of MSCs from OA donors in vitro, in this project, the cells were uncultured and hence better represented human OA conditions in vivo. Interestingly, another study from our group published in 2016 (183) found no defect in the adipogenic differentiation potential of culture-expanded MSCs either. These studies too tested the adipogenic potential of MSCs from hip OA donors in vitro under normal culture conditions, further highlighting the need to dissect any changes in OA MSCs prior plastic adhesion and culture. Overall, the expression of adipogenic transcripts were not found to be consistent with the hypothesis that there might be an exaggerated increase in OA, possibly due to other factors like lifestyle and obesity that are involved in age-related OA but were not accounted for in this project.

Recently, a study from our group investigated the expression of MSC multipotentiality transcripts in MSCs from OA FH as compared to the MSCs in healthy donors (69). Interestingly, it showed a significant increase in the expression of *OPG*, *RUNX2* and *SPARC* in patients with OA as compared to healthy donors. In this PhD project, the trend for *RUNX2* expression was found to be consistent with the previous results from our group, but the level of *SPARC* expression was not. Even though the cells used were also uncultured CD45^{low}CD271⁺ cells, the patient cohorts in Ilas et al., study were not as well age-matched. The Ilas et al., study also used a larger panel of osteogenic transcripts. While expanding the number of transcripts for investigating multipotential functionalities of MSCs would be the way forward, low numbers of CD45^{low}CD271⁺ MSCs in aged donors represent a significant limitation for future work. Using techniques like RNA seq to compare differentially expressed genes in OA as compared to healthy old donors would provide further insight into OA development.

In this project, both *RANKL* and *OPG* displayed a decline in MSCs from OA patients as compared to healthy old donors. While the decline in OPG was significant, the difference in *RANKL* failed to reach statistical significance. *RANKL* and *OPG* are

involved in bone remodelling (discussed in Introduction, section 1.1.4 and in Chapter 4, section 4.1.1) and have been extensively investigated in OA. More relevant to this study, Bitenc Logar and colleagues in 2007 investigated the levels of *RANKL, OPG* in the trabecular bone samples (uncultured) in OA patients (479). They found a significant decline in the levels of *RANKL* in the samples from OA patients compared to those with femoral neck fractures. While they used uncultured samples from donors, they did not isolate/purify any cell types for their experiments and studied mixed population of all the cell types present within the subchondral bone. The most recent study from our laboratory has shown higher *OPG* levels in OA CD45^{low}CD271⁺ MSCs compared to healthy OA MSCs whereas *RANKL* transcript levels were not altered (69). These data differ from the findings from this thesis where *OPG* was significantly lower in OA MSCs. This could be due to the fact that the age of healthy old donors and OA patients in Ilas et al., paper was not as closely age-matched and included healthy old donors in their 40s as compared to this thesis where old donors were strictly over 59 years old.

In another study, Upton and co-workers found a significant increase in the levels of both *RANKL* and OPG obtained from grade 2 OA patients compared to grade 0 OA using K/L system (480). However, Upton and co-workers had used articular cartilage and used IHC for their studies. Kwan et al showed a significant increase in the levels of OPG in OA patients cultured osteoblasts with high levels of prostaglandin E₂ (PGE₂) in comparison with OA patients with lower levels of PGE₂, but not in comparison with healthy donors. In fact, in comparison with healthy donors, the level of *OPG* was lower in both the OA groups but did not show statistical significance. They observed no significant differences in the levels *RANKL*. Considering the MSCs are pre-osteoblasts cells, the results from Kwan et al., study are very comparable to that of the current project. However, they used cultured (P1) cells making the work in this PhD project novel due to the use of uncultured cells for better *in vivo* correlation of the disease progression in humans.

This project also investigated *RANKL* and *OPG* in HLCs which showed no differences in OA patients for *RANKL*. *OPG* was not detected in HLCs from old donors and measurable in OA group suggesting a potential for increase. This project thoroughly investigated transcripts associated with bone remodelling in two pure BM cell populations for understanding bone remodelling processes in OA. This has not been done previously, making this project novel.

Transcripts investigated for MSC stromal functions in this study in MSCs and HLCs from OA patients included CXCL12 and Cx43. In the present study, there was a

significant decline observed in the expression of *CXCL12* in MSCs from OA patients as compared to MSCs from healthy old donors. However, there was no difference observed in the expression of *Cx43* in MSCs between OA patients and old donor groups. In HLCs, their differences observed were not significant. *Cx43* has been shown to be increased in synovial cells OA patients (457) and recently, *Cx43* has been suggested as a therapeutic target to halt OA progression (481). However, both of these studies were performed on cartilage and not on bone.

CXCL12 has been previously investigated in OA owing to its role in chemotaxis and movement of MSCs and other BM cells and its close links with hematopoietic compartment within the BM niche. Previous investigations of CXCL12 in human OA patients remain limited and debatable. A study of MSCs from synovial joint of OA patients showed that the levels of CXCL12 were much lower in OA as compared to those without the disease (456). Conversely, another study suggested that CXCL12 binding to its receptor CXCR4 contributed to the induction of cartilage degradation in OA (455). Even though a part of the study detected higher levels of CXCL12 in the OA patients, the model used here was cartilage explants from human OA patients. The accumulation of alternating results complicate the process of finalising the underlying mechanism that brings about the changes in the levels of CXCL12 in OA. However, with respect to OA, a study in 2017 identified CXCL12 as an important indicator of knee OA using microarray analysis (482). However, there was no difference in the expression of CXCL12 in HLCs in this project. Investigating CXCL12 at the protein level along with the gene, in more donors and comparison in different stages of OA would allow in better understanding of the gene with disease progression.

LepR was also investigated in this thesis, both at the transcript and the protein level. Data from this chapter suggested a significant 55-fold decline in the expression of *LepR* transcript in OA MSCs as compared to healthy old donor MSCs. This trend was consistent with the surface marker expression of *LepR* (CD295) which also displayed a significant decline (3-fold) in MSCs from OA donors. This potentially indicated the impairment of bone-fat balance in OA bone. Interestingly, there was no difference observed in expression of *LepR*, both at the transcript and at the protein level, in the HLCs when compared between OA patients and healthy old donors.

The hormone leptin has been found to have increased expression in the blood, plasma (483) and in the synovial fluid of OA patients and has also been shown to correlate positively with the severity of the disease (468, 484). In other studies, synovial Leptin level has been positively associated with the pain in hip and knee OA in menopausal

women (484) as well shown to be directly proportional to BMI in OA patients (485). A review in 2013 further suggested that the missing link between obesity and OA was the hormone Leptin (270). Interestingly, Leptin was also found to increase in synovial fluid of OA patients and positively correlate with the age of the OA patients (468). Leptin has also been associated with increased levels of bone forming markers in OA patients (486). This could potentially explain the loss of bone-fat balance in OA, whereby Leptin hormone potentially increases and expression of LepR decreases, as seen from data in this chapter. One study reported higher levels of both the hormone leptin and its receptor in OA, but this was investigated in the cartilage OA patients (454).

Considering all of the above literature, it is clear that a lot of work has been performed on the role of Leptin in OA patients consistently reporting elevated levels of Leptin in OA patients. However, most of the studies have been focussed on synovial fluid, cartilage or serum levels. Expression levels of LepR via which the hormone Leptin functions has largely been overlooked and LepR in uncultured MSCs from OA donors has not been studied before. Lower levels of LepR in OA MSCs could potentially be a reaction to the higher levels of leptin in blood, which has already been suggested as a biomarker of the disease. Future work to understand differences in Leptin and its receptor in OA would ideally include blood and BM samples from OA patients to investigate them together in comparison with healthy old donors. Since there was no age-related difference in the expression of CD295 (Chapter4, section 4.3.8), the decline in CD295 observed in this chapter is clearly, specific to OA.

Interestingly, there has been an increase in the number of studies associating OA with ROS and ageing (487). *SOD3* expression in human cartilage is believed to decline in the later stages of OA (445). However, this project found a non-significant 2-fold increase in the level of *SOD3* in OA MSCs as compared to healthy old donor MSCs, which was unexpected. The fact that the present study found minor age-related difference in ROS measurement and that was complemented by a lack of any major difference in *SOD3* in BM MSCs (sections 4.3.3), is in line with the observation of no aggravated differences in *SOD3* in OA MSCs. This supports the notion that levels of ROS and anti-oxidant enzymes quantified in culture expanded MSCs (288) are likely to be in a state of induced oxidative stress and do not mirror the oxidative status of MSCs *in vivo*.

Measurement of ROS alongside antioxidant enzymes genes (like Gx, SODs) in uncultured MSCs has not been performed before. However, the current study did not show any age-related or OA-related differences. Future work should include a larger panel of number of anti-oxidant enzymes (including all the SODs, Gx, Ascorbic acid) and perform these measurement in larger number of donors. Culture expanded cells are automatically exposed to oxygen concentrations that are significantly higher than in the BM niche *in vivo*, inducing oxidative stress the moment they are cultured in laboratory conditions. This could also explain why data from this project (Section 5.3.1) did not observe any differences in the level of ROS in young and old donor MSCs either.

With respect to the IFNA1, IFNB1 and IFNA receptors, all of these were found to be reduced in OA patients with IFNAR2 showing significant decline in both MSCs and HLCs. The presence of IFNA receptors has increasingly been associated with chondrocytes, synovial fibroblasts (488) and with activities their like mediating immune responses and operating various signalling networks within the bone (489). How the expression of IFN receptors change with progression of OA is a new field of research, but this can modulate binding with IFNs, and influence the immune signalling by downstream induction of various IRGs. A number of IRGs were found to be differentially expressed in both MSCs and HLCs from OA patients including LAIR1 (decreased), CASP1 (decreased) and ISG20 (increased). LAIR1 is known as leukocyte-associated immunoglobulin-like receptor 1 that has previously been proposed as a marker of OA due to significantly higher expression in OA patients (490). In this project however, the expression was found to be significantly lower in both cell types of OA patients. The difference in trend with the aforementioned research could be due to the fact that their source was all cells from the synovial fluid. CASP1 or caspase1 has also been shown to be upregulated in OA patients when investigated in peripheral blood mononuclear cells (491) or more recently, from synovium (492). However, these studies were based on patients identified with knee OA and not hip OA, and also explored these molecules in immune cells rather than in MSCs. ISG20 very interestingly, unlike other genes in this chapter, indicated a significant increase in expression (irrespective of cell type) in OA patients. While there is not enough known about the involvement of this gene in OA, data from this project merits further investigation.

Analysis of the results of the genes associated with IFN1>BID>ROS pathway revealed a non-significant decline in nearly all genes, except for *STAT1*, which showed a significant decline in MSCs from OA donors and a non-significant decline in HLCs from OA donors as compared to healthy old donors. A decline in STAT1 was aligned with the reduced expression of IFN receptor explained above. A study that compared *STAT1* in patients with rheumatoid arthritis (RA), OA as well as spondyloarthritis (SpA) found an overall increase in the level of *STAT1* in all of the 3 types of arthritis from tissue samples at the protein level (493). Another study that compared *STAT1* at the RNA and the protein level, found that *STAT1* was significantly higher in RA as opposed to OA patient synovial tissue samples (494). The opposite trends in this project can be explained by different tissue sources and techniques.

None of the anti-ageing genes or genes associated with senescence showed any significant differences in either of the cell types in OA, with *Tp53* showing a non-significant 3.5-fold decline in MSCs from OA patients. Previously, *Tp53* downregulation has been associated with reduced chondrocyte apoptosis due to strain (460) and another study suggested that miR-34a in the human chondrocytes played a vital role in OA progression via the Sirt1/p53 signalling pathway, making it a potential therapeutic target for OA. *Tp53* has also recently been labelled as one of the 'key genes' to be linked with OA, based on bioinformatics analysis (495).

Not only has *Sirt6* been suggested to prevent cellular senescence in chondrocytes (458), it has also been proposed as a potential therapeutic target for OA (459). *Sirt6* expression in this project showed no difference in OA MSCs or HLCs as compared to healthy old donors. Previous studies have indicated lower expression of *Sirt6* not only in aged mice but also in mice and humans with OA, as compared to healthy controls (496). The study also claimed to protect mice from cartilage damage using Lenti-Sirt6 intra-articular injection and to reduce chondrocyte senescence. Another study showed that inhibition of *Sirt6* from human chondrocytes *in vitro* reduced proliferation, increased senescence, DNA damage and telomere dysfunction (497). While there was no age-related or OA-related difference found in MSCs or HLCs in *Sirt6, Tp53* demonstrated an age-related decline which was further aggravated in OA (section 6.3.4) suggesting its role in ageing, senescence as well as in OA. Exploring Tp53 with p21 and p16 at gene and protein level in larger number of donors would be extremely useful in understanding the role of *Tp53* in age-related diseases.

KI displayed a trend for decline (5-fold) in both MSCs and in HLCs in OA donors compared to healthy old donors. Previously, *KI* has been shown to be associated with female Caucasian patients with hand OA wherein they found that a genetic variant in the gene was related to increased vulnerability towards hand OA (498). They also suggested that this association was potentially stemming from bone rather than cartilage. Similarly, another study of *KI* genetic variants found that this gene was closely related to knee OA in Greek population (499). Each of these investigations recruited over 200 OA patients from whom peripheral blood sample was examined for

208

the single nucleotide polymorphism in the gene. Very recently, Tilly et al., found that *KI* expression showed an age-related decline in articular cartilage (500). Although *KI* expression increased in OA chondrocytes, $KL^{-/-}$ mice did not show any OA-related changes. They suggested that *KI* potentially modulated pro-inflammatory cytokines in OA cartilage. Like *Tp53*, *KI* too displayed an age-related decline in expression which was further aggravated in OA, even though the aggravated decline failed to reach statistical significance.

Among the cytokines, *IL6* showed a significant increase in its expression in MSCs and a non-significant increase in HLCs, in OA patients in this project. IL6 is among the most well examined cytokines associated with inflammation and ageing, as shown in discussion of Chapter 5. Thus it was not surprising to observe significantly higher levels of this cytokine transcript in OA. With respect to healthy ageing, there was a 4-fold increase in the expression of *IL6* in both, MSCs and in HLCs. Previous studies have reported elevated levels of IL6 in OA patients, however, most of these studies have used blood/serum samples (501) or synovial fluids (461). Therefore, this was the first study to investigate IL6 in uncultured MSCs from OA patients and in two cell populations from healthy donors (MSCs and HLCs) simultaneously. IL7 is well-known cytokine to be secreted by BM MSCs (502). With respect to OA, this project found a trend for its decline in OA MSCs and no difference in HLCs. Data on the involvement of IL7 in OA is limited and to the best of my knowledge, has not yet been available on uncultured BM MSCs. Considering that IL7 plays an important role in cell proliferation and maturation of lymphoid cells along with having therapeutic potential as shown in a recent diabetic rat model (503), further investigations on IL7 would be interesting.

Among the surface markers, CD106 and CD295 were found to be significantly lower in OA MSCs in this project. Previously, CD106/VCAM-1 measured using ELISA from blood of OA patients who underwent surgery (n=60) revealed a very significant increase as compared to patients who did not undergo surgery (n=852) (463). Based on the findings that VCAM-1 levels are elevated in severe OA from the previous study, a very recent study investigated the serum levels of VCAM-1 in patients with hip OA (n=100) (466). They found a non-significant 8% increase in the levels of VCAM in patients with hip OA, and concluded that VCAM-1 was not able to distinguish severe OA from age and sex-matched controls. Interestingly, both these studies have tables that indicate BMIs in patients with severe OA to be significantly higher as compared to their control, potentially indicating a disruption in their bone-fat balance. In HLCs, while there was no difference in CD106, the trend for CD295 stayed the same as in MSCs, it displayed a non-significant decline. The different cell source and measuring VCAM-1

209

released protein as opposed to transcript in this study, explain the opposite results as compared to the findings in this thesis.

Cx43 and CD146 did not show any differences in young and old donors (Section 4.3.4) and neither in MSCs from OA donors. They are expressed on uncultured MSCs and definitely play important functions, however, a larger cohort of donors must be used to understand the differences in the level of expression of Cx43 and CD146 in young and old donor MSCs as well as in OA. One study focussed on the protein interactions of Cx43 with other protein interactors in OA (504). They found over 100 proteins that were closely linked with Cx43. Gene ontology data from their OA donors suggested maximum interactors of Cx43 related to cell adhesion, nucleolus, calmodulin and cytoskeleton from chondrocytes in OA cartilage.

In conclusion, there was a significant decline in adipogenic genes in OA MSCs along with a decline in OPG indicating towards increased bone resorption. *IFNAR2* also declined significantly in OA MSCs which potentially led to the decline in several IRGs in OA. This indicated the role of IFNAR2 as a marker in OA. CD106 and CD295 declined significantly in BM MSCs suggesting a decline in immune-regulatory properties and loss of the bone-fat balance, suggesting these could be potential markers of OA. Senescence associated genes did not show any statistically significant differences but the age-related *Tp53* and *Sirt6* expression decline in HLCs was further aggravated in OA. Pro-inflammatory marker *IL6* displayed an age-related increase in both MSCs and in HLCs which was further significantly aggravated in OA. Interestingly, *IFNAR2, MTCH2* and *STING* which is the stimulator of IFN genes, both displayed an age-related decline followed by an exacerbated decline in OA HLCs. This indicates towards potential markers of age-related OA that need further exploration at protein level in a larger cohort of donors.

Chapter 7 Discussion

7.1 Key findings

The aim of this project was to investigate bone resident MSCs in healthy ageing and in OA by means of enumeration and gene expression analysis. In most parts, donormatched HLCs were used as a control population. A decline in the number and proliferative capacity of MSCs was found in healthy ageing. The genes selected to reflect MSC multifunctionality and selected surface markers did not display any significant differences in old donor MSCs compared to young donor MSCs, but surprisingly, some significant age-related differences, for example in the expression of RANKL, SPARC and LepR were observed in HLCs. MSCs and HLCs were investigated further for senescence, SASP and IFN1 pathway-related exploratory genes and once again, age-related differences were more evident in HLCs as compared to MSCs. Finally, investigations in OA suggested a significant decline in CXCL12, PPAR- γ and LepR expression in MSCs, which for LepR was confirmed at the surface protein level by flow cytometry. CD106 also showed a significant decline in MSCs from OA patients. Finally, a number of candidate genes (SPARC, KI, IL6, and IL8) were found to display age-related differences in HLCs that were aggravated in OA patients, and *IL6* showed the same trend in MSCs. These candidate genes should be further investigated as potential biomarkers of age-related OA and potentially targeted for future therapeutic approaches aimed at slowing down the progression of the disease.

7.2 Clinical relevance

This project showed a significant age-related decline in the number of BM MSCs measured by the classical CFU-F assay along with a significant decline in the proliferation potential of MSCs from older donors measured by colony area and integrated density (ID) analysis. To understand proliferative capacity of MSCs in relatively more 'natural' conditions, similar experiments were performed in human serum. Growing young donor MSCs in OS conditions resulted in significantly lower colony IDs as compared to when these MSCs were grown in YS conditions. The IDs of colonies from old donor MSCs were significantly higher when grown in YS when compared to OS. This suggested that old donor MSCs could be partially 'rejuvenated' in terms of their proliferation potential when grown in YS, highlighting the importance of cell extrinsic factors in the MSC *in vivo* niche. Thus, Chapter 3 results supported the SCE theory of ageing (section 1.1.1), however, a few of the old donors exhibited MSC numbers that were as good as young donors if not better. Overall, this indicated

towards a general decline in the number and proliferative capacities of MSCs in older population.

However, the presence of old donors with a relatively good number of BM MSCs as compared to the other donors in their cohort, indicated that chronological age is a shared characteristic but biological age is personal and varies from donor to donor. From the clinician point of view, these data suggest that a higher volume of BMA needs to be aspirated from old donors to get a similar 'dose' of MSCs as compared to young donors. And even after a higher 'dose' of MSCs, the MSCs may need further stimulation and exposure to younger microenvironment in order to be used as a successful therapeutic (for fractures and other cell therapies). Rapid quantification of the number of CD45^{low} CD271⁺ cells (111) from the BMA of the donors, MSC fitness and measurement in donor serum predicted by oxidation state (320) along with the level of pro-inflammatory cytokines like *IL6* could help assess the biological age of the donor. This will potentially indicate the biological age of the donor cells to help predict the 'dose' of MSCs needed for a given patient.

Using the CD45^{low}CD271⁺ phenotype to identify MSCs, this project showed a nonsignificant age-related decline in their numbers. However, this decline has been shown to become significant when data from more donors (n=67) were analysed (505). Adipogenic bias has been suggested as the most accepted feature of 'old' donor MSCs, even though some reports of no changes in their differentiation potential, do exist. While the exact mechanism for this proposed shift remains unknown, presence of ROS in old and damaged cells has often been suggested to be the cause. In this project, no differences in the expression of selected osteogenic- and adipogeniclineage transcripts were found between young and old donors MSCs. Only some trends in the adipogenic bias were visible in old donor MSCs. In support of with this finding, when ROS and the level of anti-oxidant enzyme SOD3 were measured in MSCs, no significant difference between young and old donors were found. This indicated that scientists must be mindful when reporting on 'age-related' differences in MSCs, considering '*in vitro*' ageing and '*in vivo*' ageing may be driven by different mechanisms.

Transcripts for bone remodelling and stromal functions also did not display any significant age-related differences in MSCs, even though both *RANKL* and *OPG* increased in old donor MSCs. Surface markers also demonstrated no changes in old donor MSCs or in HLCs. Interestingly, *RANKL* and *SPARC* both showed significant declines not in MSCs, but in HLCs indicating that age-related changes in BM MSCs *in*

vivo are probably more subtle in comparison to the age-related changes in HLCs. This finding provided the first evidence that MSCs could be more resistant to age-related damages *in vivo* in comparison to HLCs, even though they may be exposed to the same levels of local cellular and intracellular insults such as ROS and DNA damage.

Taking into account recent research suggesting potential links between DNA damage, senescence, ROS, IFN1 signalling and ageing in HSCs, the question if DNA damage, senescence, ROS and ageing could also be linked to cells of our interest, the MSCs, via IFN-signalling, formed the basis of next set of experiments in Chapter 5. This study is novel (to the best of my knowledge) to investigate a large number of genes on a broad donor age range on uncultured BM MSCs and HLCs using pure cell populations. This chapter revealed very interesting findings. Firstly, while senescence and SASP associated genes displayed trends towards increased damage in old donor MSCs, they did not reach statistical significance. Second, with respect to the expression of IFN1 and related genes in MSC, some of them unexpectedly displayed very high expression in MSCs as compared to HLCs. Next, some novel genes were found to display agerelated differences in HLCs. Interestingly, in this investigation too, only one out of 96 genes showed a significant decline in MSCs from older donors, further suggesting that age-related changes are relatively subtle in MSCs in vivo. However, genes associated with IFN1>BID>ROS pathway and senescence associated genes displayed significant changes in HLCs. This could be explained by the fact that MSCs are longer living cells and may have developed molecular mechanisms to combat ageing, which is not the case for HLCs that have shorter life-span. It will be interesting to investigate in the future if other long-living bone cells, such as osteocytes, have IFN1>BID>ROS signature similar to MSCs.

Finally, investigations in MSCs from OA donors revealed a significant decline in the expression of adipogenic transcription factor PPAR- γ and stromal derived cytokine *CXCL12* in MSCs. The surface marker CD106 also displayed a significant decline in OA donors as compared to healthy old donors and *LepR* showed a significant decline at both gene and protein levels. Many genes associated with IFN1>BID>ROS pathway, including *IFNR2* and *STAT1* were also down-regulated in OA MSCs. Again, OA HLCs demonstrated more profound OA-associated changes, some of which were in the same direction as noted for normal ageing in HLCs. *IL6* was identified as the only gene which displayed an aggravated age-related change in both cell populations in OA. Thus, this investigation essentially paves the way for identifying novel candidate biomarkers for age-related OA.

213

7.3 The 'bigger picture'

A number of genes investigated in this project have been shown by others to be associated with ageing/OA and/or other age-related diseases. For example, SPARC has been established as a key player in obesity and diabetes which are already been considered to be impacted by age (506). Makhluf and colleagues have previously shown an age-related decline in the expression of OPG in older OA donors, potentially connecting age with OA (507), however the study remained limited in terms of its patient cohort, given their youngest donor was 38 years old, when the decline in MSCs already take place, as shown in this thesis (Chapter 3) and the bone remodelling dynamics begin to shift (96). Another study on human femoral head BM investigated the levels of RANKL and OPG in OA patients and found an age-related decline in the levels of OPG. They also noticed variation in the levels of OPG in the young donors but the old donor cells consistently produced very low levels of OPG. The anti-ageing gene Sirt6 was shown to be significantly reduced in the articular chondrocytes of OA patients that displayed increased cellular senescence as compared to non OA patients (508). Higher levels of IL6 and IL8 was found to be secreted from the synovial of obese OA patients as compared to normal weight OA patients (478).

In this context, my PhD has shown a decline in the expression of *OPG*, *LepR*, *PPAR-* γ and *SPARC* in MSCs from OA patients as compared to MSCs from old donors, suggesting a decline in osteogenic as well and adipogenic differentiation. *IFNAR2* declined significantly while *IL6* was found to increase significantly indicative of higher levels of SASP associated with OA MSCs. With respect to age-related differences that were aggravated in OA, the pro-inflammatory cytokine *IL6* displayed a significant increase in MSCs and the anti-ageing genes of *Sirt6* and *KI* displayed a decline in HLCs.

There has been an enormous rise in the number of publications aimed at the different aspects of OA in the last two decades on the research front which indicates towards the increased needs, availability of funding and thirst for knowledge among the medical and scientific community. However, very few of these publications have been aimed at prevention of the condition, by acting on the aspects of an individual's life that can be controlled (diet, exercise, personal well-being). Healthcare systems globally should aim at early prevention of the disease for which early diagnosis is critical. This can be achieved by enhancing basic research towards early OA biomarkers to deliver more convenient diagnostic tools and better disease-modifying drugs. At the same time, it is equally important to spread awareness and educate people early on about the lifestyle choices that could prevent the disease altogether.

214

The lack of strong and statistically significant age-related changes in uncultured BM MSCs in this project, as compared to the vast majority of literature outlining significant age-related differences in terms of MSC proliferation potential, ROS production and senescence in cultured MSCs strongly indicate towards two critical issues in experimental approach. First, a stronger emphasis should be directed towards the fact that culture expanded MSCs used in the majority of previous studies undergo *in vitro* ageing which is not the same as the cell ageing *in vivo*. *In vitro* culture forces MSCs to adapt to artificial conditions and enhances proliferation, which has an impact on their differentiation, ROS production and signalling mechanisms combating senescence. Thus, the molecular mechanisms of MSC ageing proposed, based on cultured MSCs are more likely to be applicable only to culture expanded MSCs and do not explain the mechanisms occurring in MSCs *in vivo*.

Second, the concept of biological ageing as opposed to chronological ageing (both in vivo) needs to be better defined and considered in every research work that attempts to explore the complexity of ageing. Chronological age refers to the actual age of a donor while the biological age refers to the functionality of the organs (or cells) within the donor. Two or more people with the same chronological age can display difference in the 'fitness' of their organs indicating variation in their 'biological age'. Obese teenagers have been reported to have higher bone age than their peers of same chronological age when quantified using physical examinations (509). DW Belsky and co-workers found that adults of the same age (38 years old) were growing old at different rates described as different 'pace of ageing' quantified by using physical functioning, cognitive functioning and perceptions of well-being (510). They also suggested that investigation of age-related differences in younger (<40 years old) donors may be most useful to identify factors that can be quantified as 'early signs of ageing'. These findings bring our attention back to the various parts of thesis, for example large donorvariations in both 'young' and 'old' donor groups, confirming that the biological age of a person is different to their chronological age and may be determined by diet, exercise, lifestyle and well-being.

Biomedical research for ageing at the cellular level, often tends to investigate agerelated differences in phenotypes, genotypes, senescence and other cell functions using chronological ages and *in vitro* assays. With increasing evidence of the impact of extrinsic factors like nutrition, diet and lifestyle impacting an alarmingly large number of people globally, there is a dire need to shift the approach towards considering these factors while designing experiments. While eliminating donors to suit a study design may not be an option for all, having comprehensive knowledge of the donor's nutrition intake, diet, lifestyle and well-being for at least up to 6 months prior to investigation will help connect the dots better. This is further emphasised by the WHO defining health as the 'state of complete physical, mental and social wellbeing and not merely the absence of a disease' (1). A growing body of evidence shows that good physical and mental health are essential for complete well-being of a person (511, 512), especially in the old age (513). While these factors were not accounted for in this thesis, they could potentially explain the donor variations that were observed across the experiments performed in this project.

Increasingly, more and more online tests are easily available to help determine a person's 'actual age' (i.e. biological age), based on diet, exercise, smoking, drinking, social and mental well-being, in less than 5 minutes (514, 515). Interestingly, the methodologies of these online tests are unclear, there is no known standard or normalisation and they regularly provide very different 'actual age' of the same person who answers questions to different tests. I am personally aged 7, 22 and 38 years old in three different online tests! As more researchers focus on ageing studies and investigation of age-related diseases like OA, the aim must shift towards considering diet, lifestyle and well-being for defining biological age, as well as to consider various scenarios and donor variations in their experimental design to make a study more comprehensive. This will aid better understanding towards early OA diagnosis and eventually sustainable treatment solutions.

7.4 'Comprehensive' research

Even with the advancement of science, technology and our knowledge in healthcare and ageing, most of OA therapies are still aimed at alleviating the symptoms (pain relief for OA) rather than getting to the root of the problem, which is what causes the disease. Basic science plays a crucial role in studying cell behaviour, identifying the molecular pathways and mechanisms for a given condition. For research that may have clinical implications, it is vital that after optimising techniques, the research is encouraged to be performed on uncultured cells or using *in vitro* models (like 3D models using organ-on-a-chip or organoids) that can mimic the conditions *in vivo*. Acknowledgement of the fact that biomedical or healthcare research performed with human samples will inevitably display 'donor variation' in nearly every parameter being tested for any given experiment needs to be at the core of research investigating ageing and associated diseases like OA. Clinical trials may exclude a certain population from their study (for example, obese people) to suit their parameters better,
however, that might not be an option for other researchers that have access to a limited number of human samples.

Thus, having more information about the diet, exercise, lifestyle and well-being state of the donors over at least the last 6 months will help researchers to link various factors that link ageing and OA. Ageing is inevitable and an increase in age-related diseases and disorders make 'age' an important factor in every research. To understand a specific disease or disorder, it is usually approached with a finite focus on the topic. This approach is essential to narrow down and identify molecules, factors and targets for potential therapy for a single disease. When it comes to multi-factorial conditions like ageing and OA, the research plan and experimental design should include a number of variables and parameters to represent much larger donor populations.

For example, to investigate bone age and OA, sample collection should include not only bone samples but also matched sample of peripheral blood from the same donors. It would also include designing experiments to not only quantify a receptor (*LepR*) but also the hormone itself (Leptin) to better understand the signal transduction. The same should apply to IFN 1 type pathway mediators and SASP-associated molecules. To understand early signs of ageing and age-related differences, lifestyle data helping to define donor biological age would also provide essential information about potential causes (and potential targets) of age-related diseases. With multi-factorial conditions like ageing and OA a number of symptoms, parameters, genes and markers may overlap, increasing the complexity of the research work. Having more information will thus help to extrapolate markers or parameters that require further attention.

7.5 Study novelty and contribution to the field

The novelty of this project lies in the use of minimally-manipulated and uncultured human donor cells from a large number of donors and combination of a wide variety of genes. Biomedical research implementing the use of uncultured cells has strong implications and clinical relevance as it better reflects the behaviour and characteristics of cells *in vivo*. Over 50 donor samples were used for examining age-related changes in BM MSC numbers with minimum manipulation. Similar cohort-size investigations have been performed on cultured MSCs (97), but not on uncultured MSCs. This project was also the first of its kind to perform in depth investigation of over 100 genes in two uncultured and purified cell populations co-existing with the BM niche – MSCs and HLCs. Previous studies from our group have investigated 96 or 48 genes, but the cohort sizes were significantly smaller (8 donors (188) and 15 donors (516)), respectively. Also, age-related factors were not considered in these studies.

For the first time, this project evaluated colony sizes and densities to indicate MSC proliferation and heterogeneity in relation to donor age. Previous publications investigated different colony types based on sizes but not in relation to donor age (201, 208) and colony density has never been quantified before. Also for the first time, this project studied IFN1 signature in human MSCs, previous studies have only focussed on HSCs or immune-lineage cells (335, 391, 392). For the first time, this project has shown considerable IRG and IFN1 receptor expression in MSCs, opening up a completely new research area on the role of IFN1 signalling in non-immune cells and MSCs in particular. At this stage, it can be only speculated that MSCs may use IFN1 including their receptors and intracellular machinery to combat accumulating DNA damage and control senescence, future research is needed to further explore these possibilities.

Apart from directing research towards expanding the existing horizon of knowledge on ageing and age-related risks that may eventually lead to OA, this project has also suggested considering both, chronological age and biological age in biomedical research. This project also highlighted the links between general health, ageing and risk of age-related OA at the MSCs level aimed at understanding the roles of MSCs in these processes. Discussions in each chapter outline the need for comprehensive research that must be performed in future to be able to pinpoint to specific factors and targets to better understand age-related OA. Finally, this project also provides first initial data that MSCs may be more resistant to ageing and damaging insults *in vivo*, compared to HLCs, thus making them a good choice for developing autologous cell based therapies.

7.6 Drawbacks and study limitations

Working with a rare population of cells without being able to expand them in culture has certain limitations. The low number of retrieved cells, particularly MSCs from older donors, was always a point of concern with respect to experimental design. As established in the beginning of the thesis, to understand ageing *in vivo*, it is important that the cells being used are not artificially aged *in vitro*. Apart from that, the high degree of donor-to-donor variations was something unexpected but was observed across all the experiments during the course of this PhD. This meant that for many interesting findings, particularly in relation to gene expressions, the noted trends could not be confirmed statistically using the donor numbers available. However these pilot data provided important information on the magnitudes of differences and the variations within the groups, to enable power calculations for a more comprehensive

218

study. These variations, even with tightly controlled within age groups suggested that chronological age may or may not be represent the biological age of the person. Donor differences within the groups (young or old) could be explained by different donor's BMIs, diet and other lifestyle factors, such information was not available for this study. Due to ethical approvals available, this project did not include matched blood samples, which could have allowed to quantify and compare levels of transcripts in BM cells with the corresponding proteins (like Leptin) in the circulation. Overall, only a limited number of molecules were investigated at both, the transcript and protein level.

This project also did not perform any tests to evaluate and compare MSC differentiation or senescence status between young and old donors, which would be confirmatory to the presented gene expression data. Ideally, donor-matched comparisons could be done between *in vitro* cultured and *in vivo* MSCs. Looking back, quantification of genes and surface markers from the intermediate age group could have also been useful as well as having more information about the donors' about lifestyle, medication and general well-being. As discussed above, lifestyle and well-being are increasingly being associated with age-related diseases and thus, this kind of information would be extremely useful in terms of understanding donors' biological ages.

7.7 Future Directions

With the results obtained, it is clear that a larger cohort of patients is needed. However, with experiments performed in culture expanded MSCs, due to forced oxidative stress conditions, the results are often exaggerated. This can easily lead to scientists going ahead with animal studies and eventually to clinical trials, where over 50% of drugs have been known to fail (517, 518). It is ironical, that while this is the most exciting era for scientific research, biomedical and healthcare based technologies and the potential they hold for the future generation, it is also the era, where over 50% of all new drugs fail by the time they reach phase III clinical trials. It is also ironical, that even though, a human being lives the longest today we could have ever lived before, it also brings along with it the reduced quality of life after the age of 60 in most countries, globally. Not to forget, while there is a considerable increase in the access to different types of cuisines to people everywhere in the global era of networking and travel, 71% of the death every year is actually due to lifestyle disease which includes poor choices of diet and lifestyle (519).

Taking this project forward would include investigation of the genes that have been indicated as potential 'biomarkers' (IL6, anti-ageing genes and IFNA receptors) down to the protein level in larger number of donors including donors of ages across 20-90

(from early adulthood until old age). The last 6-months record about their diet (proportions of proteins, carbohydrates, fats, vitamins, minerals and water intake), lifestyle (drinking, smoking, exercises and medications) and well-being (questionnaires on personal motivation, family-friends-relationships and work life satisfaction) would be maintained. Further, classifying OA donors on the basis of the stage of the disease would be of critical importance. Comparing the levels of these indicative genes and proteins would aid our understanding of changes in these potential biomarkers with disease progression. Techniques like single cell RNA seq and the use of 3D models to mimic bone cells and the microenvironment (520) will further aid in investigations that are representative of changes *in vivo*.

References

1. World health organisation. 2015. World report on ageing and health. 2015 [online]. [Accessed 4 June 2018]. Available from

https://apps.who.int/iris/bitstream/handle/10665/186463/9789240694811_eng.p_df;jsessionid=C891A41C20CEDE7F6BD57B42981E43ED?sequence=1

2. Samarakoon SM, Chandola HM, Ravishankar B. Effect of dietary, social, and lifestyle determinants of accelerated aging and its common clinical presentation: A survey study. Ayu. 2011;32(3):315-21.

3. Karimi L, Mattace-Raso FU, van Rosmalen J, van Rooij F, Hofman A, Franco OH. Effects of combined healthy lifestyle factors on functional vascular aging: the Rotterdam Study. J Hypertens. 2016;34(5):853-9.

4. World health organisation. 2018. Ageing and health. [Online].[Accessed 5th May 2018]. Available from: <u>https://www.who.int/news-room/fact-sheets/detail/ageing-and-health</u>

5. United Nations.2017. World Population Prospects a 2017 Revision: Ageing population | Multimedia Library - United Nations Department of Economic and Social Affairs 2017.[Online]. [Accessed on 24May 2019]. Available from:

https://www.un.org/development/desa/publications/graphic/wpp2017-ageingpopulation.

6. UK Gofs.2019. Future of an ageing population. [Online].[Accessedon 29 July 2019]. Available from

https://assets.publishing.service.gov.uk/government/uploads/system/uploads/att achment_data/file/816458/future-of-an-ageing-population.pdf

7. UK A. Briefing: Health and Care of Older People in England 2017 | A Healthier Future. 2017 2017-02-20T13:49+00:00.

8. Statistics Of N. Overview of the UK population. 2017.

9. López-Otín C, Blasco MA, Partridge L, Serrano M, Kroemer G. The hallmarks of aging. Cell. 2013;153(6):1194-217.

10. Niccoli T, Partridge L. Ageing as a risk factor for disease. Curr Biol. 2012;22(17):R741-52.

11. Boyette LB, Tuan RS. Adult Stem Cells and Diseases of Aging. J Clin Med. 2014;3(1):88-134.

12. Sobhani A, Khanlarkhani N, Baazm M, Mohammadzadeh F, Najafi A, Mehdinejadiani S, et al. Multipotent Stem Cell and Current Application. Acta Med Iran. 2017;55(1):6-23.

13. Yu J, Vodyanik MA, Smuga-Otto K, Antosiewicz-Bourget J, Frane JL, Tian S, et al. Induced pluripotent stem cell lines derived from human somatic cells. Science. 2007;318(5858):1917-20.

14. Oh J, Lee YD, Wagers AJ. Stem cell aging: mechanisms, regulators and therapeutic opportunities. Nat Med. 2014;20(8):870-80.

15. Conboy IM, Conboy MJ, Wagers AJ, Girma ER, Weissman IL, Rando TA. Rejuvenation of aged progenitor cells by exposure to a young systemic environment. Nature. 2005;433(7027):760-4.

Conboy IM, Rando TA. Heterochronic parabiosis for the study of the effects of aging on stem cells and their niches. Cell Cycle. 2012;11(12):2260-7.
 Carrel A. CONCERNING VISCERAL ORGANISMS. J Exp Med. 1913;18(2):155-61.

18. Hoeijmakers JH. DNA damage, aging, and cancer. N Engl J Med. 2009;361(15):1475-85.

19. Jin K. Modern Biological Theories of Aging. Aging Dis. 2010;1(2):72-4.

Mizushima N, Levine B, Cuervo AM, Klionsky DJ. Autophagy fights disease through cellular self-digestion. Nature. 2008;451(7182):1069-75.
 Revuelta M, Matheu A. Autophagy in stem cell aging. Aging Cell. 2017;16(5):912-5.

22. Fumagalli M, Rossiello F, Clerici M, Barozzi S, Cittaro D, Kaplunov JM, et al. Telomeric DNA damage is irreparable and causes persistent DNA-damage-response activation. Nat Cell Biol. 2012;14(4):355-65.

23. Harman D. Aging: a theory based on free radical and radiation chemistry. 1955.

24. Halliwell B. Reactive oxygen species in living systems: source, biochemistry, and role in human disease. Am J Med. 1991;91(3c):14s-22s.

25. Li TS, Marban E. Physiological levels of reactive oxygen species are required to maintain genomic stability in stem cells. Stem Cells. 2010;28(7):1178-85.

26. Bigarella CL, Liang R, Ghaffari S. Stem cells and the impact of ROS signaling. Development. 2014;141(22):4206-18.

27. Li T, Zhou ZW, Ju Z, Wang ZQ. DNA Damage Response in Hematopoietic Stem Cell Ageing. Genomics Proteomics Bioinformatics. 2016;14(3):147-54.

28. Möhrle B, Geiger H. Aging of hematopoietic stem cells: DNA damage and mutations? Exp Hematol. 2016;44(10):895-901.

29. Katlinskaya YV, Carbone CJ, Yu Q, Fuchs SY. Type 1 interferons contribute to the clearance of senescent cell. Cancer Biol Ther. 2015;16(8):1214-9.

30. Yu Q, Katlinskaya YV, Carbone CJ, Zhao B, Katlinski KV, Zheng H, et al. DNA damage-induced type I interferon promotes senescence and inhibits stem cell function. Cell Rep. 2015;11(5):785-97.

 Gross A, Tasdogan A, Fehling HJ. The IFN-1 > BID > ROS pathway: Linking DNA damage with HSPC malfunction. Cell Cycle. 2017;16(9):819-20.
 Collado M, Blasco MA, Serrano M. Cellular senescence in cancer and aging. Cell. 2007;130(2):223-33.

33. Serrano M. Unraveling the links between cancer and aging. Carcinogenesis. 2016;37(2):107-.

34. North BJ, Sinclair DA. The intersection between aging and cardiovascular disease. Circ Res. 2012;110(8):1097-108.

35. Costantino S, Paneni F, Cosentino F. Ageing, metabolism and cardiovascular disease. J Physiol. 2016;594(8):2061-73.

36. Shakeri H, Lemmens K, Gevaert AB, De Meyer GRY, Segers VFM. Cellular senescence links aging and diabetes in cardiovascular disease. Am J Physiol Heart Circ Physiol. 2018;315(3):H448-h62.

37. Liu J, Mori A. Stress, aging, and brain oxidative damage. Neurochem Res. 1999;24(11):1479-97.

38. Lardenoije R, latrou A, Kenis G, Kompotis K, Steinbusch HW, Mastroeni D, et al. The epigenetics of aging and neurodegeneration. Prog Neurobiol. 2015;131:21-64.

39. Satoh A, Imai SI, Guarente L. The brain, sirtuins, and ageing. Nat Rev Neurosci. 2017;18(6):362-74.

40. Deschenes MR. Effects of aging on muscle fibre type and size. Sports Med. 2004;34(12):809-24.

41. Gossan N, Boot-Handford R, Meng QJ. Ageing and osteoarthritis: a circadian rhythm connection. Biogerontology. 2015;16(2):209-19.

42. Li YS, Xiao WF, Luo W. Cellular aging towards osteoarthritis. Mech Ageing Dev. 2017;162:80-4.

43. Zhavoronkov A, Bhullar B. Classifying aging as a disease in the context of ICD-11. Front Genet. 2015;6.

44. Ames BN, Shigenaga MK, Hagen TM. Oxidants, antioxidants, and the degenerative diseases of aging. Proc Natl Acad Sci U S A. 1993;90(17):7915-22.

45. Bulterijs S, Hull RS, Björk VCE, Roy AG. It is time to classify biological aging as a disease. Front Genet. 2015;6.

46. Allison BJ, Kaandorp JJ, Kane AD, Camm EJ, Lusby C, Cross CM, et al. Divergence of mechanistic pathways mediating cardiovascular aging and developmental programming of cardiovascular disease. Faseb j. 2016;30(5):1968-75.

47. WHO. Priority medicines for Europe and the world, Background paper 6.12. 2004.

48. Mansell JP, Collins C, Bailey AJ. Bone, not cartilage, should be the major focus in osteoarthritis. Nat Clin Pract Rheumatol. 2007;3(6):306-7.

49. Neogi T. Clinical significance of bone changes in osteoarthritis. Ther Adv Musculoskelet Dis. 2012;4(4):259-67.

50. Li G, Yin J, Gao J, Cheng TS, Pavlos NJ, Zhang C, et al. Subchondral bone in osteoarthritis: insight into risk factors and microstructural changes. Arthritis Res Ther. 2013;15(6):223.

51. Home - ClinicalTrials.gov 2019 [Available from: <u>https://clinicaltrials.gov/</u>.

52. Stock SR. The Mineral-Collagen Interface in Bone. Calcif Tissue Int. 2015;97(3):262-80.

53. Szade K, Gulati GS, Chan CKF, Kao KS, Miyanishi M, Marjon KD, et al. Where Hematopoietic Stem Cells Live: The Bone Marrow Niche. Antioxid Redox Signal. 2018;29(2):191-204.

54. Mikkola HK, Orkin SH. The journey of developing hematopoietic stem cells. Development. 2006;133(19):3733-44.

55. Andrejs Ivanovs SR, Lindsey Welch, Richard A. Anderson, Marc L. Turner, Alexander Medvinsky. Highly potent human hematopoietic stem cells first emerge in the intraembryonic aorta-gonad-mesonephros region. Journal of experimental medicine. 2019;208(12):2417-27.

56. Ivanovs A, Rybtsov S, Ng ES, Stanley EG, Elefanty AG, Medvinsky A. Human haematopoietic stem cell development: from the embryo to the dish. Development. 2017;144(13):2323-37.

57. Doulatov S, Notta F, Laurenti E, Dick JE. Hematopoiesis: a human perspective. Cell Stem Cell. 2012;10(2):120-36.

58. Jones EA, Giannoudis PV, Kouroupis D. Bone repair with skeletal stem cells: rationale, progress to date and clinical application. Ther Adv Musculoskelet Dis. 2016;8(3):57-71.

59. Heino TJ, Määttä J. Bone Marrow Niche: Role of Different Cells in Bone Metastasis. Current Molecular Biology Reports. 2018:80-7.

60. Sugiyama T, Nagasawa T. Bone Marrow Niches for Hematopoietic Stem Cells and Immune Cells. Inflamm Allergy Drug Targets. 2012;11(3):201-6.

61. Lerner UH. Bone remodeling in post-menopausal osteoporosis. J Dent Res. 2006;85(7):584-95.

62. Demontiero O, Vidal C, Duque G. Aging and bone loss: new insights for the clinician. Ther Adv Musculoskelet Dis. 2012;4(2):61-76.

63. Boyce BF, Xing L. The RANKL/RANK/OPG pathway. Curr Osteoporos Rep. 2007;5(3):98-104.

64. Rucci N. Molecular biology of bone remodelling. Clinical cases in mineral and bone metabolism. 2008;5(1):49-56.

65. Becerikli M, Jaurich H, Schira J, Schulte M, Dobele C, Wallner C, et al. Age-dependent alterations in osteoblast and osteoclast activity in human cancellous bone. J Cell Mol Med. 2017;21(11):2773-81.

66. Theuns PJMREBCZJAMADDHM. Evidence for a diminished maturation of preosteoblasts into osteoblasts during aging in rats: An ultrastructural analysis - Roholl - 1994 - Journal of Bone and Mineral Research - Wiley Online Library. Journal of bone and mineral research. 2019;9(355-366).

67. Knothe Tate ML, Adamson JR, Tami AE, Bauer TW. The osteocyte. Int J Biochem Cell Biol. 2004;36(1):1-8.

68. Komori T. Cell Death in Chondrocytes, Osteoblasts, and Osteocytes. Int J Mol Sci. 2016;17(12):1-17.

69. Ilas DC, Churchman SM, Baboolal T, Giannoudis PV, Aderinto J, McGonagle D, et al. The simultaneous analysis of mesenchymal stem cells and early osteocytes accumulation in osteoarthritic femoral head sclerotic bone. Rheumatology (Oxford). 2019.

70. Prideaux M, Findlay DM, Atkins GJ. Osteocytes: The master cells in bone remodelling. Curr Opin Pharmacol. 2016;28:24-30.

71. Martin TJ. Historically significant events in the discovery of RANK/RANKL/OPG. World J Orthop. 2013;4(4):186-97.

72. Vaes G. Cellular biology and biochemical mechanism of bone resorption. A review of recent developments on the formation, activation, and mode of action of osteoclasts. Clin Orthop Relat Res. 1988(231):239-71.

73. Jilka RL, Noble B, Weinstein RS. OSTEOCYTE APOPTOSIS. Bone. 2013;54(2):264-71.

74. Katsimbri P. The biology of normal bone remodelling. Eur J Cancer Care (Engl). 2017;26(6).

75. Ilas DC, Churchman SM, McGonagle D, Jones E. Targeting subchondral bone mesenchymal stem cell activities for intrinsic joint repair in osteoarthritis. Future Sci OA. 2017;3(4).

76. Jones G, Nguyen T, Sambrook P, Kelly PJ, Eisman JA. Progressive loss of bone in the femoral neck in elderly people: longitudinal findings from the Dubbo osteoporosis epidemiology study. BMJ. 1994;309(6956):691-5.

77. Daly RM, Rosengren BE, Alwis G, Ahlborg HG, Sernbo I, Karlsson MK. Gender specific age-related changes in bone density, muscle strength and functional performance in the elderly: a-10 year prospective population-based study. BMC Geriatr. 2013;13:71.

78. Khosla S. Pathogenesis of age-related bone loss in humans. J Gerontol A Biol Sci Med Sci. 2013;68(10):1226-35.

79. Alswat KA. Gender Disparities in Osteoporosis. J Clin Med Res. 2017;9(5):382-7.

80. Kim M, Kim C, Choi YS, Kim M, Park C, Suh Y. Age-related alterations in mesenchymal stem cells related to shift in differentiation from osteogenic to adipogenic potential: implication to age-associated bone diseases and defects. Mechanisms of ageing and development. 2012;133(5):215-25.

81. Bellantuono I, Aldahmash A, Kassem M. Aging of marrow stromal (skeletal) stem cells and their contribution to age-related bone loss. Biochimica et Biophysica Acta (BBA)-Molecular Basis of Disease. 2009;1792(4):364-70.

82. Dominici M, Le Blanc K, Mueller I, Slaper-Cortenbach I, Marini F, Krause D, et al. Minimal criteria for defining multipotent mesenchymal stromal cells. The

International Society for Cellular Therapy position statement. Cytotherapy. 2006;8(4):315-7.

83. Friedenstein A, Chailakhjan R, Lalykina K. The development of fibroblast colonies in monolayer cultures of guinea-pig bone marrow and spleen cells. Cell Proliferation. 1970;3(4):393-403.

84. Chaudhary D, Trivedi RN, Kathuria A, Goswami TK, Khandia R, Munjal A. In vitro And In vivo Immunomodulating Properties of Mesenchymal Stem Cells. Recent Pat Inflamm Allergy Drug Discov. 2018;12(1):59-68.

85. Wagner W, Wein F, Seckinger A, Frankhauser M, Wirkner U, Krause U, et al. Comparative characteristics of mesenchymal stem cells from human bone marrow, adipose tissue, and umbilical cord blood. Exp Hematol. 2005;33(11):1402-16.

86. Fickert S, Fiedler J, Brenner RE. Identification, quantification and isolation of mesenchymal progenitor cells from osteoarthritic synovium by fluorescence automated cell sorting. Osteoarthritis Cartilage. 2003;11(11):790-800.

87. Pittenger MF, Mackay AM, Beck SC, Jaiswal RK, Douglas R, Mosca JD, et al. Multilineage potential of adult human mesenchymal stem cells. Science. 1999;284(5411):143-7.

88. Solchaga LA, Penick KJ, Welter JF. Chondrogenic Differentiation of Bone Marrow-Derived Mesenchymal Stem Cells: Tips and Tricks. Methods Mol Biol. 2011;698:253-78.

89. Horwitz EM, Le Blanc K, Dominici M, Mueller I, Slaper-Cortenbach I, Marini FC, et al. Clarification of the nomenclature for MSC: The International Society for Cellular Therapy position statement. Cytotherapy. 2005;7(5):393-5.
90. Jones EA, English A, Henshaw K, Kinsey SE, Markham AF, Emery P, et al. Enumeration and phenotypic characterization of synovial fluid multipotential mesenchymal progenitor cells in inflammatory and degenerative arthritis. Arthritis Rheum. 2004;50(3):817-27.

91. Ishii M, Koike C, Igarashi A, Yamanaka K, Pan H, Higashi Y, et al. Molecular markers distinguish bone marrow mesenchymal stem cells from fibroblasts. Biochem Biophys Res Commun. 2005;332(1):297-303.

92. Chan NL, Bourdeau A, Vera S, Abdalla S, Gross M, Wong J, et al. Umbilical vein and placental vessels from newborns with hereditary haemorrhagic telangiectasia type 1 genotype are normal despite reduced expression of endoglin. Placenta. 2004;25(2-3):208-17.

93. Narravula S, Lennon PF, Mueller BU, Colgan SP. Regulation of endothelial CD73 by adenosine: paracrine pathway for enhanced endothelial barrier function. J Immunol. 2000;165(9):5262-8.

94. Boxall SA, Jones E. Markers for characterization of bone marrow multipotential stromal cells. Stem cells international. 2012;2012.

95. Calloni R, Cordero EA, Henriques JA, Bonatto D. Reviewing and updating the major molecular markers for stem cells. Stem Cells Dev. 2013;22(9):1455-76.

96. Stolzing A, Jones E, McGonagle D, Scutt A. Age-related changes in human bone marrow-derived mesenchymal stem cells: consequences for cell therapies. Mechanisms of ageing and development. 2008;129(3):163-73.
97. Siegel G, Kluba T, Hermanutz-Klein U, Bieback K, Northoff H, Schäfer R.

Phenotype, donor age and gender affect function of human bone marrowderived mesenchymal stromal cells. BMC medicine. 2013;11(1):1. 98. Bonab MM, Alimoghaddam K, Talebian F, Ghaffari SH, Ghavamzadeh A, Nikbin B. Aging of mesenchymal stem cell in vitro. BMC cell biology. 2006;7(1):1.

99. Wagner W, Horn P, Castoldi M, Diehlmann A, Bork S, Saffrich R, et al. Replicative senescence of mesenchymal stem cells: a continuous and organized process. PloS one. 2008;3(5):e2213.

100. Ganguly P, El-Jawhari JJ, Giannoudis PV, Burska AN, Ponchel F, Jones EA. Age-related Changes in Bone Marrow Mesenchymal Stromal Cells: A Potential Impact on Osteoporosis and Osteoarthritis Development. Cell Transplant. 2017;26(9):1520-9.

101. Schneider EL, Mitsui Y. The relationship between in vitro cellular aging and in vivo human age. Proc Natl Acad Sci U S A. 1976;73(10):3584-8.

102. Sobiesiak M, Sivasubramaniyan K, Hermann C, Tan C, Orgel M, Treml S, et al. The mesenchymal stem cell antigen MSCA-1 is identical to tissue non-specific alkaline phosphatase. Stem Cells Dev. 2010;19(5):669-77.

103. Simmons PJ, Torok-Storb B. Identification of stromal cell precursors in human bone marrow by a novel monoclonal antibody, STRO-1. Blood. 1991;78(1):55-62.

104. Jones E, Schäfer R. Where is the common ground between bone marrow mesenchymal stem/stromal cells from different donors and species? Stem Cell Research & Therapy. 2015;6(1):1-8.

105. Buhring HJ, Treml S, Cerabona F, de Zwart P, Kanz L, Sobiesiak M. Phenotypic characterization of distinct human bone marrow-derived MSC subsets. Ann N Y Acad Sci. 2009;1176:124-34.

106. Tan HB, Giannoudis PV, Boxall SA, McGonagle D, Jones E. The systemic influence of platelet-derived growth factors on bone marrow mesenchymal stem cells in fracture patients. BMC Med. 132015.

107. ROGERS M-L., BEARE A., H. Z, R.A R. CD 271 (P75 NEUROTROPHIN RECEPTOR). JOURNAL OF BIOLOGICAL REGULATORS & HOMEOSTATIC AGENTS. 2008;22:1-6.

108. Jones EA, Kinsey SE, English A, Jones RA, Straszynski L, Meredith DM, et al. Isolation and characterization of bone marrow multipotential mesenchymal progenitor cells. Arthritis & Rheumatism. 2002;46(12):3349-60.

109. Harichandan A, Buhring HJ. Prospective isolation of human MSC. Best Pract Res Clin Haematol. 2011;24(1):25-36.

110. Busser H, Najar M, Raicevic G, Pieters K, Velez Pombo R, Philippart P, et al. Isolation and Characterization of Human Mesenchymal Stromal Cell Subpopulations: Comparison of Bone Marrow and Adipose Tissue. Stem Cells Dev. 2015;24(18):2142-57.

111. El-Jawhari JJ, Cuthbert R, McGonagle D, Jones E, Giannoudis PV. The CD45lowCD271high Cell Prevalence in Bone Marrow Samples May Provide a Useful Measurement of the Bone Marrow Quality for Cartilage and Bone Regenerative Therapy. J Bone Joint Surg Am. 2017;99(15):1305-13.

112. Buhring HJ, Battula VL, Treml S, Schewe B, Kanz L, Vogel W. Novel markers for the prospective isolation of human MSC. Ann N Y Acad Sci. 2007;1106:262-71.

113. Oreffo RO, Bord S, Triffitt JT. Skeletal progenitor cells and ageing human populations. Clinical Science. 1998;94(5):549-55.

114. Baxter MA, Wynn RF, Jowitt SN, Wraith J, Fairbairn LJ, Bellantuono I. Study of telomere length reveals rapid aging of human marrow stromal cells following in vitro expansion. Stem cells. 2004;22(5):675-82.

115. Li Y, Charif N, Mainard D, Bensoussan D, Stoltz JF, de Isla N. Donor's age dependent proliferation decrease of human bone marrow mesenchymal stem cells is linked to diminished clonogenicity. Biomed Mater Eng. 2014;24(1 Suppl):47-52.

116. Wagner W, Bork S, Horn P, Krunic D, Walenda T, Diehlmann A, et al. Aging and replicative senescence have related effects on human stem and progenitor cells. PloS one. 2009;4(6):e5846.

117. Geissler S, Textor M, Schmidt-Bleek K, Klein O, Thiele M, Ellinghaus A, et al. In serum veritas-in serum sanitas? Cell non-autonomous aging compromises differentiation and survival of mesenchymal stromal cells via the oxidative stress pathway. Cell Death Dis. 2013;4:e970.

118. Peffers M, Collins J, Fang Y, Goljanek-Whysall K, Rushton M, Loughlin J, et al. Age-related changes in mesenchymal stem cells identified using a multi-omics approach. Eur Cell Mater. 2016;31:136-59.

119. Tencerova M, Kassem M. The Bone Marrow-Derived Stromal Cells: Commitment and Regulation of Adipogenesis. Front Endocrinol (Lausanne). 2016;7.

120. Tissenbaum HA. Using C. elegans for aging research. Invertebr Reprod Dev. 2015;59(sup1):59-63.

121. Mitchell SJ, Scheibye-Knudsen M, Longo DL, de Cabo R. Animal models of aging research: implications for human aging and age-related diseases. Annu Rev Anim Biosci. 2015;3:283-303.

122. Yang YM, Li P, Cui DC, Dang RJ, Zhang L, Wen N, et al. Effect of aged bone marrow microenvironment on mesenchymal stem cell migration. Age (Dordr). 2015;37(2):16.

123. Yang YK, Ogando CR, Wang See C, Chang TY, Barabino GA. Changes in phenotype and differentiation potential of human mesenchymal stem cells aging in vitro. Stem Cell Res Ther. 2018;9(1):131.

124. Halfon S, Abramov N, Grinblat B, Ginis I. Markers distinguishing mesenchymal stem cells from fibroblasts are downregulated with passaging. Stem Cells Dev. 2011;20(1):53-66.

125. Tormin A, Brune JC, Olsson E, Valcich J, Neuman U, Olofsson T, et al. Characterization of bone marrow-derived mesenchymal stromal cells (MSC) based on gene expression profiling of functionally defined MSC subsets. Cytotherapy. 2009;11(2):114-28.

126. Laschober GT, Brunauer R, Jamnig A, Fehrer C, Greiderer B, Lepperdinger G. Leptin receptor/CD295 is upregulated on primary human mesenchymal stem cells of advancing biological age and distinctly marks the subpopulation of dying cells. Experimental gerontology. 2009;44(1):57-62.

127. Mauney JR, Kaplan DL, Volloch V. Matrix-mediated retention of osteogenic differentiation potential by human adult bone marrow stromal cells during ex vivo expansion. Biomaterials. 2004;25(16):3233-43.

128. Wagner W, Bork S, Lepperdinger G, Joussen S, Ma N, Strunk D, et al. How to track cellular aging of mesenchymal stromal cells? Aging (Albany NY). 2010;2(4):224.

129. Stenderup K, Justesen J, Clausen C, Kassem M. Aging is associated with decreased maximal life span and accelerated senescence of bone marrow stromal cells. Bone. 2003;33(6):919-26.

130. Campisi J. Aging, cellular senescence, and cancer. Annu Rev Physiol. 2013;75:685-705.

131. Zhai W, Yong D, El-Jawhari JJ, Cuthbert R, McGonagle D, Win Naing M, et al. Identification of senescent cells in multipotent mesenchymal stromal cell cultures: current methods and future directions. Cytotherapy. 2019.

132. D'Ippolito G, Schiller PC, Ricordi C, Roos BA, Howard GA. Age-related osteogenic potential of mesenchymal stromal stem cells from human vertebral bone marrow. Journal of bone and mineral research. 1999;14(7):1115-22.

 Zhou S, Greenberger JS, Epperly MW, Goff JP, Adler C, LeBoff MS, et al. Age-related intrinsic changes in human bone-marrow-derived mesenchymal stem cells and their differentiation to osteoblasts. Aging cell. 2008;7(3):335-43.
 Ambrosi TH, Scialdone A, Graja A, Gohlke S, Jank AM, Bocian C, et al. Adipocyte Accumulation in the Bone Marrow during Obesity and Aging Impairs Stem Cell-Based Hematopoietic and Bone Regeneration. Cell Stem Cell. 2017;20(6):771-84.e6.

135. Mueller SM, Glowacki J. Age-related decline in the osteogenic potential of human bone marrow cells cultured in three-dimensional collagen sponges. Journal of cellular biochemistry. 2001;82(4):583-90.

136. Zaim M, Karaman S, Cetin G, Isik S. Donor age and long-term culture affect differentiation and proliferation of human bone marrow mesenchymal stem cells. Ann Hematol. 2012;91(8):1175-86.

137. Madsen SD, Russell KC, Tucker HA, Glowacki J, Bunnell BA, O'Connor KC. Decoy TRAIL receptor CD264: a cell surface marker of cellular aging for human bone marrow-derived mesenchymal stem cells. Stem Cell Res Ther. 2017;8(1):201.

138. Laschober GT, Brunauer R, Jamnig A, Singh S, Hafen U, Fehrer C, et al. Age-specific changes of mesenchymal stem cells are paralleled by upregulation of CD106 expression as a response to an inflammatory environment. Rejuvenation Res. 2011;14(2):119-31.

139. Block TJ, Marinkovic M, Tran ON, Gonzalez AO, Marshall A, Dean DD, et al. Restoring the quantity and quality of elderly human mesenchymal stem cells for autologous cell-based therapies. Stem Cell Res Ther. 2017;8(1):239.
140. Shen P, Yue Y, Zheng J, Park Y. Caenorhabditis elegans: A Convenient In Vivo Model for Assessing the Impact of Food Bioactive Compounds on Obesity, Aging, and Alzheimer's Disease. <u>https://doiorg/101146/annurev-food-030117-012709</u>. 2018.

141. Gorbunova V, Bozzella MJ, Seluanov A. Rodents for comparative aging studies: from mice to beavers. Age (Dordr). 2008;30(2-3):111-9.

142. Takahashi K, Tsuboyama T, Matsushita M, Kasai R, Okumura H, Yamamuro T, et al. Modification of strain-specific femoral bone density by bone marrow-derived factors administered neonatally: a study on the spontaneously osteoporotic mouse, SAMP6. Bone Miner. 1994;24(3):245-55.

143. Moerman EJ, Teng K, Lipschitz DA, Lecka-Czernik B. Aging activates adipogenic and suppresses osteogenic programs in mesenchymal marrow stroma/stem cells: the role of PPAR-gamma2 transcription factor and TGF-beta/BMP signaling pathways. Aging Cell. 2004;3(6):379-89.

144. Stolzing A, Scutt A. Age-related impairment of mesenchymal progenitor cell function. Aging cell. 2006;5(3):213-24.

145. Katsara O, Mahaira LG, Iliopoulou EG, Moustaki A, Antsaklis A, Loutradis D, et al. Effects of donor age, gender, and in vitro cellular aging on the phenotypic, functional, and molecular characteristics of mouse bone marrow-derived mesenchymal stem cells. Stem Cells Dev. 2011;20(9):1549-61.

146. Josephson AM, Bradaschia-Correa V, Lee S, Leclerc K, Patel KS, Muinos Lopez E, et al. Age-related inflammation triggers skeletal

stem/progenitor cell dysfunction. Proc Natl Acad Sci U S A. 2019;116(14):6995-7004.

147. Asumda FZ, Chase PB. Age-related changes in rat bone-marrow mesenchymal stem cell plasticity. BMC Cell Biol. 2011;12:44.

148. Plotkin LI, Bellido T. Beyond gap junctions: Connexin43 and bone cell signaling. Bone. 2013;52(1):157-66.

149. Geißler S, Textor M, Kühnisch J, Könnig D, Klein O, Ode A, et al. Functional comparison of chronological and in vitro aging: differential role of the cytoskeleton and mitochondria in mesenchymal stromal cells. PLoS One. 2012;7(12):e52700.

150. Zhang Y, Jordan JM. Epidemiology of Osteoarthritis. Clin Geriatr Med. 2010;26(3):355-69.

151. Bortoluzzi A, Furini F, Scire CA. Osteoarthritis and its management -Epidemiology, nutritional aspects and environmental factors. Autoimmun Rev. 2018;17(11):1097-104.

152. DeFrances CJ, Lucas CA, Buie VC, Golosinskiy A. 2006 National Hospital Discharge Survey. Natl Health Stat Report. 2008(5):1-20.

153. Chen A, Gupte C, Akhtar K, Smith P, Cobb J. The Global Economic Cost of Osteoarthritis: How the UK Compares. Arthritis. 2012;2012:698709.

154. Sharma A, Kudesia P, Shi Q, Gandhi R. Anxiety and depression in patients with osteoarthritis: impact and management challenges. Open Access Rheumatol. 2016;8:103-13.

155. Im GI. Tissue Engineering in Osteoarthritis: Current Status and Prospect of Mesenchymal Stem Cell Therapy. BioDrugs. 2018;32(3):183-92.

156. Giannoudis PV, Einhorn TA, Schmidmaier G, Marsh D. The diamond concept--open questions. Injury. 2008;39 Suppl 2:S5-8.

157. Diekman BO, Guilak F. Stem cell-based therapies for osteoarthritis: Challenges and opportunities. Curr Opin Rheumatol. 2013;25(1):119-26.

158. Neogi T, Zhang Y. Epidemiology of osteoarthritis. Rheum Dis Clin North Am. 2013;39(1):1-19.

159. Hayashi D, Roemer FW, Guermazi A. Imaging for osteoarthritis. Ann Phys Rehabil Med. 2016;59(3):161-9.

160. Loeser RF, Goldring SR, Scanzello CR, Goldring MB. Osteoarthritis: A Disease of the Joint as an Organ. Arthritis Rheum. 2012;64(6):1697-707.

161. Quintana JM, Arostegui I, Escobar A, Azkarate J, Goenaga JI, Lafuente I. Prevalence of knee and hip osteoarthritis and the appropriateness of joint replacement in an older population. Arch Intern Med. 2008;168(14):1576-84.
162. King LK, March L, Anandacoomarasamy A. Obesity & osteoarthritis. Indian J Med Res. 2013;138(2):185-93.

163. Kulkarni K, Karssiens T, Kumar V, Pandit H. Obesity and osteoarthritis. Maturitas. 2016;89:22-8.

164. Spector TD, MacGregor AJ. Risk factors for osteoarthritis: genetics. Osteoarthritis Cartilage. 2004;12 Suppl A:S39-44.

165. Buckwalter JA, Anderson DD, Brown TD, Tochigi Y, Martin JA. The Roles of Mechanical Stresses in the Pathogenesis of Osteoarthritis: Implications for Treatment of Joint Injuries. Cartilage. 2013;4(4):286-94.

166. Loeser RF, Collins JA, Diekman BO. Ageing and the pathogenesis of osteoarthritis. Nature Reviews Rheumatology. 2016.

167. Anderson AS, Loeser RF. Why is Osteoarthritis an Age-Related Disease? Best Pract Res Clin Rheumatol. 2010;24(1):15.

168. Loeser RF. The Role of Aging in the Development of Osteoarthritis. Trans Am Clin Climatol Assoc. 2017;128:44-54.

169. Toh WS, Brittberg M, Farr J, Foldager CB, Gomoll AH, Hui JHP, et al. Cellular senescence in aging and osteoarthritis: Implications for cartilage repair. Acta Orthop. 872016. p. 6-14.

170. Palo N, Chandel SS, Dash SK, Arora G, Kumar M, Biswal MR. Effects of Osteoarthritis on Quality of life in Elderly Population of Bhubaneswar, India: A Prospective Multicenter Screening and Therapeutic Study of 2854 Patients. Geriatr Orthop Surg Rehabil. 62015. p. 269-75.

171. Chen Y, Hu Y, Yu YE, Zhang X, Watts T, Zhou B, et al. Subchondral Trabecular Rod Loss and Plate Thickening in the Development of Osteoarthritis. J Bone Miner Res. 2018;33(2):316-27.

172. Aigner T, Rose J, Martin J, Buckwalter J. Aging theories of primary osteoarthritis: from epidemiology to molecular biology. Rejuvenation Res. 2004;7(2):134-45.

173. van Laar M, Pergolizzi JV, Mellinghoff HU, Merchante IM, Nalamachu S, O'Brien J, et al. Pain Treatment in Arthritis-Related Pain: Beyond NSAIDs. Open Rheumatol J. 62012. p. 320-30.

174. Rannou F, Poiraudeau S. Non-pharmacological approaches for the treatment of osteoarthritis. Best Pract Res Clin Rheumatol. 2010;24(1):93-106.
175. Bliddal H, Leeds AR, Christensen R. Osteoarthritis, obesity and weight loss: evidence, hypotheses and horizons – a scoping review. Obes Rev. 2014;15(7):578-86.

176. Page CJ, Hinman RS, Bennell KL. Physiotherapy management of knee osteoarthritis. Int J Rheum Dis. 2011;14(2):145-51.

177. Roos EM, Arden NK. Strategies for the prevention of knee osteoarthritis. Nature Reviews Rheumatology. 2015;12(2):92-101.

178. NHS. Hip replacement: @nhsuk; 24 O [updated 24 Oct 201. Available from: <u>https://www.nhs.uk/conditions/hip-replacement/</u>.

179. Jazwinski SM, Kim S. Examination of the Dimensions of Biological Age. Front Genet. 2019;10.

180. Sanders JO, Qiu X, Lu X, Duren DL, Liu RW, Dang D, et al. The Uniform Pattern of Growth and Skeletal Maturation during the Human Adolescent Growth Spurt. Sci Rep. 2017;7.

181. Curtis EM, van der Velde R, Moon RJ, van den Bergh JPW, Geusens P, de Vries F, et al. Epidemiology of Fractures in the United Kingdom 1988-2012: Variation with age, sex, geography, ethnicity and socioeconomic status. Bone. 2016;87:19-26.

182. Jones E, English A, Churchman S, Kouroupis D, Boxall S, Kinsey S, et al. Large-scale extraction and characterization of CD271+ multipotential stromal cells from trabecular bone in health and osteoarthritis: Implications for bone regeneration strategies based on uncultured or minimally cultured multipotential stromal cells. Arthritis & Rheumatism. 2010;62(7):1944-54.

183. Campbell TM, Churchman SM, Gomez A, McGonagle D, Conaghan PG, Ponchel F, et al. Mesenchymal stem cell alterations in bone marrow lesions in hip osteoarthritis. Arthritis & Rheumatology. 2016.

184. Goetz JBH. Flow Cytometry: Definition, History, and Uses in Biological Research | SpringerLink. Flow cytometry basics for the non-expert: SpringerLink; 2018.

185. Adan A, Alizada G, Kiraz Y, Baran Y, Nalbant A. Flow cytometry: basic principles and applications. Crit Rev Biotechnol. 2017;37(2):163-76.

186. Rowley T. Flow Cytometry - A Survey and the Basics. Materials and Methods. 2012.

187. Ormerod MG. Fluorescence & Fluorochromes | Flow Cytometry - A Basic Introduction. Flow Cytometry - A Basic Introduction,2008.

188. Churchman SM, Ponchel F, Boxall SA, Cuthbert R, Kouroupis D, Roshdy T, et al. Transcriptional profile of native CD271+ multipotential stromal cells: evidence for multiple fates, with prominent osteogenic and Wnt pathway signaling activity. Arthritis & Rheumatism. 2012;64(8):2632-43.

189. Biotek N. Single cell RNA purification kit, Product insert. 2016.

190. Ong SE, Zhang S, Du H, Fu Y. Fundamental principles and applications of microfluidic systems. Front Biosci. 2008;13:2757-73.

191. Devonshire AS, Elaswarapu R, Foy CA. Applicability of RNA standards for evaluating RT-qPCR assays and platforms. BMC Genomics. 2011;12(1):1-10.

192. Andersson D, Akrap N, Svec D, Godfrey TE, Kubista M, Landberg G, et al. Properties of targeted preamplification in DNA and cDNA quantification. Expert Rev Mol Diagn. 2015;15(8):1085-100.

193. Korenkova V, Scott J, Novosadova V, Jindrichova M, Langerova L, Svec D, et al. Pre-amplification in the context of high-throughput qPCR gene expression experiment. BMC Mol Biol. 2015;16:5.

194. Friedenstein AJ, Piatetzky S, II, Petrakova KV. Osteogenesis in transplants of bone marrow cells. J Embryol Exp Morphol. 1966;16(3):381-90. 195. Gronthos S, Zannettino AC, Hay SJ, Shi S, Graves SE, Kortesidis A, et al. Molecular and cellular characterisation of highly purified stromal stem cells derived from human bone marrow. J Cell Sci. 2003;116(Pt 9):1827-35.

196. Justesen J, Stenderup K, Eriksen EF, Kassem M. Maintenance of Osteoblastic and Adipocytic Differentiation Potential with Age and Osteoporosis in Human Marrow Stromal Cell Cultures. Calcified Tissue International. 2002;71(1):36-44.

197. Stenderup K, Justesen J, Eriksen EF, Rattan SI, Kassem M. Number and proliferative capacity of osteogenic stem cells are maintained during aging and in patients with osteoporosis. J Bone Miner Res. 2001;16(6):1120-9.

198. Kuznetsov SA, Mankani MH, Bianco P, Robey PG. Enumeration of the colony-forming units-fibroblast from mouse and human bone marrow in normal and pathological conditions. Stem Cell Res. 2009;2(1):83-94.

199. Li J, Wong WH, Chan S, Chim JC, Cheung KM, Lee TL, et al. Factors affecting mesenchymal stromal cells yield from bone marrow aspiration. Chin J Cancer Res. 2011;23(1):43-8.

200. Oreffo RO, Bennett A, Carr AJ, Triffitt JT. Patients with primary osteoarthritis show no change with ageing in the number of osteogenic precursors. Scand J Rheumatol. 1998;27(6):415-24.

201. Gothard D, Dawson J, Oreffo R. Assessing the potential of colony morphology for dissecting the CFU-F population from human bone marrow stromal cells. Cell and tissue research. 2013;352(2):237-47.

202. Stenderup K, Rosada C, Justesen J, Al-Soubky T, Dagnaes-Hansen F, Kassem M. Aged human bone marrow stromal cells maintaining bone forming capacity in vivo evaluated using an improved method of visualization. Biogerontology. 2004;5(2):107-18.

203. Abdallah BM, Haack-Sørensen M, Fink T, Kassem M. Inhibition of osteoblast differentiation but not adipocyte differentiation of mesenchymal stem cells by sera obtained from aged females. Bone. 2006;39(1):181-8.

204. Di Bernardo G, Messina G, Capasso S, Del Gaudio S, Cipollaro M, Peluso G, et al. Sera of overweight people promote in vitro adipocyte differentiation of bone marrow stromal cells. Stem Cell Res Ther. 2014;5(1):4. 205. Stute N, Holtz K, Bubenheim M, Lange C, Blake F, Zander AR. Autologous serum for isolation and expansion of human mesenchymal stem cells for clinical use. Experimental hematology. 2004;32(12):1212-25.

206. Bieback K, Hecker A, Kocaömer A, Lannert H, Schallmoser K, Strunk D, et al. Human alternatives to fetal bovine serum for the expansion of

mesenchymal stromal cells from bone marrow. Stem cells. 2009;27(9):2331-41. 207. Cuthbert R, Boxall SA, Tan HB, Giannoudis PV, McGonagle D, Jones E. Single-platform quality control assay to quantify multipotential stromal cells in bone marrow aspirates prior to bulk manufacture or direct therapeutic use. Cytotherapy. 2012;14(4):431-40.

208. Zorin V, Zorina A, Smetanina N, Kopnin P, Ozerov IV, Leonov S, et al. Diffuse colonies of human skin fibroblasts in relation to cellular senescence and proliferation. Aging (Albany NY). 2017;9(5):1404-13.

209. dos Santos VTM, Mizukami A, Orellana MD, Caruso SR, da Silva FB, Traina F, et al. Characterization of Human AB Serum for Mesenchymal Stromal Cell Expansion. Transfus Med Hemother. 442017. p. 11-21.

210. Muschler GF, Nitto H, Boehm CA, Easley KA. Age-and gender-related changes in the cellularity of human bone marrow and the prevalence of

osteoblastic progenitors. Journal of Orthopaedic Research. 2001;19(1):117-25. 211. Mahmood S Choudhery MB, Angela Muise, John Pierce and David T Harris. Donor age negatively impacts adipose tissue-derived mesenchymal stem cell expansion and differentiation. Journal of translational medicine. 2014;12(8):1-14.

212. Sethe S, Scutt A, Stolzing A. Aging of mesenchymal stem cells. Ageing research reviews. 2006;5(1):91-116.

213. Saldias MP, Fernandez C, Morgan A, Diaz C, Morales D, Jana F, et al. Aged blood factors decrease cellular responses associated with delayed gingival wound repair. PLoS One. 2017;12(9):e0184189.

Wei J, Xu H, Davies JL, Hemmings GP. Increase of plasma IL-6 concentration with age in healthy subjects. Life Sci. 1992;51(25):1953-6.
Ferrucci L, Corsi A, Lauretani F, Bandinelli S, Bartali B, Taub DD, et al. The origins of age-related proinflammatory state. Blood. 2005;105(6):2294-9.
Ghosh K, Capell BC. The Senescence-Associated Secretory Phenotype: Critical Effector in Skin Cancer and Aging. J Invest Dermatol. 2016;136(11):2133-9.

217. Lohmann M, Walenda G, Hemeda H, Joussen S, Drescher W, Jockenhoevel S, et al. Donor age of human platelet lysate affects proliferation and differentiation of mesenchymal stem cells. PLoS One. 2012;7(5):e37839.
218. Griffiths S, Baraniak PR, Copland IB, Nerem RM, McDevitt TC. Human platelet lysate stimulates high-passage and senescent human multipotent mesenchymal stromal cell growth and rejuvenation in vitro. Cytotherapy. 2013;15(12):1469-83.

219. Guo P, Huang J, Moses MA. Characterization of dormant and active human cancer cells by quantitative phase imaging. Cytometry A. 2017;91(5):424-32.

220. Sun Y, Li W, Lu Z, Chen R, Ling J, Ran Q, et al. Rescuing replication and osteogenesis of aged mesenchymal stem cells by exposure to a young extracellular matrix. The FASEB Journal. 2011;25(5):1474-85.

221. Grezella C, Fernandez-Rebollo E, Franzen J, Ferreira MSV, Beier F, Wagner W. Effects of senolytic drugs on human mesenchymal stromal cells. Stem Cell Research & Therapy. 2018;9(1):108.

222. Churchman SM, Boxall SA, McGonagle D, Jones EA. Predicting the Remaining Lifespan and Cultivation-Related Loss of Osteogenic Capacity of Bone Marrow Multipotential Stromal Cells Applicable across a Broad Donor Age Range. Stem Cells Int. 2017;2017:6129596.

223. Kawamura H, Nakatsuka R, Matsuoka Y, Sumide K, Fujioka T, Asano H, et al. TGF-beta Signaling Accelerates Senescence of Human Bone-Derived CD271 and SSEA-4 Double-Positive Mesenchymal Stromal Cells. Stem Cell Reports. 2018;10(3):920-32.

224. Estève D, Galitzky J, Bouloumié A, Fonta C, Buchet R, Magne D. Multiple Functions of MSCA-1/TNAP in Adult Mesenchymal Progenitor/Stromal Cells. Stem Cells Int. 2016;2016.

225. Li H, Ghazanfari R, Zacharaki D, Ditzel N, Isern J, Ekblom M, et al. Low/negative expression of PDGFR-alpha identifies the candidate primary mesenchymal stromal cells in adult human bone marrow. Stem Cell Reports. 2014;3(6):965-74.

226. Wang S, Mo M, Wang J, Sadia S, Shi B, Fu X, et al. Platelet-derived growth factor receptor beta identifies mesenchymal stem cells with enhanced engraftment to tissue injury and pro-angiogenic property. Cell Mol Life Sci. 2018;75(3):547-61.

227. Tormin A, Li O, Brune JC, Walsh S, Schütz B, Ehinger M, et al. CD146 expression on primary nonhematopoietic bone marrow stem cells is correlated with in situ localization. Blood. 2011;117(19):5067-77.

228. Zhou BO, Yue R, Murphy MM, Peyer JG, Morrison SJ. Leptin-receptorexpressing mesenchymal stromal cells represent the main source of bone formed by adult bone marrow. Cell Stem Cell. 2014;15(2):154-68.

229. Maijenburg MW, Kleijer M, Vermeul K, Mul EP, van Alphen FP, van der Schoot CE, et al. The composition of the mesenchymal stromal cell compartment in human bone marrow changes during development and aging. Haematologica. 2012;97(2):179-83.

230. Quirici N, Soligo D, Bossolasco P, Servida F, Lumini C, Deliliers GL. Isolation of bone marrow mesenchymal stem cells by anti-nerve growth factor receptor antibodies. Exp Hematol. 2002;30(7):783-91.

231. Kuci S, Kuci Z, Kreyenberg H, Deak E, Putsch K, Huenecke S, et al. CD271 antigen defines a subset of multipotent stromal cells with immunosuppressive and lymphohematopoietic engraftment-promoting properties. Haematologica. 2010;95(4):651-9.

232. Venkata Lokesh Battula ST, Petra M. Bareiss, Friederike Gieseke, Helene Roelofs, Peter de Zwart, Ingo Müller, Bernhard Schewe, Thomas Skutella, Willem E. Fibbe, Lothar Kanz, Hans-Jörg Bühring. Isolation Of Functionally Distinct Mesenchymal Stem Cell Subsets Using Antibodies Against CD56, CD271, And Mesenchymal Stem Cell Antigen-1. Haematologica. 2009;94(2):12.

233. Ghazanfari R, Li H, Zacharaki D, Lim HC, Scheding S. Human Nonhematopoietic CD271pos/CD140alow/neg Bone Marrow Stroma Cells Fulfill Stringent Stem Cell Criteria in Serial Transplantations. Stem Cells and Development. 2016.

234. Cuthbert RJ, Giannoudis PV, Wang XN, Nicholson L, Pawson D, Lubenko A, et al. Examining the feasibility of clinical grade CD271+ enrichment of mesenchymal stromal cells for bone regeneration. PloS one. 2015;10(3):e0117855. 235. Mohamed-Ahmed S, Fristad I, Lie SA, Suliman S, Mustafa K, Vindenes H, et al. Adipose-derived and bone marrow mesenchymal stem cells: a donormatched comparison. Stem Cell Res Ther. 2018;9(1):168.

236. Kim M, Erickson IE, Huang AH, Garrity ST, Mauck RL, Steinberg DR. Donor Variation and Optimization of Human Mesenchymal Stem Cell Chondrogenesis in Hyaluronic Acid. Tissue Eng Part A. 2018;24(21-22):1693-703.

237. Brusnahan SK, McGuire TR, Jackson JD, Lane JT, Garvin KL, O'Kane BJ, et al. Human blood and marrow side population stem cell and Stro-1 positive bone marrow stromal cell numbers decline with age, with an increase in quality of surviving stem cells: correlation with cytokines. Mech Ageing Dev. 2010;131(11-12):718-22.

238. Takeshita S, Fumoto T, Naoe Y, Ikeda K. Age-related marrow adipogenesis is linked to increased expression of RANKL. J Biol Chem. 2014;289(24):16699-710.

239. Sugiyama T, Kohara H, Noda M, Nagasawa T. Maintenance of the hematopoietic stem cell pool by CXCL12-CXCR4 chemokine signaling in bone marrow stromal cell niches. Immunity. 2006;25(6):977-88.

240. Greenbaum A, Hsu YM, Day RB, Schuettpelz LG, Christopher MJ, Borgerding JN, et al. CXCL12 in early mesenchymal progenitors is required for haematopoietic stem-cell maintenance. Nature. 2013;495(7440):227-30.

241. Liu X, Duan B, Cheng Z, Jia X, Mao L, Fu H, et al. SDF-1/CXCR4 axis modulates bone marrow mesenchymal stem cell apoptosis, migration and cytokine secretion. Protein Cell. 2011;2(10):845-54.

242. Schajnovitz A, Itkin T, D'Uva G, Kalinkovich A, Golan K, Ludin A, et al. CXCL12 secretion by bone marrow stromal cells is dependent on cell contact and mediated by connexin-43 and connexin-45 gap junctions. Nat Immunol. 2011;12(5):391-8.

243. Del Monte U, Statuto M. Drop of connexins: a possible link between aging and cancer? Exp Gerontol. 2004;39(2):273-5.

244. Bethel M, Chitteti BR, Srour EF, Kacena MA. The changing balance between osteoblastogenesis and adipogenesis in aging and its impact on hematopoiesis. Current osteoporosis reports. 2013;11(2):99-106.

245. Geoffroy V, Kneissel M, Fournier B, Boyde A, Matthias P. High bone resorption in adult aging transgenic mice overexpressing cbfa1/runx2 in cells of the osteoblastic lineage. Mol Cell Biol. 2002;22(17):6222-33.

246. Fickert S, Schröter-Bobsin U, Gross AF, Hempel U, Wojciechowski C, Rentsch C, et al. Human mesenchymal stem cell proliferation and osteogenic differentiation during long-term ex vivo cultivation is not age dependent. J Bone Miner Metab. 2011;29(2):224-35.

247. Prall WC, Saller MM, Scheumaier A, Tucholski T, Taha S, Bocker W, et al. Proliferative and osteogenic differentiation capacity of mesenchymal stromal cells: Influence of harvesting site and donor age. Injury. 2018;49(8):1504-12. 248. Prager P, Kunz M, Ebert R, Klein-Hitpass L, Sieker J, Barthel T, et al.

Mesenchymal Stem Cells Isolated from the Anterior Cruciate Ligament: Characterization and Comparison of Cells from Young and Old Donors. Knee Surg Relat Res. 2018;30(3):193-205.

249. Chung PL, Zhou S, Eslami B, Shen L, LeBoff MS, Glowacki J. Effect of Age on Regulation of Human Osteoclast Differentiation. J Cell Biochem. 2014;115(8):1412-9.

250. Jiang Y, Mishima H, Sakai S, Liu YK, Ohyabu Y, Uemura T. Gene expression analysis of major lineage-defining factors in human bone marrow

cells: effect of aging, gender, and age-related disorders. J Orthop Res. 2008;26(7):910-7.

251. Rosset EM, Bradshaw AD. SPARC/Osteonectin in Mineralized Tissue. Matrix Biol. 2016;52-54:78-87.

252. Schwarzbauer JE, Spencer CS. The Caenorhabditis elegans homologue of the extracellular calcium binding protein SPARC/osteonectin affects nematode body morphology and mobility. Mol Biol Cell. 1993;4(9):941-52.

253. Purcell L, Gruia-Gray J, Scanga S, Ringuette M. Developmental anomalies of Xenopus embryos following microinjection of SPARC antibodies. J Exp Zool. 1993;265(2):153-64.

254. Kim JH, Liu X, Wang J, Chen X, Zhang H, Kim SH, et al. Wnt signaling in bone formation and its therapeutic potential for bone diseases. Ther Adv Musculoskelet Dis. 2013;5(1):13-31.

255. Bodine PV, Zhao W, Kharode YP, Bex FJ, Lambert AJ, Goad MB, et al. The Wnt antagonist secreted frizzled-related protein-1 is a negative regulator of trabecular bone formation in adult mice. Mol Endocrinol. 2004;18(5):1222-37. 256. Hausler KD, Horwood NJ, Chuman Y, Fisher JL, Ellis J, Martin TJ, et al. Secreted frizzled-related protein-1 inhibits RANKL-dependent osteoclast formation. J Bone Miner Res. 2004;19(11):1873-81.

257. Bodine PV, Billiard J, Moran RA, Ponce-de-Leon H, McLarney S, Mangine A, et al. The Wnt antagonist secreted frizzled-related protein-1 controls osteoblast and osteocyte apoptosis. J Cell Biochem. 2005;96(6):1212-30.
258. Bodine PV, Stauffer B, Ponce-de-Leon H, Bhat RA, Mangine A, Seestaller-Wehr LM, et al. A small molecule inhibitor of the Wnt antagonist secreted frizzled-related protein-1 stimulates bone formation. Bone.
2009;44(6):1063-8.

259. Rauner M, Sipos W, Pietschmann P. Age-dependent Wnt gene expression in bone and during the course of osteoblast differentiation. Age (Dordr). 2008;30(4):273-82.

260. Lakshman Singh TAB, Elizabeth Russell , Jung-Hoon Kim , Qijun Chen, F. Brad Johnson , Robert J. Pignolo. Aging alters bone-fat reciprocity by shifting in vivo mesenchymal precursor cell fate towards an adipogenic lineage. Bone. 2016;85:8.

261. Qian SW, Li X, Zhang YY, Huang HY, Liu Y, Sun X, et al. Characterization of adipocyte differentiation from human mesenchymal stem cells in bone marrow. BMC Dev Biol. 2010;10:47.

262. Yue R, Zhou BO, Shimada IS, Zhao Z, Morrison SJ. Leptin Receptor Promotes Adipogenesis and Reduces Osteogenesis by Regulating Mesenchymal Stromal Cells in Adult Bone Marrow. Cell Stem Cell. 2016;18(6):782-96.

263. Guerra B, Santana A, Fuentes T, Delgado-Guerra S, Cabrera-Socorro A, Dorado C, et al. Leptin receptors in human skeletal muscle. J Appl Physiol (1985). 2007;102(5):1786-92.

264. Lehrke M, Lazar MA. The many faces of PPARgamma. Cell. 2005;123(6):993-9.

265. Kotlinowski J, Jozkowicz A. PPAR Gamma and Angiogenesis: Endothelial Cells Perspective. J Diabetes Res. 2016;2016.

266. Lecka-Czernik B, Suva LJ. Resolving the Two "Bony" Faces of PPAR-γ. PPAR Res. 2006;2006.

267. Zhuang H, Zhang X, Zhu C, Tang X, Yu F, Shang GW, et al. Molecular Mechanisms of PPAR-gamma Governing MSC Osteogenic and Adipogenic Differentiation. Curr Stem Cell Res Ther. 2016;11(3):255-64.

268. Thomas T. The complex effects of leptin on bone metabolism through multiple pathways. Curr Opin Pharmacol. 2004;4(3):295-300.

269. Hamrick MW. Role of the Cytokine-like Hormone Leptin in Muscle-bone Crosstalk with Aging. J Bone Metab. 2017;24(1):1-8.

270. Vuolteenaho K, Koskinen A, Moilanen E. Leptin - a link between obesity and osteoarthritis. applications for prevention and treatment. Basic Clin Pharmacol Toxicol. 2014;114(1):103-8.

271. Wasim M, Awan FR, Najam SS, Khan AR, Khan HN. Role of Leptin Deficiency, Inefficiency, and Leptin Receptors in Obesity. Biochem Genet. 2016;54(5):565-72.

272. Spiegelman BM, Flier JS. Obesity and the regulation of energy balance. Cell. 2001;104(4):531-43.

273. Carter S, Caron A, Richard D, Picard F. Role of leptin resistance in the development of obesity in older patients. Clin Interv Aging. 2013;8:829-44.
274. Xu JC, Wu GH, Zhou LL, Yang XJ, Liu JT. Leptin improves osteoblast differentiation of human bone marrow stroma stem cells. Eur Rev Med Pharmacol Sci. 2016;20(16):3507-13.

275. Smathers RL, Petersen DR. The human fatty acid-binding protein family: Evolutionary divergences and functions. Hum Genomics. 2011;5(3):170-91.
276. Hunt CR, Ro JH, Dobson DE, Min HY, Spiegelman BM. Adipocyte P2 gene: developmental expression and homology of 5'-flanking sequences among fat cell-specific genes. Proc Natl Acad Sci U S A. 1986;83(11):3786-90.

277. Aldridge A, Kouroupis D, Churchman S, English A, Ingham E, Jones E. Assay validation for the assessment of adipogenesis of multipotential stromal cells--a direct comparison of four different methods. Cytotherapy. 2013;15(1):89-101.

278. Kanda Y, Hinata T, Kang SW, Watanabe Y. Reactive oxygen species mediate adipocyte differentiation in mesenchymal stem cells. Life sciences. 2011;89(7):250-8.

279. Atashi F, Modarressi A, Pepper MS. The role of reactive oxygen species in mesenchymal stem cell adipogenic and osteogenic differentiation: a review. Stem Cells Dev. 2015;24(10):1150-63.

280. Parvathi Kumara Reddy Thavanati KRK, 2 Aurora Escoto de Dios,3 and Jose Maria Cantu Garza1. Age-Related Correlation Between Antioxidant Enzymes and DNA Damage With Smoking and Body Mass Index. Journal of Gerontology: BIOLOGICAL SCIENCES. 2008;63A(4):360-4.

281. Davalli P, Mitic T, Caporali A, Lauriola Á, D'Àrca D. ROS, Cell Senescence, and Novel Molecular Mechanisms in Aging and Age-Related Diseases. Oxid Med Cell Longev. 2016;2016.

282. Birben E, Sahiner UM, Sackesen C, Erzurum S, Kalayci O. Oxidative
Stress and Antioxidant Defense. World Allergy Organ J. 2012;5(1):9-19.
283. Younus H. Therapeutic potentials of superoxide dismutase. Int J Health Sci (Qassim). 2018;12(3):88-93.

284. Wilkins KKKHEMKJHNSA. Mesenchymal stem cell-secreted superoxide dismutase promotes cerebellar neuronal survival - Kemp - 2010 - Journal of Neurochemistry - Wiley Online Library. Journal of neurochemistry. 2010;114:1569-80.

285. Mohyeldin A, Garzon-Muvdi T, Quinones-Hinojosa A. Oxygen in stem cell biology: a critical component of the stem cell niche. Cell Stem Cell. 2010;7(2):150-61.

286. Zhang Y, Marsboom G, Toth PT, Rehman J. Mitochondrial respiration regulates adipogenic differentiation of human mesenchymal stem cells. PLoS One. 2013;8(10):e77077.

287. Mohd Ali N, Boo L, Yeap SK, Ky H, Satharasinghe DA, Liew WC, et al. Probable impact of age and hypoxia on proliferation and microRNA expression profile of bone marrow-derived human mesenchymal stem cells. PeerJ. 2016;4:e1536.

288. Nightingale H, Kemp K, Gray E, Hares K, Mallam E, Scolding N, et al. Changes in expression of the antioxidant enzyme SOD3 occur upon differentiation of human bone marrow-derived mesenchymal stem cells in vitro. Stem Cells Dev. 2012;21(11):2026-35.

289. Nakashima T, Kobayashi Y, Yamasaki S, Kawakami A, Eguchi K, Sasaki H, et al. Protein expression and functional difference of membrane-bound and soluble receptor activator of NF-kappaB ligand: modulation of the expression by osteotropic factors and cytokines. Biochem Biophys Res Commun. 2000;275(3):768-75.

290. Bielby R, Jones E, McGonagle D. The role of mesenchymal stem cells in maintenance and repair of bone. Injury. 2007;38 Suppl 1:S26-32.

291. Gori F, Hofbauer LC, Dunstan CR, Spelsberg TC, Khosla S, Riggs BL. The expression of osteoprotegerin and RANK ligand and the support of osteoclast formation by stromal-osteoblast lineage cells is developmentally regulated. Endocrinology. 2000;141(12):4768-76.

292. Arai F, Miyamoto T, Ohneda O, Inada T, Sudo T, Brasel K, et al. Commitment and differentiation of osteoclast precursor cells by the sequential expression of c-Fms and receptor activator of nuclear factor kappaB (RANK) receptors. J Exp Med. 1999;190(12):1741-54.

293. Gohda J, Akiyama T, Koga T, Takayanagi H, Tanaka S, Inoue J. RANKmediated amplification of TRAF6 signaling leads to NFATc1 induction during osteoclastogenesis. EMBO J. 2005;24(4):790-9.

294. Trouvin AP, Goeb V. Receptor activator of nuclear factor-kappaB ligand and osteoprotegerin: maintaining the balance to prevent bone loss. Clin Interv Aging. 2010;5:345-54.

295. Yellowley C. CXCL12/CXCR4 signaling and other recruitment and homing pathways in fracture repair. Bonekey Rep. 2013;2.

296. Siwan Park HJ, Byung Soo Kim, Changmo Hwang, Gi Seok Jeong ,Yongdoo Park. Directional migration of mesenchymal stem cells under an SDF-1α gradient on a microfluidic device. Plos one. 2017;12.

297. Bara JJ, Richards RG, Alini M, Stoddart MJ. Concise review: Bone marrow-derived mesenchymal stem cells change phenotype following in vitro culture: implications for basic research and the clinic. Stem Cells. 2014;32(7):1713-23.

298. Jones E, McGonagle D. Human bone marrow mesenchymal stem cells in vivo. Rheumatology. 2008;47(2):126-31.

299. Wright LM, Maloney W, Yu X, Kindle L, Collin-Osdoby P, Osdoby P. Stromal cell-derived factor-1 binding to its chemokine receptor CXCR4 on precursor cells promotes the chemotactic recruitment, development and survival of human osteoclasts. Bone. 2005;36(5):840-53.

Wang J, Loberg R, Taichman RS. The pivotal role of CXCL12 (SDF-1)/CXCR4 axis in bone metastasis. Cancer Metastasis Rev. 2006;25(4):573-87.
Xu H, Gu S, Riquelme MA, Burra S, Callaway D, Cheng H, et al.
Connexin 43 channels are essential for normal bone structure and osteocyte viability. J Bone Miner Res. 2015;30(3):436-48.

302. Genetos DC, Zhou Z, Li Z, Donahue HJ. Age-related changes in gap junctional intercellular communication in osteoblastic cells. J Orthop Res. 2012;30(12):1979-84.

303. Yin Y, Wu R-X, He X-T, Xu X-Y, Wang J, Chen F-M. Influences of agerelated changes in mesenchymal stem cells on macrophages during in-vitro culture. Stem Cell Research & Therapy. 2017;8(1):1-14.

304. Buo AM, Stains JP. Gap junctional regulation of signal transduction in bone cells. FEBS Lett. 2014;588(8):1315-21.

305. Davis HM, Pacheco-Costa R, Atkinson EG, Brun LR, Gortazar AR, Harris J, et al. Disruption of the Cx43/miR21 pathway leads to osteocyte apoptosis and increased osteoclastogenesis with aging. Aging Cell. 2017;16(3):551-63.

306. Lecanda F, Towler DA, Ziambaras K, Cheng SL, Koval M, Steinberg TH, et al. Gap junctional communication modulates gene expression in osteoblastic cells. Mol Biol Cell. 1998;9(8):2249-58.

307. Schilling AF, Filke S, Lange T, Gebauer M, Brink S, Baranowsky A, et al. Gap junctional communication in human osteoclasts in vitro and in vivo. J Cell Mol Med. 2008;12(6a):2497-504.

308. Carter RA, Wicks IP. Vascular cell adhesion molecule 1 (CD106): a multifaceted regulator of joint inflammation. Arthritis Rheum. 2001;44(5):985-94.
309. Lu S, Ge M, Zheng Y, Li J, Feng X, Feng S, et al. CD106 is a novel mediator of bone marrow mesenchymal stem cells via NF-kappaB in the bone marrow failure of acquired aplastic anemia. Stem Cell Res Ther. 2017;8(1):178.
310. Ren G, Zhao X, Zhang L, Zhang J, L'Huillier A, Ling W, et al.

Inflammatory cytokine-induced intercellular adhesion molecule-1 and vascular cell adhesion molecule-1 in mesenchymal stem cells are critical for immunosuppression. J Immunol. 2010;184(5):2321-8.

311. Yang ZX, Han ZB, Ji YR, Wang YW, Liang L, Chi Y, et al. CD106 Identifies a Subpopulation of Mesenchymal Stem Cells with Unique Immunomodulatory Properties. PLoS One. 82013.

312. Bardin N, Anfosso F, Masse JM, Cramer E, Sabatier F, Le Bivic A, et al. Identification of CD146 as a component of the endothelial junction involved in the control of cell-cell cohesion. Blood. 2001;98(13):3677-84.

313. Hadjinicolaou AV, Wu L, Fang B, Watson PA, Hall FC, Busch R. Relationship of CD146 expression to activation of circulating T cells: exploratory studies in healthy donors and patients with connective tissue diseases. Clin Exp Immunol. 2013;174(1):73-88.

314. Sacchetti B, Funari A, Michienzi S, Di Cesare S, Piersanti S, Saggio I, et al. Self-renewing osteoprogenitors in bone marrow sinusoids can organize a hematopoietic microenvironment. Cell. 2007;131(2):324-36.

315. El-Jawhari JJ, Ganguly P, Churchman S, Jones E, Giannoudis PV. The Biological Fitness of Bone Progenitor Cells in Reamer/Irrigator/Aspirator Waste. J Bone Joint Surg Am. 2019.

316. El-Sherbiny YM, El-Jawhari JJ, Moseley TA, McGonagle D, Jones E. T cell immunomodulation by clinically used allogeneic human cancellous bone fragments: a potential novel immunotherapy tool. Scientific Reports. 2018;8(1):1-11.

317. Isidori AM, Strollo F, More M, Caprio M, Aversa A, Moretti C, et al. Leptin and aging: correlation with endocrine changes in male and female healthy adult populations of different body weights. J Clin Endocrinol Metab. 2000;85(5):1954-62.

318. Dutton HP, Borengasser SJ, Gaudet LM, Barbour LA, Keely EJ. Obesity in Pregnancy – Optimizing outcomes for mom and baby. Med Clin North Am. 2018;102(1):87-106.

319. Agrahari G, Sah SK, Kim TY. Superoxide dismutase 3 protects mesenchymal stem cells through enhanced autophagy and regulation of FoxO3a trafficking. BMB Rep. 2018;51(7):344-9.

320. Bertolo A, Capossela S, Frankl G, Baur M, Potzel T, Stoyanov J. Oxidative status predicts quality in human mesenchymal stem cells. Stem Cell Res Ther. 2017;8(1):3.

321. Yu J, Shi J, Zhang Y, Huang Y, Chen Z, Yang J. The replicative senescent mesenchymal stem / stromal cells defect in DNA damage response and anti-oxidative capacity. Int J Med Sci. 2018;15(8):771-81.

322. Gomariz A, Helbling PM, Isringhausen S, Suessbier U, Becker A, Boss A, et al. Quantitative spatial analysis of haematopoiesis-regulating stromal cells in the bone marrow microenvironment by 3D microscopy. Nature Communications. 2018;9(1):1-15.

323. Kar R, Riquelme MA, Werner S, Jiang JX. Connexin 43 channels protect osteocytes against oxidative stress-induced cell death. J Bone Miner Res. 2013;28(7):1611-21.

324. Watkins M, Grimston SK, Norris JY, Guillotin B, Shaw A, Beniash E, et al. Osteoblast connexin43 modulates skeletal architecture by regulating both arms of bone remodeling. Mol Biol Cell. 2011;22(8):1240-51.

325. Buo AM, Tomlinson RE, Eidelman ER, Chason M, Stains JP. Connexin43 and Runx2 Interact to Affect Cortical Bone Geometry, Skeletal Development, and Osteoblast and Osteoclast Function. J Bone Miner Res. 2017;32(8):1727-38.

326. Grayson WL, Zhao F, Bunnell B, Ma T. Hypoxia enhances proliferation and tissue formation of human mesenchymal stem cells. Biochem Biophys Res Commun. 2007;358(3):948-53.

327. Liu F, Akiyama Y, Tai S, Maruyama K, Kawaguchi Y, Muramatsu K, et al. Changes in the expression of CD106, osteogenic genes, and transcription factors involved in the osteogenic differentiation of human bone marrow mesenchymal stem cells. J Bone Miner Metab. 2008;26(4):312-20.

328. Horton JA, Hudak KE, Chung EJ, White AO, Scroggins BT, Burkeen JF, et al. Mesenchymal stem cells inhibit cutaneous radiation-induced fibrosis by suppressing chronic inflammation. Stem Cells. 2013;31(10):2231-41.

329. Hagmann S, Frank S, Gotterbarm T, Dreher T, Eckstein V, Moradi B. Fluorescence activated enrichment of CD146+ cells during expansion of human bone-marrow derived mesenchymal stromal cells augments proliferation and GAG/DNA content in chondrogenic media. BMC Musculoskelet Disord. 2014;15:322.

330. Moravcikova E, Meyer EM, Corselli M, Donnenberg VS, Donnenberg AD. Proteomic Profiling of Native Unpassaged and Culture-Expanded Mesenchymal Stromal Cells (MSC). Cytometry A. 2018;93(9):894-904.

331. Kim KM, Noh JH, Bodogai M, Martindale JL, Yang X, Indig FE, et al. Identification of senescent cell surface targetable protein DPP4. Genes Dev. 2017;31(15):1529-34.

332. Kuranda K, Vargaftig J, de la Rochere P, Dosquet C, Charron D, Bardin F, et al. Age-related changes in human hematopoietic stem/progenitor cells. Aging Cell. 2011;10(3):542-6.

333. Pang WW, Schrier SL, Weissman IL. Age-associated changes in human hematopoietic stem cells. Semin Hematol. 2017;54(1):39-42.

334. Lee J, Yoon SR, Choi I, Jung H. Causes and Mechanisms of Hematopoietic Stem Cell Aging. Int J Mol Sci. 2019;20(6).

335. Tasdogan A, Kumar S, Allies G, Bausinger J, Beckel F, Hofemeister H, et al. DNA Damage-Induced HSPC Malfunction Depends on ROS Accumulation Downstream of IFN-1 Signaling and Bid Mobilization. Cell Stem Cell. 2016;19(6):752-67.

336. Cecco MD, Ito T, Petrashen AP, Elias AE, Skvir NJ, Criscione SW, et al. LINE-1 derepression in senescent cells triggers interferon and inflammaging. Nature. 2019;566(7742):73-8.

337. Pang WW, Price EA, Sahoo D, Beerman I, Maloney WJ, Rossi DJ, et al. Human bone marrow hematopoietic stem cells are increased in frequency and myeloid-biased with age. Proceedings of the National Academy of Sciences. 2011;108(50):20012-7.

338. Van Zant G, Liang Y. The role of stem cells in aging. Exp Hematol. 2003;31(8):659-72.

339. Geiger H, de Haan G, Florian MC. The ageing haematopoietic stem cell compartment. Nat Rev Immunol. 2013;13(5):376-89.

340. Aqmasheh S, Shamsasanjan K, Akbarzadehlaleh P, Pashoutan Sarvar D, Timari H. Effects of Mesenchymal Stem Cell Derivatives on Hematopoiesis and Hematopoietic Stem Cells. Adv Pharm Bull. 2017;7(2):165-77.

341. Sivanathan KN, Gronthos S, Rojas-Canales D, Thierry B, Coates PT. Interferon-gamma modification of mesenchymal stem cells: implications of autologous and allogeneic mesenchymal stem cell therapy in allotransplantation. Stem Cell Rev Rep. 2014;10(3):351-75.

342. Aggarwal S, Pittenger MF. Human mesenchymal stem cells modulate

allogeneic immune cell responses. Blood. 2005;105(4):1815-22.

343. Corcione A, Benvenuto F, Ferretti E, Giunti D, Cappiello V, Cazzanti F, et al. Human mesenchymal stem cells modulate B-cell functions. Blood. 2006;107(1):367-72.

344. Borodkina A, Shatrova A, Abushik P, Nikolsky N, Burova E. Interaction between ROS dependent DNA damage, mitochondria and p38 MAPK underlies senescence of human adult stem cells. Aging (Albany NY). 2014;6(6):481-95. 345. He S, Sharpless NE. Senescence in Health and Disease. Cell. 2017;169(6):1000-11.

346. Rufini A, Tucci P, Celardo I, Melino G. Senescence and aging: the critical roles of p53. Oncogene. 2013;32(43):5129-43.

347. Rodier F, Campisi J, Bhaumik D. Two faces of p53: aging and tumor suppression. Nucleic Acids Res. 2007;35(22):7475-84.

348. Boregowda SV, Krishnappa V, Strivelli J, Haga CL, Booker CN, Phinney DG. Basal p53 expression is indispensable for mesenchymal stem cell integrity. Cell Death Differ. 2018;25(4):679-92.

349. Schellenberg A, Stiehl T, Horn P, Joussen S, Pallua N, Ho AD, et al. Population dynamics of mesenchymal stromal cells during culture expansion. Cytotherapy. 2012;14(4):401-11.

350. Gnani D, Crippa S, Della Volpe L, Rossella V, Conti A, Lettera E, et al. An early-senescence state in aged mesenchymal stromal cells contributes to hematopoietic stem and progenitor cell clonogenic impairment through the activation of a pro-inflammatory program. Aging Cell. 2019;18(3):e12933. 351. Puzianowska-Kuźnicka M, Owczarz M, Wieczorowska-Tobis K, Nadrowski P, Chudek J, Slusarczyk P, et al. Interleukin-6 and C-reactive protein, successful aging, and mortality: the PolSenior study. Immun Ageing.

2016;13.

352. de Hooge AS, van de Loo FA, Bennink MB, Arntz OJ, de Hooge P, van den Berg WB. Male IL-6 gene knock out mice developed more advanced osteoarthritis upon aging. Osteoarthritis Cartilage. 2005;13(1):66-73.

353. Franceschi C, Bonafe M, Valensin S, Olivieri F, De Luca M, Ottaviani E, et al. Inflamm-aging. An evolutionary perspective on immunosenescence. Ann N Y Acad Sci. 2000;908:244-54.

354. O'Hagan-Wong K, Nadeau S, Carrier-Leclerc A, Apablaza F, Hamdy R, Shum-Tim D, et al. Increased IL-6 secretion by aged human mesenchymal stromal cells disrupts hematopoietic stem and progenitor cells' homeostasis. Oncotarget. 2016;7(12):13285-96.

355. Corfe SA, Paige CJ. The many roles of IL-7 in B cell development; mediator of survival, proliferation and differentiation. Semin Immunol. 2012;24(3):198-208.

356. Nemoto Y, Kanai T, Takahara M, Oshima S, Nakamura T, Okamoto R, et al. Bone marrow-mesenchymal stem cells are a major source of interleukin-7 and sustain colitis by forming the niche for colitogenic CD4 memory T cells. Gut. 2013;62(8):1142-52.

357. Updyke LW, Cocke KS, Wierda D. Age-related changes in production of interleukin-7 (IL-7) by murine long-term bone marrow cultures (LTBMC). Mech Ageing Dev. 1993;69(1-2):109-17.

358. Rubenhagen R, Schuttrumpf JP, Sturmer KM, Frosch KH. Interleukin-7 levels in synovial fluid increase with age and MMP-1 levels decrease with progression of osteoarthritis. Acta Orthop. 2012;83(1):59-64.

359. Frasca D, Blomberg BB. Effects of aging on B cell function. Curr Opin Immunol. 2009;21(4):425-30.

360. Coppe JP, Patil CK, Rodier F, Sun Y, Munoz DP, Goldstein J, et al. Senescence-associated secretory phenotypes reveal cell-nonautonomous functions of oncogenic RAS and the p53 tumor suppressor. PLoS Biol. 2008;6(12):2853-68.

361. Soto-Gamez A, Demaria M. Therapeutic interventions for aging: the case of cellular senescence. Drug Discov Today. 2017;22(5):786-95.

362. Dutheil F, Trousselard M, Perrier C, Lac G, Chamoux A, Duclos M, et al. Urinary interleukin-8 is a biomarker of stress in emergency physicians,

especially with advancing age--the JOBSTRESS* randomized trial. PLoS One. 2013;8(8):e71658.

363. Todorović-Raković N, Milovanović J. Interleukin-8 in Breast Cancer Progression. J Interferon Cytokine Res. 2013;33(10):563-70.

364. Wang J, Wang Y, Wang S, Cai J, Shi J, Sui X, et al. Bone marrowderived mesenchymal stem cell-secreted IL-8 promotes the angiogenesis and growth of colorectal cancer. Oncotarget. 2015;6(40):42825-37.

365. Farr JN, Fraser DG, Wang H, Jaehn K, Ogrodnik MB, Weivoda MM, et al. Identification of Senescent Cells in the Bone Microenvironment. J Bone Miner Res. 2016;31(11):1920-9.

366. Franceschi C, Garagnani P, Vitale G, Capri M, Salvioli S. Inflammaging and 'Garb-aging'. Trends Endocrinol Metab. 2017;28(3):199-212.

367. Kim EC, Kim JR. Senotherapeutics: emerging strategy for healthy aging and age-related disease. BMB Rep. 2019;52(1):47-55.

368. Ullah M, Sun Z. Stem cells and anti-aging genes: double-edged sword do the same job of life extension. Stem Cell Res Ther. 2018;9.

369. Tasselli L, Zheng W, Chua KF. SIRT6: novel mechanisms and links to aging and disease. Trends Endocrinol Metab. 2017;28(3):168-85.

370. McCord RA, Michishita E, Hong T, Berber E, Boxer LD, Kusumoto R, et al. SIRT6 stabilizes DNA-dependent protein kinase at chromatin for DNA double-strand break repair. Aging (Albany NY). 2009;1(1):109-21.

371. Kanfi Y, Naiman S, Amir G, Peshti V, Zinman G, Nahum L, et al. The sirtuin SIRT6 regulates lifespan in male mice. Nature. 2012;483(7388):218-21.
372. Sharma A, Diecke S, Zhang WY, Lan F, He C, Mordwinkin NM, et al. The Role of SIRT6 Protein in Aging and Reprogramming of Human Induced Pluripotent Stem Cells*. J Biol Chem. 2013;288(25):18439-47.

373. Sun H, Wu Y, Fu D, Liu Y, Huang C. SIRT6 regulates osteogenic differentiation of rat bone marrow mesenchymal stem cells partially via suppressing the nuclear factor-kappaB signaling pathway. Stem Cells. 2014;32(7):1943-55.

374. Pan H, Guan D, Liu X, Li J, Wang L, Wu J, et al. SIRT6 safeguards human mesenchymal stem cells from oxidative stress by coactivating NRF2. Cell Research. 2016;26(2):190-205.

375. Zhai XY, Yan P, Zhang J, Song HF, Yin WJ, Gong H, et al. Knockdown of SIRT6 Enables Human Bone Marrow Mesenchymal Stem Cell Senescence. Rejuvenation Res. 2016;19(5):373-84.

376. Kurosu H, Yamamoto M, Clark JD, Pastor JV, Nandi A, Gurnani P, et al. Suppression of aging in mice by the hormone Klotho. Science. 2005;309(5742):1829-33.

377. Kuro-o M, Matsumura Y, Aizawa H, Kawaguchi H, Suga T, Utsugi T, et al. Mutation of the mouse klotho gene leads to a syndrome resembling ageing. Nature. 1997;390(6655):45-51.

378. Ullah M, Sun Z. Klotho Deficiency Accelerates Stem Cells Aging by Impairing Telomerase Activity. J Gerontol A Biol Sci Med Sci. 2019;74(9):1396-407.

379. Bian A, Neyra JA, Zhan M, Hu MC. Klotho, stem cells, and aging. Clin Interv Aging. 2015;10:1233-43.

380. Yamazaki Y, Imura A, Urakawa I, Shimada T, Murakami J, Aono Y, et al. Establishment of sandwich ELISA for soluble alpha-Klotho measurement: Agedependent change of soluble alpha-Klotho levels in healthy subjects. Biochem Biophys Res Commun. 2010;398(3):513-8.

381. Günther C, Stangl M, Hermann F, Kg AGC, inventors; Apceth Gmbh & Co. Kg, assignee. Genetically modified mesenchymal stem cell expressing klotho. Worldwide applications2016 2016-09-01.

382. de Oliveira RM. Klotho RNAi induces premature senescence of human cells via a p53/p21 dependent pathway. FEBS Lett. 2006;580(24):5753-8.

383. M.Chawla-Sarkar DJLY-FLRWCSHSCB. Apoptosis and interferons: Role of interferon-stimulated genes as mediators of apoptosis | SpringerLink. Apoptosis. 2019:237-49.

384. Staal FJT. Cell intrinsic regulation of external hematopoietic stem cell stress. Stem Cell Investig. 2018;5:1-3.

385. Peiffer I, Eid P, Barbet R, Li ML, Oostendorp RA, Haydont V, et al. A sub-population of high proliferative potential-quiescent human mesenchymal stem cells is under the reversible control of interferon alpha/beta. Leukemia. 2007;21(4):714-24.

386. Hertzog PJ, Williams BR. Fine tuning type I interferon responses. Cytokine Growth Factor Rev. 2013;24(3):217-25.

387. Banchereau J, Pascual V. Type I interferon in systemic lupus erythematosus and other autoimmune diseases. Immunity. 2006;25(3):383-92.

388. Wilson EB, Yamada DH, Elsaesser H, Herskovitz J, Deng J, Cheng G, et al. Blockade of chronic type I interferon signaling to control persistent LCMV infection. Science. 2013;340(6129):202-7.

389. Doherty MR, Cheon H, Junk DJ, Vinayak S, Varadan V, Telli ML, et al. Interferon-beta represses cancer stem cell properties in triple-negative breast cancer. 2017.

390. de Weerd NA, Samarajiwa SA, Hertzog PJ. Type I interferon receptors: biochemistry and biological functions. J Biol Chem. 2007;282(28):20053-7. 391. Pogue SL, Preston BT, Stalder J, Bebbington CR, Cardarelli PM. The receptor for type I IFNs is highly expressed on peripheral blood B cells and monocytes and mediates a distinct profile of differentiation and activation of these cells. J Interferon Cytokine Res. 2004;24(2):131-9.

392. Chen K, Liu J, Cao X. Regulation of type I interferon signaling in immunity and inflammation: A comprehensive review. J Autoimmun. 2017;83:1-11.

393. Suprunenko T, Hofer MJ. The emerging role of interferon regulatory factor 9 in the antiviral host response and beyond. Cytokine Growth Factor Rev. 2016;29:35-43.

394. Abb J, Abb H, Deinhardt F. Age-related decline of human interferon alpha and interferon gamma production. Blut. 1984;48(5):285-9.

395. Brzostek-Racine S, Gordon C, Van Scoy S, Reich NC. The DNA
Damage Response Induces Interferon. J Immunol. 2011;187(10):5336-45.
396. Paludan SR, Bowie AG. Immune sensing of DNA. Immunity.
2013;38(5):870-80.

397. Wu J, Sun L, Chen X, Du F, Shi H, Chen C, et al. Cyclic GMP-AMP is an endogenous second messenger in innate immune signaling by cytosolic DNA. Science. 2013;339(6121):826-30.

398. Sun L, Wu J, Du F, Chen X, Chen ZJ. Cyclic GMP-AMP synthase is a cytosolic DNA sensor that activates the type I interferon pathway. Science. 2013;339(6121):786-91.

399. Li T, Chen ZJ. The cGAS-cGAMP-STING pathway connects DNA damage to inflammation, senescence, and cancer. J Exp Med. 2018;215(5):1287-99.

400. Prabakaran T, Bodda C, Krapp C, Zhang BC, Christensen MH, Sun C, et al. Attenuation of cGAS-STING signaling is mediated by a p62/SQSTM1dependent autophagy pathway activated by TBK1. Embo j. 2018;37(8).

401. Gui X, Yang H, Li T, Tan X, Shi P, Li M, et al. Autophagy induction via STING trafficking is a primordial function of the cGAS pathway. Nature. 2019;567(7747):262-6.

402. Liu D, Wu H, Wang C, Li Y, Tian H, Siraj S, et al. STING directly activates autophagy to tune the innate immune response. Cell Death & Differentiation. 2018;26(9):1735-49.

403. Maryanovich M, Oberkovitz G, Niv H, Vorobiyov L, Zaltsman Y, Brenner O, et al. The ATM-BID pathway regulates quiescence and survival of haematopoietic stem cells. Nat Cell Biol. 2012;14(5):535-41.

404. Maryanovich M, Zaltsman Y, Ruggiero A, Goldman A, Shachnai L, Zaidman SL, et al. An MTCH2 pathway repressing mitochondria metabolism regulates haematopoietic stem cell fate. Nat Commun. 2015;6:7901.

405. Morito D, Nishikawa K, Hoseki J, Kitamura A, Kotani Y, Kiso K, et al. Moyamoya disease-associated protein mysterin/RNF213 is a novel AAA+ ATPase, which dynamically changes its oligomeric state. Sci Rep. 2014;4:4442. 406. Krabbe KS, Pedersen M, Bruunsgaard H. Inflammatory mediators in the elderly. Exp Gerontol. 2004;39(5):687-99.

407. Li G, Ju J, Weyand CM, Goronzy JJ. Age-associated Failure to Adjust Type I Interferon Receptor Signaling Thresholds after T-cell Activation1. J Immunol. 2015;195(3):865-74.

408. Gu Z, Jiang J, Tan W, Xia Y, Cao H, Meng Y, et al. p53/p21 Pathway involved in mediating cellular senescence of bone marrow-derived mesenchymal stem cells from systemic lupus erythematosus patients. Clin Dev Immunol. 2013;2013:134243.

409. Zhang W, Xue D, Hu D, Xie T, Tao Y, Zhu T, et al. Secreted klotho protein attenuates osteogenic differentiation of human bone marrow mesenchymal stem cells in vitro via inactivation of the FGFR1/ERK signaling pathway. Growth Factors. 2015;33(5-6):356-65.

410. Mostoslavsky R, Chua KF, Lombard DB, Pang WW, Fischer MR, Gellon L, et al. Genomic instability and aging-like phenotype in the absence of mammalian SIRT6. Cell. 2006;124(2):315-29.

411. Zhang P, Liu Y, Wang Y, Zhang M, Lv L, Zhang X, et al. SIRT6 promotes osteogenic differentiation of mesenchymal stem cells through BMP signaling. Scientific Reports. 2017;7(1):1-10.

412. Kovtonyuk LV, Fritsch K, Feng X, Manz MG, Takizawa H. Inflamm-Aging of Hematopoiesis, Hematopoietic Stem Cells, and the Bone Marrow Microenvironment. Front Immunol. 2016;7.

413. Arden N, Nevitt MC. Osteoarthritis: epidemiology. Best Pract Res Clin Rheumatol. 2006;20(1):3-25.

414. Loeser RF. Aging processes and the development of osteoarthritis. Curr Opin Rheumatol. 2013;25(1):108-13.

415. Manolagas SC, Parfitt AM. What old means to bone. Trends in Endocrinology & Metabolism. 2010;21(6):369-74.

416. Infante A, Rodríguez CI. Osteogenesis and aging: lessons from mesenchymal stem cells. Stem Cell Research & Therapy. 2018;9(1):1-7.

418. Lespasio MJ, Sultan AA, Piuzzi NS, Khlopas A, Husni ME, Muschler GF, et al. Hip Osteoarthritis: A Primer. Perm J. 2018;22.

419. Dagenais S, Garbedian S, Wai EK. Systematic review of the prevalence of radiographic primary hip osteoarthritis. Clin Orthop Relat Res. 2009;467(3):623-37.

420. Murphy NJ, Eyles JP, Hunter DJ. Hip Osteoarthritis: Etiopathogenesis and Implications for Management. Adv Ther. 2016;33(11):1921-46.

421. Kohn MD, Sassoon AA, Fernando ND. Classifications in Brief: Kellgren-Lawrence Classification of Osteoarthritis. Clin Orthop Relat Res. 2016;474(8):1886-93.

422. Kim C, Nevitt MC, Niu J, Clancy MM, Lane NE, Link TM, et al. Association of hip pain with radiographic evidence of hip osteoarthritis: diagnostic test study. 2015.

423. Chong T, Don DW, Kao MC, Wong D, Mitra R. The value of physical examination in the diagnosis of hip osteoarthritis. J Back Musculoskelet Rehabil. 2013;26(4):397-400.

424. Rannou F, Pelletier JP, Martel-Pelletier J. Efficacy and safety of topical NSAIDs in the management of osteoarthritis: Evidence from real-life setting trials and surveys. Semin Arthritis Rheum. 2016;45(4 Suppl):S18-21.

425. Karsdal MA, Bay-Jensen AC, Lories RJ, Abramson S, Spector T, Pastoureau P, et al. The coupling of bone and cartilage turnover in osteoarthritis: opportunities for bone antiresorptives and anabolics as potential treatments? Ann Rheum Dis. 2014;73(2):336-48.

426. Crofford LJ. Use of NSAIDs in treating patients with arthritis. Arthritis Res Ther. 2013;15(Suppl 3):S2.

427. Gandhi R, Perruccio AV, Mahomed NN. Surgical management of hip osteoarthritis. CMAJ. 2014;186(5):347-55.

428. Jeon OH, Kim C, Laberge RM, Demaria M, Rathod S, Vasserot AP, et al. Local clearance of senescent cells attenuates the development of posttraumatic osteoarthritis and creates a pro-regenerative environment. Nat Med. 2017;23(6):775-81.

429. Martin JA, Buckwalter JA. Telomere erosion and senescence in human articular cartilage chondrocytes. J Gerontol A Biol Sci Med Sci. 2001;56(4):B172-9.

430. Jeon OH, David N, Campisi J, Elisseeff JH. Senescent cells and osteoarthritis: a painful connection. J Clin Invest. 2018;128(4):1229-37.

431. Rose J, Soder S, Skhirtladze C, Schmitz N, Gebhard PM, Sesselmann S, et al. DNA damage, discoordinated gene expression and cellular senescence in osteoarthritic chondrocytes. Osteoarthritis Cartilage. 2012;20(9):1020-8.

432. Ziskoven C, Jäger M, Zilkens C, Bloch W, Brixius K, Krauspe R. Oxidative stress in secondary osteoarthritis: from cartilage destruction to clinical presentation? Orthop Rev (Pavia). 2010;2(2).

433. Lepetsos P, Papavassiliou AG. ROS/oxidative stress signaling in osteoarthritis. Biochim Biophys Acta. 2016;1862(4):576-91.

434. Xu M, Robert and Arlene Kogod Center on Aging, Bradley EW, Department of Orthopedic Surgery, Weivoda MM, Robert and Arlene Kogod Center on Aging, et al. Transplanted Senescent Cells Induce an Osteoarthritis-Like Condition in Mice. The Journals of Gerontology: Series A. 201 7;72(6):780-5.

435. Berenbaum F. Osteoarthritis as an inflammatory disease (osteoarthritis is not osteoarthrosis!). Osteoarthritis Cartilage. 2013;21(1):16-21.

436. Greene MA, Loeser RF. Aging-related inflammation in osteoarthritis. Osteoarthritis Cartilage. 2015;23(11):1966-71.

437. Livshits G, Zhai G, Hart DJ, Kato BS, Wang H, Williams FM, et al. Interleukin-6 is a significant predictor of radiographic knee osteoarthritis: The Chingford Study. Arthritis Rheum. 2009;60(7):2037-45.

438. Tan J, Xu X, Tong Z, lin J, Yu Q, Lin Y, et al. Decreased osteogenesis of adult mesenchymal stem cells by reactive oxygen species under cyclic stretch: a possible mechanism of age related osteoporosis. Bone Res. 2015;3:15003-. 439. Henrotin YE, Bruckner P, Pujol JP. The role of reactive oxygen species in homeostasis and degradation of cartilage. Osteoarthritis Cartilage. 2003;11(10):747-55.

440. Henrotin Y, Kurz B, Aigner T. Oxygen and reactive oxygen species in cartilage degradation: friends or foes? Osteoarthritis Cartilage. 2005;13(8):643-54.

441. Bolduc JA, Collins JA, Loeser RF. Reactive oxygen species, aging and articular cartilage homeostasis. Free Radic Biol Med. 2019;132:73-82.

442. Loeser RF. Aging and Osteoarthritis: The Role of Chondrocyte Senescence and Aging Changes in the Cartilage Matrix. Osteoarthritis Cartilage. 2009;17(8):971-9.

443. Martin JA, Buckwalter JA. Aging, articular cartilage chondrocyte senescence and osteoarthritis. Biogerontology. 2002;3(5):257-64.
444. Loeser RF. Aging and Osteoarthritis. Curr Opin Rheumatol. 2011;23(5):492-6.

445. Regan E, Flannelly J, Bowler R, Tran K, Nicks M, Carbone BD, et al. Extracellular superoxide dismutase and oxidant damage in osteoarthritis. Arthritis Rheum. 2005;52(11):3479-91.

446. Regan EA, Bowler RP, Crapo JD. Joint fluid antioxidants are decreased in osteoarthritic joints compared to joints with macroscopically intact cartilage and subacute injury. Osteoarthritis Cartilage. 2008;16(4):515-21.

447. Paździor M, Kiełczykowska M, Kurzepa J, Luchowska-Kocot D, Kocot J, Musik I. The Oxidative Stress in Knee Osteoarthritis Patients. An Attempt of Evaluation of Possible Compensatory Effects Occurring in the Disease Development. Medicina (Kaunas). 2019;55(5):1-10.

448. Yudoh K, Nguyen v T, Nakamura H, Hongo-Masuko K, Kato T, Nishioka K. Potential involvement of oxidative stress in cartilage senescence and development of osteoarthritis: oxidative stress induces chondrocyte telomere instability and downregulation of chondrocyte function. Arthritis Res Ther. 2005;7(2):R380-91.

449. Li D, Xie G, Wang W. Reactive oxygen species: the 2-edged sword of osteoarthritis. Am J Med Sci. 2012;344(6):486-90.

450. Mobasheri A, Rayman MP, Gualillo O, Sellam J, van der Kraan P, Fearon U. The role of metabolism in the pathogenesis of osteoarthritis. Nat Rev Rheumatol. 2017;13(5):302-11.

451. Li X, Liu L, Meng D, Wang D, Zhang J, Shi D, et al. Enhanced apoptosis and senescence of bone-marrow-derived mesenchymal stem cells in patients with systemic lupus erythematosus. Stem Cells Dev. 2012;21(13):2387-94.
452. Sharma AR, Jagga S, Lee SS, Nam JS. Interplay between Cartilage and Subchondral Bone Contributing to Pathogenesis of Osteoarthritis. Int J Mol Sci. 2013;14(10):19805-30.

453. Jayasuriya CT, Hu N, Li J, Lemme N, Terek R, Ehrlich MG, et al. Molecular characterization of mesenchymal stem cells in human osteoarthritis cartilage reveals contribution to the OA phenotype. Sci Rep. 2018;8.

454. Simopoulou T, Malizos KN, Iliopoulos D, Stefanou N, Papatheodorou L, Ioannou M, et al. Differential expression of leptin and leptin's receptor isoform (Ob-Rb) mRNA between advanced and minimally affected osteoarthritic cartilage; effect on cartilage metabolism. Osteoarthritis Cartilage. 2007;15(8):872-83.

455. Wei F, Moore DC, Wei L, Li Y, Zhang G, Wei X, et al. Attenuation of osteoarthritis via blockade of the SDF-1/CXCR4 signaling pathway. Arthritis Res Ther. 2012;14(4):R177.

456. Endres M, Andreas K, Kalwitz G, Freymann U, Neumann K, Ringe J, et al. Chemokine profile of synovial fluid from normal, osteoarthritis and rheumatoid arthritis patients: CCL25, CXCL10 and XCL1 recruit human subchondral mesenchymal progenitor cells. Osteoarthritis Cartilage. 2010;18(11):1458-66.

457. Mayan MD, Carpintero-Fernandez P, Gago-Fuentes R, Martinez-de-Ilarduya O, Wang HZ, Valiunas V, et al. Human articular chondrocytes express multiple gap junction proteins: differential expression of connexins in normal and osteoarthritic cartilage. Am J Pathol. 2013;182(4):1337-46.

458. Duarte JH. Osteoarthritis: SIRT6 prevents chondrocyte senescence and DNA damage. Nature Reviews Rheumatology. 2015;11(5):260-.

459. Ailixiding M, Aibibula Z, Okawa A, Morita S, Asou Y. SIRT6 in the crosstalk among ageing and metabolic osteoarthritis. Osteoarthritis and Cartilage. 2016;24.

460. Hashimoto S, Nishiyama T, Hayashi S, Fujishiro T, Takebe K, Kanzaki N, et al. Role of p53 in human chondrocyte apoptosis in response to shear strain. Arthritis Rheum. 2009;60(8):2340-9.

461. Doss F, Menard J, Hauschild M, Kreutzer HJ, Mittlmeier T, Muller-Steinhardt M, et al. Elevated IL-6 levels in the synovial fluid of osteoarthritis patients stem from plasma cells. Scand J Rheumatol. 2007;36(2):136-9.
462. Fiorito S, Magrini L, Adrey J, Mailhe D, Brouty-Boye D. Inflammatory status and cartilage regenerative potential of synovial fibroblasts from patients with osteoarthritis and chondropathy. Rheumatology (Oxford). 2005;44(2):164-71.

463. Schett G, Kiechl S, Bonora E, Zwerina J, Mayr A, Axmann R, et al. Vascular cell adhesion molecule 1 as a predictor of severe osteoarthritis of the hip and knee joints. Arthritis Rheum. 2009;60(8):2381-9.

464. Pulsatelli L, Assirelli E, Dolzani P, Addimanda O, Mancarella L, Pavloska B, et al. Increased serum vascular cell adhesion molecule (VCAM)-1 levels in patients with erosive hand osteoarthritis. Rheumatology (Oxford). 2013;52(2):400-2.

465. Kalichman L, Pantsulaia I, Kobyliansky E. Association between vascular cell adhesion molecule 1 and radiographic hand osteoarthritis. Clin Exp Rheumatol. 2011;29(3):544-6.

466. Oppl B, Datz C, Huber-Schonauer U, Husar-Memmer E, Brozek W, Zenz P, et al. Vascular cell adhesion molecule 1 in patients with severe osteoarthritis of the hip : A prospective cross-sectional study. Wien Klin Wochenschr. 2019;131(15-16):369-73.

467. Pottie P, Presle N, Terlain B, Netter P, Mainard D, Berenbaum F. Obesity and osteoarthritis: more complex than predicted! Ann Rheum Dis. 2006;65(11):1403-5.

468. Ku JH, Lee CK, Joo BS, An BM, Choi SH, Wang TH, et al. Correlation of synovial fluid leptin concentrations with the severity of osteoarthritis. Clin Rheumatol. 2009;28(12):1431-5.

469. Mutabaruka MS, Aoulad Aissa M, Delalandre A, Lavigne M, Lajeunesse D. Local leptin production in osteoarthritis subchondral osteoblasts may be responsible for their abnormal phenotypic expression. Arthritis Res Ther. 2010;12(1):R20.

470. Yan M, Zhang J, Yang H, Sun Y. The role of leptin in osteoarthritis. Medicine (Baltimore). 972018.

471. Stannus OP, Jones G, Quinn SJ, Cicuttini FM, Dore D, Ding C. The association between leptin, interleukin-6, and hip radiographic osteoarthritis in older people: a cross-sectional study. Arthritis Res Ther. 2010;12(3):R95.
472. Marino AA, Waddell DD, Kolomytkin OV, Meek WD, Wolf R, Sadasivan KK, et al. Increased intercellular communication through gap junctions may contribute to progression of osteoarthritis. Clin Orthop Relat Res. 2004(422):224-32.

473. Gupta A, Niger C, Buo AM, Eidelman ER, Chen RJ, Stains JP.
Connexin43 enhances the expression of osteoarthritis-associated genes in synovial fibroblasts in culture. BMC Musculoskelet Disord. 2014;15:425.
474. Shimomura S, Dept. of Orthopedics GSoMS, Kyoto Prefectural Univ. of

Med., Kyoto, Japan, Tsuchida S, Dept. of Orthopedics GSoMS, Kyoto Prefectural Only. of Prefectural Univ. of Med., Kyoto, Japan, Arai Y, Dept. of Orthopedics GSoMS,

Kyoto Prefectural Univ. of Med., Kyoto, Japan, et al. Analysis of the expression of connexin 43 in synovial tissue of patients with arthritis. Osteoarthritis and Cartilage. 2016;24.

475. Su X, Zuo W, Wu Z, Chen J, Wu N, Ma P, et al. CD146 as a new marker for an increased chondroprogenitor cell sub-population in the later stages of osteoarthritis. J Orthop Res. 2015;33(1):84-91.

476. Stiehler M, Rauh J, Bunger C, Jacobi A, Vater C, Schildberg T, et al. In vitro characterization of bone marrow stromal cells from osteoarthritic donors. Stem Cell Res. 2016;16(3):782-9.

477. de Sousa EB, Dos Santos Junior GC, Aguiar RP, da Costa Sartore R, de Oliveira ACL, Almeida FCL, et al. Osteoarthritic Synovial Fluid Modulates Cell Phenotype and Metabolic Behavior In Vitro. Stem Cells Int. 2019;2019:8169172.

478. Pearson MJ, Herndler-Brandstetter D, Tariq MA, Nicholson TA, Philp AM, Smith HL, et al. IL-6 secretion in osteoarthritis patients is mediated by chondrocyte-synovial fibroblast cross-talk and is enhanced by obesity. Sci Rep. 2017;7.

479. Logar DB, Komadina R, Prezelj J, Ostanek B, Trost Z, Marc J. Expression of bone resorption genes in osteoarthritis and in osteoporosis. J Bone Miner Metab. 2007;25(4):219-25.

480. Upton AR, Holding CA, Dharmapatni AA, Haynes DR. The expression of RANKL and OPG in the various grades of osteoarthritic cartilage. Rheumatol Int. 2012;32(2):535-40.

481. Varela-Eirín M, Varela-Vázquez A, Guitián-Caamaño A, Paíno CL, Mato V, Largo R, et al. Targeting of chondrocyte plasticity via connexin43 modulation attenuates cellular senescence and fosters a pro-regenerative environment in osteoarthritis. Cell Death & Disease. 2018;9(12):1-16.

482. Li M, Zhi L, Zhang Z, Bian W, Qiu Y. Identification of potential target genes associated with the pathogenesis of osteoarthritis using microarray based analysis. Mol Med Rep. 2017;16(3):2799-806.

483. Karvonen-Gutierrez CA, Harlow SD, Mancuso P, Jacobson J, Mendes de Leon CF, Nan B. Association of leptin levels with radiographic knee osteoarthritis among a cohort of midlife women. Arthritis Care Res (Hoboken). 2013;65(6):936-44.

484. Lubbeke A, Finckh A, Puskas GJ, Suva D, Ladermann A, Bas S, et al. Do synovial leptin levels correlate with pain in end stage arthritis? Int Orthop. 2013;37(10):2071-9.

485. Dumond H, Presle N, Terlain B, Mainard D, Loeuille D, Netter P, et al. Evidence for a key role of leptin in osteoarthritis. Arthritis Rheum. 2003;48(11):3118-29.

486. Berry PA, Jones SW, Cicuttini FM, Wluka AE, Maciewicz RA. Temporal relationship between serum adipokines, biomarkers of bone and cartilage turnover, and cartilage volume loss in a population with clinical knee osteoarthritis. Arthritis Rheum. 2011;63(3):700-7.

487. Li Y, Wei X, Zhou J, Wei L. The age-related changes in cartilage and osteoarthritis. Biomed Res Int. 2013;2013:916530.

488. Palmer G, Mezin F, Juge-Aubry CE, Plater-Zyberk C, Gabay C, Guerne P-A. Interferon β stimulates interleukin 1 receptor antagonist production in human articular chondrocytes and synovial fibroblasts. 2004.

489. Takayanagi H, Sato K, Takaoka A, Taniguchi T. Interplay between interferon and other cytokine systems in bone metabolism. Immunol Rev. 2005;208:181-93.

490. Danica D. Vance* LW, Evadnie Rampersaud, Bryson P. Lesniak, Jeffery M. Vance, Margaret A. Pericak-Vance and Lee D. Kaplan. Dynamic Gene Expression Profile Changes in Synovial Fluid Following Meniscal Injury; Osteoarthritis (OA) Markers Found. Journal of Exercise, Sports & Orthopedics. 2019;1(3):1-7.

491. Ch V, Patel SL, Mishra R, Tripathy BC, Prakash J, Gupta MTAV. Molecular signatures in peripheral blood mononuclear cells with osteoarthritis. International Journal of Clinical Rheumatology. 2016;11(6):82.

492. Wyatt LA, Nwosu LN, Wilson D, Hill R, Spendlove I, Bennett AJ, et al. Molecular expression patterns in the synovium and their association with advanced symptomatic knee osteoarthritis. Osteoarthritis Cartilage. 2019;27(4):667-75.

493. Walker JG, Ahern MJ, Coleman M, Weedon H, Papangelis V, Beroukas D, et al. Expression of Jak3, STAT1, STAT4, and STAT6 in inflammatory arthritis: unique Jak3 and STAT4 expression in dendritic cells in seropositive rheumatoid arthritis. Ann Rheum Dis. 2006;65(2):149-56.

494. Lorenz P, Ruschpler P, Koczan D, Stiehl P, Thiesen HJ. From transcriptome to proteome: differentially expressed proteins identified in synovial tissue of patients suffering from rheumatoid arthritis and osteoarthritis by an initial screen with a panel of 791 antibodies. Proteomics. 2003;3(6):991-1002.

495. Zhu N, Hou J, Wu Y, Li G, Liu J, Ma G, et al. Identification of key genes in rheumatoid arthritis and osteoarthritis based on bioinformatics analysis. Medicine (Baltimore). 2018;97(22):e10997.

496. Wu Y, Chen L, Wang Y, Li W, Lin Y, Yu D, et al. Overexpression of Sirtuin 6 suppresses cellular senescence and NF-κB mediated inflammatory responses in osteoarthritis development. Sci Rep. 2015;5.

497. Nagai K, Matsushita T, Matsuzaki T, Takayama K, Matsumoto T, Kuroda R, et al. Depletion of SIRT6 causes cellular senescence, DNA damage, and telomere dysfunction in human chondrocytes. Osteoarthritis Cartilage. 2015;23(8):1412-20.

498. F.Zhang GZ, B.S.Kato, D.J.Hart, D.Hunter, T.D.Spector, K.R.Ahmadi. Association between KLOTHO gene and hand osteoarthritis in a female Caucasian population. Osteoarthritis and cartilage. 2006;15(6):624-9.

499. Tsezou A, Furuichi T, Satra M, Makrythanasis P, Ikegawa S, Malizos KN. Association of KLOTHO gene polymorphisms with knee osteoarthritis in Greek population. J Orthop Res. 2008;26(11):1466-70.

500. Tilly EL, INSERM UMR 1229 RMaS-R, « Skeletal Tissue Engineering and Physiopathology » (STEP) Team, Nantes, France, Univ. of Nantes Fods, Nantes, France, Vinatier C, INSERM UMR 1229 RMaS-R, « Skeletal Tissue Engineering and Physiopathology » (STEP) Team, Nantes, France, Univ. of Nantes Fods, Nantes, France, et al. Towards a Role for the Antigeronic Factor Klotho in Osteoarthritis & Autophagy in Chondrocytes. Osteoarthritis and Cartilage. 2017;25.

501. Stannus O, Jones G, Cicuttini F, Parameswaran V, Quinn S, Burgess J, et al. Circulating levels of IL-6 and TNF-alpha are associated with knee radiographic osteoarthritis and knee cartilage loss in older adults. Osteoarthritis Cartilage. 2010;18(11):1441-7.

502. Amable PR, Teixeira MVT, Carias RBV, Granjeiro JM, Borojevic R. Protein synthesis and secretion in human mesenchymal cells derived from bone marrow, adipose tissue and Wharton's jelly. Stem Cell Res Ther. 2014;5(2):53.

503. Khalid RS, Khan I, Zaidi MB, Naeem N, Haneef K, Qazi RE, et al. IL-7 overexpression enhances therapeutic potential of rat bone marrow mesenchymal stem cells for diabetic wounds. Wound Repair Regen. 2019;27(3):235-48.

504. Gago-Fuentes R, Fernandez-Puente P, Megias D, Carpintero-Fernandez P, Mateos J, Acea B, et al. Proteomic Analysis of Connexin 43 Reveals Novel Interactors Related to Osteoarthritis. Mol Cell Proteomics. 2015;14(7):1831-45. 505. Ganguly P, El-Jawhari JJ, Burska AN, Ponchel F, Giannoudis PV, Jones EA. The Analysis of In Vivo Aging in Human Bone Marrow Mesenchymal Stromal Cells Using Colony-Forming Unit-Fibroblast Assay and the CD45(low)CD271(+) Phenotype. Stem Cells Int. 2019;2019:5197983. 506. Kos K, Wilding JP. SPARC: a key player in the pathologies associated with obesity and diabetes. Nat Rev Endocrinol. 2010;6(4):225-35.

507. Makhluf HA, Mueller SM, Mizuno S, Glowacki J. Age-related decline in osteoprotegerin expression by human bone marrow cells cultured in threedimensional collagen sponges. Biochem Biophys Res Commun. 2000;268(3):669-72.

508. Wu Y, Chen L, Wang Y, Li W, Lin Y, Yu D, et al. Overexpression of Sirtuin 6 suppresses cellular senescence and NF-kappaB mediated inflammatory responses in osteoarthritis development. Sci Rep. 2015;5:17602.
509. Freitas AS, Figueiredo AJ, de Freitas AL, Rodrigues VD, da Cunha AA, Deusdará FF, et al. Biological Maturation, Body Morphology and Physical Performance in 8–16 year-old obese girls from Montes Claros – MG. J Hum Kinet. 2014;43:169-76.

510. Belsky DW, Caspi A, Houts R, Cohen HJ, Corcoran DL, Danese A, et al. Quantification of biological aging in young adults. Proc Natl Acad Sci U S A. 2015;112(30):E4104-10.

511. Sharma A, Madaan V, Petty FD. Exercise for Mental Health. Prim Care Companion J Clin Psychiatry. 2006;8(2):106.

512. Penedo FJ, Dahn JR. Exercise and well-being: a review of mental and physical health benefits associated with physical activity. Curr Opin Psychiatry. 2005;18(2):189-93.

513. Delle Fave A, Bassi M, Boccaletti ES, Roncaglione C, Bernardelli G, Mari D. Promoting Well-Being in Old Age: The Psychological Benefits of Two Training Programs of Adapted Physical Activity. Front Psychol. 2018;9.

514. Take the test: What's my real age? 2019 [Available from:

https://www.bbc.co.uk/teach/take-the-test-whats-my-real-age/zrgjgwx.

515. Find out your biological-age 2019 [Available from: <u>http://www.biological-age.com/#</u>.

516. Sarah M. Churchman DK, Sally A. Boxall, Tarek Roshdy, Hiang B. Tan, Dennis McGonagle, Peter V. Giannoudis and Elena A. Jones. YIELD OPTIMISATION AND MOLECULAR CHARACTERISATION OFUNCULTURED CD271+ MESENCHYMAL STEM CELLS IN THE REAMER IRRIGATOR

ASPIRATOR WASTE BAG. European Cells and Materials. 2013;6:11. 517. Hwang TJ, Carpenter D, Lauffenburger JC, Wang B, Franklin JM, Kesselheim AS. Failure of Investigational Drugs in Late-Stage Clinical Development and Publication of Trial Results. JAMA Intern Med. 2016;176(12):1826-33.

518. Fogel DB. Factors associated with clinical trials that fail and opportunities for improving the likelihood of success: A review. Contemp Clin Trials Commun. 2018;11:156-64.

519. WHO. Non communicable diseases. 2018.

Appendix 1- List of donor cells used across the different arms of investigation in this thesis

Donors	Sex	Age	CFU-F	Colony area , ID	CFU-F in serum
PGBM002	М	19	Yes		
PGBM003	F	63	Yes		
PGBM004	Μ	56	Yes		
PGBM005	F	37	Yes	Yes	
PGBM006	F	52	Yes		
HOBM010	F	74	Yes		
PGBM007	F	51	Yes		
HOBM011	Μ	24	Yes	Yes	
PGBM008	Μ	28	Yes	Yes	
HOBM012	М	58	Yes		
PGBM009	F	68	Yes	Yes	Yes
PGBM010	F	73	Yes	Yes	
PGBM011	М	42	Yes		
PGBM012	М	40	Yes		
PGBM013	M	42	Yes		
PGBM014	F	34	Yes		Yes
PGBM015	F	20	Yes		Yes
PGBM016	М	19	Yes	Yes	
PGBM017	Μ	46	Yes		
PGBM018	F	72	Yes		
PGBM019	F	72	Yes		
KMBM020	F	46	Yes		
JAO1	M	20	Yes		
KMBM021	М	19	Yes	Yes	Yes
HOBM013	М	58	Yes		
PGBM020	M	49	Yes		
KMBM027	F	23	Yes		
PGBM021	M	22	Yes		
PGBM022	F	49	Yes		
PGBM023	F	67	Yes		
PGBM024	M	42	Yes		
PGBM025	F	38	Yes		
PGBM026	F	69	Yes		Yes
PGBM027	F	89	Yes		
PGBM028	F	76	Yes	Yes	Yes
KMBM028	M	59	Yes	Yes	

Table 1 Donor details for cells used in Chapter 3

251

HOBM014	М	44	Yes		
PGBM029	М	34	Yes		
PGBM030	F	89	Yes		
KMBM030	М	25	Yes	Yes	
JAO 25	F	55	Yes	Yes	Yes
PGBM031	F	53	Yes		Yes
PGBM032	М	46	Yes		
PGBM033	М	23	Yes		
JAO039	М	26	Yes		
PGBM034	F	36	Yes		
HOBM018	М	78	Yes		
PGBM036	М	54	Yes	Yes	
PGBM037	Μ	59	Yes	Yes	
PGBM038	F	64	Yes		
PGBM039	Μ	29	Yes		
PGBM040	F	37	Yes	Yes	Yes
PGBM041	F	26	Yes		Yes
PGBM042	М	67	Yes		
PGBM043	F	89	Yes		
PGBM044	F	79	Yes		

Age-range:19-89 years old, median: 47.5 years old
Table 2 Donor details for cells used in Chapter 4

Donors	Sex	Age	Enumeration by flow	ROS	Enriched	Cell sorting	MSC genes
PGBM005	F	37	Yes		Yes		
PGBM006	F	52	Yes		Yes		
HOBM010	F	74	Yes				
PGBM007	F	51	Yes				
HOBM011	Μ	24	Yes				
PGBM008	Μ	28	Yes		Yes		
PGBM009	F	68	Yes				
PGBM010	F	73	Yes		Yes	Yes	Yes
PGBM011	Μ	42	Yes				
PGBM012	Μ	40	Yes				
PGBM013	Μ	42	Yes				
PGBM014	F	34	Yes				
PGBM015	F	20	Yes				
PGBM016	Μ	19	Yes		Yes	Yes	Yes
PGBM017	Μ	46	Yes				
PGBM018	F	72	Yes				
PGBM019	F	72	Yes				
KMBM020	F	46	Yes				
JAO1	М	20	Yes				
KMBM021	Μ	19	Yes				
HOBM013	Μ	58	Yes				
PGBM020	M	49	Yes		Yes	Yes	
KMBM027	F	23	Yes		Yes	Yes	Yes
PGBM021	M	22	Yes		Yes	Yes	Yes
PGBM022	F	49	Yes				
PGBM023	F	67	Yes	Yes	Yes	Yes	Yes
PGBM024	M	42	Yes				
PGBM025	F	38	Yes	Yes	Yes	Yes	Yes
PGBM026	F	69	Yes	Yes	Yes	Yes	Yes
PGBM027	F	89	Yes	Yes	Yes	Yes	Yes
PGBM028		76	Yes	Yes	Yes	Yes	Yes
KMBM028	IVI	59	Yes				
HOBM014	IVI	44	Yes	N/	N/	Ň	V
PGBM029		34	Yes	Yes	Yes	Yes	Yes
PGBM030		89		Yes	Yes	Yes	Yes
KMBM030		25	Yes	Yes	Yes	Yes	Yes
JAO 25	F	55	Yes				
PGBM031		53	Yes	V	N/	V	V
PGBM033	M	23		Yes	Yes	Yes	Yes
PGBIM034	F	36		Yes	Yes	Yes	Yes
				253			

PGBM037	Μ	59	Yes	Yes	Yes	Yes	Yes
PGBM038	F	64			Yes	Yes	Yes
PGBM039	Μ	29		Yes	Yes	Yes	Yes
PGBM042	Μ	67	Yes				
PGBM047	Μ	65		Yes			
RCBM028	Μ	66			Yes	Yes	Yes
GCBM004	F	61			Yes	Yes	Yes
TRBM029	F	36			Yes	Yes	Yes

Age range: 19-89 years old, median: 47.5 years old

Table 3 Donor details for cells used in Chapter 5

Sex	Age	Enrichment	Cell sorting	Exploratory genes
F	37	Yes		
F	52	Yes		
М	28	Yes		
F	73	Yes	Yes	Yes
М	19	Yes	Yes	Yes
М	49	Yes	Yes	
F	23	Yes	Yes	Yes
Μ	22	Yes	Yes	Yes
F	67	Yes	Yes	Yes
F	38	Yes	Yes	Yes
F	69	Yes	Yes	Yes
F	89	Yes	Yes	Yes
F	76	Yes	Yes	
М	34	Yes	Yes	
Μ	25	Yes	Yes	Yes
Μ	23	Yes	Yes	Yes
F	36	Yes	Yes	Yes
Μ	59	Yes	Yes	Yes
F	64	Yes	Yes	Yes
Μ	29	Yes	Yes	Yes
Μ	66	Yes	Yes	Yes
F	61	Yes	Yes	Yes
F	36	Yes	Yes	
	Sex FFMFMMFMFFFFFMMMFMFMFF	SexAgeF37F52M28F73M19M49F23M22F67F38F69F89F76M34M25M23F36M59F64M29M66F61F36	Sex Age Enrichment F 37 Yes F 52 Yes M 28 Yes F 73 Yes F 73 Yes M 19 Yes M 19 Yes M 49 Yes F 23 Yes F 67 Yes F 67 Yes F 69 Yes F 69 Yes F 89 Yes F 76 Yes M 25 Yes M 25 Yes M 23 Yes F 36 Yes M 29 Yes M 66 Yes M 66 Yes F 61 Yes F 36 Yes	SexAgeEnrichmentCell sortingF37YesF52YesM28YesF73YesYesM19YesYesM49YesYesF23YesYesF67YesYesF67YesYesF67YesYesF89YesYesF69YesYesF76YesYesF76YesYesF34YesYesM23YesYesM59YesYesF64YesYesM29YesYesM66YesYesF61YesYesF36YesYes

Age range: 19-89 years old, median: 43.5 years old

Table 4 Donor details for cells used in Chapter 6

Donors	Sex	Age	Cell sorting	MSC genes	Exploratory genes
DiFHOA003	Μ	74	Yes		
DiFHOA008	F	81	Yes		
DiFHOA014	F	56	Yes	Yes	Yes
DiFHOA017	Μ	68	Yes	Yes	Yes
DiFHOA018	F	83	Yes	Yes	
DiFHOA019	F	74	Yes	Yes	
DiFHOA021	F	72	Yes	Yes	
DiFHOA022	F	60	Yes	Yes	Yes
DiFHOA024	F	79	Yes	Yes	Yes
DiFHOA025	F	79	Yes	Yes	Yes
DiFHOA026	Μ	68	Yes	Yes	Yes
DiFHOA027	Μ	67	Yes	Yes	
DiFHOA029	Μ	82	Yes	Yes	Yes

Age range: 56-83 years old, median: 74 years old

Appendix 2 – List of reagents, consumables, equipments, softwares, solutions and Taqman probes used throughout the thesis

Table 1 List of reagents used throughout the thesis

Tissue culture

Ammonium Chloride Anti-fibroblast beads DMEM DMSO DNAse EDTA FCS Formaldehyde MACSelect LNGFR beads Methylene blue PBS Penicillin/Streptomycin RosetteSep cocktail StemMACS MSC expansion medium Trypan blue Trypsin

Flow cytometry/FACS

7-AAD Blocking buffer BSA **CD106 PE** CD146 PE CD271 APC CD271 PeVio770 CD295 APC CD45 PE-Cy7 CD45 V450 IgG1 PE IgG2a APC IgG2b APC IgG1 PE Vio 770 CellROX (FITC) Connexin43 APC Countbright absolute counting beads FCR TBHP

Company

Stem cells Technologies MACS Miltenyi Biotec Gibco, Life Technologies Sigma Sigma Sigma Thermo Fisher Sigma MACS Miltenyi Biotec Sigma Sigma Thermo Fisher Stem Cells Technologies MACS Miltenyi Biotec Sigma Sigma

Company

BD Pharmigen Sigma Sigma **BD** Pharmigen **BD** Pharmigen MACS Miltenyi Biotec MACS Miltenyi Biotec **BD** Pharmigen **BD** Pharmigen **BD** Pharmigen **BD** Pharmigen **BD** Pharmigen R & D systems MACS Miltenyi Biotec Thermo Fisher R & D systems Life Technologies MACS Miltenyi Biotec Thermo Fisher

qPCR

Assay loading buffer Elution solution Nuclease free water PA master mix RL Buffer (lysis buffer) RNAse away Rt mastermix Sample loading buffer Single cell RNA purfication kit Taqman Universal master mix Washing solution

Company

Fluidigm Norgen Biotek Thermo Fisher Fluidigm Norgen Biotek Fluidigm Fluidigm Norgen Biotek Thermo Fisher Norgen Biotek

Table 2 List of consumables used throughout the thesis

Tissue culture

Cell strainer (70µm) Cryo vials (1.8ml) EDTA tubes (4ml) Falcon tubes (15 and 50 ml) Mr Frosty Freezing container Petri dishes (60 and 100mm) Pipette Tips (10, 200 and 1000µl) Stripettes (5,10 and 25ml)

Flow cytometry/FACS

FACS tubes Magnetic columns and plunger Tubes with cell strainers

qPCR

48.48 dynamic array IFC Collection tubes Control line fluid Eppendorfs (0.5-1.5ml) Flex Six[™] gene expression IFC RNA elution tubes Single cell RNA spin columns

Company

BD Falcon Thermo Scientific VACUETTE, Griener Bio-one Corning Thermo Scientific Corning Rainin Corning

Company

Corning MACS Miltenyi Biotec BD

Company

Fluidigm Norgen Biotek Fluidigm Eppendorf Fluidigm Norgen Biotek Norgen Biotek

Table 3 List of equipments used throughout the thesis

Tissue culture

Centrifuge Class II laminar flow biological safety cabinet CO2 Incubator Freezer (-80°C) Haemocytometer Infinity 1 Camera MACS separator MACS stand Microscope (Scanner) Microscope (TC) Nanodrop Spectrophotometer Pipetteboy Scanner Water bath

Flow cytometry/FACS

Influx 6 way cell sorter LSRII flow cytometer Pipettes

qPCR

Biomark HD system IFC controller HX Thermocycler

Company

Eppendorf

Nuaire Incusafe New Brunswick Scientific Improved Neubauer, Hawksley Lumenera MACS Miltenyi Biotec MACS Miltenyi Biotec Infinity 1, L3200B Olympus CKX41 Nanodrop Integra Epson perfection 3590 photo Leica

Company

BD Biosciences BD Pharmigen Gilson

Company

Fluidigm Fluidigm Applied biosciences

Table 4 List of softwares used throughout the thesis

Softwares

Cluster 3.0 Data collection software FACS Diva 5.02 FlowJo GraphPad Prism (version 7.0a) ImageJ Infinity analyze and capture Real-time PCR analysis Tree view

Company

Open source Biomark BD Biosciences BD Biosciences GraphPad Software, Inc. ImageJ Lumenera Biomark Open source

Table 5 List of buffers/reagents prepared and used throughout the thesis

Buffer

Ammonium chloride

FACS buffer Formaldehyde (3.7%) Freezing medium MACS buffer

Methylene blue Thawing medium Wash buffer (for enrichment)

Preparation

17.89g NH4Cl + 2g KCl + 400µl 0.5M EDTA +200ml distilled Water, pH=8.0 500ml PBS + 0.1%BSA + 0.01%Sodium Azide + 200µl of Na EDTA 1ml of 37%formaldehyde + 9 ml PBS 10%DMSO + 45%FCS + 45%DMEM 500ml PBS+ 0.1%BSA + EDTA (2ml) 1%Methylene Blue in10mM borate buffer, pH=8.8 DMEM+10%FCS + 80µl DNAse for 50ml PBS+ 2%FCS + 2ml EDTA Appendix 3 – Supplementary Figure 3.1



Age-related changes in the number of CFU-Fs in males (top panel) and females (bottom panel)

(A) Age-related change in the number of CFU-F per ml of BMA across entire donor age range in males and (B) in between age groups. (C) Age-related change in the number of CFU-F per ml of BMA across entire donor age range in females and (D) in between age groups. Each dot indicates individual donor, black dots represent male donors and empty circles represent female donors. The black line on the left indicates the slope and the on the right, the median values. Spearman non-parametric test was performed for A and C and Kruskal-wallis test with Dunn's correction was performed for B and D.

Appendix 4 List of tables indicating expression of exploratory genes

from Chapter 5 and 6

All the data was normalised with HPRT1. *p*<0.05, *p*<0.01, *p*<0.001 and *p*<0.0001, Mann-Whitney U test. NA: not-applicable, NS: non-significant

Table 5.1 Medians of expression of exploratory genes in all (young and old donors combined) donors compared between MSCs and HLCs

			Fold	
_	Median	Median	difference	_
Gene	(MSCs)	(HLCs)	(MSCs/HLCs)	<i>p</i> value
ABCA1	1.17	0.24	4.88	0.0249
ABCG1	ND	0.058	NA	NA
AICDA	ND	0.009	NA	NA
BST2	12.06	4.97	2.43	0.0083
CASP1	3.06	6.16	0.50	0.0205
CCL8	0.22	0.072	3.03	0.0591
CCND2	2.36	0.41	5.68	0.0013
CEACAM	0.31	0.007	44.29	0.0021
CHMP5	2.41	0.7	3.44	NS
CXCL10	0.25	0.40	0.61	NS
EPSTI1	1.59	0.57	2.79	0.0045
EIF2AK2	3.81	0.95	3.98	0.0023
FCGR1B	0.19	1.04	0.18	0.0031
GBP1	3.4	1.375	2.47	0.078
GUSB	2.66	3.39	0.78	NS
HERC5	0.36	0.2	1.80	0.0188
HPSE	0.09	0.73	0.13	0.0171
IFI6	3.285	0.438	7.5	0.0145
IFI16	1.66	0.81	2.06	0.0068
IFI27	1.55	0.01	155.10	<0.0001
IFI35	2.68	0.76	3.53	0.0387
IFI44	3.1	1.01	3.07	0.0205
IFI44L	1.50	0.31	4.86	0.008
IFIH1	0.45	0.17	2.65	0.0106
IFIT1	9.13	3.70	2.46	NS
IFIT2	1.53	2.67	0.57	NS
IFIT3	2.69	1.86	1.45	NS
IFIT5	0.63	0.11	5.75	0.0284
IFITM1	75.05	6.33	11.86	0.0005
IFITM3	137.9	3.79	36.39	<0.0001
IFNG	ND	ND	NA	NA
IL7R	0.09	0.18	0.51	NS
IRF2	2.48	1.83	1.36	0.0372
IRF5	0.12	1.28	0.10	0.0157
IRF7	4.73	4.97	0.95	NS
IRF9	1.70	0.99	1.72	NS

ISG15	0.38	0.55	0.69	NS
ISG20	0.39	1.12	0.35	NS
LAIR1	0.43	1.4	0.31	0.0278
LAMP3	ND	ND	NA	NA
LRP1	48.92	4.03	12.14	0.0007
LSCR1	4.12	0.99	4.13	0.0068
LY6E	24.58	2.4	10.24	<0.0001
MSR1	0.11	0.20	0.53	NS
MX1	2.889	1.71	1.69	NS
NT5C3B	0.56	0.16	3.52	0.0018
OAS1	0.61	0.13	4.66	NS
OAS2	1.19	0.67	1.78	0.059
OAS3	2.55	0.7	3.64	0.0121
OASL	1.01	0.22	4.61	0.0284
PHF11	1.79	0.98	1.82	0.0242
PPIA	14.26	13.66	1.04	NS
PRDM1	0.51	1.72	0.30	0.0148
PRDM16	0.18	0.0017	105.29	0.042
PRKRA	0.87	0.27	3.16	NS
RGS1	0.621	32.29	0.02	0.0184
RNF213	3.47	2.64	1.31	NS
RSAD2	0.43	0.13	3.30	0.0169
RTP4	0.96	0.08	12.00	0.0036
SAMD9L	1.67	0.642	2.60	0.0387
SCARB1	0.59	0.349	1.69	NS
SERPING	22	0.09	233.30	0.0003
SIGLEC1	0.13	0.38	0.34	0.0782
SOCS1	2.55	0.46	5.55	0.0003
SP100	6.78	3.86	1.76	0.049
SPATS2L	3.50	0.40	8.66	0.0001
TAP1	2.23	1.64	1.36	NS
TGFB	2.93	5.35	0.55	0.0205
TLR4	1.42	1.44	0.99	NS
TNF	0.17	3.67	0.05	0.013
TNFRSRP1	1.043	2.62	0.40	0.0512
TRIM38	0.793	0.62	1.27	NS
UBE2L6	11.86	4.5	2.64	0.0083
UNC93B	2.08	2.75	0.76	NS
USP18	0.35	0.02	17.50	<0.0001
XAF1	3.93	3.273	1.20	NS

			Age-related	
	Medians	Medians	difference	
Gene	(young)	(old)	(old/young)	<i>p</i> value
ABCA1	1.11	2.78	2.51	NS
ABCG1	ND	ND	NA	NA
AICDA	ND	ND	NA	NA
BST2	12.06	14.6	1.21	NS
CASP1	3.23	3.05	0.94	NS
CCL8	0.21	0.26	1.22	NS
CCND2	2.36	3.57	1.52	NS
CEACAM	0.1	0.35	3.50	NS
CHMP5	3.10	1.66	0.54	NS
CXCL10	0.56	0.16	0.29	NS
EPSTI1	1.59	1.69	1.06	NS
EIF2AK2	3.48	3.8	1.09	NS
FCGR1B	0.05	0.27	5.00	NS
GBP1	3.173	3.4	1.07	NS
GUSB	2.98	2.33	0.78	NS
HERC5	0.53	0.34	0.64	NS
HPSE	0.11	0.08	0.80	NS
IFI6	4.36	2.57	0.59	NS
IFI16	1.54	1.76	1.14	NS
IFI27	1.26	2.06	1.63	NS
IFI35	2.68	2.11	0.79	NS
IFI44	3.83	2.71	0.71	NS
IFI44L	1.54	1.42	0.92	NS
IFIH1	0.45	0.43	0.96	NS
IFIT1	7.04	10.06	1.43	NS
IFIT2	1.33	1.68	1.26	NS
IFIT3	1.78	3.4	1.91	NS
IFIT5	0.63	0.55	0.87	NS
IFITM1	75.05	81.38	1.08	NS
IFITM3	139.4	111.9	0.80	NS
IFNG	ND	ND	NA	NA
IL7R	0.22	0.05	0.24	NS
IRF2	2.48	2.23	0.90	NS
IRF5	0.11	0.42	3.91	NS
IRF7	4.73	4.13	0.87	NS
IRF9	1.70	1.35	0.79	NS
ISG15	0.42	0.28	0.67	NS
ISG20	0.51	0.35	0.69	NS
LAIR1	0.8	0.43	0.54	NS
LAMP3	ND	ND	NA	NA

Table 5.2 Medians of expression of exploratory genes between young and old donors in MSCs

LRP1	50.97	38.69	0.76	NS
LSCR1	4.11	6.41	1.56	NS
LY6E	24.58	21.34	0.87	NS
MSR1	0.12	0.05	0.45	NS
MX1	2.89	2.9	1.00	NS
NT5C3B	0.47	0.67	1.41	NS
OAS1	0.68	0.58	0.85	NS
OAS2	1.52	0.88	0.58	NS
OAS3	2.72	1.79	0.66	NS
OASL	1.09	0.82	0.75	NS
PHF11	2.10	1.76	0.84	NS
PPIA	15.94	12.03	0.75	NS
PRDM1	0.21	0.66	3.14	NS
PRDM16	0.29	0.17	0.6	NS
PRKRA	0.79	1.1	1.39	NS
RGS1	0.62	6.11	9.85	NS
RNF213	4.52	1.88	0.42	0.0411
RSAD2	0.394	0.47	1.19	NS
RTP4	0.96	0.71	0.74	NS
SAMD9L	1.83	0.90	0.50	NS
SCARB1	0.80	0.20	0.25	NS
SERPING	21.95	22.85	1.04	NS
SIGLEC1	0.20	0.13	0.65	NS
SOCS1	3.79	2.13	0.56	NS
SP100	6.75	6.78	1.00	NS
SPATS2L	4.09	3.38	0.83	NS
TAP1	2.11	2.53	1.20	NS
TGFB	3.35	2.59	0.77	NS
TLR4	1.2	2.16	1.80	NS
TNF	1.58	0.155	0.10	NS
TNFRSRP1	1.37	0.76	0.55	NS
TRIM38	0.80	0.73	0.91	NS
UBE2L6	11.26	12.94	1.15	NS
UNC93B	2	2.21	1.105	NS
USP18	0.31	0.36	1.16	NS
XAF1	5.81	3.03	0.52	NS

			Age-related	
Genes	Median (young)	Median (old)	difference (old/young)	<i>p</i> value
ABCA1	0.53	0.16	0.30	NS
ABCG1	0.07	0.03	0.43	0.0649
AICDA	0.009	0.01	1.11	NS
BST2	6.29	3.24	0.51	NS
CASP1	7 18	4 77	0.66	NS
CCI 8	0.09	0.03	0.00	NS
CCND2	0.85	0.291	0.34	NS
CEACAM	0.008	0.003	0.38	0.0667
CHMP5	0.87	0.63	0.72	NS
CXCL10	0.58	0.18	0.31	0.0667
EPSTI1	0.58	0.51	0.88	NS
EIF2AK2	0.80	1.05	1.30	NS
FCGR1B	1.05	1.55	1.48	NS
GBP1	1.67	1.33	0.80	NS
GUSB	4.49	2.57	0.57	NS
HERC5	0.22	0.18	0.82	NS
HPSE	0.73	0.66	0.91	NS
IFI6	0.87	0.76	0.87	NS
IFI16	0.31	0.55	1.77	NS
IFI27	0.014	0.008	0.57	NS
IFI35	0.93	0.50	0.54	0.0649
IFI44	1.01	0.87	0.86	NS
IFI44L	0.31	0.33	1.06	NS
IFIH1	0.19	0.16	0.84	NS
IFIT1	2.17	5.73	2.64	NS
IFIT2	2.51	2.76	1.10	NS
IFIT3	1.76	1.86	1.06	NS
IFIT5	0.12	0.058	0.47	0.0411
IFITM1	18.32	2.22	0.12	NS
IFITM3	7.15	3.45	0.48	NS
IFNG	ND	ND	NA	NA
IL7R	0.49	0.09	0.20	0.026
	2.09	1.11	0.53	0.0043
	1.59 6.27	1.08	U.68 0.79	
	0.37	4.97 0.62	0.70	0.026
ISC15	1.52	0.02	0.41	0.020 NS
ISG20	1 18	0.83	0.70	NS
LAIR1	1.54	1.4	0.91	NS
LAIR1	1.54	1.4	0.91	NS

Table 5.3 Medians of expression of exploratory genes between young and old donors in HLCs

LAMP3	ND	ND	NA	NA
LRP1	4.65	2.73	0.59	NS
LSCR1	1.29	0.65	0.50	NS
LY6E	2.91	1.8	0.62	NS
MSR1	0.177	0.318	1.80	NS
MX1	1.71	2.13	1.25	NS
NT5C3B	0.23	0.09	0.41	NS
OAS1	0.35	0.05	0.15	NS
OAS2	0.69	0.22	0.32	NS
OAS3	0.9	0.49	0.54	NS
OASL	0.22	0.26	1.18	NS
PHF11	0.77	1.20	1.55	NS
PPIA	16.08	12.35	0.77	NS
PRDM1	2.35	1.4	0.60	NS
PRDM16	0.0012	ND	NA	NA
PRKRA	0.97	0.2	0.21	0.0152
RGS1	28.47	32.29	1.13	NS
RNF213	3.01	1.75	0.58	NS
RSAD2	0.13	0.12	0.92	NS
RTP4	0.15	0.05	0.33	0.0152
SAMD9L	0.86	0.47	0.55	NS
SCARB1	0.431	0.131	0.30	0.0087
SERPING	0.10	0.08	0.80	NS
SIGLEC1	0.38	0.35	0.92	NS
SOCS1	0.72	0.33	0.46	NS
SP100	4.43	3.27	0.74	NS
SPATS2L	0.47	0.37	0.79	NS
TAP1	1.429	1.65	1.15	NS
TGFB	6.42	4.64	0.72	NS
TLR4	1.87	0.92	0.49	NS
TNF	2.98	3.66	1.23	NS
TNFRSRP1	2.69	2.62	0.97	NS
TRIM38	0.58	0.68	1.17	NS
UBE2L6	5.04	4.26	NS	NS
UNC93B	2.9	2.31	0.80	NS
USP18	0.027	0.025	0.93	NS
XAF1	3.212	3.61	1.12	NS

	Modiane	Modiane	Difference	
Genes	(Old)	(OA)	in OA (OA/Old)	<i>p</i> value
ABCA1	2.78	2.88	1.04	NS
ABCG1	0.02	0.21	12.35	NS
AICDA	ND	ND	NA	NA
BST2	14.60	1.36	0.09	0.0047
CASP1	3.04	0.53	0.17	0.0012
CCL8	0.25	0.12	0.48	NS
CCND2	3.57	2.43	0.68	NS
CEACAM	0.35	0.02	0.06	0.0221
CHMP5	1.66	1.78	1.07	NS
CXCL10	0.16	0.63	3.94	NS
EPSTI1	1.69	0.31	0.18	0.035
EIF2AK2	3.81	1.72	0.45	NS
FCGR1B	0.27	0.00	0.01	0.0476
GBP1	3.40	3.30	0.97	NS
GUSB	2.34	0.67	0.29	0.035
HERC5	0.34	0.20	0.59	NS
HPSE	0.08	0.06	0.75	NS
IFI6	2.57	1.45	0.56	NS
IFI16	1.76	1.16	0.66	NS
IFI27	2.15	0.82	0.38	0.014
IFI35	2.11	1.29	0.61	NS
IFI44	2.71	0.60	0.22	0.0047
IFI44L	1.43	0.35	0.25	0.0047
IFIH1	0.42	0.28	0.67	NS
IFIT1	10.06	3.90	0.39	NS
IFIT2	1.68	0.64	0.38	NS
IFIT3	3.40	0.51	0.15	0.014
IFIT5	0.55	0.26	0.47	NS
IFITM1	81.38	8.52	0.10	0.014
IFITM3	111.90	11.60	0.10	0.014
IFNG	ND	ND	NA	NA
IL7R	0.05	0.19	3.66	NS
IRF2	2.23	1.11	0.50	NS
IRF5	0.42	0.01	0.02	NS
IRF7	4.13	2.48	0.60	NS
IRF9	1.35	0.88	0.65	NS
ISG15	0.28	1.56	5.57	NS
ISG20	0.35	1.17	3.34	0.0424
LAIR1	0.43	0.13	0.30	0.0303

Table 6.1 Medians of expression of exploratory genes between old donors andOA patients in MSCs

LAMP3	ND	ND	NA	NA
LRP1	38.69	8.00	0.21	0.0221
LSCR1	6.41	2.04	0.32	NS
LY6E	21.34	4.07	0.19	0.0152
MSR1	0.06	0.11	2.00	NS
MX1	2.90	1.66	0.57	NS
NT5C3B	0.67	0.37	0.55	NS
OAS1	0.58	0.23	0.39	NS
OAS2	0.88	0.30	0.34	NS
OAS3	1.80	0.54	0.30	0.014
OASL	0.82	0.79	0.96	NS
PHF11	1.76	0.81	0.46	NS
PPIA	12.03	9.65	0.80	NS
PRDM1	0.66	0.71	1.08	NS
PRDM16	0.17	0.11	0.65	NS
PRKRA	1.10	0.82	0.74	NS
RGS1	6.12	0.18	0.03	NS
RNF213	1.88	4.07	2.16	NS
RSAD2	0.47	0.33	0.70	NS
RTP4	0.71	0.12	0.17	0.0221
SAMD9L	0.91	0.58	0.64	NS
SCARB1	0.20	0.24	1.20	NS
SERPING	22.85	8.80	0.39	0.0513
SIGLEC1	0.13	0.01	0.06	NS
SOCS1	2.13	0.76	0.36	NS
SP100	6.78	2.99	0.44	NS
SPATS2L	3.82	2.12	0.55	NS
TAP1	2.53	1.27	0.50	0.0513
TGFB	2.59	2.90	1.12	NS
TLR4	2.16	0.88	0.41	NS
TNF	0.16	0.60	3.88	NS
TNFRSRP1	0.76	0.65	0.86	NS
TRIM38	0.73	0.83	1.14	NS
UBE2L6	12.94	2.08	0.16	0.014
UNC93B	2.21	2.37	1.07	NS
USP18	0.36	0.06	0.17	0.0082
XAF1	3.03	1.51	0.50	NS

Genes	Medians (Old)	Medians (OA)	Difference in OA (OA/OId)	p value
ABCA1	0.17	0.59	3.51	0.035
ABCG1	0.03	0.17	5.28	0.0012
AICDA	0.01	0.00	0.06	NS
BST2	3.24	1.44	0.44	0.035
CASP1	4.77	1.01	0.21	0.0023
CCL8	0.04	0.02	0.61	NS
CCND2	0.29	2.29	7.90	0.0022
CEACAM	0.00	0.01	2.00	NS
CHMP5	0.60	0.21	0.34	NS
CXCL10	0.19	0.11	0.58	NS
EPSTI1	0.51	0.34	0.67	NS
EIF2AK2	1.05	0.96	0.92	NS
FCGR1B	1.55	0.07	0.04	0.0012
GBP1	1.34	0.85	0.63	NS
GUSB	2.57	0.72	0.28	0.0047
HERC5	0.18	0.22	1.22	NS
HPSE	0.66	0.16	0.24	NS
IFI6	0.55	0.48	0.87	NS
IFI16	0.76	0.56	0.74	NS
IFI27	0.01	0.07	8.75	0.0051
IFI35	0.51	0.29	0.57	NS
IFI44	0.87	0.37	0.43	NS
IFI44L	0.33	0.16	0.48	NS
IFIH1	0.17	0.20	1.20	NS
IFIT1	5.73	1.50	0.26	0.0734
IFIT2	2.76	0.49	0.18	0.0513
IFIT3	1.86	0.76	0.41	NS
IFIT5	0.06	0.13	2.19	NS
IFITM1	2.22	9.11	4.10	0.0734
IFITM3	3.45	2.47	0.72	NS
IFNG	0.07	1.29	17.42	0.0177
IL7R	0.09	2.89	32.11	0.0012
IRF2	1.11	0.65	0.59	NS
IRF5	1.08	0.14	0.13	0.0012
IRF7	4.98	3.19	0.64	NS
IRF9	0.63	0.35	0.56	NS
ISG15	0.59	0.47	0.80	NS 0.0012
ISG20 I AIR1	0.03	0.61	0.12	0.0012
	1.40 0.01	0.01	0.44 40.67	0.014
LAIVIE'S	2 72	0. 4 9 0 31	0.07	0.0023 NS
LSCR1	2.73	0.01	1 25	NS
LY6E	1.80	1.40	0.78	NS
		-	270	-

Table 6.2 Medians of expression of exploratory genes between old donors andOA patients in HLCs

MSR1	0.31	0.18	0.58	NS
MX1	2.13	1.68	0.79	NS
NT5C3B	0.09	0.07	0.78	NS
OAS1	0.05	0.02	0.44	NS
OAS2	0.22	0.44	2.00	NS
OAS3	0.49	0.65	1.33	NS
OASL	0.26	0.74	2.85	0.035
PHF11	1.21	0.86	0.71	NS
PPIA	12.35	7.68	0.62	NS
PRDM1	1.40	3.60	2.57	0.0221
PRDM16	ND	ND	NA	NA
PRKRA	0.20	0.21	1.05	NS
RGS1	32.29	21.98	0.68	NS
RNF213	1.75	2.23	1.27	NS
RSAD2	0.12	0.32	2.63	NS
RTP4	0.05	0.03	0.58	NS
SAMD9L	0.47	0.31	0.66	NS
SCARB1	0.13	0.05	0.38	NS
SERPING	0.09	0.09	1.00	NS
SIGLEC1	0.35	0.09	0.25	0.0087
SOCS1	0.33	1.03	3.11	NS
SP100	3.28	2.50	0.76	NS
SPATS2L	0.37	0.32	0.85	NS
TAP1	1.65	1.81	1.10	NS
TGFB	4.64	13.23	2.85	0.014
TLR4	0.92	0.16	0.17	0.0513
TNF	3.66	2.16	0.59	NS
TNFRSRP1	2.62	4.96	1.89	NS
TRIM38	0.68	0.24	0.35	NS
UBE2L6	4.26	1.58	0.37	0.035
UNC93B	2.32	0.51	0.22	0.0012
USP18	0.03	0.03	1.12	NS
XAF1	3.61	1.93	0.53	NS

Appendix 5 Ethics

Dr Ann Morgan Senior Lecturer and Honorary Consultant Rheumatologist Molecular Medicine Unit Clinical Sciences Building St James's University Hospital Beckett Street Leeds LS9 7TF Leeds (East) Research Ethics Committee Room 5.2, Clinical Sciences Building St James's University Hospital Beckett Street Leeds LS9 7TF

> Enquiries to Ann Prothero Ethics Secretary Direct Line/Ext: 0113 (20) 65652

> > 22 October 2004

Dear Dr Morgan,

Full title of study: Functional characterisation of the genes and proteins involved in the development and severity of autoimmune and (auto)inflammatory diseases REC reference number: 04/Q1206/107

Thank you for your letter of 08 October 2004, responding to the Committee's request for further information on the above research.

The further information has been considered on behalf of the Committee by the Chairman

Confirmation of ethical opinion

On behalf of the Committee, I am pleased to confirm a favourable ethical opinion for the above research on the basis described in the application form, protocol and supporting documentation.

The favourable opinion applies to the following research site:

Site: The Leeds Teaching Hospitals NHS Trust Principal Investigator: Dr Ann Morgan

Conditions of approval

The favourable opinion is given provided that you comply with the conditions set out in the attached document. You are advised to study the conditions carefully.

Approved documents

The final list of documents reviewed and approved by the Committee is as follows:

Application dated 15/08/2004 Investigator CV dated 15/08/2004 Protocol for 'Investigation of the molecular mechanisms involved in Tumour Necrosis Factor Receptor associated periodic syndrome (TRAPS) dated 16/08/2004

An advisory committee to West Yorkshire Strategic Health Authority

The Leeds Teaching Hospitals NHS

NHS Trust

Ref: Amy Dickinson

17/07/2014

Г

Professor Dennis McGonagle Rheumatology Department Chapel Allerton Hospital Chapeltown Road Leeds LS7 4SA

Research & Development

Leeds Teaching Hospitals NHS Trust 34 Hyde Terrace Leeds LS2 9LN

> Tel: 0113 392 2878 Fax: 0113 392 6397

r&d@leedsth.nhs.uk www.leedsth.nhs.uk

Dear Professor Dennis McGonagle

NHS Permission at LTHT for: Collection of joint Mesenchymal Stem Cells by Re: aspiration, biopsy, joint retrieval at arthroplasty or by synovium agitation during arthroscopy LTHT R&D Number: RR14/11102 (100077/WY) REC: 14/YH/0087

٦

L

I confirm that NHS Permission for research has been granted for this project at The Leeds Teaching Hospitals NHS Trust (LTHT). NHS Permission is granted based on the information provided in the documents listed below. All amendments (including changes to the research team) must be submitted in accordance with guidance in IRAS. Any change to the status of the project must be notified to the R&D Department.

Permission is granted on the understanding that the study is conducted in accordance with the Research Governance Framework for Health and Social Care, ICH GCP (if applicable) and NHS Trust policies and procedures available at

This permission is granted only on the understanding that you comply with the requirements of the Framework as listed in the attached sheet "Conditions of Approval".

If you have any queries about this approval please do not hesitate to contact the R&D Department on telephone 0113 392 2878.

Indemnity Arrangements

The Leeds Teaching Hospitals NHS Trust participates in the NHS risk pooling scheme administered by the NHS Litigation Authority 'Clinical Negligence Scheme for NHS Trusts' for: (i) medical professional and/or medical malpractice liability; and (ii) general liability. NHS Indemnity for negligent harm is extended to researchers with an employment contract (substantive or honorary) with the Trust. The Trust

Chairman Mike Collier CBE Chief Executive Maggie Boyle

The Leeds Teaching Hospitals incorporating:

use the NHS Number The second

Chapel Allerton Hospital Leeds Dental Institute Seacroft Hospital St James's University Hospital The General Infirmary at Leeds Wharfedale Hospital

Health Research Authority

NRES Committee Yorkshire & The Humber - Leeds East Yorkshire and Humber REC Office First Floor, Millside Mill Pond Lane Meanwood Leeds LS6 4RA

> Tel: 0113 3050108 Fax:

12 April 2012

Prof Peter Giannoudis Consultant Department of Trauma and Orthopaedics St James's University Hospital LS9 7TF

Dear Prof Giannoudis

 Study title:
 Biological properties of Mesenchymal Stem Cells in Fracture Healing

 REC reference:
 06/Q1206/127

 Amendment number:
 3/1

 Amendment date:
 11 April 2012

Thank you for submitting the above amendment, which was received on 12 April 2012. It is noted that this is a modification of an amendment previously rejected by the Committee (our letter of 22nd February 2012 refers).

The modified amendment has been considered on behalf of the Committee by the Vice-Chair.

Ethical opinion

I am pleased to confirm that the Committee has given a favourable ethical opinion of the modified amendment on the basis described in the notice of amendment form and supporting documentation.

Approved documents

The documents reviewed and approved are:

Document	Version	Date
Parental Agreement - Consent Form	1.1	23 March 2012
Participant Consent Form: Patient Consent Form 16 - 17 years old	1.0	23 March 2012
Participant Consent Form: Children Assent Form	1.0	01 February 2012
Participant Information Sheet: Patient Information Sheet 16 - 17 years old	1.0	23 March 2012
Participant Information Sheet: Parental Information Sheet	1.1	23 March 2012
Participant Information Sheet: Children Information Sheet	1.1	23 March 2012
Protocol	6.0	19 December 2011
Modified Amendment		11 April 2012

A Research Ethics Committee established by the Health Research Authority